NUREG/CR-5745 EGG-2650

Assessment of ISLOCA Risk-Methodology and Application to a Combustion Engineering Plant

Prepared by D. J., Kelly, J. L. Auflick, L. N. Haney

Altere als A ils of consideration and the sec

Idaho National Engineering Laboratory EG&G Idaho, Inc.

Prepared for U.S. Nuclear Regulatory Commission

> 9205180177 920430 PDR NUREG CR-5745 R PDR

AVAILABILITY NOTICE

Availability of Reference Materical Orted in NRC Publications

Most documents cited in NRC publications will be available from one of the following sources.

- 1. The NRC Public Document Room, 2120 L Street, NW. Lower Level, Washington, L., 20555
- The Superintendent of Documents: U.S. Government Printing Office, P.O. Box 37082, Washington, DC 20013-7082
- 5. The National Technical Information Service Springfield, VA 22161

Although the listing that follows represents the majority of documents ofted in NRC publications, it is not intended to be exhaustive.

Referenced documents available for inspection and copying for a fee from the NRC Public Document Room notude NRC correspondence and internal NRC memoranda: NRC bulletins, circulars, information notices, inspection and investigation notices; licensee event reports, vendor reports and correspondence; Commission papers; and applicant and licensee documents and correspondence.

The following documents in the NUREG series are available for purchase from the GPO Sales Program: formal NRC staff and contractor reports, NRC-sponsored conference proceedings, international agreement reports, grant publications, and NRC booklets and brochures. Also available are regulatory guides, NRC regulations in the Code of Federal Regulations, and Nuclear Regulatory Commission Issuances.

Documents evaluably from the National Technical Information Service Include NUREG-series reports and technical reports prepared by other Federal agencies and reports prepared by the Atomic Energy Commission, forerunner agency to the Nuclear Regulatory Commission.

Documents available from public and special technical libraries include all open literature items, such au books, journal articles, and transactions. Federal Register notices, Federal and State legislation, and congressional reports can usually be obtained from these libraries.

Documents such as thoses, dissertations, foreign reports and translations, and non-NRC conference proceedings are available for purchase from the organization sponsoring the publication ofted.

Single copies of NRC draft reports are available free, to the extent of supply, upon written request to the Office of Administration, Distribution and Mail Services Section, U.S. Nuclear Regulatory Commission, Washington, DC 20555.

Copies of industry codes and standards used in a substantive manner in the NRC regulatory process are maintained at the NRC Library, 7920 Norfolk Avenue, Bethesda, Maryland, for use by the public. Codes and standards are usually copyrighted and may be purchased from the originating organization or. If they are American National Standards, from the American National Standards institute, 1430 Broadway, New York, NY 10018.

DISCLAIMER NOTICE

This report was prepared as an account of work sponsored by an agency of the United States Government. Neither the United States Government nor any agency thereof, or any of their employees, makes any warranty, expressed or implied, or assumes any legal liability of responsibility for any third party's use, or the results of such use, of any information, apparatus, p-aduct or process disclosed in this report, or represents that its use by an other third party would not infringe privately owned rights.

NUREG/CR-5745 EGG-2650 RG, 1S, 9L

Assessment of ISLOCA Risk-Methodology and Application to a Combustion Engineering Plant

Manuscript Completed: March 1992 Date Published: April 1992

Prepared by D. L. Kelly, J. L. Auflick, L. N. Hancy

Idaho National Engineering Laboratory Managed by the U.S. Department of Energy

EG&G Idaho, Inc. Idaho Falls, ID 83415

Prepared for Division of Safety Issue Resolution Office of Nuclear Regulatory Research U.S. Nuclear Regulatory Commission Washington, DC 20555 NRC 71N B5699 Ur.der DOE Contract No. DE-AC7-76ID01570

ABSTRACT

Inter-system loss-of-coolant accidents (ISLOCAs) have been identified as important contributors to offsite risk for some nuclear power plants. A methodology has been developed for identifying and evaluating plant-specific hardware designs, human factors issues, and accident consequence factors relevant to the estimation of ISLOCA core damage frequency and risk. This report presents a detailed description of the application of this analysis methodology to a Combustion Engineering plant.

FIN B5699-Inter-System LOCA

ABS	TRAC	Τ	and the second	11				
LIST	OF F	IGURES .	New Construction and and and and an and and	ii j				
LIST	OF T	ABLES	and a second	ili				
EXE	CUTP	SUMN	MARY PROTECTION CONTRACTOR CONTRA	İX.				
ACK	NOW	LEDGME	INTS	ĸì				
ACE	IONY	MS	and and an and a summary and a second s	ili				
	INTR	ODUCTI	ON CONTRACTOR CONTRACTOR CONTRACTOR CONTRACTOR	1				
	APP	ROACH		3				
	-9.1	Accesem	ent of ISLOCA Potential	3				
	2.1	Cathorin	of Detailed Plant-Specific Information	4				
	2.E	Cautorio	g of Event Trees	á				
	2,3	Develop	Denoral Determined	5				
	2,4	Estimati	on of Rupture Potential	5				
	2,5	Human	Reliability Analysis					
	2.6	Quantifi	cation of Event Trees	0				
	2.7	Consequ	aence Evaluation					
3	DE3	CRIPTIO	N OF THE INTERFACING SYSTEMS	10				
	3.1	Interfac	ing Systems	10				
	3.2	Potentia	d ISLOCA Scenarios	10				
		3.2.1	SDC Suction Lines During Shutdown	10				
		3.2.2	SDC Suction Lines During Startup	1.5				
		3.2.3	1. DSI Cold Leg Injection Lines to KC3	15				
		3.2.4	HPSI Cold Leg Interface	1.5				
		3.2.5	SDC Suction Lines During Normal Shutdown	12				
4.	RES	SULTS		10				
	4.1 Event Trees							
		411	Premature Entry hito Shutdown Cooling-SEQ-1A	1				
		4.1.2	Shutdown Cooling System/Reactor Coolant System ISLOCA During	1				
			Startun_SEO-18					

CONTENTS

		$\begin{array}{r} 4.1.3 \\ 4.1.4 \\ 4.1.5 \\ 4.1.6 \\ 4.1.7 \\ 4.1.8 \end{array}$	RCS To LPSI Cold Leg Discharge—SEQ-2 RCS Cold Legs to HPSI (Header A)—SEQ-3A RCS Cold Legs to HPSI (Header B)—SEQ-3B RCS Hot Legs to HPSI (Header A)—SEQ-4A RCS Hot Legs to HPSI (Header B)—SEQ-4B RCS to LPSI During Shutdown—SEQ-5	17 17 20 20 20 20					
	4.2	Human	Reliability Analysis	20					
	4.3	Quanti	fication of ISLOCA Model	27					
	4,4	Risk A	ssessment	29					
	4.5	Uncerta	ainty and Sensitivity Study Results	29					
		4.5.1 4.5.2	Component Rupture Pressure Uncertainty Auxiliary Building DF Uncertainty	30 30					
5.	CONCLUSIONS								
	5.1	Plant-S	pecific Conclusions	41					
	5.2	General	Conclusions	41					
6.	REFI	ERENCE	S meren and the management of the second	43					
App	endix)	A—Syste	m Descriptions	A-1					
App	endix I	BISLO	CA Event Trees	B-I					
App	endix (Proba	C—Humi bilistic R	an Reliability Analysis for the Combustion Engineering ISLOCA	C-1					
App	endix I Analy	D—Use o sis	f Constrained Lognormal Distribution in Human Reliability	D-1					
Арр	endix I	5Core I	Uncovery Timing Calculations	E-1					
Appe	endix F	Calcul	lation of System Rupture Probability	F-1					
Appe	ndix (3—ISLO	CA Consequence Analysis	G-1					
Appe	ndix F	IComp	onent Failure Analysis	H-1					

LIST OF FIGURES

ES-1.	Approach for plant-specific evaluation of ISLOCA	<u></u>
1.	Approach for plant-specific evaluation of ISLOCA	3
2.	Example of HRA event tree	7
3.	Flow diagram of RCS cold legs to low-pressure safety injection (LPSI) pump discharge	11
4.	Flow diagram of RCS cold legs to HPSI pump discharge	12
5.	Flow diagram of RCS hot legs to HPSI pump discharge	13
6.	Flow diagram of RCS hot legs to the RWSP via the LPSI system	14
7.	SDC system ISLOCA during startup (SEQ-1B)	18
8.	Event tree for LPSI cold leg discharge to RCS (SEQ-2)	19
9.	Event tree for HPSI header A cold leg discharge to RCS (SEQ-3A)	21
10.	Event tree for HPSI header B discharge to RCS cold legs (SEQ-3B)	22
п.	Event tree for HPSI header A discharge to RCS hot legs (SEQ-4A)	23
12.	Event tree for HPSI header B discharge to RCS hot legs (SEQ-4B)	24
13.	Event tree for RCS to LPSI during shutdown (SEQ-5)	25
14.	Mean early fatality consequence results as a function of DF for the dry ISLOCA sequences	32
15.	Mean latent fatality consequence results as a function of DF for the dry ISLOCA sequences	33
16.	Mean 30-mi population dose consequence results as a function of DF for the dry ISLOCA sequences	34
17.	Mean early fatality consequence results as a function of DF for the wet ISLOCA sequences (all release elevations are 0.0 m)	35
18.	Mean latent fatality consequence results as a function of DF for the wet ISLOCA sequences (all release elevations are 0.0 m)	36
19.	Mean 50-mi population dose consequence results as a function of DF for the wet ISLOCA sequences (all release elevations are 0.0 m)	37
20.	Comparison of dry and v.et ISLOCA sequence mean early fatality consequence results as a function of <i>DF</i> (release elevation is 10.0 m for DFs of 50.0 and 100.0, otherwise elevation is 0.0 m)	31

21.	Comparison of dry and wet ISLOCA sequence mean latent fatality consequence results as a function of DF (release elevation is 10.0 m for DFs of 50.0 and 100.0, otherwise elevation is 0.0 m)	39
22.	Comparison of dry and wet ISLOCA sequence mean 50-mi population dose consequences as a function of DF (release elevation is 10.0 m for DFs of 50.0 and 100.0, otherwise elevation is 0.0 m)	40
	LIST OF TABLES	
I	List of potential ISLOCA scenarios	15
2.	Estimated mean HEPs for Sequence 2	27
3.,	Estimated mean HEPs for Sequence 5	27
4,	ISLOCA CDF (per reactor-year)	28
5.	Base case ISLOCA consequences conditional upon severe core damage	29
6.	Base case ISLOCA risk (per reactor-year)	29
7.	Distribution of DF for the base case analysis of the wet ISLOCA sequences as specified in the SEQSOR input	30
8.	Mean MACCS consequence results for each dry ISLOCA sequence sensitivity	31
9,	Mean MACCS consequence results for each wet ISLOCA sequence sensitivity	31

EXECUTIVE SUMMARY

Inter-system loss-of-coolant accidents (ISLOCAs) have been identified in some probabilistic risk assessments (PRAs) as major contributors to offsite risk at nuclear power plants (NPPs). They have the potential to result in core damage and contaiument bypass, which may lead to the early release of large quantities of fusion products to the offsite environment. Recent events at several operating reactors have been identified as ISLOCA precursors. These event have raised concerns over the frequency of occurrence. plausible initiators, and means of identifying and mitigating this potential accident. In response to these concerns, a June 7, 1989, memorandum, "Request for Office of Nuclear Regulatory Research (RES) Support for Resolution of the ISL/DCA Issue," was unasmitted from Dr. Thomas E. Murley to Dr. Eric S. Beckjord. The ISLOCA research program described in this report was initiated in response to this memorandum.

The objective of the ISLOCA research program is to provide the U.S. Nuclear Regulatory Commission with qualitative and quantitative information on the hardware, human factors, and accident consequence issues that contribute to ISLOCA risk. To meet this objective, a methodology has been developed to estimate the core damage frequency (CDF) and offsite consequences associated with an ISLOCA, and this methodology is being applied to individual NPPs. This report describes the ISLOCA methodology and the results of its application to a Combustion Engineering (CE) NPP.

An eight-step methodology was developed to evaluate the ISLOCA issue qualitatively and quantitatively. These steps and their relationships to one abother are shown in Figure ES-1. This methodology was applied to a CE plant by a team of PRA and human factors specialists. The important results, *specific to this plant*, are

 Human errors that could occur during startup and shutdown of the plant were not found to be significant contributors to ISLOCA CDF and risk.

- ISLOCA sequences initiated by hardware failures were the dominant contributors to CDF and risk.
- 3. Isolation of the break would be an important recovery action during an ISLOCA. Refueling water storage pool makeup capacity is insufficient to maintain an adequate reactor coolant inventory for breaks outside the containment that are larger than approximately 1 inch in diameter. The analysis indicates that hardware would be available to isolate these ISLOCA breaks; however, post-break procedures are not available to ensure that this hardware is used in all sequences.
- 4. At the time of the plant visit, a general survey was made of the interfacing system flow paths to qualitatively estimate the impact on equipment of ruptures in various locations. This survey could not verify that the emergency core cooling systems (ECCS) are adequately separated such that any postulated rupture would not affect redundant ECCS trains. This issue is still under study at the Idaho Natienal Engineering Laboratory.
- It appears that relatively simple changes to procedures and training could reduce ISLOCA risk substantially by reducing the initiator frequency and increasing the likelihood of successfully isolating an intersystem break.
- 6. The ISLOCA methodology has been successful in providing important insights on the relative contribution of both hardware faults and human acticas to ISLOCA CDF and risk. In particular, the extensive task analysis performed as part of the human reliability analysis provided many valuable insights that would have been missed in a less detailed analysis.



Figure ES-1. Approach for plant-specific evaluation of ISLOCA.

While caution must be exercised when using these results to draw general conclusions about the ISLOCA risk at other NPPs, the perspective provided by the aforementioned important insights,

along with suitable sensitivity studies and companion PRAs, provide additional technical bases upon which a regulator, accision for resolution of ISLOCA as a generic issue con be considered.

ACKNOWLEDGMENTS

We would like to thank all of the plant personnel who assisted in the datagathering visits. Without their assistance, this analysis could not have been completed.

At the Idaho National Engineering Laboratory, thanks are due to John Schroeder, who assisted in the rupture probability calculations, to Craig Kullberg, who provided the core uncovery time estimates, and to Curtis Smith, who assisted in the sequence screening analysis and developed the simplified system flow diagrams. Special thanks go to operator examiner Mark Jones for his invaluable assistance in interpreting the plant operating procedures and developing the sequence progression. Bob Richards aided in programming the software used in the human reliability analysis.

Finally, we would like to thank all the members of the inter-system loss-ofcoolant accident inspection team for their aid in gathering and interpreting plant data.

ACRONYMS

ACI	Auto-closure interlock	LPSI	Low-pressure safety injection
AFW	Auxifiary feedwater	MAAP	Modular Accident Analysis Program
ARP	Annunciator response procedure	MACCS	MELCOR Accident Consequence Code System
BNL	Brookhaven National Laboratory	MOV	Motor-operated valve
CDF	Core damage frequency	NPP	Nuclear power plant
CE	Combustion Engineering	NRC	Nuclear Regulatory Commission
CR	Control room	NSSS	Nuclear steam supply system
CRS	Control room supervisor	P&ID	Piping and instrumentation diagram
CVCS	Chemical and volume control system	PIV	Pressure isolation valve
DF	Decontamination factor	PRA	Probabilistic risk assessment
DHR	Decay heat removal	PSF	Performance shaping factor
ECCS	Emergency core cooling systems	PWR	Pressurized water reactor
EPRI	Electric Power Research Institute	PZR	Pressurizer
GLP	Gross leak pressure	RCS	Reactor coolant system
HCR	Human cognitive reliability	RO	Reactor operator
HEP	Human error probability	RPV	Reactor pressure vessel
HOV	Hydraulically operated valve	RWSP	Refueling water storage pool
HPSI	High-pressure safety injection	SDC	Shutdown cooling
HRA	Human reliability analysis	SHARP	Systematic Human Action Reliability Procedure
INEL	Idaho National Engineering Laboratory	S1	Safety injection
ISLOCA	Inter-system loss-of-coolant accident	SIT	Safety injection tank (accumulator)
IST	In-service testing	TALENT	Task Analysis-Linked Evaluation Technique
LER	Licensee Event Report	THERP	Technique for human error rate
LOCA	Loss-of-coolant accident		prediction

Assessment of ISLOCA Risks—Methodology and Application to a Combustion Engineering Plant

1. INTRODUCTION

The Reactor Safety Study-An Assessme a of Accident Risks in U.S. Commercial Nuclear Power Plants (WASH-1400)1 identified a class of accidents that can result in overpressurization and rupture of systems that interface with the reactor coolant system (RCS). These events were postulated to be caused by failure of the check valves and motor-operated valves (MOVs) normally used for system isolation. In a subset of these inter-system loss-of-coolant accidents (ISLOCAs), called V-sequences or event V, the system rupture occurred outside of the containment building. In cases where the rupture led to severe core damage, ISLOCAs were found to be significant contributors to risk because fission products released from the RCS bypassed the containment and were discharged directly to the environment. Subsequent probabilistic risk assessments (PRAs), including the NUREG-11502 results for Surry and Sequoyah, have identified ISLOCAs as important contributors to public health risk. Researchers at Brookhaven National Laboratory (BNL) have evaluated the vulnerability of several reactor designs to an ISLOCA and identified improvements that could reduce ISLOCA frequency.3,4

Recent events at several operating reactors have been identified as precursors to an ISLOCA. These events have raised concerns over the frequency of occurrence, potential initiators, and means of identifying and mitigating this potential accident. In response to these concerns, a June 7, 1989, memorandum, "Request for Office of Nuclear Regulatory Research (RES) Support for Resolution of the ISLOCA Issue," was transmitted from Dr. Thomas E. Murley to Dr. Eric S. Beckjord. The ISLOCA research program described in this report was initiated in response to this memorandum.

The objective of the ISLOCA research program is to provide the U.S. Nuclear Regulatory Commission (NRC) with qualitative and constitutive information on the hardware, human factors, and accident consequence issues that contribute to ISLOCA risk. This information is to be used in

- Developing a PRA framework for evaluating the ISLOCA issue and identifying insights with respect to the risk contribution from both hardware and human factors, along with recommendations for reducing the ISLOCA risk.
- Highlighting the effects of specific types of human errors and their root causes on ISLOCA risk, along with recommendations for risk reduction.
- Evaluating the fragility of low-presses e systems exposed to high-pressure, high-temperature reactor coolant. This evaluation will include identification of likely failure locations and failure probabilities.
- Identifying and describing potential ISLOCA sequences with respect to timing, possible accident management strategies, and effects on other plant equipment and systems.
- Estimating the fission product source terms and offsite consequences for postulated ISLOCAs. Again, important issues will be identified and recommendations will be made on possible consequence reduction actions.

To meet the above program objectives, a methodology has been developed to estimate the ISLOCA core damage frequency (CDF) and offsite risk, and this methodology is being applied to a limited sample of nuclear power plants (NPPs) of different design. This report describes the ISLOCA methodology and documents the results from its application to a Combustion Engineering (CE) plant. These results emphasize the effect of hardware failures and human actions on the ISLOCA CDF. The offsite risk measures are considered useful in comparing results from the sensitivity studies. Major uncertainties in this estimate are also identified.

Section 2 of this report describes the methodology developed to evaluate ISLOCAs,

the approach taken for its application to a specific plant, and a description of the plant systems that were identified as potential ISLOCA flow paths. Section 3 describes the interfacing systems and the possible ISLOCA scenarios. Section 4 describes the plant-specific results and Section 5 contains the conclusions and recommendations from this assessment. Appendices A–H are used to document the details of the separate analyses.

2. APPROACH

The general approach that is being used to evaluate ISLOCA risk and plant vulnerabilities to ISLOCA is to perform a detailed analysis for a small but diverse sample of plants and, to the extent possible, extrapolate and generalize these results for additional plants. A detailed plant analysis methodology was developed to meet the program objectives discussed in the previous section. The steps in this individual plant methodology are illustrated in Figure 1. Subsections 2.1 through 2.7 discuss each of these steps briefly.

Before beginning the individual plant evaluations, historical plant operating information was reviewed to provide insights on potential ISLOCA issues. The major emphasis of this evaluation was identification and evaluation of Licensee Event Reports (LERs) that (a) involved valve failures resulting from either hardware failures or human errors or (b) indicated that an ISLOCA had occurred. The results from this search provided information on the causes and frequencies of valve failures and provided important insights on the systems involved and the potential causes of taose ISLOCAs that have occurred. This information was used during the plant visits to help identify systems to be reviewed, develop the event trees, and quantify the failure rates of some interfacing system valves. Appendix A to Assessment of ISLOCA Risks—Methodology and Application to a Babcock and Wilcox Plant summarizes the results of this evaluation.⁵

2.1 Assessment of ISLOCA Potential

The first step in the individual plant evaluation approach is a preliminary assessment of the potential for an ISLOCA to occur. Plant-specific



Figure 1. Approach for plant-specific evaluation of ISLOCA.

Approach

information on the systems that could be involved in an ISLOCA is obtained during a short datagathering visit to the plant. Detailed information is obtained on the hardware and operation of a range of low- and high-pressure interfacing systems. Examples of information collected include plant procedures, piping and instrumentation diagrams (P&IDs), isometric drawings, and training manuals. This information is then reviewed by a team of PRA and human factors specialists to familiarize them with the systems and operations that have the potential to initiate, prevent, or mitigate an ISLOCA. All systems that interface with the RCS are identified during this preliminary assessment. A determination is then made of the maximum interfacing system break size that would not be expected to result in core damage. The systems are screened to identify those with interfacing pipe sizes larger than this maximum value with the potential to bypass containment. The systems that meet this screening criterion are analyzed further to identify specific ISLOCA initiators and scenarios. The identified scenarios are developed in sufficient detail to guide the team in obtaining detailed information during a subsequent extended plant visit.

2.2 Gathering of Detailed Plant-Specific Information

An extended visit to the plant allowed the analysts to gather the information needed to complete the above reviews and to develop and analyze the candidate ISLOCA scenarios. Members of the team that developed the candidate scenarios obtained information by interviewing plant personnel and walking down the systems of interest. This task was performed in conjunction with an ISLOCA inspection conducted by the NRC Office of Nuclear Reactor Regulation. The types of information that were obtained during this visit include detailed information on

 Hardware that would be involved in an ISLOCA. For example, data were collected on control valves, relief valves, piping, flanges pumps, and heat exchangers.

- Procedures, guidelines and practices followed by plant personnel during startup, normal power operation, and shutdown of the plant, as well as detailed information on maintenance and in-service testing.
- Factors that could influence performance of plant personnel as related to initiation, detection, diagnosis, prevention, or mitigation of an ISLOCA.

2.3 Development of Event Trees

After the plant-specific information was collected, a final list of low-pressure interfaces and scenarios was compiled and the detailed accident sequence analysis begun. This analysis was a joint effort of the PRA and human factors specialists. The scenarios were modeled using (primarily) component level event trees combining the bardware faults and the human errors that constitute each sequence in the scenario. In general, each event tree comprised dree phases:

- The initiating events, which are those combinations of failures, both hardware and human-related, that result in a failure of the RCS pressure isolation boundary and expose the low-pressure interfacing system to the RCS.
- The rupture events, which model a break in the interfacing system, its size, and location.
- 3. The post-rupture events, which model the performance of the control room and auxiliary operators in recovering from or mitigating the consequences of an ISLOCA. The list of possible ISLOCA sequences contains both hardware-based sequences (as found in typical PRAs) and sequences initiated by human error. The potential human errors in both types of sequence comprised errors of omission, commission, and pre-existing or latent errors.

2.4 Estimation of Rupture Potential

It is important to realistically assess the performance of those components designed for lowpressure conditions that are exposed to the beyond-design pressures associated with an ISLOCA. The basic approach for performing this assessment is

- The failure probability of each piece of equipment in the interfacing system is described by a lognormal distribution with a specified median failure pressure and logarithmic standard deviation
- Thermal-hydraulic response of the systems is simulated, if necessary, to estimate the pressure distribution in the system based on the expected initiating event, initial primary system conditions, and the expected performance of relief valves designed to protect the systems from overpressurization
- The failure pressure of each component is compared with the calculated pressure at that point in the system to estimate the component failure probability (see Appendix F for details)
- The individual component failure probabilities are combined to give an estimate of the system rupture probability.

The component and piping failure pressures and distributions used in the rupture calculations were developed from an independent structural analysis performed by Impell Corporation (see Appendix H). Not only were failure pressure: calculated, but likely leak rates and leak areas, also. In this respect, flanges are somewhat unique in that there are actually two failure pressures of interest. First, there is the estimated gross leak pressure (GLP) at which a measurable leak area develops. At lower pressures, leakage is possible but only at very small rates (measured in mg/s) caused by seepage around the flange gasket. Once the GLP is exceeded, the flange bolts begin to stretch (elastic deformation) and the flange surfaces begin to separate. At some yet higher pressure (P_0), the bolts begin to undergo plastic deformation. At this point, large leak areas begin to develop with correspondingly large leak rates. These three pressure regions (below GLP, between GLP and P_0 , and greater than P_0), are associated with three leak sizes, respectively: spray leaks, small leaks, and large leaks.

2.5 Human Reliability Analysis

The predominant human errors for each scenario in the ISLOCA PRA were modeled using the techniques of human reliability analysis (HRA). HRA is a methodological tool used for prediction, evaluation, and quantitative analysis of work-oriented human performance. As a diagnostic tool, HRA can estimate the error rate anticipated for individual tasks and can identify where errors are likely to be most frequent.

The general methodological framework for the ISLOCA HRA was based on guidelines (under development) from the NRC-sponsored Task Analysis-Linked Evaluation Technique (TALENT) Program.6 which recommends the use of task analyses, time-line analyses, and interface analyses in a detailed HRA, NUREG/CR-1278, Handbook of Human Reliability Analysis with Emphasis on Nuclear Power Plant Applications [which discusses the technique for human error rate prediction (THERP)],7 recommends similar techniques and, in addition, provides a data base that can be used for estimating human error probabilities (HEPs). Finally, the ISLOCA HRA integrated the steps from the Systematic Human Action Reliability Procedure (SHARP)8 and A Guide for General Principles of Human Action Reliability Analysis for Nuclear Power Generation Stations (IEEE Standard P1082/D7).6

From this combination of approaches, the analysts identified 11 basic steps to be followed in performing the HRA:

 Select the analysis team and train them on relevant plant functions and systems (IEEE P1082).⁹

Approach

- Familiavize the team with the plant through the use of system walkdowns, simulator observations, etc. (IEEE P1082).⁹
- Ensure that the full range of potential human actions and interactions is considered in the analysis (SHARP) (IEEE P1082).^{8,9}
- Construct the initial model of the relevant systems and interactions (IEEE P1082).⁹
- 5. Identify and screen specific human actions that are significant contributors to safe operation of the plant. This was accomplished through detailed task analyses, timeline analyses, observations of operator performance in the plant and in the simulator, and evaluations of the human/machine interface (SHARP and IEEE P1082).^{8.9}
- Develop a detailed description of the important human interactions and associated key factors necessary to complete the plant model. This description should include the key failure modes, an identification of errors of omission/commission, and a review of relevant performance shaping factors (SHARP) (IEEE P1082).^{8,9}
- Select and apply the appropriate HRA techniques for modeling the important human actions (SHARP).⁸
- Evaluate the impact on ISLOCA of significant human actions identified in Step 6 (SHARP).⁸
- Estimate error probabilities for the various human actions and interactions, determine sensitivities, and establish uncertainty ranges (SHARP and IEEE P1082).^{8,9}
- Review results for completeness and relevance (IEEE P1082).⁹
- Document all information necessary to provide an audit trail and to make the information understandable (SHARP).⁸

Because most of the human actions in this HRA involved the use of various written normal, abnormal, and emergency operating procedures. THERP-type HRA event trees were used to model most of the human actions in the detailed analyses.⁷ However, not all ISLOCA scenarios were best represented by THERP event trees alone. In those cases, HRA fault trees were used in conjunction with the THERP event trees. The fault trees and THERP event trees were used in a detailed analysis to estimate the probability of human error for each of the dominant human actions.

Traditionally, human reliability analysts model human performance through the use of an event tree like that shown in Figure 2, with operator error generally placed along the descending right branches of the event tree, and successful operafor actions sequenced on the left side of the tree. For example, on the top left, event a, [RO (reactor operator) detects decreasing pressurizer (PZR) level and pressure] is the success path. Failure to accomplish this task is modeled as event A. (RO fails to detect decreasing pressurizer level and pressure). When a second operator, or group of operators, is involved, such as in event B, [control room (CR) fails to detect PZR Hi-Lo alarm], the action of this second operator may be modeled in a recovery branch, as shown in Figure 2. Event b models how the control room also has an opportunity to detect the PZR Hi-Lo alarm. If the control room does detect the alarm, this becomes a recovery action because it would bring the model back to the success path (via the dotted lines in Figure 2).

The basic, or unmodified, HEPs for branches in the HRA event trees were estimated using techniques from THERP and human cognitive reliability (HCR).¹⁰ These basic HEP estimates were then revised by using performance shaping factors (PSFs) to more realistically model the work process at the plant. Each PSF was either positive or negative and, accordingly, either decreased or increased the likelihood of a given human error. For example, an analog meter, such as a pressure gauge, if it does not have easily seen limit marks, would be judged to be a negative PSF. Thus, there would be a higher-than-normal probability for



7

Approach

Approach

error in reading the gauge. Individual PSFs were derived from task analyses, time-line analyses, evaluation of the human/machine interface, and direct observations of operator performance. They are presented as part of the *ISLOCA Inspection Report*.¹¹

Specific PSFr that were investigated include

- Quality of the human/machine interface
- Written procedures (emergency, abnormal, mai, 'enance, etc.)
- P&IDs
- Response times for systems and personnel
- Communication requirements
- Determination of whether the operator actions were skill-, rule-, or knowledgebased
- Crew experience
- Levels of operator stress in different scenarios
- Feedback from the systems in the plant
- Task dependence and operator dependence
- Location of the task (control room, auxiliary building, etc.)
- Training for individual operator actions including those required in ISLOCA situations.

Finally, the combination of all identified failure paths (i.e., sequences that included either single or multiple human errors leading to a failure of the action modeled by the HRA tree) gave the failure probability for the action modeled in the HRA tree. The guidelines of THERP were followed in identifying the individual error paths. As depicted by Figure 2, each human error event tree may have several unique error paths. For example, event A, event B, and event C together constitute an error path in which the reactor oper-

ator fails to detect decreasing PZR level and pressure, followed by individual failures of the control room to detect two PZR level marms. In a similar manner, failure path A o-D-E models a sequence where the reactor operator fails to detect decreasing PZR level and pressure, then recovers (e.g., the control room detects the PZR Hi-Lo alarm) from this first failure (event b), only to have both the control room supervisor (CRS) and shift supervisor (SS) fail to enter Procedure OP-901-046, "Shutdown Cooling Maifunction." Probabilities for each unique error path were calculated by multiplying each 'HEP on a given error path by other HEPs on the same path. For example, the error rate for path A-B-C would be calculated by multiplying the HEP of failure A by that for failure B and then by the HEP for failure C, resulting in a nominal HEP for that specific path. Other error paths for this event tree include A-B-c-D-E and a-D-E. The individual error path failure probabilities were then summed to give the total event tree failure probability. Comprehensive details of this process are provided in Appendix C for each event, and the results are summarized in Section 4.2.

A detailed HRA was conducted for each of the significant scenarios identified in the ISLOCA PRA. See Section 4.2 or Appendix C for details of the results of these analyses. (The tables in Section 4 summarize the results.) These tables provide the identifier and description for each significant human error, as well as both nominal and mean HEPs. Nominal HEPs in these tables are assumed to be *median point estimates from a lognormal distribution* (using guidelines from *Handbook of Human Reliability Analysis with Emphasis on Nuclear Power Plant Applications*).⁸ while mean values are *mean HEPs from a lognormal distribution*, which were derived using the following formula:

mean
$$HEP = exp\left(\mu + \frac{\sigma^2}{2}\right)$$

where

$$u = lnx$$

$$\hat{x} = the median HEP$$

$$5 = \frac{ln(error factor)}{1.645}$$

The conversions to mean values were carried out as a result of mathematical concerns where median values from a lognormal distribution should not be multiplied by mean values in estimating the mean CDF, a process that has been followed in some past HRAs.

2.6 Quantification of Event Trees

The top events on the ISLOCA accident sequence event trees are quantified by separate calculations that generate the conditional probabilities of occurrence of each event for each path through the tree. The means of obtaining the rupture event probabilities and the probabilities relating to failure of plant personnel have already been discussed. Hardware failure probabilities were developed using the data base documented in Appendix B of Assessment of ISLOCA Risks--Methodology and Application to a Babcock and Wilcox Plant.⁵ The ISLOCA event trees were constructed using the ETA-II personal computer code.¹²

2.7 Consequence Evaluation

The ISLOCA CDFs are multiplied by the corresponding consequences (conditional on the occurrence of core damage) calculated using the MELCOR Accident Consequence Code System (MACCS) code to obtain the ISLOCA annual risk estimates.13 The conditional consequences were generated with MACCS using a hybrid input deck. The fission product source terms were obtained from the SEQSOR parametric source term generation code.14 The source terms generated are the ones identified with the containment bypass V-sequence in NUREG-1150.2 The site information was taken from the Surry MACCS input deck used in the NUREG-1150 program. The Surry site was chosen by reviewing Technical Guidance for Siting Criteria Development¹⁵ and calculating an average site based on weatherweighted population density. This average population density was then compared to the five NUREG-1150 sites and Surry was chosen because it most closely matched the calculated average population density. Further details of the consequence calculations can be found in Appendix G.

3. DESCRIPTION OF THE INTERFACING SYSTEMS

The unit analyzed is a 3,390 MWt pressurized water reactor (PWR), with a two-by-four loop (two hot legs and four cold legs) CE nuclear steam supply system (NSSS). It is equipped with a large, dry, atmospheric-pressure containment and a separate reactor auxiliary building and turbine building. An overview of the interfacing systems is presented in the following section. For more details on the interfacing systems, see Appendix A

3.1 Interfacing Systems

A'l interfacing systems were screened to identify those systems that required further evaluation. The first criterion used in screening was that any system with an interfacing pipe diameter larger than 1 inclushould be evaluated. The 1-inch pipe size was selected based on an estimation of the discharge from a 1 inch high-pressure pipe break, which was about 200 gpm. A 200-gpm leak rate outside of the containment is considered to be critical based on the capacity of the refueling water storage pool (RWSP) (minimum Technical Specification volume of 443,000 gal), the capacity of the three charging pumps (132 gpm), and the normal makeup rate to the RWSP (~150 gpm). Based on these considerations and the number of hours it would take for the plant to achieve cold shutdown (conservatively assumed to be about 10 hours), leak rates of 200 gpm or less were judged no, to be risk significant. The second crite .. n was that systems whose lowpressure policions were isolated from reactor pressuce by three or more normally closed valves or periodically leak-tested check valves in series would not be walyzed. The basis for screening Ad in its the lasy expected frequency of out sa dure of the pressure isolation boursa

The initial screening resulted in the selection of the safety injection (SI) system for further analysis, including the high- and low-pressure safety injection pumps and the shutdown cooling (SDC) lines. Figures 3 the show simplified flow diagrams of the hardware configuration. Additional details on these systems are provided in Appendix A. The SI system interface compliants 12 ceparate reactor pressure vessel (RPV) injection lines, eight high pressure and four low pressure. Starting from the RPV, each injection line contains two check valves in series, a normally closed motor-operated flow control valve and the SI pump discharge check valve. The SDC interface consists of four low-pressure injection lines to each of the RCS cold legs, two high-pressure recirculation lines to the RCS hot legs, and two suction lines from the RCS hot legs used for shutdown cooling.

3.2 Potential ISLOCA Scenarios

Potential scenarios were developed by examining the system interfaces and plant operational information. A team of PRA and human factors specialists was involved in the scenario development. In some cases (e.g., the SI system injection lines), the sequences are hardware drivea; that is, the ISLOCA potential is a function of the hardware failure rates of the pressure isolation boundary valves. In other cases (e.g., the SDC suction lines), human errors can initiate an ISLOCA. Table 1 summarizes the ISLOCA scenarios identified.

3.2.1 SDC Suction Lines During Shutdown. During the plant shutdown process, the operators will open MOVs £1-401 and \$1-407 and hydraulically operated valve (HOV) \$1-405 to allow for the removal of decay heat. Sequence 1A investigates the fikelihood that the valves (that will be opened by the operators) are opened prematurely, that is, at an RCS pressure greater than the procedural limit of 396 psig.

3.2.2 SDC Suction Lines During Startup. Sequence 1B is similar to Sequence 1A, except that the plant is undergoing a startup. Thus, failure to close MOVs SI-401 and HOVs SI-405 is modeled, as opposed to Sequence 1A where the failure mode for valves SI-401 and SI-405 was premature opening.

Description of the Interfacing Systems



Figure 3. Flow diagram of RCS cold legs to low-pressure safety injection (LPSI) pump discharge.

Description of the Interfacing Systems



Figure 4. Flow diagram of RCS cold legs to high-pressure safety injection (HPSI) pump discharge.



Figure 5. Flow Alagram of RCS hot legs to HPSI pump discharge.

Description of the Interfacing Systems



Figure 6. Flow diagram of RCS hot legs to the RWSP via the LPSI system.

Interface	Description	Notes
SDC section lines	Failure to close valves during startup or premature opening during shutdown	Two scenarios investigated: one stanop and one shutdown (SEQs-1A and 1B)
CP51 cold leg injection lines	Failure of two check valves with stroke-testing of normelly closed MOV	Initiated by hardware failures in conjunction with MOV stroke testing (SEQ-2)
HPSI cold leg interface	Failure of two pressure isolation check valves and stroke test of normally closed MOV, plus failure or safety injection pump discharge check vidve	Only hardware feitures considered (SEQs-3A and 2B)
HPSI hot leg interface	Failure of two pressure i adation obeck valves and stroke test of normally closed MOV, plus failure of safety injection pump discharge check valve	Only hardware failures considered (SEQs-4A and 4B)
SDC suction	Failure of two check valves	Sequence initiated during normal shutdown (SEQ-5)

Yabie 1. Lia of potential ISLOCA scenarios.

3.2.3 LPSI Cold Leg Injection Lines to

FICS. Through the normal reactor operating year, MOVs S1-138 (A/B) and S1-139 (A/B) are stroke tested quarterly. Thus, the accident sequence for the LPS1 pump discharge is based upon the fact that the MOVs will be opened once each quarter.

3.2.4 HP91 Cold Leg Interface. These scenazios are similar to Sequence 2. Once each quarter, MOVo S1-225 through S1-228 (A/B) are stroke-tested while the plant is operating at neuroal power. Thus, the accident sequence path for the HPS1 pump cold leg discharge is based upon the flict that the MOV's will be opened once each guarter. Note that Sequence 3B is similar to Sequence 3A, but there is one less clicck value to protect the low-pressure portions of the S1 system.

3.2.5 HPSI Hot Leg ..., **ce.** Once every quarter, MOVs SI-502A and SI-506A are stroke tested. Therefore, Sequence 4A is based on the opening of MOV SI-502A. Since valve SI-502A is opened and closed beform valve SI-506A is opened, the opening of SI-502A is defined as the induming event for the sequence. Once again, the assumption of no prior knowledge of the condition of the system is used. Sequence 4B is similar to Sequence 4A except that check valve SI-216 is missing from piping header B.

3.2.6 SDC Suction Lines During Normal Skutdown. When the CE plant enters the shutdown mode, the operators rely on check valves SI-108 and SI-1071 closing when the RCS pressure exceeds the interfacing system design pressure. Thus, Sequence 5 is based upon failure of the two check valves.

Because of the unique neture of the ISLOCA sequence, a detailed understanding of the capabilities of the plant hardware and personnel is needed to accurately analyze the ISLOCA challenge. For this report, an ISLOCA is considered to involve a loss of reactor coolant outside containment. Since the supply of water available for nyakeup to the RCS is essentially limited to the available inventory in the RWSP, a high-priority item for the control room operators should be to isolate the break expeditiously and terminate the loss of reactor coolant. If the break were isolated in a timely manner and the loss of RCS inventory terminated, the plant could be cooled down safely using the auxiliary feedwater (AFW) system (secondary cooldown) or SDC (primary

Before discussing the detailed results, some general continents can be made that are applicable to all the postulated ISLOCA scenarios. During the course of the plant visit, particular attention was paid to the issue of local environ-facing systems, so the time of Ce plant visit, the probabilistic system rupture calculations had not been completed, so a general survey was made of the interfacing system flow paths to qualitatively estimate the impact of ruptures on equipment in various locations. This s. vey included walkdowns of the emergency core cooling systems (ECCS) to examine likely break locations. For example, the assumption was made for this analysis that all equipment in the compartment where a break occurs will be rendered unavailable for use in isolating/mitigating the ISLOCA. Therefore, equipment in compartments judged to be candidate locations for an ISLOCA break was inventoried. This survey could not verify that the ECCS are adequately separated such that any postulated rupture would not affect redundant ECCS trains. If there were a piping break and blowdown of steam from the RCS into one of the safeguards pump rooms, the plant configuration may not ensure that at least one train of ECCS would still be available after a rupture had occurred. However, we stress that these conclusions are not

based on mechanistic heat and mass transfer calculations and are therefore qualitative in nature. This issue is still under investigation at the INEL.

4.1 Event Trees

The following sections describe the event trees developed for the postulated ISLOCA scenarios. The event trees are quantified on a yearly or quarterly basis, as reflected in the frequency of the initiating event. The event trees are constructed such that the downward branch depic ts the failure event listed at the top of the event tree and the upward branch denotes the complement of the event (i.e., typically success). The top events are a combination of individual component failures, human errors, and functional failures that describe the progression of the ISLOCA from initiation to core damage or recovery.

All event tree qu. "ification is performed using mean failure probabilities. The derivations of the event tree split fractions are presented in Appendices B and C. Note that detailed (i.e., nonscreening) failure probabilities were calculated only for Sequences 2 and 5, all of the other sequences had CDFs <10 8 /year in the screening analysis and were not developed further.

Finally, each event tree end state was assigned to one of the source term bins listed below:

- OK—No overpressurization of the lowpressure system occurred (no fission product release).
- OK-op—Scenario results in overpressurization of the interfacing system but the system does not rupture or leak (no fission product release).
- LK-ncd—Scenario results in RCS leakage from the interfacing system, through either a break or an open relief valve, but severe core damage (sufficient to cause offsite health effects) does not occur because the leak is either isolated before core uncovery or the leak is t. o small to interfere with core cooling (no fission product clease).

- LOCA-ic—Identifies scenarios that produce a loss-of-coolant accident (LOCA) inside containment Because these sequences are enveloped by the design basis analysis of the plant, they are not fully developed on the event trees and are not considered to be core damage events for the purposes of this analysis.
- PEL-mit—An ISLOCA with core damage occurs but the radioactive fission product release is mitigated through some means, such as scrubbing through an overlying water rool or general area fire sprays in the acciliary building.
- REL-Ig—An ISLOCA with core damage occurs and results in a large unmitigated radioactive release. Note that this does not necessarily mean that the break size is large.

4.1.1 Premature Entry Into Shutdown Cooling--SEQ-1A. A risk-significant scenario at the Babcock and Wilcox plant5 involved premature entry into shutdown cooling, with RCS pressure and temperature above the open permissive set point of the decay heat removal (DHR) system suction isolation valves. This scenario was considered credible at the Babcock and Wilcox plant because the plant procedures allowed operators to bypass the open permissive interlock for one of the twe ... itdown cooling isolation valves. This allowed an error of commission to be postulated in which, or a decision is made to enter shutdown coc and and the the other valve, also, even though the procedure does not instruct them to do so. For the CE plant, the HRA did not reveal any circumstances that would lead to an analogous scenario. Therefore, this scenario was not developed further.

4.1.2 Shutdown Cooling System/Reactor Coolant System ISLOCA During Startup-

SEQ-1B. Figure 7 shows the event tree used to model an intersystem LOCA between the RCS and SDC system during startup. The SDC suction isolation valves from the RCS are open initially because the SDC system is being used to remove

decay heat from the reactor. Since startup is a "low-pressure" procedure compared to normal full RCS pressure, it is assumed that any overpressurization that causes the relief valve to open will not cause an ISLOCA. The event tree models one flow line (out of two) on a mission time of one year.

4.1.3 RCS To LPSI Cold Leg Discharge— **SEQ-2.** Through the normal reactor operating year, MOV's SI-138 (A/B) and SI-139 (A/B) are stroke-tested quarterly. Thus, the accident sequence path for the LPSI pump discharge is based upon the fact that the MOV's will be opened .nce each quarter. Figure 8 st wis the event tree used to model this sequence.

Obviously, if the two isolation check valves (SI-335/336 (A/B) and SI-142/143 (A/B)] protecting the MOVs had failed, it would not be desirable to open the MOVs. But, for analyzing this sequence, it is assumed that no prior information (for example, a high-pressure reading on the pressure indicator between the two isolation check valves) is known for the system. This assumption is made because the stroke-testing procedure does not direct the operators to check pressure between the PIV check valves before performing the stroke test. In addition, the annunciator card for this pressure indicator was found to be deactivated during the plant inspection,11 so no credit was given for this annunciator. Therefore, for the model, it is postulated that internal failure of the two isolation check valves will automatically lead to an overpressurization of the interfacing system when the MOV is stroke tested. Note that the event tree evaluates one flow path (out of four possible) for a mission time of one quarter. Thus, to get the failure frequency for the complete system based on a one-year mission time, the sequence end state frequencies must be multiplied by 16.

4.1.4 RCS Cold Legs to HPSI (Header A)— SEQ-3A. This scenario is similar to Sequence 2. Once each quarter, MOVs SI-225 through SI-228 (A/B) are stroke tested while the plant is operating. Thus, the accident sequence path for the HPSI pump discharge is based upon



Figure 7. SDC system ISLOCA during startup (SEQ-1B).



Figure 8. Event tree for LPSI cold leg discharge to RCS (SEQ-2).

NUREG/CR-5745

19

Results

the fact that the MOVs will be opened once each quarter.

Once again, if the isolation check valves protecting the MOVs had failed, it would not be desirable to open the MOVs. But, for this analysis, no prior knowledge for the system is assumed (see discussion in Section 4.1.3). Thus, for the model, random failure of the two isolation check valves is assumed to lead automatically to a demand on check valve SI-216. Figure 9 shows the event tree fc⁺ this sequence.

4.1.5 RCS Cold Legs to HPSI (Header B)— **SEQ-3B.** Sequence 3B is comparable to Sequence 3A with the exception that Sequence 3B has one less check valve to protect the interfacing system. Whereas header A has check valve SI-216, header B does not have the corresponding check valve in the piping design. Figure 10 shows the event tree for this sequence.

4.1.6 RCS Hot Legs to HPSI (Header A)— SEQ-4A. Once every quarter MOVs SI-502A and SI-506A are stroke-tested. Therefore, Sequence 4A is based on the opening of MOV SI-502A. Since valve SI-502A is opened and closed before valve SI-506A is opened, the opening of SI-502A is defined as the initiating event for the sequence. Once again, the assumption of no prior knowledge of the condition of the system is used. Figure 11 shows the event tree for this sequence.

4.1.7 RCS Hot Legs to HPSI (Header B)— **SEQ-4B.** Sequen: 4B is similar to Sequence 4A except that check valve SI-216 is missing from piping header B. The initiating event for Sequence 4B is the opening of MOV SI-502B. The initiating event probability is identical to that of Sequence 4A and is assumed to be 1.0. Figure 12 shows the event tree for this sequence.

4.1.8 RCS to LPSI During Shutdown— SEQ-5. When the analyzed plant enters shutdown cooling, the operators rely on check valves SI-108 and SI-1071 closing when the RCS pressure exceeds the interfacing system design pressure. Thus, Sequence 5 is based upon failure of the two check valves. Figure 13 shows the event tree for this sequence.

4.2 Human Reliability Analysis

This section summarizes the results of the ISLOCA HRA efforts. Appendix C provides detailed information regarding HRA fault trees, event trees, tabulated HEP values, and discussions of the HRA process. The HEPs presented as part of the HRA are estimates based upon the best contemporary models and quantitative techniques. As in any HRA, these HEPs must be considered in light of hard ware failure information contained within this report and should not be used in isolation.

HRA was used to model the predominant human errors for each significant scenario in the ISLOCA PRA. As discussed in Section 2.5. HRA is a methodological tool that involves the quantitative analysis, prediction, and evaluation of work-oriented human performance. The ISLOCA HRA diagnosed those factors within the plant's systems that could lead to less than optimal human performance in the initiation, detection, diagnosis, and mitigation of ISLOCA scenarios. HRA was used as a diagnostic tool to isolate the error rate anticipated for individual tasks and to determine where errors were likely to be most frequent.

Because most of the human actions in this HRA involved the use of various written normal, abnormal, and emergency operating procedures, THERP-type HRA event trees were chosen for modeling most of the human actions in the detailed analysis. However, in several ISLOCA scenarios, HRA fault trees were used in conjunction with the typical THERP event trees to provide the best representation of the modeled events. Detailed analyses were conducted using the fault trees and/or THERP event trees to estimate the error probabilities and uncertainty ranges of the dominant human actions.

Individese error branches for each of the HRA event trees (see Section 2.5 or Appendix C for details) were quantified using techniques from

CLARKS		õ	ő	ð	a		1	deli-14	100	5 - 19U	1.4	101 - 138	
8004 039		0.005-50	2 346-07	4+ 385 S	a saesaa	0.006+30	0. 008 - CH	9	0.055+00	8 426 - 24	2 - 205 + 30	1.246-13	
Release is not not.	Rines.							1.002+30		1000-000		1 205-20	
50 81 81 01 8 6 81 5 5 0 1 50 - 81 8 1 50 - 81 8	1.8.4						* 305-20						
Operatora 1811 to 1810 to 15000 h	FIDGH												
Operators fair 10 detect 15LDCA	570												
toterfacing System Ruptures (tr)	1583									* 952×30			
Check Valve Si 227A Tails to close	CHIVEN						1.405.03						
Check Value 51 216 faris to close	CHWE3					1.030.03							
0.5 8 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	0d			1 006+00									
CHBER Valves S: 3351336 abd 51-245	CHWP.2		2 562 07										

Figure 9. Event tree for HPSI header A cold leg discharge to RCS (SEQ-3A).

NUREG/CR-5745

Results

Results



Figure 10. Event tree for HPSI header B discharge to RCS cold legs (SEQ-3B).





23

Results



Figure 12. Event tree for HPSI header B discharge to RCS hot legs (SEQ-4B).


25

THERP and HCR. Specific human actions on each error branch were assigned an estimate of the basic HEP. These basic HEP estimates were then modified using PSFs to realistically describe the work process at the plant. Finally, possible failure paths (i.e., sequences that included either single or multiple human errors leading to a failure of the action modeled by the HRA tree) were identified and combined to estimate the total failure probability for the HRA tree, in accordance with the THERP guidelines. Individual PSFs were derived from task analyses, time-line analyses, evaluation of the hum_n/machine interface. and direct observations of operator performance. The majority of these PSFs were presented in the ISLOCA Inspection Report for the analyzed plant.11 Each PSF was seen as casting either a positive or negative influence on the basic HEP. that is, as either decreasing or increasing the probability of failure for a given human action. For example, some of the positive PSFs in evaluations of the CE plant included the following.

- "The team did not identify any significant deficiencies in the man-machine interface that might significantly increase the probability of an operator error initiating an ISLOCA."¹¹
- "The team found emergency operating procedures to be well written although they lacked some human factors considerations (see the second item in the negative PSFs listed below).¹¹
- "Although training specific to ISLOCAs was not part of the licensee's training program, operators indicated, during walkthroughs and simulator exercises, that they were generally well prepared to cope with losses of RCS inventory."¹¹

Examples of negative PSFs were

 "The team identified weaknesses in the man-machine interface that could adversely affect the ability of the operators to mitigate an ISLOCA because of poor equipment labeling and the inaccessibility of some equipment."¹¹

- Even though emergency operating procedures were generally well written, the RCS leak procedure, OP-902-002, does not provide relevant guidance with respect to requisite a tont for the isolation of ISLOCAs. As a esult, operators and supervisory personnel would be required to rely on knowledge-based actions, outside of normal procedures.
- Within the context of the prior finding operator training (based on Three Mile Island scenarios) emphasized that operators should not override a safety injection signal occurring in conjunction with an unisolated RCS leak (see Sequence 2). This training could lead control room personnel away from the actions necessary in Sequence 2 to isolate a break in the safety injection lines (e.g., operators would have to sequentially close each HPSI and LPSI safety injection valve on the affected SI train).
- Operators' ISLOCA diagnostic abilities were centered on Attachment 1 of GP-962-0C2, which verifies a LOCA outside containment but directs operators to a procedure (OP-962-002) that does not provide relevant guidance for the isolation of an ISLOCA.

A detailed HRA was conducted for each of the significant scenarios identified in the ISLOCA PRA. Tables 2 and 3 summarize the results of these analyses, which are extensively described in Appendix C. These tables provide the identifier and description for each significant human error, as well as the mean HEPs.

Inspection of these tables reveals that HEPs increase with time following an interfacing system rupture. These increasing error rates reflect the following circumstances identified for the CE plant. First, procedures may not effectively lead operators to the control room indications that are most relevant for detection of an ISLOCA, and do not provide definitive guidance for necessary and sufficient actions to isolate an ISLOCA in the two sequences that were modeled in detail.

Identifier	Human error	Mean HEP
FTD-LOCA	Control room fails to detect LOCA	0.018
FTDGN	Control room fails to c .agn. 2 ISLOCA	0.02
FTI	Control room fails to isolate ISLOCA	1.00

Table 2. Estimated mean HEPs for Sequence 2

Toble 2	Detimated	mean L	IEDe fo	r Secu	ance 5
Table 5.	Commanco	HICHU I	1471 3 10	a mode	POTTOR ALL ALL

Identifier	Human error	Mean HEP
FTD	Operators fail to detect loss-of-coolant	0,0076
FTDGN	Operators fail to diagnose system leakage	0,0076
FTI-A	Fail to isolate (1 SDC in service)	0.0233
FTI-B	Fail to isolate (both SDC in service)	0.0233

Second, diagnostic abilities in the control room (e.g., procedures and training) rely on the diagnostic flow chart in Attachment 1 of OP-902-002, the RCS leak procedure. That flow chart can successfully diagnose a LOCA outside of containment, but also directs operators to use OP-902-002, with the drawbacks mentioned above. Therefore, operator workload is increased and significant stress (threat level) is likely to be experienced by the operators at the time when ISLOCA isolation actions are required.

4.3 Quantification of ISLOCA Model

Based on the event trees described in Section 4.1 (and in more detail in Appendix B), the total mean ISLOCA CDF for the plant is estimated to be 2.0×10^{-6} per reactor-year of operation. Table 4 provides a breakdown of this frequency by sequence and release category. The dominant scenario is hardware dominated, involving failure of the pressure isolation check valves in the LPSI cold leg discharge to the RCS (SEQ-2). This scenario is equivalent to the classical event-V category of core damage sequences that has been examined in some past PRAs. Note, however, that this sequence could be eliminated by modifying the stroke test procedure to require the operators to check for pressurization of the header between the discharge check valves prior to performing the stroke test. The relative insignificance of the human error-initiated sequences is due to the excellent administrative controls and safety culture present at the plant, for example, the practice of not defeating (jumpering out) equipment interlocks during normal operations and the tight control of keys needed to restore power to isolation valves.

The likely failure locations for Sequence 2 are the schedule 40 piping and the 10-inch, 300-psi flange at the discharge flow element. The flange failure probability was relatively high (0.69) because of the "soft" SA 193-B8 bolts that are used. Upgrading these bolts to SA 564 grade 630 would eliminate flange failure from consideration in this sequence. [As discussed in Appendix F, the probability of flange failure was partitioned into small leaks and large failures. The large failure probability of 0.12 was used in calculating the split fraction used in the event tree for this

Scenario	CDF	REL-lg	REL-mit	LOCA-ic	LK-ncd	OK-op
1A	£ ⁸	£	0.0	£	£	E
IB	ε	ε	0.0	E	E	ε
2	2.0E - 06	2.0E-06	0.0	0.0	1.4E-06	7.8E-07
3A	ε	ε	0.0	0.0	ε	0.0
3B	ε	ε	0.0	0.0		0.0
4A	٤	ε	0.0	0.0		0.0
4B	٤	٤	0.0	0.0	- e	0.0
5	ε	ε	0.0	0.0	9.6E - 08	1.0E - 03
Totals	2.0E - 06	2.0E - 06	0.0	E	1.5E-06	1.0E - 03
a. ε <10 ⁻⁸ /y	ear.					

Table 4. ISLOCA CDF (per reactor-year).

sequence. Small leaks were judged to be recoverable by the operator (see Appendix C for more details) and were binned into the LK-acd end state on the event tree.]

Sequence 5 was the major contributor to ISLOCA risk in the initial screening analysis because the frequency of pressurizing the lowpressure system beyond its design pressure was approximately 10-3/year; however, detailed analysis of the component pressure fragility showed that small flange leaks are the only credible overpressure failure modes for this sequence. and the probability of even these small leaks is extremely small (<10-4/fiange). The components contributing to the rupture probability are the 150-psi flanges at check valves SI-107 (suction from RWSP) and SI-407 (suction from containment sump). (Refer to Appendix H for the details or the component pressure fragility analysis.) This sequence would also appear to be driven by hardware failures; however, the hardware failure

of concern is the demand failure of check valves SI-108 A and B. A demand failure probability of 1.0 was assumed for these valves, based on their as-found condition at the plant and the complete lack of testing or maintenance on these valves at the time of the inspection. Were these valves to receive regular leak-testing and some form of periodic maintenance (e.g., disassembly to inspect for boric acid precipitation and corrosion), a generic demand failure probability of 10-3 could be justified. The Impeli analysis of these flanges also showed that the leak rates would be far too small to threaten core cooling. However, the HRA for this sequence was done before these results were available, so operator response to a small break was modeled. Even with this additional conservatism, the CDF from this sequence is <10⁻⁸/year. Had the flange failure probability been higher, the probability that the failure results in a leak large enough to threaten core cooling would have had to be factored into the CDF calculation.

4.4 Risk Assessment

As described in Section 2.6, the offsite consequences of ISLOCA core damage sequences were estimated using the V-se-uence source term from the June 1989, NUREG-1150 analysis of Sequoyah (see Appendix G for details of the consequence analysis). The conditional consequences for the base case analysis are listed in Table 5. Based on information from the NUREG-1150 program that estimated decontamination factors (DFs) for both dry and wet containment bypass releases, a DF of 1 (no decontamination) is assumed for the release from the auxiliary building (large dry release) in the base case. Additional work on estimating DFs for the auxiliary building has been sponsored by the Electric Power Research Institute (EPRI) u. ing the Modular Accident Analysis Program (MAAP) code.ª This work would seem to support DFs for a dry release in the range of 3 to 80. depending on the specific configuration of the auxiliary building. Wet release DFs, either due to a flooded break location or scrubbing by general area fire sprays in the auxiliary building, ranged from 40 on up. However, the MAAP code, when the core flow blockage feature is used (as it was for the EPRI work), tends to predict relatively little hydrogen generation in-vessel. Hydrogen generated in-vessel, it released into the auxiliary building, could burn, potentially opening pathways for free convective exchange with the outside environment, thus reducing the effective auxiliary building DF. The effect of a credible range of E/Fs on the offsite consequences is examined in Section 4.5.

When reviewing the ISLOCA consequence and risk estimates, several aspects of this calculation should be kept in mind. Many measures of risk are available and have been used in recent studies. However, to produce these estimates, many sequence-specific and site-specific assumptions must be made, from the cost of land to the warning time available to activate the offsite emergency response plan before a catease recurs. These assumptions can have significant effects on the consequences calculated with MACCS. The base case ISLOCA risks are shown in Table 6.

4.5 Uncertainty and Sensitivity Study Results

No uncertainty analysis was performed for the dominant ISLOCA scenarios because the bestestimate CDF is relatively low and almost all of the uncertainty is contained in the initiating event of Sequence 2, failure of the two series check valves. The error factor on this initiating event is 100; therefore, the CDF distribution will have a

Mean early fatalities	Mean latent fatalities	(person-rem)
0.00E + 1	5.36E+3	6.12E+6

Table 5. Base case ISLOCA consequences conditional upon severe core damage.

		Mean 50-mi dose	
Mean early fatalities	Mean latent fatalities	(person-rem)	
2.0E-4	1.1E-2	1.22E + 1	

a. Electric Pov cr Research Insuitute, Evaluation of the Consequences of Containment Bypass Scenarios, EPRI-NP-6586-L, Not ember 1989. This report contains proprietary information that is not available to the general public; however, the results of this study were made available to the INEL analysts for review.

large, positive skewness coefficient, indicating that the reported mean CDF will be close to the 95th percentile value.

Based on similar results obtained for the Westinghouse plant,¹⁶ the uncertainty in CDF should span approximately four orders of machitude.

4.5.1 Component Rupture Pressure Uncertainty. The base case analysis used a logarithmic standard deviation of 0.36 for the pipe rupture pressure distribution (which was modeled as lognormal). As discussed in Appendix H, this value is derived by assuming that the probability of component failure is 10⁻³ when applied stress equals yield stress. This may be an overly conservative assumption; however, consitivity cases were examined in References 5 and 16 in which the probability of component fail 'e at yield stress was taken to be 10-4 and 10-5. These values correspond to a logarithmic standard deviation for the pipe rupture pressure of 0.30 and 0.26. respectively. The rupture probabilities were recalculated, with the result that there was not a significant effect on CDF. Because the piping materials are the same in the CE plant as in the Babcock and Wilcox and Westinghouse plants, this result should apply to the CE plant, also. Therefore, no detailed calculations were performed.

4.5.2 Auxiliary Building DF Uncertainty.

Uncertainty due to lack of knowledge was treated via a sensitivity analysis that examined the effects of a range of credible auxiliary building DFs on fission product source terms and offsite consequences. The details of this analysis can be found in Appendix G. The important aspects and results of this analysis are summarized below.

For the dry ISLOCA sequences, the base case DF for all release classes had a uniform value of 1 and the release was at ground-level. The sensitivity analysis involved calculating new fission product source terms with uniform-valued DFs of 5, 10, 50, and 100 for all release classes except the noble gases, for which the base case DF of 1 was retained For the first two calculations, with DFs of 5 and 10, a ground-level release was assumed. For the last two calculations, with DFs of 50 and 100, the release elevation was specified as 10 m. The reason for the change in release elevation is that industry-sponsored analyses of auxiliary building DFs show that higher DFs, in the absence of water-scrubbing, are produced when there is a lack of free convective exchange with the outside environment. This generally corresponds to having an opening high up in the auxiliary building, with a release from the reactor coolant system in the lower elevations of the building.

Sensitivity cases also were run to examine the potential effects of auxiliary bailding fire sprays on the release (there are no such sprays at the plant analyzed in this report). In these so-called wet ISLOCA sequences, the base case DF was specified as a distribution that was sampled upon, as in the NUREG-1150 analysis of Sequoyah. The distribution is shown in Table 7 for the base

 Table 7.
 Distribution of DF for the base case

 analysis of the wet ISLOCA sequences as
 specified in the SEQSOR input,

Distribution of DF (%)	DF
0	5.1E+03
1	4.5E ± 03
5	$4.1E \pm 0.3$
25	$1.3E \pm 0.2$
50	6.2E + 00
75	3 0E + 00
95	$1.8E \pm 00$
99	$1.7E \pm 00$
100	$1.6E \pm 00$

case analysis of the wet sequences. As was done for the dry sequences, sensitivity analyses were performed with uniform-valued DFs of 5, 10, 50, and 100 for all release classes except the noble gases, for which the DF remained at the base case value of 1. For the first two sensitivity calculations, with DFs of 5 and 10, a ground-level release was assumed. For the calculations with DFs of 50 and 100, calculations were performed with the release elevation specified both at ground level and 10 m.

Tables 8 and 9 present the mean conditional consequence results for the dry and wet ISLOCA sequence sensitivity cases, respectively. These results are also presented graphically in Figures 14 through 19. Comparisons of the dry and wet sequence consequence results are also presented graphically in Figures 20 through 22.

Sens	itivity Case					
DF	Release elevation	Mean early fatalities	Mean latent fatalities	Mean 50-mi dose (person-rem)		
1	0.0	9,99E+01	5.36E+03	6.12E+06		
5	0.0	1,63E+00	1.82E + 03	$2.15E \pm 06$		
10	0,0	4.03E-01	1.17E + 03	1.55E+06		
.50	10.0	7.12E 02	5.16E+02	8.83E + 05		
100	10.0	6.10E-02	$3.97E \pm 02$	7.24E + 05		

Table 0	A. S	AATTO.	the second for the second state in the	and the second second	in and de	A DO 191	a summariant	constitutity
lable o.	Mean A	NACTO	consequence	TCSUITS I	on onen en y	131444	seducure	sentenninit.

Table 9. Mean MACC	S consequence results fo	r each wet ISLOCA se	quence sensitivity.
--------------------	--------------------------	----------------------	---------------------

Sensitivity case					
DF	Release	Mean early fatalities	Mean latent fatalities	Mean 50-mi dose (person-rem)	
Base ^a	0.0	$3.69E \pm 00$	1.71E + 03	$2.08E \pm 06$	
5	0.0	9.92E - 01	$1.49E \pm 0.3$	1.79E + 06	
10	0.0	4.01E - 01	9.85E + 02	1.37E + 06	
50	0.0	1.76E-01	4.43E + 02	7.78E + 05	
50	10,0	$1.18 \text{E} \simeq 01$	4.51E+02	$7.86E \pm 05$	
100	0.0	1.59E-01	3.51E+02	$6.40 \text{E} \pm 0.5$	
100	10.0	1.06E - 01	3.56E + 02	$6.46E \pm 05$	

a. Wet sequence base case DF is a sampled distribution as given in Table 7.



NUR2:G/CR-5745

32





Mevn 30-mi population dose consequence results as a function of DF for the dry ISLOCA sequences.



35









37



NUREG/CR-5745



NUREG/CR-5745

39





5. CONCLUSIONS

The methodology for evaluating ISLOCA risk that was developed in Assessment of ISLOCA Risks-Methodology and Application to a Babcock and Wilcox Plant5 has been applied to a CE plant with a dry, atmospheric-pressure containment. This methodology has been successful in providing insights regarding the relative contributions of both hardware faults and human actions to ISLOCA CDF. The results indicate that human errors of commission, latent faults of equipment, and normal procedural tasks can combine in an ISLOCA sequence. However, the methodology also was used to identify potential means of reducing these contributions to risk. Conclusions are presented below, followed by a preliminary discussion of the relationship of these results to the general population of NPPs.

5.1 Plant-Specific Conclusions

A publicly available PRA of the analyzed plant has not yet been completed by the licensee. Therefore, no comparison could be made of the results of this analysis with results obtained by the licensee.

In the pressure fragility analysis of the interfacing systems, existing relief valves were found to provide very little protection against the dominant ISLOCA initiator (SEQ-2). Typically, relief valves in the interfacing systems are designed to mitigate the occasional pressure transient associated with routine valve realignments and pump starts and stops. The pressures generated in ISLOCA events simply overwhelm the relatively small relief capacity of these valves.

The Idaho National Engineering Laboratory HRA found that operator error could contribute to ISLOCA initiators. However, risk-significant human error initiators were judged unlikely during operations involving interfacing systems. This is due, in part, to the existence of significant administrative procedures and related operator training, as well as the presence of wellcontrolled interlocks that prevent the inadvertent operation of those pressure isolation valves under operator control. The HRA found a higher probability for operator error during detection, diagnosis, and isolation of an ISLOCA. This results from an interaction of the following variables:

- A limited number of clear control room indications for ISLOCA
- Emergency procedures that do not address isolation of a break outside containment
- Limited amount of time in the dominant scenario for the detection, diagnosis, and isolation of ISLOCA before core uncovery
- Operator exposare to high workload and threat stress at the time when actions to isolate an ISLOCA are needed.

5.2 General Conclusions

Extreme caution should be exercised when attempting to extrapolate the results of a single analysis to estimate the performance of the entire commercial nuclear power industry. The analysis of the CE plant in this report has identified some potential ISLOCA issues, but the completeness and typicality of the results, even for other CE plants, has not been determined. The analysis of this plant indicates that the most important concern regarding ISLOCA risk centers on the lack of procedural guidance for responding to an ISLOCA, rather than on the plant personnel. However, it is imprudent to conclude that human errors, while not important ISLOCA initiators at the plant analyzed in this report, will not dominate the ISLOCA risk for other plants. Therefore, a major emphasis in any evaluation of ISLOCA should be the assessment of the potential for human error initiators. Specifically, this involves judging the adequacy of plant procedures and personnel training and awareness of the potential for and consequences of an ISLOC7. To generalize, the plant personnel's understanding of the importance of maintaining the pressure isolation

Conclusions

boundary, and recognizing the potential for an ISLOCA and its consequences, can have a dramatic impact on ISLOCA risk. In the case of the CE plant analyzed in this report, the detailed HRA considering these effects eliminated human

error-initiated ISLOCAs from detailed consideration, whereas at the Babcock and Wilcox plant,⁵ the effect was just the opposite; the dominant ISLOCA sequence was initiated by a human error of commission.

6. REFERENCES

- U.S. Nuclear Regulatory Commission. Reactor Safety Study—An Assessment of Accident Risks in U.S. Commercial Nuclear Power Plants, WASH-1400, October 1975.
- U.S. Nuclear Regulatory Commission, Severe Accident Risks: An Assessment for Five U.S. Nuclear Power Plants, NUREG-1150, June 1989.
- G. Bozokiet, al., Interfacing Systems LOCA: Pressurized Water Reactors, NUREG/CR-5102, BNL-NUREG-52135, February 1989.
- T-L. Chu, S. Stoyanov, and R. Fitzpatrick, Interfacing Systems LOCA: Boiling Water Reactors, NUREG/CR-5124, BNL-NUREG-52141, EGG-2608, February 1989.
- W. J. Galyean and D. I. Gertman. Assessment of ISLOCA Risks—Methodology and Application to a Babcock and Wilcer Nuclear Power Plant, NUREG/CR-5604, EGG-2608, April 1992.
- T. G. Ryan, "A Task Analysis Linked Approach for Integrating the Human Factor in Reliability Assessment of NPP," *Reliability Engineering and System Safety*, 22, 1988.
- A. Swain and H. Gutman, Handbook of Human Reliability Analysis with Emphasis on Nuclear Power Plant Applications, NUREG/CR-1278, August 1983.
- G. Hannaman and A. Spurgin, Systematic Human Action Reliability Procedure (SHARP), EPRI NP-3583, Electric Power Research Institute, 1984.
- Institute of Electrical and Electronics Engineers, A Guide for General Principles of Human Action Reliability Analysis for Nuclear Power Generation Stations, IEEE Draft Standard P1082/D7, August 1989.
- G. W. Hannaman, A. Spurgin, and Y. Lukic, Human Cognitive Reliability Model for PRA Analysis, NUS-4531, Electric Power Research Institute, 1984.
- J. Ball, ISLOCA Inspection Report 50-382/90-200, U.S. Nuclear Regulatory Commission, Washington D.C., September 14, 1990.
- 12. Science Applications International Corporation, ETA-II, Version 2.0, Los Altos, California, 1990.
- 13 D. I. Chanin, et al., MELCOR Accident Consequence Code System (MACCS), NUREG/CR-4691, February 1990.
- J. Gregory, et al., Evaluation of Severe Accident Risks: Sequoyah Unit 1, NUREG/CR-4551, Vol. 5, Rev. 1, draft report for comment, June 1990 (available from Public Document Room Washington D.C.).
- D. C. Aldrich, et al., Technical Guidance for Siting Criteria Development, NUREG/CR-2239, SAND81-1549, 1982.
- D. L. Kelly, J. L. Auflick, and L. N. Haney, Assessment of ISLOCA Risks—Methodology and Application to a Westinghouse Four-Loop Ice Condenser Plant, NUREG/CR-5744, EGG-2649, April 1992.

Appendix A

System Descriptions

Appendix A System Descriptions

The plant analyzed for this report is a single-unit site. The unit is a 3390 Mwt pressurized water reactor (PWR), with an NSSS supplied by Combuation Engineering (CE). The unit has a large dry containment that is maintained at atmospheric pressure and a separate reactor auxiliary building and turbine building.

The plant is similar to other CE plants in the number and type of charging and safety injection pumps.

A.1 Reactor Coolant System

The reactor coolant system (RCS) transfers energy from the reactor core to the secondary water in the steam generators. The RCS pressure boundary acts as a barrier (one of several) against the uncontrolled release of radicactive material from the reactor core and primary coolant.

During power operation, primary coolant in the RCS is circulated by one reactor coolant cump in each of the four cold legs. Pressure is maintained within a prescribed band by the combined action of the pressurizer heaters and sprays. RCS inventory is maintained within a prescribed band by the chemical and volume control system (CVCS), otherwise known as the charging system.

Component Information

Α,	RCS 1. 2.	Volume: 10,300 ft ³ excluding pressurizer and surge line Nominal operating pressure: 2235 psig	
Β.	Steam 1. 2.	generators (2) Type: vertical shell and U-tube Model: CE	

A.2 Interfacing Systems

All interfacing systems were screened to identify those systems that needed further evaluation. The criterion used in screening was that any system with an interfacing pipe diameter larger than one inch should be evaluated. The one-inch pipe size was selected based on an estimation of the discharge from a one-inch high pressure pipe break, which was about 200 gpm. A 200-gpm leak rate outside of the containment is considered to be critical based on: the capacity of the RWSP (Technical Specification minimum volume of 443,000 gal), the capacity of three charging pumps (132 gpm), and the normal makeup rate to the RWSP (-150 gpm). Based on these considerations and the number of hours it would take for the plant to achieve cold shutdown (conservatively assumed to be about 10 hours), leak rates of 200 gpm or less were judged not to be risk-significant.

The initial screening resulted in the selection of the safety injection (SI) system, including the low pressure safety injection system, and the shutdown cooling system. Figures A.1 through A.6 are schematic diagrams showing the hardware configuration of the safety injection system.



Schematic.



A-6





A-7





Figure A.F Shutdown Cooling Suction Line Schematic.



Figure A.6 RCS Hot Legs to RWSP via the LPS1 System

A.2.1 Safety Injection System

The safety injection system provides high and low pressure coolant injection capability, as well as the ability to remove residual heat from the reactor core when the plant is shutdown and at low pressures. There are two low pressure safety injection system trains, each with one safety injection pump, and two high pressure trains with three pumps. The safety injection pumps are normally aligned for cold leg injection (to all four RCS cold legs). but are capable of supplying flow to the hot legs, also. The pumps start automatically upon receipt of a safety injection actuation signal. During injection, the RWSP supplies borated water which the safety injection pumps deliver to the cold legs via a common discharge header that branches into four lines, one for each RCS loop.

The low pressure safety injection pumps also provide the motive force for shutdown cooling flow. In this mode of operation they take suction from the RCS hot legs and discharge to the cold legs through the shutdown cooling heat exchangers.

The safety injection system also contains four cold leg safety injection tanks (SITs). Each SIT contains borated water with a pressurized cover gas. The borated water is forced into the respective cold legs when RCS pressure decreases below the cover gas pressure.

Testing of the safety injection system is specified in the plant Technical Specifications and in the in-service testing (IST) program. The pumps are flow-tested on a quarterly basis. The normally closed discharge MOVs are stroke-tested quarterly. Functional actuation tests of the safety injection system are performed during cold shutdown.

Table A.1 Low Pressure Safety Injection Pump Data

Туре	Single-stage, vertical, centrifugal
Design pressure	650 psig

Design temperature	400° F.
Design flow	4050 gpm
Maximum flow	5500 gpm
Design head	342 ft
Head at maximum flow	265 ft

.

Table 4.2 High Pressure Safety Injection Pump Data

Туре	Relti-stage, horizontal, centrifugal
Design pressure	1950 psig
Design temperature	400° F.
Design flow	380 gpm
Maximum flow	910 gpm
Design head	2830 ft
Head at maximum flow	1275 ft

Appendix B

ISLOCA Event Trees

Appendix B ISLOCA Event Trees

This appendix describes in detail the ISLOCA scenarios developed for the CE plant and the event trees used to calculate the core damage frequency. The event trees are quantified on a yearly basis. The downward branch at each node depicts the failure event listed at the top of the event tree and the upward branch denotes the complement of the event (typically success). The top events represent a combination of individual component failures, human errors, and functional failures that describe the ISLOCA progression.

All event tree quantification is performed using mean failure probabilities. An uncertainty analysis of core damage frequency is presented in Section 4.5 of the main report. In addition, sensitivity studies were performed for selected issues that are believed to dominate the risk or about which there is significant uncertainty.

Note that only screening failure probabilities are shown for Sequences 1B, 3A, 3B, 4A, and 4B. Sequences 2 and 5 were analyzed in detail, so the failure probabilities shown in the event trees for these sequences are the mean values calculated in the detailed analysis.

Finally, each event tree end state was assigned to one of the consequence bins listed below.

OK - No overpressurization of the low pressure system occurred.

OK-op - Scenario results in overpressurization of the interfacing system but the system does not rupture or leak.

LK-ncd - Scenario results in RCS leakage from the interfacing system, through either a break or an open relief valve, but no severe core damage (sufficient to cause offsite health effects) occurs because the leak is either isolated before core uncovery or the leak is too small to interfere with core cooling. LOCA-ic - Identifies scenarios that produce a LOCA inside containment. Because these sequences are enveloped by the design basis analysis of the plant, they are not fully developed on the event trees and these scenarios are not considered to be core damage events.

REL-mit - An ISLOCA with core damage occurs but the radioactive release is mitigated through some means, such as scrubbing through an overlying water pool or general area fire sprays in the auxiliary building.

REL-1g - An ISLOCA with core damage occurs and results in a large unmitigated radioactive release. Note that this does not necessarily imply that the broak size is large.

B.1 Premature Entry Into Shutdown Cooling - SEQIA

A risk-significant scenario at the Babcock and Wilcox (B&W) plant (see [B-1]) ir olved premature entry into shuidown cooling, with RCS pressure and temperature above the open permissive set point of the decay heat removal (DHR) system suction isolation valves. This scenario was considered credible at the B&W plant because the plant procedures allowed operators to bypass the open permissive interlock for one of the two shutdown cooling isolation valves. This allowed an error of commission to be postulated in which, once the decision is made to enter shutdown cooling early, the operators will be led to bypass the interlock for the other valve, also, even though the procedure does not instruct them to do so. For the CE plant, the HRA did not reveal any circumstances that would lead to an analogous scenario. Therefore, this scenario was not developed further.

B.2 RCS To SI System ISLOCA During Plant Startup - SEQIB

In Sequence 1B the plant is undergoing a startup from cold shutdown. Thus, failure to close MOVs SI-401 and HOVs SI-405 prior to raising RCS pressure above 396 psig is the initiating event. The event tree for this sequence is contained in Figure B.2, while the corresponding flow diagram is displayed in Figure B.1. Since startup is a "low-pressure" procedure compared to normal full RCS pressure, it is assumed that any overpressurization that causes the relief valve to open will not cause an ISLOCA. The event tree models one flow line (out of two) on a mission time of one year.

PSUM. The initiating event for this sequence is a startup from cold shutdown. Such a startup is estimated to occur once every 18 months (cold shutdown does not necessarily occur at every shut down), the frequency of event PSUM is assumed to be 0.67/year.

MOVLO. Once the plant is in a startup mode, MOV SI-401 (A/B) and HOV SI-405 (A/B) must both be left open after the RCS pressure exceeds 396 psig for an ISLOCA to occur. Thus, event MOVLO represents the probability that both SI-401 and SI-405 are left open. The screening probability assumed for MOVLO was 1.0×10^{-4} .

ACIF. If both values SI-401 and S1-405 are left open, the automatic closure interlock (ACI) is designed to shut both values automatically when the RCS pressure exceeds 700 psig. Event ACIF models the probability that the ACI fails. Compounding the evaluation of this event is the fact that the analyzed plant is petitioning the NRC for permission to remove the ACI (this is being done because of concerns about losing SDC inadvertently due to value closure). Thus, two separate probabilities for this event were used. With the ACI in place, the screening probability was assumed to be 1.0×10^{-3} . For the case of removal of the ACI, the probability of failure for this event would obviously be 1.0. The Sequence 18 event tree shows the sequence with the ACI in place.

8-5



Figure B.1 : Shutdown Cooling Suction Line Schematic.



Sequence 1B

D

F. re B.2 : Shutdown Cooling Suction Line ISLOCA Sequence Event Tree (plant in startup mode).
RVF1. This event models failure of relief value SI-406 to open on demand. The failure probability was taken to be 1.0×10^{-3} .

OFNP. This event models failure of the operators to detect the overpressure condition in time to prevent damage. A screening probability of 0.5 was used for this and all other human error probabilities in this sequence.

OFIP. This event models operator failure to isolate the SDC system from the RCS prior to damage. A screening failure probability of 1.0 was used.

ISR3. Event ISR3 models a break in the low pressure portion of the SDC system outside containment. The conditional probability of a break is taken to be 1.0 for the screening analysis.

FTD. Event FTD models failure of the operators to detect the loss of coolant and enter the correct emergency procedure. A screening failure probability of 1.0 was used.

FTDGN. This event models failure of the operators to diagnose that the break is outside containment. A screening failure probability of 1.0 was used.

FTI. This event models operator failure to isolate the break given that it has been detected and diagnosed. A screening failure probability of 1.0 was used.

RNM. Based on the walkdowns performed during the plant visit, the probability that the release would not be mitigated by flooding or auxiliary building fire sprays was judged to be 1.0.

B.3 RCS To LPSI Cold Discharge - SEQ2

Through the normal reactor operating year, MOVs SI-138 (A/B) and SI-139 (A/B) are stroke-tested guarterly. Thus, the accident sequence path for the

low pressure safety injection (LPSI) pump discharge is based upon the fact that the MOVs will be opened once each quarter. Figure B.3 illustrates the simplified flow diagram for the LPSI pumping path. The corresponding event tree for this sequence is contained in Figure B.4. The event tree evaluates one flow path (out of four possible) for a mission time of one quarter. Thus, to get the failure frequency estimate for the complete system based on a oneyear mission time, the sequence end state frequencies must be multiplied by 16.

Obviously, if the two isolation check valves (SI-335/336 (A/B) and SI-142/143 (A/B)) protecting the MOVs had failed, it would not be desirable to open the MOVs. But, for analyzing this sequence, it is assumed that no prior information (for example, a high pressure reading between the two isolation check valves) is known for the system. This assumption is made because the stroke-testing procedure does not direct the operators to check pressure between the PIV check valves before performing the stroke test. Therefore, for the model, it is postulated that internal failure of the two isolation check valves will automatically lead to an overpressurization of the interfacing system when the MOV is stroke-tested.



Figure B.3 : RCS Cold Legs to Low Pressure Safety Injection Pump Discharge Schematic.



Figure B.4 : RCS Cold Legs to Low Pressure Safety Injection Discharge ISLOCA Sequence Event Tree.

CHVF1. The failure of the two isolation check values (SI-335/336 (A/B)and SI-143/144 (A/B) is modeled as the initiating event. Since it is assumed that both values are closed as in a non-failed state at the beginning of the mission time, only a time-dependent failure mode is presumed.

Even though the two check values are not the same size (SI-335/336 are 12-inch values, while SI-143/144 are 8-inch values) and the environmental conditions are not identical for the two check values, an assumption is made that the failure rate λ is constant and the same for the two values. This is a conservative assumption that reflects the lack of detail available in the value failure rate database. The failure rate is assumed to be lognormally distributed, with a mean value of 8.7×10^{-8} /h and error factor of 10. The probability that one of the check values fails in a time T is defined as



Since the model for CHVF1 assumes both check valves failure rates are identical, the underlying probability failure distributions for the two valves must be interchangeable, which leads to the assumption that the valves need to be treated as if the failures were correlated. Thus, for the two check valves in series, the probability that they both fail in the mission time is given by the probability of the first check valve failing intersected with the probability of the second check valve failing, or

> $E(CHVF1) = E[(CHVF)_1 \bigcap (CHVF)_2]$ = $\tau^2 E(\lambda^2)$ = $\tau^2([E(\lambda)]^2 + Var(\lambda))$

Given that the mission time is 2190 hours, the probability of failure for the two check valves is calculated to be 2.58x10⁻⁷. This probability is multiplied by 16 (four quarters per year times four injection lines) in calculating the end state frequencies for this sequence.

PO. This event models the opening of MOVs SI-138 (A/B) and SI-139 (A/B). Since the MOVs are opened once each quarter for stroke-testing, the probability of event PO in the mission time is assumed to be 1.0.

ISR3. Event ISR3 models a break in the low pressure portion of the LPSI system outside containment. This probability is calculated in Appendix F. The conditional probability of a break is found to be 1.0.

FTD. Event FTD models failure of the operators to detect the loss of coolant and enter the correct emergency procedure. The quantification of this event can be found in Appendix C. The mean failure probability is estimated to be 1.8×10^{-2} .

FTDGN. This event models failure of the operators to diagnose that the break is outside containment. The quantification of this event is presented in Appendix C. The mean failure probability is estimated to be 2.0×10^{-2} .

FTI. This event models operator failure to isolate the break given that it has been detected and diagnosed. In quantifying this event (see Appendix C for details), the flow of the plant's existing emergency procedures was strictly modeled. Because the emergency procedures to not contain steps that would direct the operators to isolate the break (by terminating LPSI flow), and because the operators have received post-TMI training that cautions against overriding a valid safety injection signal, the failure probability of this event is 1.0. It is possible that the operators could take knowledgebased actions outside of the emergency procedures, but such actions were not modeled in the base case analysis.

B-13

RNM. Based on the walkdowns performed during the plant visit, the probability that the release would not be mitigated by flooding or auxiliary building fire sprays was judged to be 1.0.

8.4 RCS Cold Legs to High Pressure Safety Injection (Header A) - SEQ-3A

This scenario is similar to Sequence 2. Once each quarter, MOVs SI-225 through SI-228 (A/B) are stroke-tested while the plant is operating. Thus, the accident sequence path for the high pressure safety injection (HPSI) pump discharge is based upon the fact that the MOVs will be opened once each quarter. Figure 8.5 depicts the simplified flow diagram for the HPSI pumping path. The matching event tree for this sequence is in Figure 8.6. The event tree models one flow path (out of four) on a mission time of one quarter.

Once again, if the isolation check valves in the sequence protecting the MOVs had failed, it would not be desirable to open the MOVs. Tut. for this analysis, it is assumed that there is no prior knowledge for the system. Thus, for the model, it is assumed that random failure of the two isolation check valves will automatically lead to a demand on check valve SI-216.

CHVF2. The initiating event for this sequence is the failure of the two isolation check valves (SI-335/336 (A/B) and SI-241 through SI-244 (A/B)). The event is modeled as a single event, similar to event CHVF1. It is assumed that both valves are closed and in a non-failed state at the beginning of the mission time, leading to only a time-dependent failure mode for the check valves.

Although the two check valves are not the same size (SI-335/336 are 12inch valves, while SI-241 through SI-244 are 8-inch valves) and the



Figure B.5 : RCS Cold Legs to the High Pressure Safety Injection Pump Discharge Flow Diagram.



Figure 8.6 : RCS Cold Legs to the Righ Pressure Safety Injection (Header A) ISLOCA Event Tree

environmental conditions are not identical for the two check values, it is assumed that the failure rate λ is constant and the same for the two values. The failure rate is assumed to be lognormally distributed, with a mean of 8.7×10^{-8} /h and an error factor of 10. Using the analysis from Sequence 2, the quarterly failure probability of event CHVF2 is found to be 2.58 $\times 10^{-7}$.

PO. Event PO is similar to that for Sequence 2, except the MOVs that are opened during stroke-testing are SI-225 through SI-228 (A/B). Since the MOVs are opened once each quarter, the event probability for one mission time is assumed to be 1.0.

CHVF3. Once the two isolation check valves fail, the interfacing system will become pressurized, putting a demand on check valve SI-216. Thus, event CHVF3 models the probability of the check valve SI-216 failing to close upon demand. This failure probability is taken to be 1.0×10^{-3} /demand.

CHVF4. This event is the same as CHVF4 except that check valv' SI-207A must close. The failure rate of CHVF5 is taken to be 1.0×10^{-3} /demand, also.

B.5 RCS Cold Legs to HPSI (Header B) - SEQ-3B

Sequence 3B is comparable to Sequence 3A with the exception that Sequence 3B has one less check valve to protect the interfacing system. Whereas header A has check valve SI-216, header B does not have the corresponding check valve in the piping design.

The piping diagram for this sequence is shown in Figure B.5, the event tree in Figure B.7. The only difference between the event tree for Sequence 3A and that for Sequence 3B is that event CHVF3 has been deleted in sequence 3B. As in Sequence 3A, the event tree analyzes one flow path (out of four possible) on a mission time of one quarter.

After the deletion of event CHVF3, the remaining events in the event tree are identical for the two sequences.



Figure B.7 : RCS Cold Legs to High Pressure Safety Injection (Header B) ISLOCA Event Tree

B.6 RCS Hot Legs to HPSI (Header A) - SEQ-4A

Once every quarter MOVs SI-502A and SI-506A are stroke-tested. Therefore, Sequence 4A is based on the opening of MOV SI-502A. Since valve SI-502A is opened and closed before valve SI-506A is opened, the opening of SI-502A is defined as the initiating event for the sequence. Once again, the assumption of no prior knowledge of the condition of the system is used. The system flow diagram in shown in Figure B.8. The event tree for Sequence 4A is shown in Figure B.9. The event tree analyzes one flow path (out of four) on a mission time of one quarter.

CHVF5. The initiating event is similar to event CHVF1 from Sequence 2, except the two check valves that are modeled are SI-512A and SI-510A. The modeling of the two check valves results in a quarterly failure probability of 2.58x10⁻⁷.

PO. Event PO models the plant operating at normal power and the MOV SI-502A being opened for stroke-testing. Since valve SI-502A is stroke-tested once a guarter, the probability of event PO is assumed to be 1.0.

MOVF2. Event MOVF2 models the internal random failure of MOV SI-506A. The failure rate is assumed to be 1.0×10^{-7} /hr. Thus, for a mission time of 2190 hours, the probability of failure for the closed MOV is 2.19×10^{-4} .

The remaining events are the same as for Sequences 3A and 3B.

B.7 RCS Hot Legs to HPSI (Header B) - SEQ-4B

Sequence 4B is similar to fince 4A except that check valve SI-216 is absent from piping header B.w diagram is contained in Figure B.8, while the event tree is contailed in Figure B.10. As in Sequence 4A, the event tree models one flow path (out of four) on a mission time of one quarter. The initiating event for Sequence 4B is the opening of MOV SI-502B. The initiating event probability is identical to that of Sequence 4A, and is assumed to be 1.0.

CHVF6. This event is similar to event CHVF5, except the two check valves that are modeled are SI-512B and SI-510B. The quarterly failure probability is found to be 2.58x10⁻⁷.

PO. Event PO models the plant operating and valve SI-50%B opening. Since SI-502B is tested every quarter, the probability of this event is assumed to be 1.0.

MOVF3. Event MOVF3 is like event MOVF2, except that the valve that is modeled is SI-506B. The probability of failure for this event is 2.19x10⁻⁴.

The remaining sequence events and event probabilities have previously been defined.



8,-

.

T



9

. . .

.

3

1 In 19



. .



Figure B.9 :

RCS Hot Legs to High Pressure Safety Injection Discharge (Header A) ISLOCA Event Tree.



Figure B.10 : RCS Hot Legs to High Pressure Safety Injection Discharge (Header B) ISLOCA Event Tree

B.8 RCS Hot Legs to the LPSI System During Shutdown - SEQ-5

When the analyzed plant enters shutdown cooling, the operators rely on check valves SI-108 and SI-1071 closing when the RCS pressure exceeds the interfacing system design pressure. Thus, Sequence 5 is based upon failure of the two check valves. The simplified flow diagram for this sequence is contained in Figure B.11. The ISLOCA event tree is contained in Figure B.12.

PSM. The plant is assumed to enter shutdown cooling using the LPSI system once a year on average. During the shutdown, MOVs SI-401, SI-405, and SI-407 (A/B) are opened. Therefore, the probability of one initiating event in the mission time is assumed to be 1.0.

CHVF10. This event models the failure of check valve SI-108. Due to the as-found degraded condition of valve SI-108, no credit is taken for this valve. Thus, the demand failure probability of event CHVF10 is assumed to be 1.0.

CHVF11. Event CHVF11 models the failure of check valve SI-1071 to close on demand. The demand probability is assumed to be 1x10⁻³.

The remaining events have already been defined. The human error probabilities were calculated in Appendix C. The probability of a break in the interfacing system was calculated in Appendix F.



Figure B.11 : RCS Hot Legs to RWSP via the LPSI System



Figure B.12 :

 RCS Hot Legs to RWSP via the LPSI System ISLOCA Event Tree

References

B-1. W. J. Galyean, et al., Assessment of ISLOCA Risks - Methodology and Application: Babcock and Wilcox Plant, NUREG/CR-5604, to be published.

Appendix C

Human Reliability Analysis for the Combustion Engineering ISLOCA Probabilistic Rick Assessment

CONTENTS

CE ISLOCA Human Reliability Analysis	- 5
Scenarios and Human Actions for the CE ISLOCA HRA	- 26 - 26 - 26 - 27 - 27
Human Actions For this ISLOCA HRA	-28
Modeling Of Human Actions And Estimated Human Error Probabilities C FTD: Sequence 2 (Using Procedure Only)	- 29 - 29 - 30 - 34 - 38 - 38
FT1 Sequence 2 (Using Knowledge-based Behavior)	- 42
Behavior)	-43 -43 -47 -51 -55 -72
Summary of HRA Results	-89
Conclusions	-91
References	- 92

Figures and Related Tables

Figure 1: ISLOCA Data Collection Form, page 1	1
Figure 3: HRA Event Tree for Sequence 2, FTD (Fail to Detect)C-16 Table C1: HEPS for FTD Sequence 2C-19 Table C2: Failure Paths & Total Failure Probability (Seq.2,FTD)C-20 Table C3: Revised HEPs for the CE ISLOCA HRAC-29	5 7 3 5
Table C4: Human Actions for the CE ISLOCA HRA	6
Figure 4: HRA Event Tree for Sequence 2, FTDGN (Procedure Only)	123
Figure 5: HRA Event Tree for Sequence 2, FTI (Using Procedure Only)C-3 Table C7: HEPS for FTI, Sequence 2 (Using Procedure Only)C-3 Table C8: Failure Paths & Total Failure Probability (Seq.2,FTI)C-3	5 5 7
Figure 6: HRA Event Tree for Seq.2, FTDGN (Proc. & Knowledge-based)C-38 Table C9: HEPS for Seq. 2, FTDGN (Proc. & Knowledge-based)C-40 Table C10: Failure Paths & Total Failure Probability (FTDGN)C-4	8 0 1
Figure 7: HRA Fault Tree for Sequence 2, FTI (Using Knowledge-based)C-44 Table C11: HEPs for Seq. 2, FTI (Knowledge-based Behavior)C-44 Table C12: Failure Paths & Total Failure Probability (FTI)C-44	456
Figure 8: HRA Fault Tree for Sequence 5, FTD	8 9 0
Figure 9: HRA Event Tree for Sequence 5, FTDGNC-5 Table C15: HEPS for Sequence 5, FTDGNC-5 Table C16: Failure Paths & Total Failure Probability (Seq.5,FTDGN).C-5	234
Figure 10: HRA Event Tree for Sequence 5, FTI-A (1 Train SDC)C-5 Table C17: HEPS for Seq. 5, FTI-A (1 Train SDC)C-5 Table C18: Failure Paths & Total Failure Probability (Seq.5,FTI-A).C-6	6 8 0
Figure 11: HRA Event Tree for Sequence 5, FTI-B (2 Trains SDC)C-7 Table C19: HEPS for Seq. 5, FTI-B (2 Trains SDC)C-7 Table C20: Failure Paths & Total Failure Probability (Seq.5,FiI-B).C-7	3 5 7
Table C21: Summary of Results for CE ISLOCA HRA	0

CE ISLOCA Human Reliability Analysis

This appendix describes in detail the methodology and results of the human reliability analysis (HRA) for the third ISLOCA probabilistic risk assessment (PRA). HRA was used to model the predominant human errors for each significant scenario in the PRA. HRA is a methodological tool for analyzing, predicting, and evaluating work-oriented human performance in quantitative, that is, probabilistic terms. As a diagnostic tool, HRA can be used to identify those factors in the system which lead to less than optimal human performance and can estimate the error rate anticipated for individual tasks. In a given system, or sub-system, HRA can also be utilized to determine where human errors are likely to be most frequent. Traditionally, HRA analysts model human performance through the use of event trees like those found later in this appendix.

The general methodological framework for this ISLOCA HRA was based on guidelines (under development) from the NRC-sponsored Task Analysis-Linked Evaluation Technique (TALENT) Program [C-1] which recommends the use of task analyses, time line analyses, and interface analyses in a detailed HRA. NUREG/CR-1278, the Handbook of Human Reliability Analysis with Emphasis on Nuclear Power Plant Applications (THERP) [C-2], recommends similar techniques and, in addition, provides a data base that can be used for estimating human error probabilities (HEPs). Finally, this ISLOCA HRA integrated the steps from the Systematic Human Action Reliability Procedure (SHARP) [C-3], and A Guide for General Principles of Human Action Reliability Analysis for Nuclear Power Generation Stations (draft IEEE standard P1082/D7 [C-4]).

From this combination of approaches, the analysts identified 11 basic steps, summarized below, which were used as guidelines for this HRA. Following this brief summation of the 11 steps is a detailed explanation of how each step was applied to the HRA process. The 11 basic steps are as follows:

> Select the team and train them on relevant plant functions and systems. (IEEE P1082)

- Familiarize the team with the plant through the use of system walkdowns, simulator observations, etc. (IEEE P1082)
- Ensure that the full range of potential human actions and interactions is considered in the analysis. (SHARP) (IEEE P1082)
- Construct the initial model of the relevant systems and interactions. (IEEE P1082)
- 5. Identify and screen specific human actions that are significant contributors to the safe operation of the plant. This was accomplished through detailed task analyses, time line analyses, observations of operator performance in the plant and in the simulator, and evaluations of the humanmachine interface. (SHARP and IEEE P1082)
- 6. Develop a detailed description of the important human interactions and associated key factors necessary to complete the plant model. This description should include the key failure modes, an identification of errors of omission/commission, and a review of relevant performance shaping factors. (SHARP) (IEEE P1082)
- Select and appl_appropriate HRA techniques for modeling the important human actions. (SHARP)
- Evaluate the impact on ISLOCA of significant human actions identified in Step 6. (SHARP)
- Estimate error probabilities for the various human actions and interactions, determine sensitivities, and establish uncertainty ranges. (SHARP) (IEEE P1082)
- Review results (for completeness and relevance). (IEEE P1082)
- Document all information necessary to provide an audit trail and to make information understandable. (SHARP)

The following paragraphs explain in detail how each of the preceding steps was completed. Since the PRA/HRA process is iterative in nature, the reader should note that several sections of this 11 step method were repeated to refine the analysis. The first two steps in this process required the selection of a PRA/HRA team and their subsequent training on the plant and its relevant systems. The PRA/HRA team from the INEL was composed of three members: a nuclear engineer (for the PRA), a human factors engineer (for the HRA), and an electrical engineer (with extensive experience in both the PRA and HRA approaches). To familiarize, or train themselves, the team members reviewed the following:

- mechanical and electrical system descriptions (e.g., the reactor coolant, residual heat removal, safety injection, and chemical and volume control systems),
- a sourcebook of plant systems and schematic drawings (NRC-03-87-029, FIN D-1763 [C-5]),
- the plant's Final Safety Analysis Report (FSAR),
- the plant's Technical Specifications [C-6],
- plant procedures (operating, abnormal, emergency, maintenance, administrative, etc.), station directives, and operational practices,
- piping and instrumentation diagrams (P&IDs),
- the types, capacities, and locations of check valves/motoroperated valves identified as being pressure isolation valves,
- training materials such as flow charts, lesson plans, etc.,
- crew composition (for control room and auxiliary building operators) and level of training/experience,
- significant precursor information From general ISLOCA-related LERs, as well as a plant-specific ISLOCA-related LER (NRC Inspection Report 50-412/90-10 and 50-414/90-10 [C-7]).

This training/familiarization process for the plant's systems was enhanced by a two-week visit to the plant.

Step #3 required that significant human actions and interactions be incorporated into the ISLOCA PRA analysis. This was accomplished through an extensive data collection process during the plant visit. As part of the data collection, the utility provided written procedures, training materials, and P&ID drawings. This data was supplemented by interviews and detailed task analyses with both licensed and non-licensed nuclear operators in the plant. Observations of control room personnel, the use of the utility's simulator, and system walkdowns with licensed and non-licens. 'operators supplied additional information. Further information was supplied this igh interviews and walkthroughs with a former shift supervisor (with over 10 years experience) from this particular plant.

The initial plant models were constructed in the fourth step. Using the plant-specific data gathered in Step #3, the HRA analysts worked with the PRA analyst and systems engineering personnel to specify human actions related to the postulated ISLOCA scenarios. As a consequence of several findings from the earlier ISLOCA PRA of the B&W plant, significant attention was given to latent, or precursor, human errors during normal operations which could lead to inoperable equipment or misaligned valves. Examples of these precursor actions included: jumpering of valves to defeat protective interlocks, maintenance procedures, in-service testing practices, and administrative procedures governing the generation and completion of work packages.

The HRA analysts also examined active, or initiator, failures which could lead to an ISLOCA, and post-initiating human errors during responses to abnormal situations. Examples of initiator failures included violations of Technical Specifications, procedural violations (such as early entry into decay heat removal), selection of the incorrect vent path, and reconfiguring plant equipment. For post-initiating errors, the HRA team examined operator responses following a significant break outside containment. Specifically, the HRA analysts looked at operator actions entailing detection, diagnosis, recovery, and isolation.

The fifth step required the HRA analysts to identify those human actions which are significant contributors to the effective operation and safety of the plant. Using the data collected in Step #3, in conjunction with a review of operational procedures and training materials, the HRA team screened the various human actions, identifying those which had a significant impact on plant operations and/or safety with respect to ISLOCA. These significant human actions were included in the PRA event trees, and they helped guide the activities in the next step.

The output from the preceding step (i.e., Step #5) was a group of important human actions, for specific ISLOCA scenarios, which were described in generic, functional terms (e.g., operators recover system). In the sixth step, the analysts expanded the description of each of these key human actions from a functional description into specific operator tasks and subtasks (e.g., operator opens valve SI-401A, or operator closes valve SI-407B). By breaking down the human actions into specific tasks and subtasks associated with individual equipment and procedures, the analysts began to identify specific failure modes, root causes, and failure effects. The description of each task/subtask was enhanced by referencing significant performance shaping factors (PSFs) which affected a given task. These PSFs were derived from the task analyses, time line analyses, evaluation of the human-machine interface, and direct observations of operator performance. Examples of PSFs included:

- 1 the quality of the human-machine interface,
- 2 written procedures (emergency, abnormal, maintenance, etc.),
- 3 P&IDs,
- 4 response times for systems and personnel,
- 5 communication requirements,
- 6 whether the operator actions were skill, rule, or knowledgebased.
- 7 crew experience,
- 8 levels of operator stress in different scenarios,
- 9 feedback from the systems in the plant,
- 10 task dependence and operator dependence,
- 11 location of the task (e.g., control room, auxiliary building, etc.),
- 12 training for individual operator actions, including ISLOCA situations.

Each PSF was seen as casting either a positive or negative influence on the basic HEP, that is, as either decreasing or increasing the probability of failure for a given human action. For example, some of the positive PSFs found at the plant included the following:

- 1 "The team did not identify any significant deficiencies in the man-machine interface that might significantly increase the probability of an operator error initiating an ISLOCA." [C-7]
- 2 "The team found emergency operating procedures to be well written although they lacked some human factors consideration. (see #2, negative PSFs)." [C-7]
- 3 "Although training specific to ISLOCAs was not part of the licensee's training program, operators indicated, during walkthroughs and simulator exercises, that they were <u>generally</u> well prepared to cope with losses of RCS inventory."[C-7]

Examples of negative PSFs were:

- 1 "...the team identified weaknesses in the man-machine interface that could adversely affect the ability of the operators to mitigate an ISLOCA because of poor equipment labeling and the inaccessibility of some equipment." [C-7]
- 2 Even though EOPs were generally well-written, the RCS Leak Procedure, OP-902-002, does not provide relevant guidance with respect to requisite actions for the isolation of ISLOCAs. As a result, operators and supervisory personnel would have to rely on knowledge-based actions, outside of normal procedures.
- 3 Within the context of the prior finding, operator training (based on Three Mile Island scenarios) emphasized that operators should not override a safety injection occurring in conjunction with an unisolated RCS leak (see Sequence 2). This training could lead control room personnel away from the necessary actions in Sequence 2 to isolate a break in the safety injection lines (e.g., operators would have to sequentially close each HPSI and LPSI safety injection valve on the affected SI train).
- 4 Operators' ISLOCA diagnostic abilities were focused on Attachment 1 of OP-902-002, which verifies a LOCA outside containment but directs operators to a procedure (OP-902-002) which does not provide relevant guidance for the isolation of an ISLOCA.

For this HRA analysis, the majority of influences from specific PSFs were implicitly modeled as each HEP was identified and quantified using various THERP tables. A careful examination of these tables will show how

individual basic HEPs can only be identified after associated PSFs are specified. Stress and dependence were explicitly modeled (using T^{1} RP) as two of the more significant PSFs. From a human performance perspect. ... high levels of stress lead to higher probabilities of human error. Generally, a person's short-term memory (STM) can retain from five to nine items of information for brief periods. However, as stress increases, this capacity shrinks to levels where STM can only hold three to five items. This well documented finding interacts with a phenomenon called cognitive tunnel vision where high levels of stress cause an operator's visual and perceptual abilities to begin shrinking into a limited focus so that only one or two salient aspects of his environment are featured. Also, as stress continues to increase, the operator begins to retreat from current conditions, relying on previously learned (perhaps incorrect) patterns of behavior. In Sequence 2, for operator actions FTD-LOCA (fail to detect LOCA) and FTDGN (fail to diagnose ISLOCA), stress levels were modeled as moderately high due to required procedural responses during a reactor trip and/or safety injection. For Sequence 5, FTD (fail to detect loss of coolant) stress was initially modeled as optimal until entry into OP-901-046, the Shutdown Cooling Malfunction procedure, when it increased to moderately high. For FTDGN in Sequence 2, stress remained at moderately high levels, but was increased slightly (e.g., a PSF modifier of 3) in FTI-A/FTI-B. This slight increase was modeled to reflect this plant's concern about a loss of shutdown cooling (based on a significant past LER.

In several of the ISLOCA scenarios, low (LD), moderate (MD), and high (HD) levels of dependence were assigned between the control room supervisor (CRS) or shift supervisor (SS) and the licensed reactor operator (RO). As used in THERP, dependence refers to the level of interaction between two or more workers. Dependence is usually modeled on a scale which ranges from complete dependence (where a second worker fails on a given task because of the failure of a primary worker on the same task) to complete independence (zero dependence or ZD).

A detailed data collection form (see Figures #1 and #2 in this appendix) was developed as an aid in the HRA data collection, task analyses, and the

decomposition and description activity just mentioned. This data form served as a template which guided the collection of the requisite information, in sufficient detail, for each task or subtask in the dominant ISLOCA sequences. Additional items of information, for each human action, were added to these forms as new details surfaced (i.e., details from follow-up telephone conversations with plant personnel, the ISLOCA inspection report for this plant, and a comparison of procedural steps to P&IDs).

The output from the preceding step (#6) is an extensive list of operator tasks and subtasks (with their associated PSFs) for each human action in the dominant PRA sequences. These detailed tasks are the required input for the seventh step, where appropriate HRA techniques for modeling the significant human actions were selected and applied. For each human action, the analysts selected an appropriate technique for task modeling and quantification. Because most of the human actions in this HRA involved the use of various wr ten procedures, THERP-type HRA event trees were used in modeling a majority of the human actions in the detailed analysis. However, not all ISLOCA scenarios were best represented by THERP event trees alone. In those cases, HRA fault trees were used in conjunction with the typical THERP event trees. The fault trees and THERP event trees were used in a detailed analysis to estimate the probability of human error for each of the dominant human actions. Quantification techniques included THERP and Human Cognitive Reliability (HCR) [C-8]. For each human failure, basic HEPs were calculated using THERP or HCR and were then modified using performance shaping factors (PSFs) to realistically describe the work processes at the utility.

Prior to the quantification, or estimation of human error probabilities, the PRA and HRA specialists reviewed and evaluated the significant human actions, and their associated PSFs, for each of the dominant ISLOCA sequences (Step #8). After this evaluation, the HRA analysts developed the HRA event trees and fault trees used to model the significant human actions, and their associated PSFs, for each of the dominant ISLOCA sequences (Step #8). After this evaluation, the HRA analysts developed the HRA event trees and fault trees used to model the significant human actions in each sequence. According to the SHARP method, the development and use of these HRA fault and event

Sequence ID Task ID Subtask ID
Crew size & composition
Who does task/subtask?
Crew experience: Low Optimal Moderate High
is time limit important for this task/subtask? "es or No
Time to perform task/subtask (after diagnosis/decision)
Median response time for whole task Std. Dev
Plant/system time available
If task not successfully completed, what is next action?
z and type of alarms competing for attention
Quality of plant interface: Excellent Good Fair Poor Very Poor
Operators Stress Low Optimal Moderate High
Type of instrument/control
HF notes on controls
Consequence of Improper performance High Medium Low
Explain:
Feedback/system response to operator action
Operation routine: Yes or No Operation/transient understood 'Yes or No
Proc Read Yes or No Proc covers case. Yes or No
Proc well written res or No Proc understood Yes or No
Proc practiced: Yes or No How much practice/training on task?
Coonitive Behavior SkillRuleKnowledge
Tagging Yes or No Describe
Recovery Actions Checklists Inspections 2nd Person
Feedback from Annunciators Alarms Displays

Figure 1: ISLOCA Data Collection Form, page 1

Local or Remote operation? Explain: ______

Tasks or subtasks done step-by-step_____ or Dynamic_____ Dependence: is the order of the tasks critical Yes or No Does the success/failure of one action affect the success/failure of the next Yes or No Explain_____

If 2 men do the job, does the action of either one affect the success/failure of the next? Yes or No Explain:

is the job done with rest stops_____ or continuous performance_____?

Is there any radiation safety or caution for this job? Yes or No.

If yes, what dosage?_____mrem

HF comments of plant-specific PSF's

Additional Comments/observations:

Figure 2: ISLOCA Data Form, page 2

trees "provides a disciplined approach for explicitly evaluating alternative actions and, if properly interpreted, may provide the rational for including some human errors known as acts of commission in the event trees." This HRA modeled errors of commission and omission, which are identified on specific branches of the event trees seen later in this appendix.

Assigning HEP estimates to each of the subtasks was the major activity in Step #9 - Quantification. Traditionally, HRA analysts model human performance through the use of an event tree like Figure 3, which represents "FTD," fail to detect LOCA for Sequence 2. Operator error was generally placed along the descending right branches of the event tree. Successful operator actions were sequenced on the left side of the tree. For example, on the top lefi, Event "a" - Control Rcom (CR) Detects Dropping Pressurizer (PZR) Pressure, is the success path. Failure to accomplish this task is modeled as Event "A" - CR Fails to Detect Pressurizer (PZR) P. ssure Dropping. When a second operator, or group of operators, is involved, such as in Event "B" -Control Room Fails to Detect PZR Low Pressure Flarms, the action of this second operator, or group, may be modeled in a recovery branch, as shown in Figure 3. Event "b" models the opportunity for the control room to detect the pressurizer low pressure alarms. If the control room does detect the alarm, this becomes a recovery action because it would bring the model back to the success path (via the dotted lines is Figure 3).

Basic HEPs for each individual failure were calculated using THERP or HCR. These basic HEPs were then modified using PSFs to realistically describe the work process. Each PSF either increased or reduced the likelihood of a given human error action. For the event tree in Figure 3, the following PSFs were used to modify the basic HEPs:

The control room is assumed to be in Mode 1 operations with one RO performing a quarterly stroke test on the safety injection (SI) valves. It is assumed that two PIV check valves have failed and that, as the RO opens the associated SI valve, there is an immediate overpressurization and break, which results in a reactor trip and safety injection actuation signal.

C-15



- Stress levels were modeled as being moderately high following the break.
- The crew was judged to be experienced.

High dependency (HD) was used to model the relationship between the CR's ability to detect decreasing pressurizer pressure (or level) and the CR's ability to detect the subsequent pressurizer alarms. High dependence was also modeled between the CRS and SS as they decide to enter OP-902-002, the RCS Leak Keduction Procedure.

Individual error paths were identified and failure probabilities were estimated using the HEPs and tables from THERP or estimates from HCR. (The probabilistic values in the THERP tables are to be considered as median values from a lognormal distribution. The estimates from HCR are assumed to be point estimates). For example, path "A" in Figure 3 leads to failure by the CR to detect pressurizer pressure dropping (the first branch on the right side of the tree). This particular failure had a basic median HEP of 0.006 (from Table C1) and an error factor of 3. This information came from THERP Table 20-10 #3, item #2, and was modified by a PSF of 2, for moderately high stress. The basic median HEP is converted to a basic mean HEP which is modified by the same PSFs. This results in a nominal mean HEP of 0.015 and an error factor of 3.

Each event tree has several unique error paths. For example, event "A" and event "B" togeth - constitute an error path wherein the CR fails to detect uropping pressurizer pressure and the same CR crew fails to detect the subsequent alares. In a similar manner, failure path "A-b-C-D" models a sequence of events in which the CR fails to detect dropping pressurizer pressure, then detects the subsequent low pressurizer pressure alarms, but fails to detect decreasing pressurizer level and the subsequent low level alarms. Probabilities for each unique error path were calculated by multiplying each nominal mean HEP on a given error path by any other nominal mean HEP on the same error path (see Table C2). For example, the error rate for path "A-B" would be calculated by multiplying the HEP of failure "A" (0.015) by the HEP for failure "B" (0.279), resulting in a nominal HEP for
that path equaling 0.004 (0.015 x 0.279 = 0.004). NOTE: the 6-digit accuracy for numerical values in the following tables is an artifact of the software used for quantification and does not imply 6-digit precision for the HEP estimates. Other examples of error paths for this event tree include: "A-b-C-D", "a-c-E-F", and "A-b-C-d-E-F." The failure probabilities for individual error paths were summed to give the total failure probability for that event tree. The resulting error factor for the total failure probability was calculated from an uncertainty analysis using IRRAS (the Integrated Reliability and Risk Analysis System [C-9]).

Table C1 lists the basic median HEPs and nominal mean HEPs for the event tree depicted in Figure 3 (FTD-LOCA, Sequence 2). This table enumerates the basic human actions/errors, the basic or unmodified HEPs (median and mean), their sources from the table and item number in THERP, whether the action was modeled as being performed in a step-by-step mode or dynamically, PSF modifier values and the related THERP source, level of dependency, and finally, the nominal, or modified, mean HEP with its error factor (derived from THERP HEPs or THERP Table 20-20).

	Human Action / Error	Basic Median HEP	Error Factor	Source/ THERP Table #	Step-by- Step or Dynamic	Modifier for PSFs	Modifier Source	THERP Depend- ency	Basic Mean HEP	Nominal Mean HEP	Error Factor
A	Control room fails to detect dropping PZR pressure	0.006	3.0	T20-10 #3	SBS	2	T20-16 #4a	ZD	0.007499	9.614998	3.0
в	Conirol room fails to detect PZR low pressure atarms	0.0001	10.0	T20-23#1a	SBS	2	T20-16 #4a	HD	6.090266	8.50(365	5,0
										# 0.278928	2.5
C.	Control room fails to Criect dropping P7R level	0.006	3.0	T26-10#3	\$85	2	T20-16 #4a	ZD	0.007499	0.014998	3.0
Ð	Control room fails to detect PZR low level alareas	6.0001	10.0	T20-23 #1	SBS	2	T20-16 #4a	HD	8.000266	0.500266	5.0
					1.000					# 0.278928	2.5
E.	CRS fails to enter OP-902-002	0.005	10.9	T20-6 #4	SBS	2	T20-16#4#	ZD	0.013317	0.026635	10.0
ę.	SS fails to direct CRS to enter OP-902-002	0.1	5.3	T20-22 #1	SBS	2	T20-16#2#	HD	0.161383	0.661383	5.0
					1 1 1	1.00				# 0.33997	2.3

		Table C2 : Failure Paths and Total Failure Probabilities	
		FTD LOCA - Sequence 2	
	Failure Path	Calculations	Results
1	AB	0.014998 x 0.278928	0.004183
	ABCD	0.014998 x 0.014998 x 0.278928	0.000062
5	AbCdEF	0.014998 x 0.014998 x 0.026635 x 0.33097	0.000001
1	AbcEF	0.014998 x 0.025635 x 0.33097	0.000132
5	aCD	3.014998 x 0.278928	0.004183
5	#C4EF	0.814998 z 0.026635 z 0.33097	0.000132
7	acEF	0.026635 x 0.33097	0.00881
		Total Failure Probability	0.918
		Error Factor	4.26

As discussed in Section 4.2 of the main report, the estimates of human error probabilities obtained from THERP are generally treated as point estimates with a given error factor. The authors of THERP indicate that there is insufficient data, at this time, to accurately determine the true shape of the underlying probability distribution associated with these point estimates and that these distributions are unimportant. Quoting from THERP (pages 7-6 through 7-8):

"Although we would like to have data clearly showing the distributions of human performance for various NPP (nuclear power plant) tasks, there is ample evidence that the outcomes of HRAs are relatively insensitive to assumptions about such distributions...."

The authors then provide several examples to support a general conclusion:

"the assumption of normal, lognormal, or other similar distributions will make no material difference in the results of HRA analyses for NPP operations. In some cases, this insensitivity may result from a well designed system that has so many recovery factors that the effect of any one human error on the system is not substantial.... For computational convenience, one might wish to assume the same distribution for probabilities of human failure as the one used for probabilities of equipment failure, as was used in WASH-1400."

To summarize, the authors of THERP "suggest" that iRA analysts "assume" the point estimates from THERP are medians from a lognormal distribution, even though such an assumption is "speculative" at best.

While the THERP approach (treating the HEPs as median values from a lognormal distribution) has certain computational and interpretational advantages, it has one distinct drawback, with respect to PRAs. In most PRAs, hardware failure probabilities are assumed to be lognormally distributed. The HEPs are multiplied by hardware failure probabilities when calculating core damage frequencies. This requires a median to be multiplied by a mean, a procedure which does not result in a mean value of the core damage frequency. A mean core damage frequency can be obtained by converting the <u>median</u> HEP values (from an assumed lognormal distribution) to <u>mean</u> HEP values, thereby al'owing the necessary multiplications.

This HRA ~lopted THERP's recommendation to treat each HEP as a median value from a) rmal distribution. Detailed HRA analyses were conducted for each of the significant scenarios identified in this ISLOCA PRA. Tables C1 and C2 summarize the results of these analyses, i.e., by converting the median HEPs to mean HEPs using the following formulas:

> Mean HEP = $\exp(\mu + \frac{\sigma^2}{2})$; where \Re = the Median HEP; μ = $\ln \Re$; and, $\sigma = \frac{\ln(ErrorFactor)}{1.645}$

Converting median HEPs (from an assumed lognormal distribution) to mean HEPs allowed uncertainties in human error to be included in calculations of the uncertainty in .ore damage frequency. The actual conversions to mean HEPs were accomplished by inserting the basic, median HEPs in each event tree int. the equations above. The resulting mean HEPs were then modified by appropriate PSFs and used ir the appropriate error branch on specific event trees to calculate error path and total failure probabilities for each event tree.

A careful review of Table C1 will show that the conversion from median to mean HEPs can cause problems with the resulting confidence interval. The reader may recall that individual HEPs are considered a point estimate with some uncertainty, e.g., a confidence interval, surrounding it. Generally, this confidence interval is defined by calculating the upper bound (95th percentile) and lower bound (5th percentile) for each HEP. The upper bound is found by multiplying the nominal (modified-median) HEP by its associated error factor (EF) and the lower bound results by dividing the nominal 'modifiedmodian) HEP by the same F1. For example, when the basic median HEP for event "A" (Table C1) is modified, it becomes a value of 0.012 (the nominal mean HEP equals 0.015), the resulting upper bound is 0.036 (0.012 x an EF of 3).

However, when a basic HEP is modified by several PSFs, including dependency, problems with the confidence interval begin to arise. For example, examine event "B" on Table C1. The basic median HEP for this event is 0.00C1 with an EF of 10. When this HEP is converted to a mean value and modified for stress and high dependence, the resulting nominal mean HEP is 0.5 with an EF of 5 (from THERP able 20-20, #5). If one calculates the upper bound for this HEP by multiplying this value by the EF (or more correctly by multiplying the modified median value, 0.5, by the EF' the result is a value of 2.5; this value is an anomaly, because the maximum value for a probability is constrained to be less than or equal to one (i.e., unity). To correct this difficulty, the nominal mean HEP and EF were adjusted using a constrained lognormal distribution (see Appendix D for details). The resulting revised nominal mean HEP and EF are shown in Table C1 as the values with a "#", just

below the old values for event "B". The revised mean HEP is 0.279 with an EF of 2.5. Similar adjustments have been made for other events in Sequence 2, as well as for several human actions in Sequence 5. Table C3 lists all of the HEP revisions for each sequence and individual actions in this HRA. This table lists the sequence number and associated HRA tree, the human action from that tree, and the following humanical values:

- 1 basic median HEP with its EF,
- 2 modified median HEP (using associated PSFs) with suggested EF.
- 3 Mean HEP and its EF (from lognormal distribution),
- 4 Revised mean HEP and EF (from constrained lognormal distribution),
- 5 50th percentile value (constrained logno mal distribution), and
- 6 95th percentile value (constrained lognormal distribu/.ion).

This example focused on the event tree for Sequence 2, event FTD-LOCA; a similar process was followed for each of the remaining human actions in the HRA. Specific details, including event trees and HEP tables are provided in the following section.

In the final two steps (#9 and #10) of the HRA process, the analysts reviewed the results of the HRA and documented all of the information needed to provide an audit trail. As final HRA failure probabilities were generated for each ISLOCA sequence, the HRA analysts consulted with the PRA analyst and a systems engineer regarding the validity, completeness, and relevance of the results. During these reviews, several questions arose which required more information. Several telephone calls were placed to operations personnel at the plant and detailed interviews or walkthroughs were conducted with a past shift supervisor from the plant.

The last step necessitated the documentation of the data, methodology, and results from thi: HRA to provide an audit trail. This was accomplished by creating a data notebook containing the completed data forms, pertinent procedures, working notes from the ISLOCA inspection, and the NRC ISLOCA inspection report.

-			
	_ 1	 	-
	- 12		C. P
	-	 	

C3: Revised HEPs for the CE ISLOCA HRA

T	IEP Revisions Using a Con-	straine	od Log	norm	al Dist	ribut	ion
Seq. #	Human Action	Median HEP (EF)	Modified Med. (EF)	Mech HEP(EF)	Revised Mean(EF)	50 %ile	95 %lie
2-FTD	B-CR fails to delect PZR low press alarms D-CR fails to detect PZR low level alarms F-SS fails to direct CRS to enter OP-902-002	0.0001(10) 0.0001(10) 0.1 (5)	0.5 (6) 0.5 (5) 0.6 (5)	0.5 (5) 0.5 (5) 0.5 (5) 0.361(5)	0.279(2.5) 0.279(2.5) 0.331(2.3)	0.237 0.237 f 791	0.608 0.608 0.672
2.FTDGN	B-SS laits to determine if SG pressure low/dropping D-SS tails to determine contain, press, not rising F-SS tails to detect no activity in stream plant H-SS tails to verity entry into OP-902-002	01 (5) 01 (5) 01 (5) 01 (5)	06 (5) 06 (5) 06 (5) 06 (5)	0.661(5) 0.661(5) 0.661(5) 0.661(5)	0 331(2 3) 0 331(2 3) 0 331(2 3) 0 331(2 3)	6 10 10 0 20 0 20	0.672 0.672 0.672 0.672
6-5TD	C-CR fails to detect PZR Lo-Lo level alarm E-SS fails to direct CRS to enter OP-901-046	0.001(10) 0.005(10)	0.5 (6) 0.51 (6)	05(5) 051(5)	0.279(2.5)	0.237	0.608
6-FTDGN	E-SS tails to direct CRS to enter Sec. 6.1 (Op-901-046)	0.005(10)	0.51 (5)	0.51(5)	0.284(5)	0.241	0.614
6+TI-A	B-RO tails to remind CRS to close SI401 (A/B) D-CRS fails to remind RO to close SI401 (A/B) F-RO fails to remind RO to close SI401 (A/B) H-CRS fails to tell Ro to check RCS not stabilizing J-Ro fails to remind CRS to place standby SDC L-CRS fails to verify RO placed Standby SDC N-RO fails to remind CRS to close SI401 (A/B) on P-CRS fails to verify RO closed correct SI401	01 (5) 01 (5) 01 (5) 01 (5) 01 (5) 01 (5) 01 (5)	0.65 (5) 0.4 (5) 0.65 (5) 0.65 (5) 0.4 (5) 0.4 (5) 0.4 (5) 0.4 (5)	0.74(5) 0.56(5) 0.74(5) 0.74(5) 0.56(5) 0.56(5) 0.74(5) 0.56(5)	0.354(2.2) 0.299(2.4) 0.354(2.2) 0.354(2.2) 0.299(2.4) 0.299(2.4) 0.299(2.4)	0.315 0.257 0.315 0.257 0.257 0.315 0.315 0.315	0.633 0.633 0.633 0.633 0.633 0.633 0.633 0.633
n + 0	B HO tails to remind CHS to close SI401 (A/B) D-CRS fails to remind RO to close SI401 (A/B) F RO fails to remind CHS to check RCS not stabilizing H-CRS fails to tell Ro to check RCS not stabilizing J-Ro fails to remind CRS to place standby SDC L-CRS fails to verify RO placed Standby SDC. P-CRS fails to verify RO closed Standby SDC. P-CRS fails to verify RO closed Standby SDC.	01 (5) 01 (5) 01 (5) 01 (5) 01 (5) 01 (5) 01 (5) 01 (5)	0.4 (5) 0.4 (5) 0.45 (5) 0.45 (5) 0.45 (5) 0.45 (5) 0.45 (5)	0.74(5) 0.56(5) 0.56(5) 0.74(5) 0.74(5) 0.74(5) 0.74(5) 0.56(5)	0.299(2.4) 0.299(2.4) 0.354(2.2) 0.354(2.2) 0.299(2.4) 0.354(2.2) 0.299(2.4) 0.299(2.4)	0.315 0.257 0.315 0.257 0.257 0.257 0.257 0.257 0.257	0.633 0.633 0.633 0.633 0.633 0.633 0.633 0.633 0.633

Scenarios and Human Actions for the CE ISLOCA HRA

This section describes the scenarios and summarizes the human actions analyzed in the CE ISLOCA HRA. Human actions for the sequences were initially identified in a cooperative effort by PRA and HRA analysts based on plantspecific information. The sequences were selected for analysis by using screening HEPs in the PRA modeling to determine likely scenarios in terms of ISLOCA risk (i.e., core damage frequencies greater than $10^{-8}/yr$). A screening HEP of 0.5 was used, except in some cases where r screening HEP of 1.0 was judged appropriate. Following are brief descriptions of the selected scenarios from an HRA perspective and specific tables of the human actions relevant to the scenarios.

Sequence 1A: Premature Entry Into Shutdown Cooling (SDC)

During the plant shutdown process, the operators will open MOVs SI-401 (A&B) and SI-407 (A&B) and HOV SI-405 (A&B) to place both trains of SDC into service. Sequence 1A investigates the likelihood that the operators prematurely open these valves when RCS pressure is above SDC entry procedural limits (396 psig and 350° F). To open the valves, operators would have to override interlock permissives, disregard administrative barriers, and take actions beyond those specified by operating procedures. Further human actions for this scenario were not analyzed in the HRA, as premature opening of the SDC system suction isolation valves (the scenario initiator) was estimated by the HRA to be not credible, having a negligible probability.

Sequence 1B: Startup with Shutdown Cooling Valves Open

Sequence 1B is similar to Sequence 1A, except that the plant is undergoing a startup. In this sequence operators must fail to close MOVs SI-401 (A&B) and SI-007 (A&B) and HOV SI-405 (A&B) leaving one, or both trains of SDC in service. An extensive review of administrative barriers, operating procedures, and plant systems indicated that the plant has well defined procedural guidance, in conjunction with redundant systems and multiple alarms which would warn operators about any of the MOVs or HOVs being left open during plant startup and pressurization. In addition, the auto-closure interlocks on the SI-401 and SI-405 valves (a feature which the utility wants to remove) would automatically close both valves when RCS pressure exceeded 700 psia. Further human actions for this scenario were not analyzed in the HRA, since startup of the plant with these valves left open was estimated by the HRA to be not credible, having a negligible probability.

Sequence 2: RCS Cold Leg/LPSI Discharge Interfacing LOCA

In this scenario the reactor is operating at power and a quarterly stroke test is being performed on the LPSI discharge motor-operated flow control valves [SI-138 (A&B) and SI-139 (A&B)]. Two check valves protecting one of the four MOVs are assumed to have failed. When the MOV is cycled open during the stroke test, the LPSI piping is exposed to RCS operating pressure. This results in overpressurization of one LPSI train and a rupture of the system in one of the reactor auxiliary building (RAB) safeguard pump rooms, creating a large RCS leak outside of containment. A reactor trip and safety injection automatically occur upon low pressurizer level before the MOV is stroked shut. LPSI components in either the overpressurized or remaining train may be adversely affected and fail due to their proximity to the leak. The leak may be isolated by closing the MOV that was stroked open (which created the overpressurization). Operator failure to detect, diagnose, and isolate the leak may result in core damage.

Sequence 5 (A&B): LPSI/RWSP Suction Interfacing LOCA During Shutdown

In this scenario, the reactor is shut down and SDC is being brought into service to remove decay heat. Low pressure piping from the refueling water storage pool (RWSP) is protected from higher pressure in the LPSI piping (used during SDC) by two check valves in each line. Failure of both check valves in one line is assumed to result in overpressurization and rupture of the RWSP suction piping, thus creating a reactor coolant leak in the RAB, outside of containment. Version A of the scenario begins with one SDC train in service and one in standby. Version B of the scenario begins with both SDC trains in service. The RCS leak can be isolated by shutting one of the suction isolation valves (MOVs SI-401, SI-407, or HOV SI-405) in the affected SDC train. Failure of the operators to detect, diagnose, and isolate the break will lead to core damage.

Human Actions For this ISLOCA HRA

The following table lists the PRA identifier and a brief description of each relevant human action identified for analysis in this HRA.

Seq. #	IDENTIFIER	DESCRIPTION
2	FTD	Control room (or operators) fail to detect LOCA
2	FTDGN	Control room fails to diagnose ISLOCA
2	FTI	Control room/operators fail to isolate break
5	FTD	Control room/operators fail to datect LOCA
5	FTDGN	Control room/operators fail to diagnose ISLOCA
5	FTI-A	Control room fails to isolate break with one train of shut down cooling in service
5	FTI-B	Control room fails to isolate break with both trains of shut down cooling in service

Table C4: Human Actions for the CE HRA

Modeling Of Human Actions And Estimated Human Error Probabilities

This section describes the HRA event trees and the HEP estimates for the human actions identified as significant for this ISLOCA HRA. An HRA event tree (with any associated fault trees), subtask HEP tables documenting HEP estimation for each subtask branch on the tree, and tables providing failure path calculations and total failure probability estimates are presented for each human action. Each set of trees and tables for a human action is preceded by a brief discussion relevant to the modeling and HEP estimation for that action.

FTD: Sequence 2

Event FTD for Sequence 2 models operator or control room failure to detect significant indications for a loss of coolant and enter OP-902-002, the Loss of Coolant Accident Recovery Procedure. The critical subtasks, detecting symptoms of a LOCA, i.e., decreasing pressurizer level and pressure, for FTD are modeled according to step B.2 of this procedure. The HRA modelir assumes that the control room is in Mode 1 operation with one RO performing a quarterly stroke test on the safety injection (SI) valves. It also assumes that two PIV check valves have failed and that as the RO opens the associated SI valve, there is an immediate overpressurization and break, resulting in a reactor trip and safety injection actuation signal (SIAS). Furthermorstress levels were modeled as being moderately high and the crew was judged to be experienced. High dependency (HD) was used to model the relationship between the CR's ability to detect decreasing pressurizer pressure (or level) and the CR's ability to detect the subsequent pressurizer alarms. High dependence was also modeled between the CRS and SS as they decide to enter OP-902-002. The HRA event tree, subtask quantifications, and total failure probabilities for this event have already been presented in Figure 3 and Tables C1 and C2. As listed in Table C2, the total mean failure probability for event FTD in this sequence is 0.0175 with an EF of 4.26. No credit was given to the CR for an alarmed pressure indicator (PI) between the two check valves because the CR generally disabled this annunciator by pulling its card, and because the stroke test procedure does not direct operators to check the pressure on this PI before stroking the valve open.

FTDGN: Sequence 2 (Using Procedure Only)

Event FTDGN for Sequence 2 models failure of the control room to correctly diagnose an ISLOCA using the diagnostic flow chart from OP-902-000 (and Attachment 1 of OP-902-002), the Emergency Entry Procedure. The critical subtasks for event FTDGN are modeled according to the requisite actions for the CRS and SS to correctly diagnose an ISLOCA. HRA modeling includes critical procedural steps from the diagnostic flow chart and verification of entry into OP-902-002, the Loss of Coolant Accident Recovery Procedure. Recovery paths are also modeled. Stress was assigned a PSF value of 2, i.e., moderately high, and high dependence was assessed between the SS and CRS during diagnosis of the event. The HRA event tree, subtask quantifications, and total failure probabilities are presented in Figure 4 and Tables C5 and C6. As listed in Table C6 the total mean failure probability for event FTDGN (Using procedure only) is 0.02 with an EF of 3.96.



C-31

i. 3man Action / Error	Basic Median HEP	Error Factor	Source/ THERP Table #	Step-by- Step or Dynamic	Modifier for PSFs	Modifier Source	THERP Depend- ency	Basic Mean HEP	Nominal Mean HEP	Error Facto
CRS fails to determine SG pressure not abnormally loss or dropping	0.006	3.0	T26-10 #3	585	2	720-16 <i>#</i> 4a	ZD	9.007499	0.014998	3.0
SS fails to determine if SG pressure not abnormally low or dropping	6.1	5.0	T20-22 #1	SBS	2	729-16 #4a	RD	0.161383	0.461383	5.8
									# 0.33097	2.3
CRS fails to determine that containment pressure not rising	5.095	3.0	T20-10#3	SBS	z	T20-15#4e	ZD	6.007499	0.014998	3.0
SS fails to determine containment pressure not rising	8.1	5.0	T20-22 #1	SBS	2	T20-16 #4#	HD	0.161383	8.661383	5.0
									# 0.33097	2.3
CRS fails to detect no activity in steam plant	0.006	3.8	T20-10#3	585	2	T29-16 #4s	ZD	6,007499	0.014998	3.0
SS fails to detect no activity in steam plant	0.1	5.6	T20-22 #1	SBS	2	T20-16 #44	HD	0.161,383	9.661385	5.0
	6.5								# 0.33097	2.3
CRS fails to enter OP-902-002 per attachment #1	0.006	3.0	T20-10 #3	SBS	2	T20-16 #4a	ZÐ	8.067499	0.014990	3.0
SS fails to verify entry into OP-902-802 per	9.1	5.0	T20-22 #1	SBS	2	T20-16 #4a	HD	0.161383	0.661383	5.0
	21.00				1.1					

Table C5: HE

Table C6: Failure Paths and Total Failure Probabilities (Seq. 2, FTDGN)

FT1: Sequence 2 (Using Procedure)

2

An extensive review of OP-902-002, the Loss of Coolant Accident Recovery Procedure, and multiple interviews with a past SS from this utility indicate that this scenario is beyond the scope of the appropriate procedure (which was written to cover a design basis LOCA only). As a result, there are no actions within this procedure that would guide operators to isolate the leak in this scenario (FTI - Sequence 2). Therefore, HRA modeling only included one subtask, "Control Room Fails to Isolate Break," with a failure probability of 1.0.

Since procedure OP-902-002 does not provide guidance pertaining to the isolation of an ISLOCA, the question becomes: "What will an average Control Room (SS, CRS, Shift Technical Advisor, and Technical Support Center) do to isolate this ISLOCA break?" Figures 6 and 7 and Tables C-9 - C-12, (following Table C-8) present a sensitivity analysis for Sequence 2 actions: events FTDGN and FTI. The sensitivity analysis is based on interviews with a past shift supervisor from the CE plant. It attempts to model the "average" CRS, SS, and Control Room response using the diagnostic flow chart from OP-9C2-002 and knowledge-based behavior, which lies outside of procedural guidance.



Figure 5: HRA Event Tree for Sequence 2 - ril (Using Procedure Only)

<u>ه او</u> ا		
Eac		
Nominal Mean REP	02	
Basic Mean HEP		
THERP Depend- ency		
Modifier Source		
Madifier for PSFs		
Step-by- Step or Dynamic	SBS	
Source/ THERP Table 7		
Busic Error Median Factor HEP	10 1.0	
Human Action / Ecror	Control room falls to lookete ISLOCA following OP-902-002	
	Human Action / Error Busic Error Source/ Step-by- Modifier Modifier Modifier Modifier Medifier Resc Nominal Error Median Factor THERP Step or for PSFs Source Depend- Mean Factor Rep	Human Action / Error Basic Error Step-by- Modifier THERP Rasic Nominal Error Median Factor THERP Step-by- Step-by- for PSFs Source Depend- Mean Factor HEP Table /* Dynamic Step-or Tor PSFs Source Depend- Mean Factor Op :902.4002 OP :902.4002 10 1.0 Step Step-by- Step-by- Lo

Table C7: HEPs for FTI - Sequence 2 (Using Procedure Only)

Press not atmortally low/or opping Press not abnormalil low or dropping -SS fails to determine if 36 -CHS fails to determine if SG containment press not rising -SS fails to determine that cotationent press, not rising -55 falls to detect no activity -CRS fails to determine that a - CRJ correctly determined b - 55 sea in steam plant not tow or dropping in steen plant Flace very -CAS fails to detect no 5 - Control room fails to diagnose that 50 pressure not abnormally /a - 55 see that conta sent press SOL FISIO Hecovery) activity location of break c - DRS determines that - 55 seeg tow or dropping containment pressure no activity steam pla-. 501 1 1 1011 e - CRS detects Sequence the activity in of 1 Steam plant Using procedure Know ledge-based g - Control room diagn location of brack FTDGN -(not vehav HRA Event Tree for Seq. 2, FTDGN-Using Procedure & Knowledge-based

Figure 6: Behavior

Table C9: 1	HEPs for	Seq. 2,	FTDGN-Us	ing	Pro	cedur	·e &	Kno	owle	edge	e - t -	be	Behavio
		Error Factor	30	5.0	2.3	91	8.8	57	3.0	8.6	2.3	5.0	a
	(0r)	Nonsiaal Mean HEP	860F 1010	0.661383	# 0.13097	966F10-0	08(19970	# 0.330MT	Bosch 10.0	0.661383	# 0.33097	0.774642	# 0.363672
	l behavi	Basic Mean HEP	0.007499	0.162383		0.907499	0.161383		0.007499	0.161383		0,774642	
	e-based	THERP Depend- cacy	82	GH .		0Z	CH .		402	HD.		62	
	cnowledg	Modifier Source	120-16 #4e	720-16 #4a		T.29-16 #4a	T20-16 #4a		T20-16 #4s	T29-16.#4s			
	roc. & I	Modifier for PSFs	-	*		N	24		ы	11		-	
	(using p	Step-by- Step or Dynamic	SBS	SBS		SBS	SBS	3	SBS	SBS		SBS	
	FTDGN	Source/ THERP Table #	T26-10 #3	120-22 #1		T20-10 #3	129-22 #1		120-10 #3	T20-22 #1		HCR. knowledge	
	nce 2;	Error	ę.	20		3.6	20		3.0	5.0		5.0	
	Seque	Basic Mediat HEP	8.00K	10		0.006	0.1		0.006	1.0		0.45	
	Table C9: HEPS for S	Human Action / Error	CRS fails to determine SG pressure not abnormally low or dropping	SS fails to determine if %C pressure not schenermally low or dropping		CRS fields to determine that conteinmont pressure not risibig	SS fiells to determine contributent pressure not ricknes		(RS fails to detect no activity in steam plant	SS fails to detect no activity in steam plant		Constrol room fails to diagnose correct location of beeak	
		1000		45		0	0		- 14	Sec		o.	

1 19 14	nie C.W. Failure Paths and Total Failure Probabilities IDGN (using proc. & knowledge-based behavior) - Sequence 2	
	Calculations	Results
	0.0145998 ± 0.33057	0.064963
	0.01 4995 x 0.01 4995 x 0.3 M97	0.000074
	6.01 4998 z 0.01 4995 z 0.01 4995 x 0.13697	6.060001
	0.014995 x 0.014998 x 0.014998 z 0.3672	D. OKNOKORI
	0.61 4996 x 0.03 49905 x 0. 36.2672	180000-0
	0.01 4998 x 0.01 4998 x 0.30097	0.000074
	0.01 4998 x 0.01 4998 x 0.362672	0.0600081
	0.014998 x 0.362672	0.005439
	0.654998 z 0.33697	0.00496.1
	0.01 49908 x 0.01 49968 x 0.33897	0.0666074
	0.014998 x 0.014998 x 0.362672	0.000081
	0.014998 x 0.362672	6.005439
	0.014998 x 3.33057	6.004963
	0.614998 x 0.362672	0.005439
	0.362672	0.362672
	Total Failure Probability	670
	Error Factor	2.02

Table C10: Failure Path & Total Failure Probability (Seq. 2, FTDGN-Knowledge-

FTI: Sequence 2 (Using Knowledge-based Behavior)

For event FTI, Sequence 2, using knowledge-based behavior, outside of procedures, entry into this action assumes successful diagnosis of an ISLOCA (using Attachment 1 of OP-902-002), successful diagnosis of the break's most probable location (i.e., the safeguards pump room on the minus 35-foot level of the RAB), <u>AND</u> an understanding that the break occurred when the RO opened one of the LPSI flow control valves (FCVs) on the affected safety injection train during a quarterly stroke test. Successful isolation of the break depends on the operators going outside of procedures, overriding their training (to not terminate SI without meeting SI termination criteria) and closing the correct LPSI FCV [SI-138, SI-139 (A or B)]. This could be done by successively closing individual LPSI FCVs, then monitoring the pressurizer to see if leve; and pressure are stabilizing.

Figure 7 shows an HRA fault tree which models this sequential closing of the four LPSI FCVs. FP#1 (failure path #1) models the probability of an RO failing to close the correct FCV valve by sequentially opening and closing each one: FCV1 is the probability of an RO failing to select the correct FCV on the first try (1 chance in 4 of being correct); FCV2 is the probability of an RO failing to select the correct FCV on the second try (1 chance in 3 of being prrect); FCV3 is the probability of an RO failing to select the correct FCV on the third try (1 chance in 2 of being correct); and HEP, which is the probability of selecting the correct valve on the fourth try but incorrectly activating the controls for that valve. FP#2 models correctly selecting the proper FCV on the first attempt (/FCV1, i.e., the complement, or probability of success, for FCV1) and the human error probability (HEP) for incorrectly activating the FCV controls. FP#3 models the probability of an RO failing to select the correct FCV on the first try (FCV1), correctly selecting the valve on the second attempt (/FCV2, the complement of FCV2), and the human error probability (HEP). FP#4 models the probability of an RO failing to select the correct FCV on the first try, the failure of an RO to select the correct valve on the second attempt, selecting the correct valve of the third attempt (/FCV3 - the complement of FCV3) and the human error probability (HEP). The total failure probability for this action is 0.019 with an EF of 2.97.

Small Break FTDGN: Sequence 2 (Using Knowledge-based Behavior)

For a small break ISLOCA (flange failure), event FTDGN is modeled in the same manner as before with the addition of subtask "G" - Control Room Fails to Diagnose Location of Break. This particular subtask was re-quantified using the HCR technique assuming that the operators had 26970 seconds available and it would take 1830 seconds to complete the task; the crew had average (nominal) training; the situation was a potential emergency; and the quality of the man-machine interface was fair. The knowledge-based non-response probability for this subtask was estimated to be 0.001. The resulting total failure probability was estimated to be 0.016.

Small Break FTI: Sequence 2 (Using Knowledge-based Behavior)

For event FTI (small break), Sequence 2, using knowledge-based behavior, outside of procedures, entry into this action once again assumes successful diagnosis of an ISLOCA (using Attachment 1 of OP-902-002), successful diagnosis of the break's most probable location (i.e., the safeguards pump room on the minus 35-foot level of the RAB), <u>AND</u> an understanding that the break occurred when the RO opened one of the LPSI flow control valves (FCVs) on the affected safety injection train during a quarterly stroke test. As before, successful isolation of the break depends on the operators going outside of procedures, overriding their training (to not terminate SI without meeting SI termination criteria) and closing the correct LPSI FCV [SI-138, SI-139 (A or B)]. This could be done by successively closing individual LPSI FCVs, then monitoring the pressurizer to see if level and pressure are stabilizing. The HRA event trees and total failure probability (0.019) for this action have already been presented in Figure 7 and Tables C-11 and C-12.



-	Table C11: HE	PS for	Seque	ence 2 ; F	TI-Usin	g Know	rledge-b;	ised Bel	havio		
	FIGHER ACTION / CITOR	Mediat HEP	Factor	THERP Table #	Step-or Dynamic	for PSFs	Source	Depend- ency	Mean HEP	Mean HEP	Facto
	FCV1-Operators fail to select correct FCV on 1st try	0.75	£.0						0.75	6.75	2,6
	$F(\nabla 2)$ Operators full to select correct $F(\nabla \mbox{ on } 2nd \mbox{ try}$	9.667	0.1						6.567	0.667	1.0
	FCV3-Operators fall to select connect FCV are 3rd key	9,5	1.0						0.5	0.5	1.9
	HEF Operators incorrectly activate PCV control	0.625	3.0	T39-12 #10	585	5	T25-16#5	2.0	8.633745	6.018747	3.0
	/FCV1-e.g. the complement of "A"	9.25	1.6						0.25	0.25	1.0
	IEP Operators incorrectly activitie control	2.003	3.0	7-2-12 610	SBS	5	T20-14#5	230	0.003747	0.028747	3.0
	FCVs-Operators fail to select cornect FCV	8.75	1.0						0,75	0.75	1.0
	/FCV2-Complement of "E"	0.353	6.0						0.333	6.333	1.0
	HEP Operators inco, recity accessic control	0.063	3.0	T29-12 #10	SBS	5	T29-16-#3	ZD	0.003749	0.018747	3.9
	FCV1-Operators fails to so leti correct FCV	6.75	£.0			A DECEMBER OF			8.75	9.75	1.0
	FCV2-Operators fail to select correct FCV	0.667	1.0						0.667	0.667	1.0
	/FCV3-Complement of "C"	0.5	1.0						0.5	0.5	1.0
15	HEP-Operators bicorrectly activate cuplered	0.003	3.6	720-12 #10	SBS		720-16 #5	70	0.963749	0.018747	3.0

ure l'robabiities	avier	ns Results	0.20040599	0.16046566	8.5946652	8(034689	e Probability 0.819	2.37
Me C12: Failure Paths and Total Fail	FTI-Using Knowledge-based 8	Calcula	位75米位467米日5大台3大台35米	0.25 x 0.5187 x7	0.75 x 0.335 x 0.0558747	0.75 > 0.667 ± 8.5 × 0.86814747	Total Failur	Error Factor
Tah		Failure Path	1 ABCD	2 #5F	1 2045/381	a murdk1.M		

Table C12: Failure Paths & Total Failure Probability (Seq.2, FTL-Knowledge-

63

FTD: Sequence 5

Event FTD for Sequence 5 represents operator failure to detect a reactor coolant leak resulting from failure of low pressure flanges in a suction line from the RWSP. The leak results from a failure of two check valves protecting RWSP piping from higher pressure in the LPSI lines used during shutdown cooling. HRA modeling includes detection by the RO of decreasing pressurizer level and pressure, and subsequent entry into procedure OP-901-046 (Shutdown Cooling Malfunction) at the direction of the control room supervisor (CRS). Control room (CR) personnel are given recovery credit for detecting decreasing pressurizer level (Hi-Lo and Lo-Lo PZR level alarms). Recovery credit for entering OP-901-046 is given for the shift supervisor (SS) advising the CRS to enter the procedure. The HRA event tree, subtask quantifications, and total failure probability are presented in Figure 8, Table C13, and Table C14, respectively. Table C14 lists the total mean failure probability for this event tree as 0.00758.



.....

C-47

to all

411

Jul 14



Table	C13:	HEPs 1	for S	eq. 1	5. FTD
1 10 10 1 10	N & W *	1.1.0.1.1.1.1	- Mar. 1. 1997		N

	A NAME AND A AND A AND A						U INCIECT		COULAI		
E	Human Action / Ecror	Basic Median HEP	Error Factor	Source/ THERP Table #	Step-by- Step or Dynamic	Modifier for PSFs	Modifier Source	TEERP Depend- ency	Basic Mean REP	Nominal Mean HEP	Error
1.00											
	R() fails to detect decreasing PZR level & pressure	6:003	3.0	120.9 #4	SBS	*	T20-16 #2a	0Z	0.003749	0.003749	3.0
	Control room fails to detect PZR Hi-Lo level alarm	100970	10.0	T20-23 #2a	SHS	-	T20-16 #2#	8	0.0002046	99230010	10.0
	Control room fails to detect PCR Lo-Lo level starts	0.001	16.0	128-23 6 25	SBS		T20-16 #2x	BD	0.0025463	0.561331	8
										à c.279364	12
1.1	C.R.S. fails to enter OP-991-046 - NDC Malfunction	9,005	10.0	139-4-01	SBS	ea	120-16 #6	92	0.013117	0.036635	10.0
	SS fails to direct CRS to enter OP 361,566	500 ú	10.0	120-6-83	SBS	**	120-16 #4	HD	0.013347	0.513317	405
										# 0.287565	si el

Table C14: Failure Paths and Total Failure Probabilities

Sequence 5: FTD-Operators Fail to Detent Loss of Coolant

Failure Path	Calcul	Calculations					
4 ABC	0.003749 x 0.000266 x 0.279304		•				
2 ABcDE	0.003749 x 0.000266 x 9.026635 x 0.283505						
3 ALDE	9.003749 x 0.02663. x 0.283505		0.000028				
4 aDE	0.826635 x 0.283505		0.007551				
	Total F	ailure Probability	0.008				
	Error F	actor	11.68				

-

Table

C14:

1

.....

FTDGN: Sequence 5

Event FTDGN for Sequence 5 represents operator failure to diagnose that the reactor coolant leak is outside of containment (in the RAB) and enter the section of procedure OP-901-046 (Shutdown Cooling Malfunction) relevant to the isolation of the leak. HRA modeling includes denction of the safeguards room flooding alarm by control room (CR) personnel, with recovery credit for detection of RAB radiation alarm(s) and increasing waste tank level by CR personnel. Step 2 of SUBSEQUENT OPERATOR ACTIONS (OP-901-046) directs operators to Attachment 6.1, System Leakage, based on relevant CR indications. Sequence 5 HRA modeling for event FTDGN includes entry into Attachment 6.1 on the direction of the control room supervisor (CRS) and recovery credit for the shift supervisor (SS) ad ing the CRS to do so. The HRA event tree, subtask quantifications, and total failure probability are presented in Figure 9 and Tables C15 and C16, respectively. Table C16 also lists the total mean failure probability for this event tree as 0.00755.



	Human Action / Error	Basic Median HEP	Error Factor	Source/ THERP Table #	Step-by- Step or Dynamic	Modifier for PSFs	Modifier Source	THERP Depend- ency	Basic Mean HEP	Nominal Mean HEP	Error Factor
	Control room fails to detect Safegnards Room Flooded alarm	0.9001	18.0	T20-23 #2a	SRS	2	T20-16 #4	ZÐ	8.000756	0.090532	10.0
8	Control room fails to detect RAB radiation alarm	6.0001	10.0	T20-?3#2s	SBS	2	T20-16 #4	ZÐ	0.000266	9.000532	16.9
	Control room falls to detect increasing waste tank level	6.003	3.0	T20-9 #4	SBS	2	T26-16 #4	ZÐ	0.003745	6.007499	3.6
5	CRS fails to entry section 6.1 of OP-901-046 - SDC Malfunction	0.005	10.0	T20-6 #3	SBS	2	T20-16 #4	ZÐ	9.013317	0.026635	10.0
	SS fails to direct CRS to enter section 6.1 of OP- 901-846	0.005	10.0	T20-6 #3	SRS	2	T20-16 #4	HD	0.013317	0.513317	5.0
										# 0.283505	2.5

Table C15: HEPS for Sequence 5, FTDGN-Operators Fail to Diagnose ISLOCA
Ti	able C16: Failure Paths and Total Failure Probabilities	
	Sequence 5: FTDGN Opcrators Fail to Diagnose ISLOCA	
Failure Path	Calculations	Results
ABC	0.000532 x 0.000532 x 0.007499	*
ABcDE	0.0/00532 x 0.000532 x 0.026635 x 0.283505	•
Abity	0.000532 x 0.026635 x 0.283505	0.000604
aDF	0.026635 x 0.283505	0.007554
	Total Failure Press Vility	0.008

2

.

FTI-A: Sequence 5 (1 Train of SDC)

Event FTI-A for Sequence 5 represents operator failure to isolate the leak (see Sequence 5 description in Sequences and Human Actions section) by performing the relevant steps of Attachment 6.1 (System Leakage) of procedure OP-901-046 (Shutdown Cooling Malfunction), when one SDC train is in service and one is in standby. The HRA event tree for event FTI-A models the critical steps, substeps, and actions represented in steps 9 and 10 of Attachment 6.1 related to isolation of the leak. Steps/substeps modeled include: closing the SDC suction isolation valve on the train in standby, observation of RCS level for stabilization, placing the isolated SDC train in service (referencing OP-009-005, system operating procedure, Shutdown Cooling System), and closing the SDC suction isolation valve on the initially operating SDC train. The HRA modeling conservatively assumes that the leak is in the second train isolated. Omission and commission errors are modeled for each step/substep. Omission errors are modeled as errors by the control room supervisor (CRS), with recovery credit for the reactor operator (RO). A high level of dependence was modeled between the two (CRS & RO), when the CRS was directing RO actions. Commission errors are modeled as errors by the RO, with recovery credit for the CRS. For these actions, a moderate level of dependence was modeled between the RO performing the action and the CRS who would be concurrently performing other dynamic actions. The HRA event tree, subtask quantifications, and total failure probability are presented in Figure 10, Table C17, and Table C18, respectively. Table C18 also lists the total mean failure probability for this event tree as 0.0233.

Modeled for one SOC train in service & one in standby valve SI 4014(B) per step 9.1 of 0P-901-046 A - CHS fails to close SDC Loop Suc. 180 201 CHS. D - CRS fails to remind RU to close H - HO fails to remind close SI 401 A(B) RCS F - RO fails to remind CRS to determine if E - CRS falls to determine RCS level not stabilizing 150 C + RO fails to close SDC loop suc. velve St 401 A(B) (Commission) (unissim) step 18 of OP-901-046 (Omission) SI 401 ACB) level not stabilizing H - CHS fails to fail AD to determine IF RCS tevel is not stabilizing Suc. 150 Valva 401 A(8) 6 - R0 remin for SDC train in standby/ CRS to close SI 401 ACB) G - RD fails to correctly determine if (Hecovery) ACS level not stabilizing d - CRS tel a - CRS remembers to close SDC Loop RD to clos SI 401 A(B) (Recovery) (Commission) per c - R0 closes SDC toop suc. ISC. VAIVE SI 401 A(B) NOTE: - RO remi CRS to chec RCS level Isclate [Hecovery] ** e - CHS remembers to check if RCS level AXX ACS level not stabilizing h - CAS tel RD to check RCS level Recovery) Operators Fail to 11 g - R0 correctly determines philidels for Sequence Tree for Sequence 5, FTI-A (1 SDC Train) Figure HRA Event 10:

Modeled for one SDC train in service & one in standby J - RO fails to remind CRS to place - CRS fails to place isolated SOC train in isolated SDC train in service service (0P-009-005) per step 10.1 of N - RD fails to remind CRS to close Si 401 A(B) on L - CRS fails to verify RO placed correct SDC train in service K - RO fails to place isolated SDC train 0P-901-046 (Omission) operating SDC train per step 12 3 (Omission) M - CRS fails to close Si 401 ACB3 on Initially P - CRS fails to verify R0 closed SI 401 A(B) initially operating SOC train in service (Commission) 0 - RO fails to close SI 401 A(B) (Commission) A - R0 remines 5 SDC train (Hecovery) Service i - CRS remembers to place (solated SIX A 1 - CPS te RO to place, correct SDC ain in serv (Recovery) k - R0 correctly places isolayed - RO remind ** NOTE: CRS to close 51 401 ACB) SDC train in service to Isolate train in service Recovery) m - CRS remembers to close SI 401 A(B) XXX p - CRS tel BO to clo SI 401 ACE (Recovery) FT I - U AO correctly closes SI 401 AVB) -- 64 (continued Operators Sequence Figure 10 (cont.):

	Human Action / Error	Basic Mediat HEP	Error Factor	Source/ THERP Table #	Step-by- Step or Dynamic	Modifier for PSFs	Modifier Source	THERF Depend- ency	Basic Mean HEP	Nominal Mean HEP	Erroi Facto
	CRS falls to close SDC loop Soc. Iso. valve SI- 401 A(B) (Omission)	0.003	3.0	T20-7 #3	SBS	3	T20-16 #2	ZD	0.003749	0.011248	3.0
ŝ	RO fails to remind CRS to close SI-401 $A(B)$	0.1	5.0	T26-22 #1	SBS	3	T20-16 #2	HD	0.161383	0.742075	5.0
			1944		B. 314	8-19 st		17.71		# 0.353909	2.2
	RO fails to close SI-401 A(B) (Commission)	6.001	3.0	720-12#3	SBS	3	T20-16 #2	ZD	0.001249	0.003749	3.8
	CRS fails to remind RO to close SI-401 $A(B)$	0.1	5.0	T20-22 #1	SBS	3	T20-16 #2	MD	0.161383	0.557844	5.8
										# 0.298605	2.4
	CRS fails to determine in RCS level not stabilizing (Omission)	6.003	3.0	T20 7 #3	SBS	3	T20-16 #2	ZÐ	0.003749	6.011248	3.0
	RO fails to remind CRS to determine if RCS level not stabilizing	0.1	5.0	T20-22 ∉1	SBS	3	T20-16 #2	ĦD	0.161383	0.742075	5.0
		ET :	14.4			199				# 0.353909	2.2
	RO fails to correctly determine if RCS level not stabilizing (Commission)	0.003	3.6	T26-10 #1	SRS	3	T20-16 #2	ZD	0.003749	0.011248	3.0
	CRS fulls to tell RO to determine if RCS level	0.1	5.0	T20-22 #1	SBS	3	T26-16 #2	MD	0.161383	0.557844	5.0
	not stantszing									# 0.395405	
			100		1.00					# 0.1748040	2.4

C-58

Table C17: HEPS

Table C17: HEPS for Sequence 5, FTI-A; Fail to Isolate (1 SDC Train) Human Action / Error Basic Error Source! Step-by- Modifier Modifier THERP Basic Nominal Error Median Factor THERP Step or for PSFs Source Depend-Mean Mean Factor NEP Table # Dynamic ency HEP HEP 3 RO fails to reinind CRS to place standby SDC 1.0 5.0 T20-22 #1 SBS T20-16 #2 HD 3 0.161 -73 0.742075 5.8 train in service # 0.353909 2.2 K RO fails to place isolated SDC train in service 6.001 3.0 720-12#3 SBS T20-16 #2 9.001249 3 20 0.003749 3.0 1. CRS fails to verify RO placed correct SDC tr in 0.1 5.0 T20-22 #1 SBS 3 T20-16 #2 MD 0.141383 0.557844 5.0 to service # 0.298605 2.4 NO. CRS fails to close SI-401 A(B) on initially 0.003 3.0 T20-7 #3 SBS T20-16 #2 ZD 0.007749 0.011248 3 3.0 operating SDC train 86 RO fails to i - and CRS to close SE-401 A(B) on 0.1 5.0 T20-22 #1 SBS 3 T20-16 #2 HD. 5161383 0.742075 5.0 initially operaling SDC train # 0.353909 2.2 0 RO fails to close correct SI-401 A(B) valve 9.001 3.6 T28-12#3 SBS 3 T20-16 #2 ZD 0.001249 0.003749 3.0 (Commission) P . CRS fails to verify RO closed correct SI-401 0.557844 0.1 5.0 T20-22 #1 SBS T20-16 #2 MD 0.161383 3 5.0 A(B) valve # 0.298505 2.4

	Tab	ole C18: Failure Paths and Total Failure Probabilities Sequence 5: FTI-A; Fail to Isolate (1 SDC Train)	
	Failure Path	Calculations	Results
I AB		0.011248 x 0.353909	0.00398
z AbCD	,	0.011248 x 0.003749 x 0.298605	0.00012
3 SbCd	EF	0.011248 x 0.003749 x 0.011248 x 0.353909	•
4 AbCd	EIGH	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.298605	
s AbCd	EfGhIJ	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.311248 x 0.353909	•
i AbCd	ERGAJJKL	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.001248 x 0.003749 x 0.298605	
7 AbCd	ERGEUKIMN	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	
8 AbCd	EfGhljKIMnOP	0.011248 x 0.903749 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	•
9 AbCd	EfGhijKimOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.298605	•
10 AbCd	ERGHURMIN	0.011248 x 9.063749 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.353909	· Charles in
11 AbCd	EKbijkMeOP	0.011248 x 0.003749 x 0.011248 x 0.011245 x 0.011248 x 0.011248 x 0.003749 x 0.295605	•
га Арса	EfGhijkneOP	0.011245 x 0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.063749 x 0.298605	•
13 AbCd	ERGHIKL	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
14 AbCd	ERGHIKIMN	0.611248 x 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	
15 AbCd	ERGhiKIMnOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.228605	•
16 Ab-Cd	EfGhiKimOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.298605	
17 AbCd	ERGhikMN	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.011248 z 0.353909	•
18 AbCd	I/SIGhikMnOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	
19 AbCd	iE/GhikmOP	0.011248 x 0.003749 x 0.01248 x 0.011248 x 0.003749 x 0.298605	
20 AbCd	IE/gLj	0.011248 x 0.003749 x 0.011248 x 0.011245 x 0.353909	

Failure Path	Calculations	Results
11 AbCdEigijki.	0.011248 x 0.003749 x 0.011245 x 0.011248 x 0.003749 x 0.298605	
22 AbCdEIgljKIMN	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	•
23 AbCdEfgljKiMnOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	*
4 AbCdEIgljKlmOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.298605	•
5 AbCdEIgIjkMN	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.353909	
6 AbCdEigijkMnOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	
7 AbCdElgljkmOP	011248 x 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605	*
adcafigiki.	0.011748 x 0.003749 x 0.011248 x 0.003749 x 0.298605	*
19 AbCdFIgiKIMN	0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.353909	*
M AbCdEIgiKIMBOP	0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	
31 AbCdEIgiKImOP	0.011248 0.003749 x 0.011248 x 0.003749 x 0.003749 x 0.298605	
12 AbCdEighMN	0.011248 x 0.003749 x 0.011246 x 0.011248 x 0.353909	
3 AbCdEIglkMaOP	0.011248 ± 0.003749 ± 0.011248 ± 0.011248 ± 0.003749 ± 0.298605	•
4 AbCdEIgikmOP	0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	•
35 AbCdeGH	0 711248 x 0.003749 x 0.011248 x 0.298605	•
36 AbCdeGhIJ	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.353909	*
37 AbCdeGbIJKL	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
38 AbCdeGhIJKIMN	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	•
39 AbCdcGhIJKIMnOP	0.011248 x 0.003749 x 0.011248 x 0.01248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	
40 AbCdeGhljKtmOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.063749 x 0.003749 x 0.298605	
41 AbCdeGhljkMN	0.011248 x 0.00374y x 0.011248 x 0.011748 x 0.011248 x 0.353909	•
42 AbCdeGhlikMaOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	

÷.

Table C18: Failure Paths and Total Failure Probabilities

Sequence 5: FTI-A; Fail to Isolate (1 SDC Train)

	Sequence 5: FTI-A; Fail to Isolate (1 SDC Train)	
Failure Path	Calculations	Results
kmOP	0.011248 x 0.003749 x 0.012248 x 0.011248 x 0.903749 x 0.298605	•
1	0.0E1248 x 0.303749 x 0.011248 x 0.003749 x 0.298605	
IMN	0.011248 x 0.063749 x 0.011248 x 0.063749 x 0.011248 x 0.53909	•
OMnOF	0.011248 x 0.303749 x 0.511248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	•
GmoP	0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.003749 x 0.003749 x 0.298605	
MN	0.011248 ± 0.063749 ± 0.011248 ± 0.611248 ± 0.353909	•
MaOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298685	
mOP	0.011 x 0.003749 x 0.011248 x 0.303749 x 0.298605	•
	0.011248 x 0.003749 x 0.011238 x 0.353909	
	0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	•
MN	0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.353909	•
IMnOP	0.011248 x 0.003749 x 0.011245 x 0.003749 x 0.011245 x 0.003749 x 0.295405	
IntOP	0.011248 x 0.003749 x 9.031248 x 0.003749 x 0.003749 x 0.208605	•
46N	0.011248 x 0.063749 x 0.011248 x 0.011248 x 0.051248 x 0.352909	•
AuOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
#OP	0.011248 ± 0.003/49 ± 0.011248 ± 0.003749 ± 0.298605	•
	0.011248 z 0.003749 x 0.003749 x 8.298605	¥
MN	0.011248 x 0.003749 x 0.003749 x 0.011248 x 0.353909	•
MaOP	0.011248 x 0.003749 x 0.003749 x 0.011248 x 0.003749 x 0.003749 x 0.298605	
NOP	0.011248 x 0.003749 x 0.003749 x 0.003749 x 0.03749 x 0.238605	
(N	0.011248 x 0.003749 x 0.011248 x 0.353909	

in.

.

Table C18 (cont.):

0

	Table C18: Failure Paths and Total Failure Probabilities	
	Sequence 5: FTI-A; Fail to Isolate (1 SDC Train)	
railure Pa	th Calculations	Results
δ5 AbCdegikmOP	0.011248 x 0.003749 x 0.003749 x 6.298605	
66 AbcEF	0.011248 x 0.011248 x 0.353909	0.000044
67 AbcERGH	9.011248 x 0.011248 x 0.011248 x 0.298605	
68 AbcERGhij	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.353909	•
69 AbcEfGhijKl.	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	
70 AbcElGhIJKIMN	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	
71 AbcElGhIJKIMnOP	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.296605	•
72 AbcERGhljKimOP	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.0037 +> x 0.295605	
73 Ab ERGhijkMN	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0 353909	•
74 AbcEfGhljkMnOP	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
75 AbcErChlikmOP	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
76 AbcElGhiKL	0.011245 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
77 AbcEfGbiKIMN	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	
78 AbcE#GhtKIMnOP	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	•
79 AbcEfGhiKimOP	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.298605	•
80 AbcEfGhikMN	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.353909	•
81 AbcEfGhikMnOP	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	*
82 AbcEfGbikm()P	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	*
83 AbcEfgIJ	0.011248 x 0.011248 x 0.011248 x 0.353909	•
84 AbcE/gljKL	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
85 AbcElgIJKIMN	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	•
S6 Aby Efgelik (MenOP	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	

.

	Sequence 5: FTI-4; Fail to Isolate (1 SDC Train)	
Failure Path	Calculations	Results
87 AbcEfgljKimOP	0.01 248 x 0.011248 x 0.013248 x 0.003749 x 0.003749 x 0.298605	
88 AbcEIgIJAMN	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.353909	•
89 AbcEIgijkMaOP	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
90 AbcEfgljkmO?	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
91 AbcFIgiKL	0.0%12~8 ± 0.011248 ± 0.003749 ± 0.298605	ales en carre de la com
92 AbcElgiKIMN	0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	• • •
93 AbcEfgiKiMaOP	6.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	· · · · · · · · · · · · · · · · · · ·
94 AbcElgtKlimOP	0.011248 x 0.011248 x 0.003749 x 0.063749 x 0.298605	•
95 AbcEfgikMN	9.011248 x 0.011248 x 0.011248 x 0.353909	
96 AbcEfgikMinOP	0.011248 x 0.011248 x 0.011249 x 0.003749 x 0.298605	The state water and
97 AbcEIgikarOP	0.011248 x 0.011248 x 0.003749 x 0.298605	(1) • • • • • • • • •
95 AliceGH	3,011248 x 0.011248 x 0.298605	0.000037
99 AbceGhU	0.011248 x 0.011248 x 0.011248 x 0.353909	•
100 AbceGhljKL	0.017248 x 0.011248 x P.01248 x 0.003749 x 0.298605	e
101 AbceGhljKIMN	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011243 x P 573909	•
102 AbceGhljKiMnOP	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003 49 x 0.298605	
103 AbceGhljKimOP	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.298605	
104 AborChijkMN	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.353909	
195 AbceGhijkMnOP	0.011248 x 0.0112-3 x 3.011248 x 0.011248 x 0.003749 x 0.298605	
106 AbceGhljkmOP	0.011248 x 0.011248 x 9.011248 x 0.003749 x 0.298605	•
107 AbceGhiKL	0.011248 x 0.011248 x 0.003749 x 0.298605	•
108 AborChiKIMN	9.911248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	

*

Table C18: Failure Paths and Total Failure Probabilities

Table C18: Failure Paths and Total Failure Probabilities Sequence 5: FTI-A; Fail to Isolate (1 SDC Train) Failure Path Calculations Results 109 Abcellbik (MinOV) 0.011248 x 0.019248 x 0.003749 x 0.0112348 x 0.003749 x 0.298605 . 110 AbceGhiKima... 0.011248 x 9.011248 x 0.003749 x 0.903749 x 0.298605 . 111 Abc-OhikMN 8.011248 x 0.051248 x 0.011248 x 0.353909 . AborChikMnOP 0.011248 x 0.011248 x 0.011245 x 0.003749 x 0.298405 112 . E13. AbceGhlkonOP 8.011248 x 0.011245 x 9.003269 x 0.298605 * 114 Abcegij 0.011248 ± 6.011248 ± 0.353909 O (NEW)44 AhorgijK1. 115 8.011248 x 8.011248 x 8.003749 x 9.798505 . AbcegtikiMN £ 011248 ± 6.011348 ± 0.003749 ± 0.011248 ± 0.353909 æ 126 AbcegljKIMcOF 117 8-911246 x 0 011248 x 0.003/49 x 0.012245 x 0.003749 x 0.298685 . AhregUKImOP 0.011348 x 0.011245 x 0.003749 x 0.003749 x 0.298605 118 * 119 AborgUANIN 0.011248 + 0.011247 + 0.011248 + 0.353700 . 6.011248 x 0.011248 x 0.011245 x 0 003749 x 0.298605 1.20 AbceglikMnOP ÷ 121 Aborg(JkmOP 0.011248 x 0.011248 x 0.003749 x 0.298605 10 Abcegth1. 8.011248 x 0.003749 x 0.298685 6,000037 122 AbergikiMN 6.011258 x 0.003749 x 0.011248 x 0.953909 . 123 AborgiklisteOF 0.021248 ± 0.003749 x 0.011248 x 0.003749 x 0.298665 . 124 AbcogtXimOP 0.011248 x 0.005749 x 0.003749 x 0.298605 * 125 AbregikNiN 8.02 - 248 x 8.011248 x 0.353999 0.000044 1.26 0.011248 x 3/011248 x 0.003749 x 0.298605 Abergelahtselle . 127 AbceptimOP 4.011288 ± 0.003749 ± 0.298605 0.069012 128 £.003749 x 0.298605 0.001119 125 af 13 0.003749 x 0.011248 x 0.353909 0.000013 130 aCdEF

	Tab	le C18: Failure Paths and Total Failure Probabilities	
		Sequence 5: FTI-A; Fail to Isclate (1 SDC Train)	
	Failure Path	Calculations	Results
131	oCdEKCR	0.003749 x 0.611740 x 0.011243 x 0.298605	
132	«CdErGb3	0.063749 x 0.011248 x 0.011248 x 0.011248 x 0.353905	•
133	⊯CdElGhHKI.	0.201749 x 0.611248 x 0.011248 x 0.012248 x 0.003749 x 0.298603	
136	*CdErGhljKIMN	0.003749 x 0.011242 x 0.011245 x 0.011243 x 0.003749 x 0.011248 x 0.553999	•
135	aCdESGhijXAtoAF	0.0K 37.49 ± 0.011248 ± 0.011248 ± 0.011248 ± 0.001248 \pm 0.001248	*
1.36	*CdEfGhljKhwOP	0.603749 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 9.0.3749 x 0.298605	•
137	*CAFRIEIJEMN	6.003749 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.353909	•
138	sCdEIGhljkMoO?	0.06'3749 x 0.611248 x 0.011247 x 0.611248 x 0.011248 x 0.003749 x 0.295605	•
1.39	*CdERGhijkmOP	0.003747 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
1.40	sCdERGBIKL	0.003749 x 0.011248 x 0.011248 x 0.003749 x 7.298605	• • • •
141	*CdErGBIRIMN	0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	•
142	3C4ERGHKIMaOP	0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 8.298605	•
143	aCdErGhiKhmOP	0.501749 x 0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.293605	•
145	aCdF#GhikMN	0.003749 x 0.011248 x 0.011248 x 0.721248 x 0.353909	*
1.45	#CdEIGhikMoGP	0.003749 x 0.011248 x 0.011248 ; 0.011248 x 0.003749 z 9.299605	
146	aC4EEGhikmOP	0.003749 x 0.011248 x 0.011248 x 0.03749 x 0.298505	•
147	eCdFfel]	0.063779 x 0.011248 x 0.011248 x 0.353909	
1.48	aCdEbyl(K).	6.003749 x 0.011248 x 9.011248 x 0.003749 x 9.298695	•
149	•CdEfellKIMN	0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	•
. 50	af dEfelik MaOP	0.903749 x 0.611248 x 0.051248 x 0.063749 x 0.051248 x 0.003749 x 0.298605	•
151	aCdE2aliKimOP	0.003.49 x 0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.208605	•
152	oCdEIgljkMN	6.003749 x 0.011248 x 0.011248 x 0.011245 x 0.353909	

÷

Table C18

(cont.):

C-66

5

Failure Path Calculations Ref 53 #CdE2gljBAM-OP 6.003749 x.0.011248 x.0.011248 x.0.003749 x.0.298605 6.003749 x.0.298605 6.003749 x.0.298605 6.003749 x.0.298605 6.003749 x.0.298605 6.003749 x.0.298605 6.003749 x.0.003749 x.0.298605 6.003749 x.0.003749 x.0.298605 6.003749 x.0.003749 x.0.003749 x.0.298605 6.003749 x.0.003749 x.0.003749 x.0.298605 6.003749 x.0.003749 x.0.003749 x.0.298605 6.003749 x.0.011248 x.0.003749 x.0.298605 6.003749 x.0.011248 x.0.003749 x.0.298605 6.000012 54 cdE2glkMNOP 0.003749 x.0.011248 x.0.003749 x.0.298605 6.000012 6.000012 55 cdE2glkMNOP 0.003749 x.0.11248 x.0.003749 x.0.298605 6.000012 56 cdE2glkMNOP 0.003749 x.0.11248 x.0.003749 x.0.298605 6.000012 56 cdE4cBlkMOP 0.003749 x.0.11248 x.0.011248 x.0.003749 x.0.298605 6.000012 56 cdE4cBlkMOP 0.003749 x.0.11248 x.0.003749 x.0.298605 6.0000012 56 cdE4cBlkMOP 0.003749 x.0.11248 x.0.003749 x.0.298605 6.0000000 56 cdE4cBlkMOP 0.003749 x.0.011248 x.0.003749 x.0.298605 6.000000000000000000000000000000000000		Sequence 5: FTI-A; Fail to Isolate (1 SDC Train)	
NGAElegitykawoP 0003749 x0.011248 x 0.001749 x 0.298605 • 54 ACGAElgitykawoP 0003749 x0.011248 x 0.003749 x 0.298605 • 55 ACGAElgitykawoP 0003749 x0.011248 x 0.003749 x 0.298605 • 56 ACGAElgitykawoP 0003749 x0.011248 x 0.003749 x 0.298605 • 57 ACGAElgitykawoP 0003749 x0.011248 x 0.003749 x 0.298605 • 58 ACGElgitykawoP 0003749 x0.011248 x 0.003749 x 0.298605 • 59 ACGElgitykawoP 0003749 x0.011248 x 0.003749 x 0.298605 • 60 ACGElgitykawoP 0.003749 x0.011248 x 0.003749 x 0.298605 • 61 ACGElgitykawoP 0.003749 x0.011248 x 0.003749 x 0.298605 • 61 ACGAElgitykawoP 0.003749 x0.011248 x 0.003749 x 0.298605 • 61 ACGAElgitykawoP • • • 61 ACGAElgitykawoP • • • 62 ACGElgitykawoP • • • 63 ACGElgitykawoP • • • 64 ACGAElgitykawoP <	Failure Path	Calculations	Kesults
Sol and spin Biological Biolo	ca of defails MaOP	0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	
sin strange security security security 55 acdaErgikKi 0.000779 x 0.011248 x 0.003749 x 0.011248 x 0.033749 x 0.298605 security 56 acdaErgikKiMaOP 0.000379 x 0.011248 x 0.003749 x 0.003749 x 0.298605 security 57 acdaErgikKiMaOP 0.00379 x 0.011248 x 0.003749 x 0.003749 x 0.298605 security 58 acdaErgikKiMaOP 0.003749 x 0.011248 x 0.003749 x 0.298605 security 50 acdaErgikKiMaOP 0.003749 x 0.011248 x 0.003749 x 0.298605 0.000012 50 acdaErgikKiMAOP 0.003749 x 0.011248 x 0.003749 x 0.298605 0.000012 51 acdaErgikKiMAOP 0.003749 x 0.011248 x 0.003749 x 0.298605 0.000012 52 acdaErgikKiMAOP 0.003749 x 0.011248 x 0.003749 x 0.298605 security 53 acdaErgikKiMAOP 0.003749 x 0.011248 x 0.003749 x 0.003749 x 0.01248 x 0.003749 x 0	sa «CdFlatikmiDP	8.003749 x 6.011248 x 6.011248 x 0.003749 x 0.298605	
Sol average everage everage 50 average 0.002749 x 0.011248 x 0.003749 x 0.21368 x 0.003749 x 0.298605 ele 51 average 0.003749 x 0.011248 x 0.003749 x 0.298605 ele 52 average 0.003749 x 0.011248 x 0.003749 x 0.298605 ele 53 average 0.003749 x 0.011248 x 0.003749 x 0.298605 ele 54 average 0.003749 x 0.011248 x 0.003749 x 0.298605 ele 54 average 0.003749 x 0.011248 x 0.003749 x 0.298605 ele 54 average 0.003749 x 0.011248 x 0.003749 x 0.298605 ele 55 average 0.003749 x 0.011248 x 0.01749 x 0.298605 ele 54 average 0.003749 x 0.011248 x 0.01749 x 0.298605 ele 55 average 0.003749 x 0.011248 x 0.01749 x 0.298605 ele 56 average 0.003749 x 0.011248 x 0.01749 x 0.298605 ele 56 average 0.003749 x 0.011248 x 0.01749 x 0.298605 ele 56 average 0.003749 x 0.011248 x 0.01749 x 0.298605 ele 56 average 0.003749 x 0.011248 x 0.01749 x 0.298605 ele 56 average 0.003749 x 0.011248 x 0.01749 x 0.298605 ele 56 avev	st and grant	0.003749 x 0.011245 x 0.003749 x 0.298605	
30 ac.def Sig Mixing OP 0.003749 x 0.011248 x 0.003749 x 0.01349 x 0.298603 e 57 ac.def Sig Ki Mixo OP 0.003749 x 0.011248 x 0.003749 x 0.098605 e 59 ac.def Sig Ki Mixo OP 0.003749 x 0.011248 x 0.003749 x 0.298605 e 60 ac.def Sig Ki Mixo OP 0.003749 x 0.011248 x 0.003749 x 0.298605 e 61 ac.def Sig Ki Mixo OP 0.003749 x 0.011248 x 0.013749 x 0.298605 0.000012 62 ac.def Sig Ki Mixo OP 0.003749 x 0.011248 x 0.013749 x 0.298605 e 63 ac.def Sig Ki Mixo OP 0.003749 x 0.011248 x 0.013749 x 0.298605 e 64 ac.def Sig Ki Mixo OP 0.003749 x 0.011248 x 0.013749 x 0.298605 e 65 ac.def Sig Ki Mixo OP 0.003749 x 0.011248 x 0.013749 x 0.298605 e 66 ac.def Sig Ki Mixo OP 0.003749 x 0.011248 x 0.013749 x 0.298605 e 67 ac.def Sig Ki Mixo OP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 e 68 ac.def Sig Ki Mixo OP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 e 69 ac.def Sig Ki Mixo OP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 </td <td>ez aCatiotKMN</td> <td>0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.353909</td> <td>•</td>	ez aCatiotKMN	0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.353909	•
3 acdatogenemics e 53 acdatogenemics e.0003749 x.0.003749 x.0.003749 x.0.098605 e.0 54 acdatogenemics e.0003749 x.0.011248 x.0.003749 x.0.298605 e.0 50 acdatogenemics e.0003749 x.0.011248 x.0.003749 x.0.298605 e.0 61 acdatogenemics e.0003749 x.0.011248 x.0.003749 x.0.298605 e.0 61 acdatogenemics e.0003749 x.0.011248 x.0.098605 e.0 62 acdatogenemics e.0003749 x.0.011248 x.0.093749 x.0.298605 e.0 63 acdatogenemics e.0003749 x.0.011248 x.0.013248 x.0.093749 x.0.298605 e.0 64 acdatogenemics e.0003749 x.0.011248 x.0.013749 x.0.298605 e.0 64 acdatogenemics e.0003749 x.0.011248 x.0.013749 x.0.298605 e.0 65 acdatogenemics e.0003749 x.0.011248 x.0.0013749 x.0.298605 e.0 66 acdatogenemics e.0003749 x.0.011248 x.0.0013749 x.0.298605 e.0 67 acdatogenemics e.0003749 x.0.011248 x.0.0013749 x.0.298605 e.0 68 acdatogenemics e.0003749 x.0.011248 x.0.0013749 x.0.01248 x.0.003749 x	20 acting internet	0.003749 x 0.611248 x 0.003749 x 0.011248 x 0.003749 x 0.298663	•
Sa R. Groger March e. C. de Sge March e. S. de Sge March	F8 =C4EtalKind)P	0.003749 x 0.011248 x 0.003749 x 0.003749 x 0.298605	*
53 a. Carbagais HX 0.003749 x 0.011248 x 0.003749 x 0.298605 • 60 a/CdE2gikMoOP 0.003749 x 0.011248 x 0.003749 x 0.298605 \$ 61 a/CdE2gikMoOP 0.003749 x 0.011248 x 0.093749 x 0.298605 \$ 62 a/CdeCH 0.003749 x 0.011248 x 0.01248 x 0.353909 \$ 63 a/CdeChUKI 0.003749 x 0.011248 x 0.01248 x 0.0298605 \$ 64 a/CdeChUKI 0.003749 x 0.011248 x 0.01248 x 0.0298605 \$ 65 a/CdeChUKINN 0.003749 x 0.011248 x 0.01248 x 0.0298605 \$ 66 a/CdeChUKINN 0.003749 x 0.011248 x 0.01248 x 0.003749 x 0.298605 \$ 67 a/CdeChUKINNOP 0.003749 x 0.011248 x 0.01248 x 0.003749 x 0.298605 \$ 68 a/CdeChUKINOP 0.003749 x 0.011248 x 0.01248 x 0.003749 x 0.298605 \$ 69 a/CdeChUKINNOP 0.003749 x 0.011248 x 0.01248 x 0.003749 x 0.298605 \$ 60 a/CdeChUKINNOP 0.003749 x 0.011248 x 0.01248 x 0.003749 x 0.298605 \$ 60 a/CdeChUKINNOP 0.003749 x 0.011248 x 0.003749 x 0.298605 \$ 60 a/CdeChUKINNOP 0.0037		0.003749 x 0.011248 x 3.011248 x 0.353909	*
80 a. degab. Mick 9 61 wCdEfgikamOP 0.0003749 x 0.011248 x 0.009749 x 0.298605 0.000012 62 wCdeGH 0.0003749 x 0.011248 x 0.098605 0.000012 63 mCdeGhU 0.0003749 x 0.011248 x 0.011248 x 0.053749 x 0.298605 0.000000 64 wCdeGhUKL 0.0003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 0.000000 65 wCdeGhUKIMNOF 0.0003749 x 0.011248 x 0.001749 x 0.011248 x 0.003749 x 0.298605 0.0000000 66 wCdeGhUKIMNOF 0.0003749 x 0.011248 x 0.001749 x 0.011248 x 0.003749 x 0.298605 0.00000000000000000000000000000000000	Sy at designation	0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
61 #C. AP 200 MIN PA 0.0003749 x 0.011248 x 0.299605 0.000012 162 #C.der GH 0.0003749 x 0.011248 x 0.0153909 # 163 #C.der Gh LJ 0.0003749 x 0.011248 x 0.011248 x 0.053799 x 0.298605 # 164 #C.der Gh LJKI. 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 # 165 #C.der Gh LJKI. 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 # 166 #C.der Gh LJKI.MnOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 # 167 #C.der Gh LJKI.MNOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 # 168 #C.der Gh LJKI.MNOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 # 169 #C.der Gh LJKI.MOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 # 169 #C.der Gh LJKI.MOP 0.003749 x 0.011248 x 0.003749 x 0.298605 # 170 #C.der Gh LJKI.MOP 0.003749 x 0.011248 x 0.003749 x 0.298605 # 171 #C.der Gh LJKI.MN 0.003749 x 0.011248 x 0.003749 x 0.298605 # 171 #C.der Gh LJKI.MN 0.003749 x 0.011248 x 0.003749 x 0.298605	and an an approximation	0.003749 x 0.011248 x 0.903749 x 0.298605	•
62 #CdeGhT 0.003749 x 0.011248 x 0.01248 x 0.353909 # 163 #CdeGhLJ 0.003749 x 0.011248 x 0.01248 x 0.353909 # 164 #CdeGhLJKL 0.003749 x 0.011248 x 0.01248 x 0.063749 x 0.298605 # 165 #CdeGhLJKLMN 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.03749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.0298605 # 166 #CdeGhLJKLMnOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 # 167 #CdeGhLJKLMOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 # 168 #CdeGhLJKLMOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 # 169 #CdeGhLJKMNOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 # 170 #CdeGhLJKMNOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 # 171 #CdeGhLKJ 0.003749 x 0.011248 x 0.003749 x 0.298605 # 171 #CdeGhLKMN 0.003749 x 0.011248 x 0.003749 x 0.298605 # 172 #CdeGhLKMNOP 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.803749 x 0.298605 # 173 #CdeGhLKMNOP 0	61 margaan	0.003749 x 0.011248 x 0.298605	0.000012
6.63 #C.derOhlj # 164 #C.derOhljKI. 6.003749 x 0.011248 x 0.01248 x 0.003749 x 0.298605 # 165 #C.derOhljKIMN 6.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.0353909 # 165 #C.derOhljKIMnOP 8.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 # 166 #C.derOhljKIMnOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 # 167 #C.derOhljKIMNOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 # 168 #C.derOhljKMNOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 # 169 #C.derOhljKMOP 0.003749 x 0.011248 x 0.003749 x 0.298605 # 170 #C.derOhljKMOP 0.003749 x 0.011248 x 0.003749 x 0.298605 # 171 #C.derOhlKI. 0.003749 x 0.011248 x 0.003749 x 0.298605 # 171 #C.derOhlKIMNOP 0.003749 x 0.011248 x 0.003749 x 0.298605 # 172 #C.derOhlKIMNOP 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 # 173 #C.derOhlKIMnOP 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 #	62 BLOROTI	0.003749 x 0.011248 x 0.011248 x 0.353909	
64 #CaeGbiljK1 8.003749 x 0.011248 x 0.003749 x 0.011248 x 0.353909 * 165 #CdeGbiljK1MmOP 8.003749 x 0.011248 x 0.001749 x 0.011248 x 0.003749 x 0.298605 * 166 #CdeGbiljK1MmOP 0.003749 x 0.011248 x 0.001749 x 0.003749 x 0.03749 x 0.298605 * 167 #CdeGbiljK1MOP 0.003749 x 0.011248 x 0.001248 x 0.003749 x 0.298605 * 168 #CdeGbiljK1MOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 * 169 #CdeGbiljkMnOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 * 170 #CdeGbiljkmOP 0.003749 x 0.011248 x 0.003749 x 0.298605 * 171 #CdeGbiljkmOP 0.003749 x 0.011248 x 0.003749 x 0.298605 * 171 #CdeGbilKJ 0.003749 x 0.011248 x 0.003749 x 0.298605 * 172 #CdeGbilKIMN 0.003749 x 0.011248 x 0.003749 x 0.298605 * 173 #CdeGbilKIMnOP 0.603749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 *		0.003749 x 0.011248 x 0.011248 x 0.063749 x 0.298605	•
initial activity initial activity 0.003749 x 0.011248 x 0.001248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 * 166 wCdeGhtljKlimOP 0.003749 x 0.011248 x 0.001248 x 0.003749 x 0.298605 * 167 wCdeGhtljKlimOP 0.003749 x 0.011248 x 0.001248 x 0.003749 x 0.298605 * 168 wCdeGhtljKlimOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 * 169 wCdeGhtljKMnOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 * 170 wCdeGhtljKMnOP 0.003749 x 0.011248 x 0.003749 x 0.298605 * 171 wCdeGhtlKJM 0.003749 x 0.011248 x 0.003749 x 0.298605 * 171 wCdeGhtKJM 0.003749 x 0.011248 x 0.003749 x 0.298605 * 172 wCdeGhtKJMN 0.003749 x 0.011248 x 0.003749 x 0.298605 * 173 wCdeGhtKIMNOP 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 *	64 M. MEGBURL	6.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	
666 ac. dec Bit JK INTOP 0.003749 x 0.011248 x 0.001749 x 0.003749 x 0.298605 * 167 #CdeG Bit JK INTOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 * 168 #CdeG Bit JK MNOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 * 169 #CdeG Bit JK MNOP 0.003749 x 0.011248 x 0.001248 x 0.003749 x 0.298605 * 170 #CdeG Bit JK MNOP 0.003749 x 0.011248 x 0.003749 x 0.298605 * 171 #CdeG Bit JK MNOP 0.003749 x 0.011248 x 0.003749 x 0.298605 * 171 #CdeG Bit JK MNOP 0.003749 x 0.011248 x 0.003749 x 0.298605 * 172 #CdeG Bit JK MNOP 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.353909 * 173 #CdeG Bit KIMNOP 0.603749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 *	65 # decing simil	0.003749 x 0.011248 x 0.011248 x 0 003749 x 0.011248 x 0.003749 x 0.298605	•
167 #CdeGhtjkLinkOF 0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.353909 * 168 #CdeGhtjkMnOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 \$ 169 #CdeGhtjkMnOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 \$ 170 #CdeGhtjkMnOP 0.003749 x 0.011248 x 0.003749 x 0.298605 \$ 171 #CdeGhtKJ. 0.003749 x 0.011248 x 0.003749 x 0.298605 \$ 172 #CdeGhtKJMN 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.353909 \$ 173 #CdeGhtKJMnOP 0.603749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 \$	66 at desiright with or	0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.298605	•
#68 #CdeGhijkMinOP 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605 \$ 169 #CdeGhijkMinOP 0.003749 x 0.011248 x 0.001749 x 0.298605 \$ 170 #CdeGhijkminOP 0.003749 x 0.011248 x 0.003749 x 0.298605 \$ 171 #CdeGhiKJI. 0.003749 x 0.011248 x 0.003749 x 0.298605 \$ 172 #CdeGhiKJMinOP 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.353909 \$ 173 #CdeGhiKIMinOP 0.603749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 \$	167 statingtimor	0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.353909	•
169 #CdeGbiljkm/03P 0.003749 x 0.011248 x 0.003749 x 0.298605 0 170 #CdeGbiljkm/0P 0.003749 x 0.011248 x 0.003749 x 0.298605 0 171 #CdeGbiljkm/0P 0.003749 x 0.011248 x 0.003749 x 0.298605 0 172 #CdeGbilKIMN 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.353909 0 173 #CdeGbilKIMnOP 0.603749 x 0.011248 x 0.003749 x 0.011248 x 0.663749 x 0.298605 0	168 el.deGhijkMN	0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	8
aCdeGhi[ktmOP # 171 aCdeGhi[ktmOP 0.003749 x 0.011248 x 0.003749 x 0.298605 # 171 aCdeGhi[ktmN 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.353909 # 172 aCdeGhi[ktmOP 0.603749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 #	169 @CdeGhljkMm3?	9.003749 x 0.011248 x 0.021248 x 0.003749 x 0.298605	
IncdeeGhtKJA 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.353909 * 172 #CdeGhtKIMN 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.353909 * 173 #CdeGhtKIMnOP 0.603749 x 0.011248 x 0.003749 x 0.011248 x 0.063749 x 0.298605 *	170 CdrGhljkmOP	0.003749 x 0.011248 x 0.003749 x 0.298605	*
172 #CdeGhlKIMN 0.007749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 # 173 #CdeGhlKIMnOP 0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 #	171 sCdeGhiKL	0.001749 x 6.011248 x 0.003749 x 0.011248 x 0.353909	
173 #CdeGhiKiMaOP	172 wCdeGhlKIMN	0.003749 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	•
0.001749 - 0.011 49 - 0.001749 - 0.001749 x 0.298605	173 aCdeGhlKIMnOP	0.003749 x 0.011 249 x 0.003749 x 0.298605	

9

	Sequence 5: FTI-A; Fail to Isolate (1 SDC Train)	
Failure Path	Calculations	Results
175 eCdeGhikMN	0.003749 x 0.011248 x 0.011248 x 0.353909	
176 sCdeGhikMnOP	0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
177 sCdeGhikmOP	0.003749 x 0.011248 x 0.003749 x 0.298605	
175 sCdeglj	0.003749 x 0.011248 x 0.353909	0.000014
179 #CdeglJKI.	0.003749 x 0.911248 x 0.903749 x 0.298605	•
180 eCdegljKIMN	0.063749 x 0.011248 x 0.003749 x 0.011248 x 0.353909	•
181 #CscgljKIMnOP	0.003749 x 0.011248 x 0.003749 x 9.011248 x 0.003749 x 0.298605	•
182 aCdegljKimOF	0.003749 x 0.011248 x 0.003749 x 0.003749 x 0.298605	•
183 sCdegljkMN	0.003749 x 0.011248 x 0.011248 x 0.353909	
184 æCdegljkMnOF	0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
185 sCdegljkmOP	0.003749 x 0.011248 x 0.003749 x 0.298605	
186 sCdegtKL	0.003749 x 0.003749 x 0.298605	0.999994
187 aCdegiKiMN	8.003749 x 0.003749 x 0.011248 x 0.353909	•
188 sCdeglK1MmOP	0.003749 x 0.003749 x 0.011248 x 0.003749 x 0.298605	
189 aCdegiKImOP	0.003749 x 0.003749 x 0.003749 x 0.298605	•
190 #CdegikMN	0.003749 x 0.011248 x 0.353909	0.000014
191 aCdegikMnOP	0.003749 x 0.011248 x 0.803749 x 0.298605	•
192 #CdegikmOP	0.003749 x 0.003749 x 0.298505	0.000004
193 acEF	0.011248 x 0.353909	0.00398
194 acE2GH	0.011248 x 0.011248 x 0.298605	0.000037
195 scEfGhij	0.611248 x 0.011248 x 0.011248 x 0.353909	•
196 acE/GhljKL	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	

.

Table C18: Failure Paths and Total Failure Probabilities

Table C18

(cont.):

Table C18: Failure Paths and Total Failure Probabilities

Sequence 5: FTI-A; Fail to Isolate (1 SDC Train)

Failure Path	Calculations	Results
197 scEfGbljKiMN	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	
198 scEfGhljKlMnOP	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	· · · · · · · · · · · · · · · · · · ·
199 scEPGhljKlmOP	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.298605	•
200 scErGhljkMN	6.011248 x 0.011248 x 0.011248 : 0.011248 x 0.353909	•
201 acE#GhljkMnOP	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	
202 acEFihljkmOP	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	
203 acEfGhiKL	8.011248 x 0.011248 x 0.003749 x 0.298605	
204 acEfGhiKiMN	0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	
205 acERGhiKiMnOP	0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	
206 scEfGhlKimOP	0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.298605	•
207 acEIGhikMN	0.011248 x 0.011248 x 0.011248 x 0.353909	
208 acE2GhikMmOP	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	
209 scEfGhikmOP	0.011248 x 0.011248 x 0.003749 x 0.298605	•
210 scEfgij	0.011248 x 0.011248 x 0.353909	0.000044
211 scEIgIj' L	0.011248 z 0.011248 x 0.003749 x 0.298605	•
212 scEigljKIMN	0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	
213 acEIgIjKIMnOP	0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	*
214 acElgljKimOP	6.611248 x 0.611248 x 0.003749 x 0.003749 x 9.298605	•
215 scEigljkMN	0.011248 x 0.011248 x 0.011248 x 0.353909	
216 scElgljkMnOP	0.011248 x 0.011248 x 0.011248 x 0.06 ³ 749 x 0.298665	
217 acEfglikmOP	0.011248 x 0.011248 x 0.003749 x 0.298605	•
218 acEfgIK1.	6.011248 x 0.003749 x 0.298605	0.000012

Tab		
	ie C18: Failure Paths and Total Failure Probabilities	
	Sequence 5: FTI-A; Fail to Isolate (1 SDC Train)	
ure Path	Calculations	Results
	0.011248 x 0.005749 x 0.011268 x 0.153909	
	6.011248 x 3.003749 x 9.011248 x 0.003749 x 0.298605	•
	0.011248 x 0.003749 x 0.003749 x 0.298605	•
	0.911248 ± 0.011248 ± 0.353909	0.000044
	0.011248 x 0.011248 x 0.003749 x 0.298605	•
	0.011248 x 0.003749 x 0.250505	0.6996912
	0.011245 x 0.298605	0.003358
	0.011248 ± 0.011248 ± 0.353909	0.000044
	0.011248 x 0.011248 x 0.003749 x 0.298605	•
	0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	•
	0.011248 x 0.011248 x 0.003749 x 0.011245 x 0.003749 x 0.290605	
	3.011248 x 9.011248 x 9.003749 x 9.003749 x 9.296005	•
	0.011246 z 0.011248 z 0.011248 z 0.353999	•
	0.011248 x 0.011238 x 0.021248 x 0.003749 x 0.296605	*
	0.011248 x 0.011248 x 0.003749 x 0.298605	
	6.011248 x 9.003749 x 0.298635	0.000012
	6.011248 x 0.003749 x 0.011248 x 0.353909	•
	0.011248 x 0.003749 x 9.011246 x 0.003749 x 0 296005	•
	0.011248 x 0.003749 x 0.003749 x 0.298665	•
	0.011248 x 0.011248 x 0.353909	0.000044
	0.011245 x 0.01124 j x 0.003749 x 0.298605	•
	0 011348 v 0.00149 v 0.298405	0.000012

Table C18 (cont.):

Table C18: Failure Paths and Total Failure Probabilities Sequence 5: FTI-A; Fail to Isolate (1 SDC Train) **Failure** Path Calculations Results 241 accept] 0.011248 x 0.353909 0.00398 242 acreljK1. 0.011248 x 0.003749 x 0.298605 0.090012 243 scegljKIMN 0.011248 x 0.003749 x 0.011248 x 0.353909 . 244 aceglJKIMnC.? 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605 . 345 accgljKlmOP 0.011248 x 0.003749 x 0.003749 x 0.298605 . 246 accellEMN 0.011248 x 0.011248 x 0.353909 0.000044247 scegljhMnOP 0.011248 x 0.011248 x 0.003749 x 0.298605 e. 248 acegljkmOP 0.011248 x 0.003749 x 0.298605 0.000012 249 acegiKL 0.003749 ± 0.298605 6.001119 scrgikiMN. 0.003749 ± 0.011248 ± 0.353909 250 0.00.015 8.003749 ± 0.011248 ± 0.003749 ± 0.298605 . 251 aceglKIMnOF 252 ncegiKimOP 0.803749 ± 0.003749 ± 0.198605 0.000064 253 acegikMN 0.011248 x 0.353909 0.00398 254 aceglkMnOP 0.011248 x 0.003749 x 0.298695 0.000012 0.003749 x 0.299405 255 acegikmOP 0.001119 **Total Failure Probability** 64 .1) 2.77 Error Factor

FTI-B: Sequence 5 (2 Trains of SDC)

Event iTI-B for Sequence 5 represents operator failure to isolate the leak (see Sequence 5 description in Sequences and Human Actions section) by performing the relevant steps of Attachment 6.1 (System Leakage) of procedure OP-901-046 (Shutdown Cooling Malfunction), when both SDC trains are in service. The HRA event tree for FII-B models the critical steps, substeps, and actions represented in steps 11 and 12 of Attachment 6.1 relating to isolation of the leak. Steps and substeps modeled include closing the SDC suction isolation valve on one SDC train, observing RCS level for stabilization, placing the isolated SDC train in service (referencing OP-009-005: system operating procedure, Shutdown Cooling System), and closing the suction isolation valve on the opposite SDC train. The HRA modeling conservatively assumes that the leak is in the second train isolated. Omission and commission errors are modeled for each step/substep. Omission errors are modeled as errors by the control room supervisor (CRS), with recovery credit for the reactor operator (RO). A high level of dependence was modeled between the two, when the CRS was directing RO actions. Commission errors are modeled as errors by the RO, with recovery credit for the CRS. For these actions, a moderate level of dependence was modeled between the RO performing the action and the CRS who would be concurrently performing other dynamic actions. The HRA event tree, subtask quantifications, and total failure probability are presented in Figure 11, Table C19, and Table C20, respectively. Table C20 also lists the total mean failure probability for this event tree as 0.0233.



Modelad for both 520 trains in service, 194,004 in 2nd train isolated 21 - 30 faits to remind CRS to place - CHS fails to place iscinted 500 train in isoleted SOC trein in service service (0P009-005) per step 12 1 of - AD fells to remind CAS to close SI 421 M(B) on . opposite SOC frain L . CRS fails to verify RD placeo Correct SDC train in service K ~ RO fails to place isolated 500 train OP-901-946 [Onitston] a - CRS foils to close Si ablace, on opposite o - Dog fails to verify AC closed at 401 ACBS in service (Commission) SOC train per step (2(2m/ds/pm) 0 - 40 fairs to close 51 ADT AFB) (Connission) A - UD remined on appropriate SDC train SOC Eroin Harlower w Service - CRS remembers to place isclated SX R0 to place ally in service on Appropriate 300 train Correct 5000 (Recovery) k - RO correctly piblows lealsyed - PO realm cars to close SI 404 ACB) SDC train in service train in Service Recovery3 to |Solat FT1 - 8** m - EHS remembers to close St AUT ACED, an co clory 51 401 ACR ** NOTE: (Avevoue) AD concrets in citraes St ADT AVAL on apposite 200 train Operators Fail an appropriate SDC stain (continued) Sequence Figure 11 (cont.):

Hamaa Action / Error	Basic Median HEP	Errer Factor	Source/ TitERP Tabat #	Step-hy- Step er Syszesic	Moduler for PSF:	Midifier Saurce	THER? Prpend- ency	Besic Mean HEP	Nominal Mean KEP	Errst
 CRS fails to close one SDC loop Suc. Iso. valve	0.003	3.4	T24-7 #3	85.5	,	129-46 #2	ZD	6.003749	3,611248	3.8
S1-401 A(B) (Ourlesion) RO fails to resulted CRS to close S1-401 A(B) for according trains	0.1	5.8	₹20-22 #1	SBC	3	520-16 02	КD	9.161.383	6.742075	5.9
RO fails is close 50-46. A(B)in appropriate	8.001	3.0	T28-12 #3	SBS	3	T20-16 #2	τÐ	6,001249	# 0.353909 0.003749	2.2 3.0
SDC state (Commission) CRS fails to renefind RO to Oose SI-401 A(B) in an ecopylistic train	8.1	5.8	119-22 ₹1	585	3	T20-14 #2	MD	0.161383	6,557844	5.0
CRS fails is determine to RCS level not	0.003	3.5	T20-7 #3	585	3	726-16-#2	ZÐ	0,003749	# 6.195525 9.011248	2.5
stabilizing (Omiasion) RO fails to remind CRS to determine if RCS	0.i	5.0	T30-22 #1	S25	3	T26-16 #2	1253	0.161383	0.743075	5,8
level not stabilizine									\$ 0.353999	2.2
RO Julls to correctly determine if RCS level not	0.003	3.9	T20-10#1	SBS	3	T20-16 #3	ZĐ	0.003749	0,011248	3.0
CRS fails to tell RO to determine if RCS level	0.1	5.0	T20-22 #1	\$85	3	T29-16 #2	MD	0.162 383	0.557844	5.8
not stabilizing				1					# 0,298605	2.4

T. LL. CIR. SILDS for Semionce 5, UTLR: Fail to Isolate (2 SINC Trains)

.

Table C12:

WEP's

	Human Action / Error	Basic Media HEP	Error Factor	Source/ THERP Table #	Step-by- Step or Dynamic	Modifier for PSFs	Modifier Source	THERP Depend- ency	Basic Mean HEP	Nominal Mean HEP	Error Factor
I	CRS fails to place isolated train in service (step 12.1 of OP901-046)	6.003	3.0	T20-7 #3	SRS	3	T20-16 #2	ZÐ	6.963749	0.011248	3.0
3	RO fails to remind CRS to place isolated SDC train in service	0.1	5.0	T20-22 #1	SBS	- 3	T20-16 #2	HD	0.161383	0.742075	5.6
										# 0.353909	2.2
к	RO fails to place isolated SDC train in service (Commission)	0.001	3.0	T20-12#3	SBS	3	T26-16 #2	ZÐ	0.001249	6.003749	3.0
L	CRS fulls to verify RO placed correct SDC train in service	0.1	5.0	T20-22 #1	888	3	120-16 #2	MD	0.161383	0.557841	5.0
										# 0.295605	2.4
м	CRS fails to close S1-401 $\mathrm{A}(B)$ on opposite SDC train	0.003	3.0	T20-7 #3	SBS	3	T20-16#2	ZÐ	8.903749	0.011248	3.0
N	RO falls to remind CRS to close SI-401 A(B) on opposite SDC train	0.1	5.0	T20-22 #1	SBS	3	T20-16 #2	HD	0.161383	9.742075	5.0
										# 0.353909	2.2
0	RO fails to close correct SL401 $A(B)$ value on appropriate train (Commission)	0.001	3.0	T20-12 #3	SBS	3	T20-16#2	ZD	0.901249	0.003749	3.0
P	CRS fails to verify RO closed correct SI-401 A(B) velve	6.1	5.0	T20-22 #1	885	3	T20-16#2	MD	0.161383	0.557844	5.0
	김 씨는 것을 가려갔어?									# 0.298605	2.4

C-76

Table C19 (cont.):

	Tal	de C20 : Failure Paths and Total Failure Probabilities	
		Sequence 5: FTI-B; Fail to Isolate (2 SDC Trains)	
	Failure Path	Calculations	Results
	AB	0.011248 x 0.353909	0.00398
	AbCD	0.011248 x 0.003749 x 0.298605	0.00012
	AbCdEF	0.011248 x 0.003749 x 0.011248 x 0.353909	•
	ABCRENGH	0.011248 x 0.003749 x 0.01248 x 0.011248 x 0.011248 x 0.298605	•
	AbCdERGhi3	0.011248 x 0.003749 x 0.01248 x 0.011248 x 0.011248 x 0.353909	
	AbCdEROMJKL	R.011248 x 0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
	AbCdERGhijkEMIN	$0.011248 \ge 0.003749 \ge 0.011248 \ge 0.011248 \ge 0.011248 \ge 0.001749 \ge 0.001248 \ge 0.353909$	•
	AbCdEKthUKIMBOP	$0.011248\ge 0.003749\ge 0.011245\ge 0.011245\ge 0.011245\ge 0.011245\ge 0.003749\ge 0.011248\ge 0.003749\ge 0.298605$	•
	AbCdERchijkLimOP	0.011248 ± 0.003749 ± 0.011248 ± 0.011248 ± 0.011248 ± 0.003749 ± 0.003749 ± 0.298605	•
	AMCAERGAIJAMN	8.011248 x 0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.353909	
	AbCGERShijkMmOP	0.011248 x 0.003749 x 0.011246 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
	AbCdERchljtmOP	0.011248 ± 0.003749 ± 0.011248 ± 0.011248 ± 0.011248 ± 0.003749 ± 0.298605	•
	AbCdERCHIKL.	0.011248 x 0.003749 x 0.013248 x 0.011248 x 0.003749 x 0.298605	•
	AMCAEMCMIKIMIN	0.011248 x 0.003349 x 0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.011248 x 0.353909	•
	AMCRETCHISCOP	0.011248 x 0.003749 x 0.01248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.003749 x 0.298605	*
	AbCdERchiktmOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.298605	•
	AbCdErCbitkMIN	0.011248 x 0.003749 x 0.011248 x 0.01.248 x 0.011248 x 0.011248 x 0.353909	•
	AbCdERthRMmOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
1	AbCdEAChikanOP	0.011248 x 0.0023749 x 0.011248 x 0.011248 a 0.001749 x 0.298605	•
The second second	AbcdEigIJ	0.011245 x 0.003749 x 0.011248 x 0.011248 x 0.351909	•
1	And the second se		the second

Table C20: Failure Paths & Total Failure Probability (Seq. 5, FTI-B)

1			
	Tat	de C20 : Failure Paths and Total Failure Probabilities	
		Sequest of FTI-B; Fail to Isolate (2 SDC Trains)	
	Failure Path	Calculations	Results
1.1	ABCAEJBUKL	6.0112448 x 0.001749 x 0.011248 x 0.011248 x 0.011248	
1	AbCABJE! KEMN	0.011248 x 0.003749 x 4.011248 x 0.0(1248 x 0.003749 x 0.011248 x 0.353999	
	AbCdElgfjKEMmOP	0.011248 x 0.003749 x 0.511245 x 0.511245 x 0.003749 x 0.011248 x 0.503749 x 0.298035	4
1000	AbC4E3g3K1mOP	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.012248 x 0.003749 x 0.003749 x 0.003749 x 0.2	
	ABCAEJEUMAN	0.011248 x 0.903749 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.353999	
1	AbCdElgIjkMoOP	0.051248 x 0.003749 x 0.011245 x 0.01246 x 0.01246 x 0.011248 x 0.003749 x 0.298605	
1.	Abs' dElgijkenOF	0.011245 x 0.903749 x 0.011245 x 0.011245 + 0.003749 x 0.295605	•
	AMCAETgGKU.	0.011248 x 0.063749 x 9.011248 x 0.063745 x 0.208605	
	AMCAERSINGNAN	0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.002749 x 0.011248 x 0.353909	•
	AMCAERGIKEMANOP	0.011248 x 0.003749 x 0.011248 x 9.003743 x 0.011248 x 0.093749 x 0.093749 x 0.298605	•
100	AbC4E2g3K1mOF	2.911248 x 0.803749 x 6.011245 x 0.903749 x 0.003749 x 0.296605	
1.1	AbCaEIgaMIN	0.011248 z 0.003749 z 0.01248 z 0.011246 z 0.353999	
1.1.1	AbCdElgikMeOF	0.0411248 x 0.003749 x 0.01248 x 0.011248 x 0.011248 x 0.003749 x 0.2421260	•
1.00	AbCdEIgthmOF	30.051248 x 0.003749 x 0.011246 x 0.003749 x 0.205005	•
1.1	Pris acGH	0.3111248 x 0.603749 x 0.011248 x 0.295605	
100	AbCabchU	6.011248 ± 0.003749 ± 0.011248 ± 0.011248 ± 0.051248 ± 0.353009	
1	ABCARCELIKE	3.011248 x 0.003749 x 0.011246 x 0.011246 x 0.01246 x 0.003749 x 0.295605	•
1	ABCARCHIJKEMIN	0.011248 x 0.003749 x 0.01248 x 0.011245 x 0.003749 x 0.011248 x 0.353900	
1.	AbCaeChijkIMnOP	0.0117-38 x 0.003749 x 0.011248 x 0.011-48 x 0.02749 x 0.011248 x 0.063749 x 0.298605	
	AbCdeCht[KImOP	0.051248 x 0.003749 x 0.011248 x 0.011248 x 0.0013749 x 0.003749 x 0.003749 x 0.298605	
1	ABCARCHERENEN	0:011248 x 0:003749 x 0.011248 x 0.011248 x 0.011248 x 0.01248	
	AbCdeChildMmOP	0.051245 x 0.063749 x 0.011245 x 0.011245 x 0.011248 x 0.011248 x 0.003749 x 0.205665	•

Table C20 (cont.):

Tabie C20	re Probabilities)C Trains)	Results	# S1999657 %	•	* 8.353909 a	x 0 9057745 x 0.298005 a	* 8. 298485	•	x 6. 2094aut5 *		•		z 0.353969	x 0:00C3.189 x 0: 2989435 8	* SBM 2 5		a 509967.0 ×				x 0.298605	•		
	ble C20 : Failure Paths and Total Fail	Sequence 5: FTI-B; Fail to Isolate (2 S	Calculation	B.011248 x 0.903749 x 0.921248 x 0.011248 x 0.001248	0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.29860	0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.0124	0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.01124	0.011248 x 0.003749 x 0.011248 x 0.003745 x 0.00374	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.312390	0.011248 x 0.003749 x 0.011248 x 0.011248 x 0.00374	0.011248 ± 0.003749 ± 0.011248 ± 0.003749 ± 0.20840	0.015 - 01 T49 x 0.011245 x 0.353999	0.011	0.01124. 0.011248 # 0.003749 # 0.01124	0.011248 # 1.mm	0.011246 x 0.002749 x 0.011248 x 0.001349 x 0.003749 x 0.00374	0.011248 ± 0.003749 ± 0.01248 ± 0.011248 ± 0.15390	0.011248 x 0.003749 x 0.01248 x 0.011248 x 0.0913	0.051.248 x 9.003749 x 9.011246 x 9.051246 x 0.003749 x 9.29640	0.011248 x 0.003749 x 0.003749 x 0.298605	0.051248 x 0.003749 x 0.003749 x 0.011248 x 0.35399	0.011248 x 0.003749 x 0.003749 x 0.011248 x 0.00124	0.011248 x 0.003749 x 0.003749 x 0.003749 x 0.003749 x 0.29564	0.011248 x 0.002749 x 0.011248 x 0.151909	20000 0 x 0007300 x 2x 2 2 2 2 0 0 x 0 x 0 0 0 2 0 0 0 0
	Tai		Failure F2'	Abd" dad 2h (Rewel)P	and and here a	ANCANCHARINGN	ALCACHIKIMMOP	abCdeGhiKJmOP	ABCARCHIEMIN	AbCdeChikMmOF	AbCdeGbikmOP	AbCdegU	AbCdegUKL	AbCdegUFMGN	AbCdegIJKIMoOF	AbCdegUKImOP	AbCdegIJEMN	AbCdegIJEMinOP	AbCdegUkmOP	AbCdegtKL.	AMC der WIMIN	Abr.AcetKIMmOP	AbCdegtKtmOP	AbCdeglkMN	
				-	2 3	5 4	2			-	8	15	52	0	3	35	9	5	5	5	3	-	1 23	63	

٩

		Table C2
re Probabilities C Trains)		0 (cont.):
	Results	
	0.000044	
0.298605		
0.011248 0.353909		
0.01124E x 0.003749 x 0.298605	*	
6.003749 x 6.295605	•	
		1

Table C20: Failure Paths and Total Failu

Sequence 5: FTI-B; Fail to Isolate (2 SD

	Failure Path	Calculations	Results
6.5	AbCdegikms()P	0.011240 x 0.003749 x 0.003749 x 0.298605	•
66	AlteEF	0.011248 x 0.011248 x 0.353969	0.000044
67	AbeFJGH	0.011248 x 0.011248 x 0.011248 x 0.298605	
65	AbeEKGhU	0.011248 x 0.011248 x 0.011245 x 0.011248 x 0.353909	•
69	AbeFRGhijk1.	0.051248 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
70	AMERGEIJKIMN	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 0.353909	•
71	Ale ERGHIKIMBOP	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011245 x 0.003749 x 0.298605	
72	AbcEJGhijKimOP	0.011248 x 0.011248 x 0.011248 x 0.011348 x 0.003749 x 0.003749 x 0.295605	• 567 Substra
73	AbcEJGhIjkMN	0.911248 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.353909	•
74	AbeF#GhijkM#OP	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
75	AbcERGhljkmOP	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298605	•
76	AbcEBGbiKL	0.011245 x 0.011245 x 0.011245 x 0.003749 x 0.298695	
77	AbcERGBIKIMN	0.011248 x 0.011248 x 0.011248 x 0.013749 x 0.011248 x 0.353999	•
78	AbcERGhiKIMnOP	6.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	
79	AbcERGhiKInsOP	0.01124b x 0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.298605	
80	AbcERGhikMN	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.353909	• • • • • • • • • • • • • • • • • • •
51	AbcEfGhikMaOP	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.296605	•
82	AbcEfChikmOP	0.611245 x 0.011248 x 0.011248 x 0.003749 x 0.298605	• 5/ 5-1/ 4/ 300
23	AlscEigli	0.011248 z 0.011248 z 0.011248 z 0.353909	•
24	AbcEIgljKL	6.011248 x 0.011248 x 0.011248 x 0.003749 x 0.295605	
85	AbcEIgIJKIMN	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	
86	AbcEfgljKJMeOP	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.298605	
			the state of the s

-

4

9

	Tab	IC U.S.S. & GHAIC & BHO GHA & MGI & CHAICE & LANGTHING	
		Sequence 5: FTI-B; Fail to Isolate (2 SDC Trains)	
	Failure Path	Calculations	Results
1.00	AberChik(Muc)P	0.011248 z 0.011248 z 0.063749 z 0.011248 z 0.003749 z 0.298665	
1	AberGhiKimOP	0.011246 x 0.011248 x 0.003749 x 0.068 \ en - 0.295405	•
	AbsecchdaMN	0.011248 x 0.011248 x 0.011248 x 0.153999	•
1.000	Abrevick-MartOP	0.011248 ± 0.011248 ± 0.011248 ± 0.011248 ± 0.005749 ± 0.295005	•
1.1.1	AbovChikumCP	4.011248 x 0.011248 x 0.003749 x 0.298405	
	AbergU	0.011248 x 0.011248 x 0.253409	0.000044
	AbergUKI.	a.011248 x 0.011248 x 0.003749 x 0.298485	
	AbougUKIMIN	w 651246 x 0.011246 x 0.003749 x 0.011248 x 0.353909	
	AborgijkiMistor	0.01 (248 ± 0.011248 ± 0.503749 ± 0.011248 ± 0.003749 ± 0.298665	•
	AbecgljX(ImO?	0.011246 x 0.011246 x 0.003749 x 0.003749 x 0.296605	•
	AboogUAMIN	0.051248 ± 0.021248 ± 0.011248 ± 0.3529999	•
	AborgUkMistOF	8.0112468 x 0.0511248 x 0.011248 x 0.001749 x 0.295605	•
	AbcogUkunOP	8.811248 x 0.011248 x 0.063749 z 0.295405	
	Aborg/K.I.	0.0112405 x 0.003749 x 0.208605	0.000012
	AbergikIMIN	0.011240Fx0.0005749 x 0.011246Fx0.359699	•
	AbergRUMaOF	0.01.248 x 0.002749 x 0.011248 x 0.003749 x 0.063749 x 0.2956855	
	AbceglKlmOP	8.4122345 x 0.003749 x 0.003749 x 0.298685	•
	AbeeglkMN	0.0112468 x 0.0112488 x 0.353909	0.000044
	AbecghMmOP	0.011248 s 9.011248 z 9.003749 z 0.298605	•
	AbergikanOP	0.011248 x 0.003749 x 0.298605	9.6666612
	a(1)	8 (8013148 # 0.298985 0	611100'0

1

ŝ,

Table C20 (cont.):

á

.....

k

Table C20 Failure Path Seq Failure Path Cercolus	Failure Paths and Total Failure Probabilities	tence 5: FTI-B; Fail to Isolate (2 °DC Trains)	Calculations	4 0:01349 x 0:011348 x 0:011248 x 0:208605	(80.3749 ± 0.011.248 ± 0.011.248 ± 0.011.248 ± 0.353999	e 0.05749 x 0.051248 x 0.051248 x 0.051248 x 0.005749 x 0.295605	1.0017749 x 0.011246 x 0.011246 x 0.911246 x 0.003749 x 0.011246 x 0.353999	1.90.3749 x 0.011248 x 0.011248 x 0.011248 x 0.00.3749 x 0.01248 x 0.003749 x 0.205605 a	8.06.37 49 x 0.011.248 x 0.011.248 x 0.012.348 x 0.00.3749 x 0.000.749 x 0.298405	0.60.27.2 * 8.40.11.2468 × 0.051.2468 × 0.051.248 × 0.051.248 × 0.0539009	0.003749 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.298405	0.043749 ± 0.011248 ± 0.011248 ± 0.011248 ± 0.003749 ± 0.208405	6.00.3749 x 6.011245 x 6.011248 x 6.003749 x 9.295605	0.001749 x 0.011248 x 0.013248 x 0.003749 x 0.01248 x 0.051248 x 0.05009	0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.011248 x 0.003749 x 0.003749 x 0.003749 x 0.00000	0.003749 ± 0.311246 ± 0.011246 ± 0.003749 ± 0.3037	0.003749 x 0.011248 x 0.011248 x 0.353009	0.003749 x 0.011248 x 0.011248 x 0.003749 x 0.299605	0.0013749 ± 0.011248 ± 0.011248 ± 0.003749 ± 0.298605	0.001749 ± 0.011248 ± 0.01248 ± 0.353%99	0.003749 x 0.011248 x 0.011248 x 0.001749 x 0.295605	0.003749 ± 0.051248 ± 0.051246 ± 0.05149 ± 0.011248 ± 0.353909	0.90.5749 ± 0.611248 ± 0.601349 ± 0.011248 ± 0.001248 ± 0.003749 ± 0.295605	209842 0 % 0.001 201 0 % 0.001 201 0 % 0.001 0
8 8 8 8 8 8 4 4 4 4 4 8 8 8 8 8 8 8 8 8	Table C20: Failur	Sequence 5: F	Failure Path	8.003749 s 0.01	acaron 0.001749 z 0.01 acaroni	assessions assession(sk1 0.00	48.54148.84V	atticklijk.1Med. P	48.96.36.46.0F	457-54J&M6N	4E9Ch[]kMmOP	4EX/hijkmO7 8.963749 ± 0.0	45.805749 x 6.0	46.965749 x 6.0	46%/hikIMmOP	dERChiKlenOF 6.963749 x 6.3	dEPE2bitkMN	CERChRAMo(2P	065%:httkmOP	C4F3w13 0.903749 ± 0.4	0.963749 x 0.3	0.003749 x 0.6	CatStellKEMeOP 2.04	2 001749 x 0.0

 $^{\odot}$

	Table C20): Failure Paths and Total Failure Probabilities	
	8	guence 5: FTI-R; Fail to Isolate (2 SDC Trains)	
	Failure Path	Calculations	Results
153	ac designed to the	0.003749 x 0.011248 x 0.013248 x 0.013248 x 0.013248 x 0.003749 x 0.298605	•
554	#CdF3gjjkmOP	0.0003749 x 0.011248 x 0.011248 x 0.001349 x 0.2003249 x 0.20030	•
155	«CAEIgKU.	8.00.3749 x 6.012 2.000 x 0.000.3749 x 0.2906465	•
135	SCAPE_BROWN	8.662749 x 8.611248 x 8.861749 x 8.611248 x 8.3512489	
157	wCdE3g(KJMwOP	0-003749 x 0-051248 x 0.003749 x 0.01248 x 0.003749 x 0.298605	•
158	#CdE3giK1weOP	0.603749 x 0.011248 x 0.903749 x 0.803749 x 0.298405	•
159	#C4EJgRMN	0.063749 x 0.011248 x 0.011248 x 0.353999	•
3	acids/glaMmOP	8.003749 x 8.011248 x 9.012248 x 9.003749 x 8.298495	•
161	#C4EJgitumOF	8.(HC1749 ± 8.011248 ± 6.061749 ± 6.298605	•
162	MCM4CH	8.663749 ± 0.8511248 ± 0.2956685	0.000012
163	#C#CHC	6.003749 x 0.012248 x 0.012348 x 0.353909	
164	aCdeChIJKL	6.001749 x 9.011248 x 9.011248 x 8.003749 x 8.298485	•
165	actackersternes	0.003749 ± 0.011248 ± 0.011248 ± 0.003749 ± 0.01248 ± 0.01248 ± 0.153000	•
Les	#C#ChilkIMeO?	\$5.663749 ± 9.911248 ± 9.921248 ± 0.003749 ± 0.011248 ± 3.003749 ± 0.298605	
\$67	wCdwChA3K3mWP	8.003/49 x 0.011348 x 0.011348 z 0.002749 x 0.003749 x 0.208405	
16.0	wcreatistismus	8.003749 x 0.011248 x 0.012248 x 0.012248 x 0.051248 x 0.353009	
169	ec. appende	0.005749 x 0.011248 x 0.011248 x 0.011245 x 0.003749 x 0.298005	
170	#C4-CbUkmOF	3.663749 x 0.051148 x 0.011248 x 0.063749 x 0.298605	
171	#C#cDhKL	0.001349 ± 0.011248 ± 0.0033749 ± 0.298605	•
172	aCdeChIKIMIN	0.003749 x 0.031246 x 0.003749 x 0.011246 x 0.353M09	•
173	wCdwChtK3MmOP	0:003749 x 6.011248 x 0.003749 x 0.011248 x 0.003779 x 0.298605	•
174	aCdeCP-SOmOP	0.003749 ± 0.011248 ± 0.003749 ± 0.063749 ± 0.298005	*

Table C20 (cont.):

	Tab	le C20 : Failure Paths and Total Failure Probabilitie	\$
		Sequence 5: FTI-B; Fail to Isolate (2 SDC Trains)	
	Failure Path	Calculations	Results
175	#CdeGhikMN	0.003749 x 0.011248 x 0.011245 x 0.353909	
176	aCdeGhlkMaO?	0.003749 x 0.01. *48 x 0.011248 x 0.003749 x 0.298605	
177	#CdeGhikmOP	0.002749 x a.011248 x 0.003749 x 0.298605	
178	#CdegU	0.003749 x 0.011248 x 0.353909	0.000014
179	øCdegijKL	0.003749 x 0.011248 x 0.003749 x 0.298605	
1.80	#CdegljKIMN	0.903749 x 0.011248 x 0.003749 x 0.0v1248 x 0.353909	
151	eCdegIJKIMnO?	6.003749 x 0.011245 x 0.003749 x 0.011245 x 0.003749 x 0.298605	*
182	aCdegljKimOP	0.003749 x 6.011248 x 6.003749 x 0.003749 x 0.298605	
185	#CdegljkMN	0.603749 ± 6.911248 ± 0.911248 ± 0.353909	*
184	aCdegIJkMmOP	0.003749 x 6.011248 x 0.011248 x 0.003749 x 0.298605	
185	øCdegIJkmOF	C-003749 x 0.011248 x 0.063749 x 0.298505	
156	øCdegiKL	0.003749 x 0.003749 x 0.298605	0.000064
187	#CdeglKIMN	0.003749 x 0.003749 z 0.011248 x 0.353909	
188	aCdegik.IMmOP	0.003749 x 0.003749 x 0.011248 x 0.003749 : 0.298605	*
185	aCdegiKtur()P	0.003749 ± 0.003749 ± 0.003749 ± 0.298605	
190	#CdegikMN	0.003749 n 0.01124' x 0.353909	0.000014
191	øCdeglkMnOP	0.003749 x 0.0? 1248 x 0.003749 x 0.2* \$605	*
192	#C.legikmOP	0.003749 x 4.003749 x 0.298605	0.000004
193	actif	0.011244 x 0.353902	0.00398
194	acEIGH	6.011248 ± 0.011748 × 0.298605	0.000037
19%	scERGbiJ	0.011248 x 0.0112/8 x 0.011248 x 0.353909	*
196	*FRSHJKI	0.011248 - 0.011248 x 0.011248 x 0.003749 x 0.298605	

 \mathbf{w}

	Tab	de C20: Failure Paths and Total Failure Probabilities	
		Sequence 5: FTI-B; Fail to Isolate (2 SDC Trains)	
	Failure Path	Calculations	Results
	acEAChIJKIMN	0.011248 x 9.011248 x 0.012148 x 0.003749 x 0.001240 x 9.353909	
	active and the other of the other of the other o	0.011248 ± 0.011248 ± 0.011248 ± 0.003749 ± 0.011248 ± 0.003749 ± 0.298405	
	active and a second sec	0.011248 x 0.011248 x 9.011248 x 0.003749 x 0.003749 x 0.295605	•
1.0	#FRUEJKMN	0.051248 x 0.011248 x 0.011248 x 0.021248 x 0.021248 x 0.351909	
	weEnchildeMan. P	0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.295605	•
	actERchilphanOP	0.011248 x 0.011248 x 0.011248 x 0.003749 x 0.299605	•
	scrohikl.	0.011248 x 0.011248 x 0.003749 x 0.298605	
	*CEXCIERTINN	6:011248 x 0.011248 x 0.003749 x 0.011248 x 0.353909	•
2	ac ERChild March	G.011248 z 0.011248 z 0.903749 z 0.011248 z 5.003749 z 0.298605	•
	ac@KchiKlenO?	0.011248 x 0.011246 x 0.003749 x 0.501749 x 0.205605	•
	*CENCIMIENTN	0.0112.2 x 41011248 x 0.011248 x 0.353969	
	action Ward?	0.011248 x 0.011248 x 0.011248 x 9.003749 x 0.298605	•
	ac SPC hikoro OP	0.011248 x 0.011248 x 0.003749 x 0.295605	•
	*Ebjl	0.011248 s 0.011246 s 0.253909	0.000044
	actigitikt.	0.011245 x 0.011245 x 0.003749 x 0.298605	•
	*ElgljKIMN	0.011248 x 0.011248 x 0.003749 x 0.011248 x 3.53909	•
1	acElgIjKIMnOP	0.011248 x 0.011248 x 0.003749 x 0.951248 x 0.083749 x 0.298605	•
1	acEJg1JKJmrOP	0.011248 x 0.011248 x 0.003749 x 0.003749 x 0.298605	•
	*ETELIARIN	0.011248 x 0.011248 x 0.011248 x 0.353909	•
	acE1g(jkMixOP	0:011248 x 0.011248 x 0.015248 x 0.003749 x 0.003749 x 0.298605	•
	ecEfgljkmOP	9.011248 x 0.011248 x 0.003749 x 0.298605	*
1	ectalglammer		

1 .

1

1

a

Table C20 (cont.):

4

1

1

.

Table C20 (cont.):

r,

-

4 V 3

Tal	le C20: Failure Paths and Total Failure Probabilities	
	Sequence 3: FTI-B; Fail to Isolate (2 SDC Trains)	
e Path	Calculations	Results
	0.011248 x 0.353909	0.08398
	0.011248 x 0.003749 x 0.296005	0.909912
	0.011248 z 0.003749 z 0.011246 z 0.353%	•
	0.011248 x 0.003749 x 0.011245 x 0.003749 z 0.2065405	•
	0.011248 x 0.003749 x 0.003749 x 0.295605	•
	0.011248 x 0.011248 x 0.353909	G.SKMAG44
	0.031248 x 0.031248 x 0.003749 x 0.298605	
	0.0' US x 0.903749 z 0.298685	0,000612
	0.005749 z 0.296605	6.001119
	0.003149 ± 8.011248 ± 0.353909	0.000014
	0.003749 x 0.6t1248 x 0.063749 x 0.298605	•
	0.003749 x 0.003749 x 9.298685	0.000004
	0.011248 x 0.153909	0.00398
	0.011238 ± 0.003745 ± 0.258605	0.000012
	0.003745 x 0.296605	0.001119
	Total Failure Probability	0.023
	Error Factor	277

Table C20 (cont.):

Summary of CE ISLOCA HRA

Detailed HRA analyses were conducted for each of the significant scenarios identified in this ISLOCA PRA. Extensive documentation for each analysis has been provided above (e.g., descriptions of the scenarios, human errors, and both median and mean HEPs).

Data from each human action in each sequence was then used in an uncertainty analysis. HEPs for individual tasks/subtasks, in individual event trees, were entered into IRRAS, the Integrated Reliability and Risk Analysis System. Using the Latin Hypercube Sampling Methodology in IRRAS, with a sample size of 5,000, the HRA analysts were able to propagate uncertainty throughout each analysis and were then able to calculate a final EF (error factor) for each total failure probability. Results from the traditional HRA and uncertainty analysis are summarized in Table C21. Individual human actions are listed in the first column (with sequence number) followed by:

- 1- the HRA point estimate (total mean failure probability) using THERP,
- 2- the suggested EF for that probability (from THERP Table 20-20),
- 3- the calculated mean value from the uncertainty analysis (IRRAS),
- 4- the calculated EF from the uncertainty analysis,
- 5- the median value (from IRRAS), and finally,
- 6- both the 5th and 95th percentile value from the uncertainty analysis.
| Table C21: Summary of Results for CE ISLOCA HRA | | | | | | | |
|---|-----------------------|--------------------|-----------------|-------------|-------------------|---------------------|----------------------|
| Seq. #/ Action | Point Est.
(THERP) | EF-THERP
T20-20 | Mean
(IFRAS) | EF
Calc. | Median
(IRRAS) | Sth %ile
(IRRAS) | 95th %ile
(IRRAS) |
| 2 - FTD | 0.018 | 5 | 0.0167 | 4.26 | 0.0111 | 0.0029 | 0.0472 |
| 2 - FTDGN (Proc.) | 0.02 | 5 | 0.0192 | 3.96 | 0.0136 | 0.0036 | 0.0540 |
| 2 - FTI (Proc.) | 1.00 | | 1.2.3.3 | | | | |
| 2 - FTDGN (Know.) | 0.39 | 5 | 0.3617 | 2.02 | 0.3306 | 0.1656 | 0.6685 |
| 2 - FTI (Know.) | 0.019 | 5 | 0.0188 | 2.97 | 0.0151 | 0.0051 | 0.0449 |
| 2 - FTDGN (Small Break) | 0.016 | 5 | 0.016 | 3.73 | 0.011 | 0.0036 | 0.041 |
| 2 - FTI (Small Break) | 0.019 | 5 | 0.0188 | 2.97 | 0.0151 | 0.0051 | 0.0449 |
| 5 - FTD | 0.008 | 5 | 0.0074 | 11.86 | 0.0025 | 0.0002 | 0.0293 |
| 5 - FTOGN | 0.008 | 5 | 0.0074 | 12.11 | 0.0024 | 0.0002 | 0.0294 |
| 5 - FTI (1 Train) | 0.02 | 5 | 0.0219 | 2.77 | 0.0179 | 0.0072 | 0.0497 |
| 5 -FTI (2 Trains) | 0.02 | 5 | 0.0219 | 2.77 | 0.0179 | 0.0072 | 0.0497 |

Table C21:

C-90

8

Conclusions

The HRA found that operator error could contribute to ISLOCA scenarios. Risk-significant human error initiators were found to be unlikely during normal operations involving interfacing systems. This is due, in part, to the existence of vital administrative procedures and relat d operator training, as well as the presence of controlled interlocks and procedural guidance which prevent the inadvertent operation of pressure isolation MOVs. The CE HRA analysis also found a significantly higher probability for operator error during detection, diagnosis, and isolation of a LOCA following a break outside containment, as compared to human error initiating actions. This general finding results from an interaction of the following variables:

- 1 For Sequence 2, the applicable procedure (OP-902-002) does not provide any guidance for requisite operator action to isolate the break.
- 2 For Sequence 2, operator training was focused on scenarios in which operators were directed not to override safety injection actuations. Because of this, the operators' situation awareness may be directed away from required actions to isolate the affected LPSI train (i.e., by closing the appropriate SI flow control valve).
- 3 There is a limited amount of time after the sequence initiator, before a break, for the detection, diagnosis, and isolation of ISLOCA before core damage.
- 4 Operators are exposed to a higher workload and elevated levels of stress at the time when actions to detect, diagnose, and isolate an ISLOCA are needed.

References

- C-1 T. G. Ryan, "A Task Analysis Linked Approach for Integrating the Human Factor in Reliability Assessment of NPP", <u>Reliability Engineering and</u> <u>System Safety</u>, vol. 22, 1988.
- C-2 A. Swain and H. Gutman, <u>Handbook of Human Reliability Analysis with</u> <u>Emphasis on Nuclear Power Piant Applications</u>, NUREG/CR-1278, August, 1983.
- C-3 G. Hannaman and A. Spurgin, <u>Systematic Human Action Reliability</u> <u>Procedure (SHARP)</u>, EPRI NP-3583, 1984.
- C-4 R. Hall (ed.), A Guide for General Principles of Human Action Reliability Analysis for Nuclear Power Generation Stations, IEEE Draft Standard P1082/D7, August, 1989.
- C-5 U.S. Nuclear Regulatory Commission, Technical Specifications, Docket number 50-382 W3, SES FSAR, January, 1979.
- C-6 U.S. Nuclear Regulatory Commission Inspection Report 50-412/90-10 and 50-414/90-10.
- C-7 J. Ball, Interfacing System LOCA Inspection Peport, Docket number 50-382/90-200, September 14, 1990. U. S. Nuclear Regulatory Commission, Washington, D. C.
- C-8 G. Hannaman, A. Spurgin, and Y. Lukic, <u>Human Cognitive Reliability Model</u> for PRA Analysis, NUS-4531, Electric Power Research Institute, 1984.
- C-9 K. Russell, M. McKay, M. Sattison, N. Skinner, S. Wood, and D. Rasmuson, Integrated Reliability and Risk Analysis System (ver. 2.5), February, 1991. EG&G Idaho, Inc. Idaho Falls, ID.
- C-10 Technical Specification Change Request NPF-38- (Draft), Docket number 50-382.

Appendix D

Use of Constrained Lognormal Distribution in Human Reliability Analysis

Appendix D Use of Constrained Lognormal Distribution In Human Reliability Analysis

1. Motivation

As discussed in [D-1], use of a lognormal distribution can sometices lead to erroneous results when it is used to represent a probability. These problems arise because a probability, which must lie between 0 and 1, is being modeled by a lognormal random variable, whose only constraint is that it take on values \geq 0; there is no upper limit. One area where this problem sometimes arises is human reliability analysis, or HRA. One of the frequently used tools in NRA is [D-2], also known as THERP. Human reliability analysts use tables of human error probabilities (HEPs) in THERP to obtain median estimates and error factors (HEPs are assumed in THERP to be lognormally distributed), which can be used in an uncertainty analysis. However, THERP will, under some circumstances, provide combinations of medians and error factors which are not meaningful. For example, THERP may recommend the use of a median HEP of 0.5 with an associated error factor of 5, producing a 95th percentile probability greater than unity. The method documented below is one approach to solving this problem.

2. The Constrained Lognormal Distribution

An intuitive approach to solving this problem is suggested in [D-1]. There the authors recommend the use of a constrained lognormal distribution, whose random variate assumes values between 0 and 1. The cumulative distribution function (cdf) of such a random variable is obtained as follows. First, the probability that a lognormal random variable (unconstrained) is less than some value x is given by

$$P(X \le x) = \Phi\left(\frac{\ln x - \mu}{\sigma}\right) \tag{1}$$

where μ and σ are the parameters of the distribution and ϕ is the cumulative

normal function. Constraining the random variable X to be less than 1 is heuristically equivalent to asking for the conditional probability that X is less than or equal to some value x, given that it is less than or equal to 1. Thus, one has

$$P(X \le x | X \le 1) = \frac{\Phi(\frac{\ln x - \mu}{\sigma})}{\Phi(\frac{-\mu}{\sigma})}, \quad 0 \le x \le 1$$

$$= 1, \quad x \ge 1.$$
(2)

This expression allows one to calculate percentiles and moments of the distribution. For example, the expected value of X will be given by

$$E(X) = \int_{0}^{1} \frac{\frac{1}{\sqrt{2\pi\sigma}} \exp\left[-\frac{(\ln x - \mu)^{2}}{2\sigma^{2}}\right]}{\Phi\left(-\frac{\mu}{\sigma}\right)} dx.$$
 (3)

This integral, and the integrals for the higher moments, can be evaluated numerically for specific values of μ and σ . This problem of truncating the probability values can also be approached through a transformation of variables process as described in the next section.

3. A Transformation of 'iriables Approach

Section 2 presented a somewhat heuristic approach to constraining the THERP HEPs. This section presents an alternative approach. One starts by defining a transformation of variables that will restrict the unconstrained lognormal random variable to lie between 0 and 1. If Y is a random variable that is lognormally distributed with parameters μ and σ^2 , denoted Y ~ $\Lambda(\mu, \sigma^2)$, then the random variable X, defined by the transformation

$$Y = \frac{X - \tau}{\theta - X} \tag{4}$$

will be a truncated lognormal random variable constrained to values between r and θ [D-4]. In the present case one wants X to take on values between 0 and 1, so the transformation becomes

$$Y = \frac{X}{1-X}.$$
 (5)

This is a one-to-one transformation from X onto Y with a Jacobian given by

$$\mathcal{J} = \frac{1}{(1-X)^2} \,. \tag{6}$$

The pdf cf Y is

$$g(y) = \frac{1}{\sqrt{2\pi}\sigma y} \exp\left[-\frac{(\ln y - \mu)^2}{2\sigma^2}\right].$$
 (7)

Therefore, the pdf of X is given by

$$f(x) = \frac{1}{\sqrt{2\pi}\sigma(\frac{x}{1-x})} \exp\left[-\frac{\left[\ln\left(\frac{x}{1-x}\right) - \mu\right]^{2}}{2\sigma^{2}}\right] \frac{1}{(1-x)^{2}}$$

$$= \frac{1}{\sqrt{2\pi}\sigma x(1-x)} \exp\left[-\frac{\left[\ln\left(\frac{x}{1-x}\right) - \mu\right]^{2}}{2\sigma^{2}}\right],$$
(8)

202

The graph of f(x) is shown in the figure below.



Truncated Lognormal pdf

No simple analytic expression for the mean value of this distribution exists (see [D-3] for the exact result). However, the expression for the expected value of X can be integrated numerically without difficulty, or tables in [D-3] can be used to approximate the mean, skewness, and kurtosis. The resulting mean value for this example (median = 0.5, error factor = 5) is E(X) = 0.36.

The percentiles of this distribution can be found by integrating the pdf as follows. Let x_p be the value of X such that $P(X \le x_p) = p$. One has

$$P(X \le x_p) = \int_{0}^{x_p} \frac{1}{\sqrt{2\pi\sigma x (1-x)}} \exp\left[-\frac{\left[\ln\left(\frac{X}{1-x}\right) - \mu\right]^2}{2\sigma^2}\right] dx$$

$$= \Phi\left[\frac{\ln\left(\frac{X_p}{1-x_p}\right) - \mu}{\sigma}\right].$$
(9)

The integral was evaluated by making the substitution

$$z = \ln\left(\frac{x}{1-x}\right). \tag{10}$$

One can compare percentiles obtained from (9) to those from (2). Using (9), one finds that the median value of X is 0.33, compared with 0.37 calculated with (2). Thus, the percentiles obtained with (2) are somewhat conservative relative to the percentiles obtained with (9).¹

Some might question the use of a (truncated) lognormal distribution; a beta distribution, in which the random variable is constrained, a priori, to lie between 0 and 1 might seem more appropriate. In answer to this potential objection the authors show in the following paragraphs that the truncated lognormal distribution <u>is</u> a beta distribution, so any possible objection on this basis is simply a question of terminology.

The "proof" utilizes a Pearson plot [D-5] of β_2 vs. β_1 where β_1 and β_2 are measures of skewness and kurtosis, respectively. They are defined by

$$\sqrt{\beta_1} = \frac{E[X - E(X)]^3}{[Var(X)]^{\frac{3}{2}}}$$
(11)

¹The reason the distribution functions, and hence the percentiles, are different is because the random variable defined in this section is different from the random variable defined in Section 2. The cdf of a random variable is unique; therefore, the conditional cdf in Section 2 will differ from the cdf of the constrained random variable defined in Section 3.

and

$$\delta_{2} = \frac{E[X - E(X)]^{4}}{[Var(X)]^{2}}.$$
(12)

The third and fourth moments of X about its mean that are required for these calculations were obtained by numerical integration (they could also be found from tables given in [D-3]). One finds that

$$\sqrt{\beta_1} = 0.50$$
 (13)

and

$$\beta_2 = 2.50.$$
 (14)

Plotting β_1 and β_2 shows that the truncated lognormal distribution is indeed a beta distribution.

Since uncertainty analysis software typically does not allow the use of a truncated lognormal distribution, but does allow use of a beta distribution, the following paragraphs will show how to estimate the beta distribution parameters using the mean and variance of the truncated lognormal distribution.

First, recall that the pdf of a beta-distributed random variable X is given by

$$f(x) = \frac{x^{\alpha-1}(1-x)^{\gamma-1}}{B(\alpha,\beta)}$$
for $0 \le x \le 1$
(15)

where $B(\alpha,\beta)$ is the beta function, defined as

$$B(\alpha, \beta) = \int_{0}^{1} x^{\alpha-1} (1-x)^{\beta-1} dx$$

$$= \frac{\Gamma(\alpha) \Gamma'(\beta)}{\Gamma(\alpha+\beta)}.$$

$$\Gamma(\alpha) = \int_{0}^{m} x^{\alpha-1} e^{-x} dx.$$
 (16)

It can be shown that the mean and variance of X are related to the parameters of the beta distribution by

$$E(X) = \frac{\alpha}{\alpha + \beta}$$
(17)

and

$$Var(X) = \frac{\alpha\beta}{(\alpha+\beta+1)(\alpha+\beta)^2}.$$
 (18)

By integrating numerically, or using the tables in [D-3], one finds that E(X) = 0.36 and Var(X) = 0.04. This gives two equations in two unknowns, which can be solved to find c = 1.88 and $\beta = 3.36$. These parameters are then used to specify the beta distribution in the uncertainty analysis software. The graph on the next page compares the beta and truncated lognormal pdfs. Note that the method of moments does not produce an exact fit. However, the fit is judged to be sufficient for the purposes of this analysis.

One can generalize these results, as Johnson has done [D-3]. The truncated lognormal distribution is really a special case of the bounded Johnson distribution defined in [D-3]. Johnson shows that the random variable defined by

$$Z = \gamma + \delta \ln\left(\frac{x - \epsilon}{\lambda + \epsilon - x}\right)$$
(19)



where

$$Y = \frac{X - \epsilon}{\lambda + \epsilon - \pi} \sim \Lambda\left(\frac{-\gamma}{\delta}, \frac{1}{\delta^2}\right)$$
(20)

has a standard normal distribution. The pdf of X is given by

$$f(x) = \frac{\delta\lambda}{\sqrt{2\pi} (\lambda + \epsilon - x) (x - \epsilon)} \exp\{-\frac{1}{2} [\gamma + \delta \ln \frac{x - \epsilon}{\lambda + \epsilon - x}]^2\}$$
(21)

for $\varepsilon \leq x \leq \varepsilon + \lambda$. In this particular case, one has $\varepsilon = 0$, $\lambda = 1$, $\delta = \sigma^{-1}$, and $\gamma = \mu/\sigma$. Johnson also shows in [D-3] that β_1 and β_2 lie between the lognormal line and the line $\beta_2 - \beta_1 - 1 = 0$ on a Pearson plot. This is precisely the region that encompasses the beta distribution (note that the beta distribution is a Type I Pearson distribution), so one has an alternative demonstration that the so-called truncated lognormal distribution is really a bata distribution. In addition, the distribution is skewed to the right because, as is shown in [D-3], positive values of γ correspond to positive skewness. Positive skewness is desirable as it reflects the lognormality of the data in THERP.

Another potential approach to constraining the range of the random variable would be to retain the unconstrained median value from THERP but reduce the error factor so that the 95th percentile probability is unity. However, this is an *ad hoc* approach which does not give consistent, justifiable results in all cases. In addition, for median values greater than about 0.5, retaining the unconstrained median gives a very drastic reduction in the coefficient of variation (ratio of the standard deviation to the mean), implying far more knowledge of the range of values about the mean than is justified (i.e., a reduction in the uncertainty, perhaps leading to an underestimation of the contribution of this variable to overall sequence uncertainty). This is shown graphically in the attached figures, which illustrate probability density functions for three cases





Figure 4 Median of 0.7, Error Factor of 5



4. Approximating the Constrained Distribution

Because of limitations in the software used for the uncertainty analysis, the basic approach will be to obtain percentiles of the constrained distribution using (8), then use an unconstrained distribution passing though these points to estimate the constrained mean. An example of this is shown builder.

Assume that THERP has provided a median HEP of 0.5 with an associated error factor of 5. This value is constrained to be loss than 1 as follows. First, one needs to find the parameters, μ and σ , of the unconstrained distribution. These are obtained from

$$\mu = \ln X_{0.5} \tag{22}$$

and

$$\sigma = \frac{\ln EF}{1.645} \,. \tag{23}$$

Using the values of 0.5 and 5, one finds that μ = -0.693 and σ = 0.978.

Next use (8) to find the 50th and 95th percentiles of the constrained distribution. Solving for x in (ε), and denoting constrained values of X by X^{*}, one finds that X^{*}_{0.5} = 0.334 a J X^{*}_{0.95} = 9.714. The constrained error factor is then 2.14.

Now find the <u>unconstrained</u> lognormal distribution that passes through these percentiles. Using (21) and (22), one obtains $\mu' = -1.097$ and $\sigma' = 0.462$. The mass of this distribution is given by

$$E(X') = \exp\left(\mu' : \frac{\sigma^2}{2}\right)$$

= 0.37. (24)

To repeat, this is the mean of an unconstrained lognormal distribution passing

chrough the median and 95th percentile of the constrained distribution. Because the new unconstrained distribution includes values of the random variable that are greater than 1, the mean of this distribution is somewhat greater than the mean of the constrained distribution (0.36), so the estimate is somewhat conservative.

As a check, one can calculate the probability that the new unconstrained random variable is greater than 1. This is given by

$$P(X'>1) = 1 - P(X' \le 1) - \Phi\left(\frac{-\mu'}{\sigma'}\right)$$

$$= 0.01.$$
(25)

This provides some confidence that the unconstrained estimate of the mean is reasonably close to the true constrained mean.

The three distributions utilized in the analysis are shown in the following figure.



5. Conclusions

The use of a truncated lognormal (beta or bounded Johnson) distribution would appear to be a reasonable way to deal with situations in which THERP produces a credible best estimate HEP but provides an uncertainty range that is not meaningful. For calculational ease, one can utilize the unconstrained lognormal approximation to this distribution or, if more rigor is desired and resources are available to do the necessary calculations, one can use the truncated lognormal distribution directly to propagate uncertainties.

The method is based on the assumption that the HEPs is THERP are lognormally distributed (i.e., positively skewed) over the range where data is available for estimation. Using a truncated lognormal distribution (i.e, a beta distribution in which both parameters are greater than 1) preserves the general lognormal shape (i.e., positive skewness) and is therefore compatible with the assumptions in THERP regarding lognormal 'y.

An alternative approach would be to retain the unconstrained median value from THERP but adjust the error factor so that the majority of the distribution lies to the left of 1. This approach has the advantage of being simple to implement. It also leaves the original median value unchanged. This approach does reduce the mean value, but not by quite as much as using the bounded Johnson distribution. The principal objection to this approach is that it leaves the range of the random variable unbounded. Of course, one can make the probability that the random variable assumes values greater than one as small as desired by suitable choice of an error factor; however, there is an accompanying decrease in the coefficient of variation, which is undesirable. The authors' preference is for a more rigorous method.

The method of transforming variables is felt to be desirable because it retains the order-of-magnitude point estimate implied by the original median value. In the case of human error probabilities from THERP, more than order-of-magnitude accuracy is not implied (e.g., 0.01 - 0.1, 0.1 - 1). The method also maintains the positive skewness of the original lognormal distribution, but employs a random variable whose range is restricted, a priori, to values

between 0 and 1.

Finally, the authors feel that the constrained lognormal distribution is a natural model for cases where the random variable being modeled represents a probability, and the values of the variable are skewed to the right (mean > median). For actual empirically derived data, the maximum likelihood method for estimating the distribution parameters is tractable and provides minimum variance unbiased estimators. Therefore, fitting a constrained lognormal distribution to probability data (e.g., observed human error rates) is no more difficult than fitting an unconstrained lognormal distribution. However, the constrained lognormal distribution wil? produce a reasonable uncertainty range, even for median probabilities well above 0.1, for which an unconstrained lognormal distribution is really not an appropriate model.

References

- D-1. G. Apostolakis and S. Kaplan, "Pitfalls In Risk Calculations," <u>Reliability Engineering</u> 2 1981, pp. 135-45.
- D-2. A. Swain and H. Gutman, Handbook of Human Reliability Analysis with Emphasis on Nuclear Power Plant Applications, NUREG/CR-1278, August 1983.
- D-3. N. L. Johnson, "Systems of Frequency Curves Generated by Methods of Translation," <u>Biometrika</u> 36, 1949, p. 149.
- D-4. J. Aitchison and J. A. C. Brown, The Lognormal Distribution With Special Reference To Its Uses In Economics, Cambridge University Press, 1957.
- D-5. G. J. Hahn and S. S. Shapiro, Statistical Models In Engineering, John Wiley & Sons, Inc., 1968.

Appendix E

Core Uncovery Timing Calculations

Appendix E

Large Break ISLOCA Core Uncovery calculation for a CE PWR

E.1 INTRODUCTION

This appendix documents the bounding calculation used to estimate the time for the onset of core uncovery for a large break interfacing system LOCA (ISLOCA) sequence. This accident sequence was applied to a CE PWR rated at a core power of 3390 Mw. In particular, the core uncovery time estimate was made after ECCS pump suction was lost from the refueling water storage pool (RWSP). It is possible that core uncovery may occur prior to losing suction to the RWSP, but this issue was not addressed in this document. The incovery time is defined as the time at which the collapsed vessel liquid level will drop below the top of the core fuel rods. The break was assumed to be outside of containment on one of the LPI injection lines; was approximated as an equivalent (30 % of an 8 inch break) primary system break. Large break LOCA assumptions were made; including the assumption that the remaining ECCS trains would immediately reach their rated runout values, and that RCS pressure feed back from decay and piping stored energy would not significantly perturb the ECCS injection rates. The Mathematica computer code was used to do the numerical calculations presented in the following analysis.

E.2 Estimate of time to lose ECCS

This section presents a time estimate to lose suction from the RWSP. A time correction was added to this estimate to account for the presence of SIT injection. This time correction is believed to be very conservative. A number of assumptions were incorporated into the suction loss time estimate and included:

• The ECCS water supply was limited to 443,000 gal with a nominal density of 62 lbm/ft^3 .

• The SIT's were available and the corresponding delay time contribution to core uncovery was estimated by dividing the total ECCS pump runout flow rate by the total volume of the SIT liquid inventory.

• It was assumed that the transient is initiated at 100 % power conditions and that the reactor was immediately scramed.

• Three HPI, three charging pumps, and one LPI were available to recover the primary system. The run-out rate for each HPI equals 910 gpm and for the LPI 5500 gpm. The charging pumps were rated at 42 gpm.

• The RCS depressurization was large enough to cause all ECCS pumps to immediately transition to their run-out injection rates.

The following variables were defined:

Tank = is the mass of the RWSP tank in 1bm

RUNOUTFLOW = is the total ECCS run out flow rate

TIME1 =is the time in hours for the RWSP to empty

Equating the ECCS runout rate to the break flow rate, the estimated time to empty the RWSP is calculated as follows:

Tank= 443000. GAL * .1337 ft³/GAL * 62.0 lbm/ft³ RUNOUTFLOW= (3*910.+ 3*42. +5500.)*62.0*.002228 lbm/sec TIME = Tank/RUNOUTFLOW

TIME1=TIME/(3600. SEC/hr) (*time to empty RWSP in hr*) time to empty RWSP = 0.883729 hr

Sources of uncertainty in this estimate are as follows:

• The ECCS injection rates will not immediately relat to their rated run out discharge rates.

•There is some uncertainty in the known ECCS pump run out mass flow rates.

•The simultaneous use of 3 HPI pumps is an off normal configuration. Generally only two pumps are in operation. It was judged that the assumption of 3 pumps in operation is conservative.

•SIT injection may temporarily alter the RCS pressurization response and reduce LPI flow.

• Primary to secondary heat transfer may also induce RCS pressure feed back that could reduce ECCS injection flows.

•Flooding the auxiliary building may disrupt or destroy the ECCS pumping equipment before the RWSP tank is empty.

An additional time correction to core uncovery was made for the loop safety injection tanks (SIT's). The minimum liquid volume of each SIT equals 1679 ft³ which is representative of some CE plants. We will assume that one SIT empties directly out the break and will not contribute to delaying core uncovery. Once the accumulators are activated we will assume that they empty at an average rate equal to the total run-out rate of all the ECCS pumps. As the accumulators empty, it is assumed an equal amount of liquid is forced out the break. Hence, the actual dwell time for the accumulator liquid in the RCC is equated with the accumulator injection time. This assumption is conservative since this time scale is shown to be small. In order to make the dwell time estimate, which is equated with the time correction, we define the following variables: SITVOL = the total liquid volume of the SIT 's TIME = the estimated time for the SIT's to drain in hours SITVOL= 3*1679. ft³ (*total SII' liquid volume *) TIME2 = (SITVOL/RUNOUTFLOW)*(62.0 lbm/ft³)/(3600. SEC/hr) additional delay time = 0.075 h r TIME3 = TIME1+TIME2

effective time to lose ECCS = 0.959 hr

Therefore the time to lose ECCS is approximately one hour.

E.3 Estimate of core uncovery time

The remainder of this narrative will detail how the time to core uncovery was calculated. This time is referenced to the beginning of the transient. In order to estimate the uncovery time the following assumptions were made:

• No credit for cooling is given to any possible liquid remaining in the reactor coolant loops. This assumption is conservative since after the RWSP has drained some liquid in the primary loops will drain back into he vessel region. Unless a full scale systems calculation is done, there is no way of making a reliable estimate of how much liquid would actually drain into the vessel and help extend the time to core uncovery.

• It was assumed that the vessel upper head and upper plenum region above the loops are completely voided by the time the RWSP has emptied. This assumption is reasonable since most PWR large break LOCA scenarios leau to voiding in these regions.

• At the instant ECCS is lost it was assumed that the partially filled vessel is at bulk subcooled conditions of 100 F and 20 psia. The assumed vessel liquid temperature is not based on rigorous quantitative arguments. Unless a full scale simulation is performed to model the ECCS/vessel liquid mixing dynamics it is not clear that an exact p.essure and bulk temperature can be easily defined. • Assumed vessel volume distributions are based on a Westinghouse 3411 MW 4-loop plant. Volume data for a 3390 MW CE plant were not available. However, since the power levels of these two system are close, it is expected that the volume scaling will also be very similar.

Using steam table data the following variables are defined:

h120 = the enthalpy of the subcooled vessel liquid at 20 psia, 100 F.

hls20 = the saturation enthalpy of the vessel liquid at 20 psia, 100 F.

rho = the bulk density of the liquid at 100 F.

VOLREMAIN = the volume of the reactor vessel minus the regions above the hot and cold legs. This volume was estimated to be 3300 ft^3 .

TOTALEN1 = the total sensible heat needed to heat the reactor vessel coolant from subcooled to liquid saturated conditions.

The above variable definitions were used to estimate the time to reach bulk saturation conditions. Some uncertainty exists here, since temperature gradients may result in non-uniform heating, leading to significant heat up of the core liquid, with the liquid in the lower plenum remaining relatively cool.

```
h120 = 68.04 \text{ btu/!bm}
h1s20 = 196.3 \text{ btu/lbm}
rho=62. (1bm/it^3)
VOLREMAIN= 3300. \text{ ft}^3
TOTALEN1= rho*(h1s20-h120)*VOLREMAIN*1055. J/btu
sensible heat need to reach saturation = 2.769 10<sup>10</sup> J
```

Additional energy added to the vessel liquid is assumed to result in liquid vaporization. In order to initiate core voiding, all the liquid above the core is assumed to be vaporized. From an actual physics standpoint, bulk boiling in the vessel will cause core voiding before the liquid in the upper plenum is boiled off. This will not be of concern because the core mixture level will still be above the fuel rod region. The total energy needed to vaporize the liquid region above the fuel rods is equated to the energy needed to vaporize the saturated liquid in the upper plenum. To complete this energy balance calculation the following variables are defined.

volvap = the remaining upper plenum volume of saturated liquid to be vaporized. deitah =the latent heat change rhosat = the liquid saturation density TOTALEN2 = the total energy needed for upper plenum vaporization.

With the appropriate substitutions and units changes we have:

volvap= 857. ft^3 deltah = (1156.3-196.3) btu/lbm rhosat = 59.4 lbm/ft³ TOTALEN2= volvap*deltah*rhosat*1055. J/btu energy to boiloff liquid in upper plenum = 5.156 10¹⁹ J

Therefore, the total energy needed to initiate core uncovery becomes:

TOTE = $(TOTALEN1 + TOTALEN2) = 7.92 \ 10^{10} \text{ J}$

This analysis used a core decay response which approximates an ANS decay power response for a reactor with an infinite operation period. The normalized decay rates are given as follows:

Time(s)	No.malized	Power
100.	.0331	
400.	.0235	
800.	.0196	
1000.	.0185	
2000.	.0157	

1

E-8

4000.	.0128			
8000.	.0105			
10000.	.00965			
20000.	.00795			
60000.	.00566			
100000.	.00475			

Using the above data, we shall define decay(t) as the normalized core decay power. The expression decay(t) was calculated as a polynomial function of time using supplied Mathematica curve fit routines. The expression for the normalized power was approximated as:

 $decay(t) = 0.01590 - 8.75 \ 10^{-7} \ t + 2.5*10^{-11} \ t^2$

Note that decny(t) is a good approximation in the range 4000-10000 sec, and becomes progressively less accurate outside of this time interval.

To do the energy balance formulation the total integrated core decay power between the time TIME3 and some unknown time t was equated to the total energy TOTE needed for core uncovery. Recall that the core scram was coincident with beginning of the transient and that TIME3 is the time at which ECCS cooling has ended. To calculate the core decay power the normalized power function decay(t) is multiplied by the initial core power. The nominal 100 % power for the CE plant in this analysis is 3390 Mw. The energy balance formulation was written as follows:

TOTE = $P \int_{TIME3}^{t} decay(t) dt$

Where **P** is the initial core power. Using the appropriate numerical and symbolic algorithms supplied with Mathematica and making the correct units changes, the uncovery time solution is t = 1.476 hr.

Appendix F

13

٦

Calculation of System Rupture Probability

Appendix F Calculation of System Rupture Probability

Interfacing system rupture probability is calculated from system pressure and component fragility data. The component fragility data are obtained from structural analyses performed by the ABB Impell Corporation (see Appendix H). The following sections describe the component fragility data, and the methods used to calculate rupture probabilities. The first section describes the system pressure data. This is followed by a description of the component pressure fragility data and limiting assumptions. Then the individual component failure probability calculations are described, followed by the method used to obtain the system rupture probability from the individual failure probabilities.

SYSTEM PRESSURE

Interfacing system rupture probabilities are required for the event trees in Appendix B. Two significant sequences survived the screening process. These are Sequences 2 and 5. Sequence 2 involves pressurization of the low pressure safety injection (LPSI) system to normal operating reactor coolant system (RCS) pressure as a real tof failure of the isolation check valves and opening of the isolation MOV during quarterly stroke-testing. Pressure relief valves in the LPSI pump discharge lines have very small capacities (5 gpm each). Therefore, the discharge piping will rapidly pressurize to the cold leg pressure of 2306 psig. Table F.1 provides a list of the low-pressure rated components that will be subjected to RCS pressures.

In Sequence 5 the operators enter shutdown cooling as per procedure, and valves SI-401, SI-405, and SI-407 are opened while the plant is at -400 psig and 350° F. Note that this is a conservative assumption; the entry into shutdown cooling may be made at lower pressures and temperatures, but these are the maximum allowable by procedure. If check valves SI-108 and SI-1071 fail to seat, 150 psig-rated piping and components will be subjected to higher-than-design pressure. The components that may fail as a result are listed in Table F.1.

F-3

COMPONENT PRESSURE FRAGILITY DATA

The pressure capacities for each type of component in the interfacing systems have been evaluated by the ABB Impell Corporation (see Appendix H). The pressure capacities for these component types have been developed under the assumption of quasi-steady state pressure conditions. While a variety of analytical techniques have been used to obtain pressure fragility data for the above components, they all are expressed as follows:

$$P = \hat{P} * M * S$$

where

- \hat{P} = the component median pressure capacity
- M = a lognormally distributed random variable having unit med $\beta < \tau$, logarithmic standard deviation, β_m , representing modelin uncertainties

(1)

S \sim a lognormally distributed random variable having unit median and a logarithmic standard deviation, $\beta_{\rm s},$ representing material property uncertainties

The random variable P is lognormally distributed with logarithmic standard deviation β_c . In the tabulations provided in Appendix H, the separate uncertainty parameters, β_m and β_s , are not routinely provided. Instead one composite value, β_c , is used to represent both kinds of uncertainty, where β_c is given by

$$\beta_{c} = \sqrt{\beta_{m}^{2} + \beta_{c}^{2}},$$

The component pressure capacities and uncertainty parameters from Appendix H are summarized in Table F.1. Application of the above data for determining system failure probabilities requires some understanding of the failure modes represented. The following discussions provide a summary of the important failure modes and data limitations.

Piping

Piping failure is assumed to result when pressure-induced hoop stress in the unflawed pipe causes failure strain to be reached. Variability in the

Company, st	Med. Fail.	Mean Fail	Ln	Ln	Error	A
	Pressure	Pressure	Mean	Std. Dev.	Factor	P(A)=.001
Sequence 2	at 600° F.					
Flange - 10 inch, 300 ps	si 1888	2150	7.543	0.41	1.96	529.7
Piping - 6 inch sched.	40s 3790	4044	8.240	0.36	1.81	1241.6
Piping - 8 inch sched.	40s 3345	3569	8.115	0.36	1.81	1095.8
Piping - 10 inch sched.	40s 3044	3248	8.021	0.36	1.81	997.2
Sriver te 5	at 400 ⁰ F.					
Flange - 10 inch 150 psi	i flange 941	949	6.855	0,13	1.24	628.9
Flange - 24 inch 150 psi	i flange 640	646	6.471	0,13	1.24	627.7
Piping - 10 inch sched.	40s piping 3271	3490	8.093	0,36	1.81	1071.5
Piping - 14 inch sched.	30 piping 2545	2715	7.842	0,36	1.81	833.7
Piping - 20 inch sched.	20 piping 1752	1869	7.469	0,36	1.81	573.9
Piping - 24 inch sched.	20 piping 1450	1547	7.279	0,36	1.81	475.0

Table F-1. LPSI system component pressure capacity data

pressi e stress results from many factors, such as variability in material strength and wall thickness. Most important among these factors is the existence, size, and orientation of preexisting flaws or cracks in the piping system. Data on these factors are not currently available, so Impell has developed variability data for the lognormal pressure capacity distribution by assuming that there is a failure probability of 0.001 at the yield stress of the piping. This implies that the probability of existence of a large flaw is 0.001. The authors of the Impell report believe this may be an overly conservative assumption and recommend that this assumption should be reevaluated if it is found that a significant contribution to risk occurs from pipe rupture at pressures well below the median burst pressure.

The data provided in Table F.1 include no allowance for the number of feet of piping in the system. The data are also dependent on both the degree to which piping has been corroded and on the temperature to which the piping is exposed. In this analysis, a corrosion allowance of 0.020 in. is used. System temperature is, however, time-dependent, with the lowest temperatures occurring before peak pressure is reached. Since failure pressures generally decrease with increasing temperature, it is conservative to assume a higher temperature than is expected at peak pressure. In this analysis, a system temperature of 600° F. is used in Sequence 2, 400° F. in Sequence 5.

Flanges

The flanges subject to overpressurization in Sequences 2 and 5 are 300 psi-rated 10-inch, and 150 psi-rated 24-inch flanges. The data provided in Appendix H are tabulated according to the initial bolt stress, percent joint relaxation, and bolt yield stress. The initial bolt stress is calculated from data in Table 3-1 of Appendix H, and from the plant's general torquing procedure. Appendix H uses a median initial bolt stress of 25,000 psi, and 25% median joint relaxation. From Table 3-7 in Appendix H, the gross leak pressure (GLP) is found to be 1888 psi. This GLP is a median value for a lognormal distribution with a composite logarithmic standard deviation of 0.12, representing both uncertainty in modeling and data variability.

Similarly, for 24-inch, 150-psi flanges, Appendix H gives a median GLP of 640 psi. For 10-inch, 150-psi flanges, the median GLP is 941 psi.

Leakage from flanged connections can occur at pressures below the GLP. Leak rates for 10-inch, 300-psi flanges are on the order of 1 mg/sec at GLP. For 24-inch flanges, the leak rate at the GLP is approximately 37 mg/sec. Above GLP the flange gasket can separate from the flange surface and significant leak areas can develop. For 10-inch, 300-psi flanges, these leak rates range from 0.03 in.² at 1.25*GLP to 7.39 in.² at 2.0*GLP. The corresponding leak rates for 24-inch flanges are 0.05 and 2.79 in.².

In Sequence 5, flange failures made a negligible contribution to the system rupture probability; therefore, no distinction was made between small and large leaks. In Sequence 2, however, failure of the 10-inch, 300-psi flange at the LPSI discharge flow indicator was a significant contributor to the rupture probability of the interfacing system. In this case, flange failures were grouped into two classes, based on the expected leak area. From Appendix H, at twice the GLP, the leak area is 7.39 in. This is equivalent to a break diameter of slightly more than three inches. Therefore, two times the GLP was used as the criterion for defining a large flange failure. Failures at less than twice the GLP were classified as small leaks, which were judged to be recoverable by the operators in time to prevent core damage (see Appendix C).

F-6

In developing the above pressure capacity and leak area data impell had to make assumptions about the maximum value of the ratio of incremental gasket load to total pressure load. The authors of the Impell report caution that because high strength flange bolts are used at the study plant, some flange inclastic behavior can be expected at higher pressures. This may affect the calculated leak rates. Therefore, they caution that additional study of flange behavior may be required if leakage through flanged connections is found to be a significant contributor to risk.

Valves

Impell postulated three failure modes for valves within the LPSI system. They include failure of the valve body, failure of the stem packing, and failure of the bolted bonnet seal. Impell judged that failure of the valve body would not be likely because the wall thickness of the valve body is greater than the surrounding piping. Packing failure was dismissed because overpressurization would tend to improve the packing seal, and failure, if it were to occur, would result in a negligible leak rate. Failure of the bolted bonnet was considered to be the only credible failure mode for valves but could not be evaluated because the detailed information needed on the bonnet seals was not available. However, it is expected that the fragility of the bonnet seals at the CE plant will be similar to those at the Westinghouse and B&W plants, where all valves affected by overpressurization have large enough leak capacities to justify not including them in the component failure calculations.

One check valve was of particular interest at the CE plant. This was SI-107 (A and B) which prevents reverse flow in the LPSI suction line to the RWSP. This valve is rated to 150 psig, so it could fail in Sequence 5. The failure mode of interest was internal failure that would allow reverse flow into the RWSP. Because detailed information on the dimensions of the internal hinged plates was unavailable, the pressure capacity for this failure mode could not be calculated. However, based on the limited available information, Impell judged that the pressure capacity for this failure mode would be high; therefore, our analysis of Sequence 5 assumed that SI-107 closed on demand, preventing reverse flow to the RWSP.
Calculational Methods

Determination of system rupture probability requires "summing" the probability of component failure over the number of components in the system. The following discussion presents the method used to calculate single component failure probabilities and the resulting system rupture probability.

Convolution

The pressure capacity of a system component is characterized in terms of a random variable P_f , which is assumed to be distributed lognormally with a median and logarithmic standard deviation. The probability density function of P_f is

$$f(p_f) = \frac{1}{\sqrt{2\pi\beta_c}p_f} \exp\left[\frac{(\ln p_f - \mu_f)^2}{2\beta_c^2}\right]$$

where $P_{\star} \geq 0$ and μ_{\star} is the natural logarithm of the median failure pressure.¹

The system pressure is characterized in terms of a normally distributed random variable P_s , with a probability density function given by

$$g(p_s) = \frac{1}{\sqrt{2\pi\sigma_s}} \exp\left[\frac{(p_s - \mu_s)^2}{2\sigma_s^2}\right]$$

where $\mu_{\rm s}$ is the mean (median) system pressure and $\sigma_{\rm s}$ is the standard deviation. 2

A random variable representing the component failure margin is then

¹In the following discussion random variables will be denoted with capital letters and values assumed by random variables will be denoted with lowercase letters, e.g., P_f is the random variable representing failure pressure and p_f represents actual values of the failure pressure.

²In Sequence 5, system pressure was characterized by a single value of 400 psig (the distribution is a delta function). This represents a bounding pressure for the calculation of system rupture probability. Because of the extremely low probability of system rupture at this bounding pressure, this simplification is justified.

defined as

$$Z = P_f - P_g.$$

The probability density function of Z is obtained from P_{f} and $\mathsf{P}_{\mathsf{s}},$ by convolution, as follows

$$h(z) = \int_{0} f_{P_{s}}(p_{f}-z) f_{P_{f}}(p_{f}) dp_{f}$$
$$= \int_{0}^{\infty} \frac{1}{2\pi\sigma_{s}\beta_{c}p_{c}} e^{-\frac{(p_{f}-z-\mu_{s})^{2}}{2\sigma_{s}^{2}}} e^{-\frac{(\ln p_{f}-\beta_{c})^{2}}{2\beta_{c}^{2}}} dp_{f}$$

The probability of system failure is obtained by finding the probability that $Z \leq 0$:

$$P(failure) = P(Z \le 0) = \int_{-\infty}^{0} h(z) dz.$$

To simplify the above calculations, the system probability was characterized by a single pressure, rather than a distribution, thus allowing the rupture probability to be calculated by hand.³ Based on comparison calculations with the full Monte Carlo simulations performed for the previous two ISLOCA analyses, this is a valid approximation. The reason it is valid is that system pressure is defined fairly accurately in the cases of interest here; assigning a standard deviation of 50 psi, as has been done in the previous analyses, does not affect the results significantly.

The probability of system rupture is calculated from the Boolean union of the individual component failure probabilities as expressed in the following equation

 $^{^{3}\}mbox{The system pressure distribution in this case can be described by a delta function.$

$$P(component failure) = \bigcup_{i=1}^{n} P_{f_i}$$

where P_{fi} is the failure probability of the ith component in the system. Assuming the failures to be independent allows us to simplify this equation to

$$P(component \ failure) = \bigcup_{i=1}^{n} P_{f_i}$$
$$= 1 - \prod_{i=1}^{n} (1 - P_{f_i})$$

1 4

Results

The results from the component failure analysis are provided in Tables F.2 and F.3, along with the calculated system rupture probability.

Component	Median Failure Pressure (psig)	Rupture Provability**
10-in., sch 40 pipe	3044	0.22
6-in., sch. 40 pipe	3790	0.08
8-in., sch. 40 pipe	3345	0.16
10-in., 300-psi flange	1888	0.12*
System rupture prob	ability	0.47

Table F.2 Rupture Probabilities for Sequence 2

^{*}This is the probability of a large flange failure. The probability of a small flange failure is 0.57.

**Based on a system pressure of 2306 psig.

Component	Median Failure Pressure (psig)	Rupture Probability*
10-in., sch. 40 pipe	3271	negligible
14-in., sch. 30 pipe	2545	negligible
20-in., sch. 20 pipe	1752	negligible
24-in., sch. 20 pipe	1450	negligible
10-in., 150-psi flange	941	negligible
24-in., 150-r iange	640	negligible
System rupture prot	pability	negligible
*Based on a system pressu	ire of 400 psig.	

Table F.3 Rupture Probabilities for Sequence 5

Appendix G

ISLOCA Consequence Analysis

ISLOCA Consequence Analysis

1.0 INTRODUCTION

This appendix analyzes the effect of decontamination factors (DFs) on the offsite consequences of an ISLOCA accident sequence. The fission product source terms in this analysis are those generated for the Event V accident sequences at the Sequoyah plant for the June 1989 draft of NUREG-1150. In using these ISLOCA accident sequences, source term release information was obtained from the SEQSOR program and partitioned into a small number of source term groups using PARTITION. The consequences of these releases were calculated with the MELCOR Accident Consequence Code System (MACCS), using the Surry plant site input decks developed for the June 1989 draft of NUREG-1150. For this report consequences were calculated for the base case sequences and eight sensitivity studies in which the effect of varying the sequences and of values was examined. Two additional sensitivities examined the effect of increasing the release elevation from the base case value of 0.0 m to 10.0 m.

2.0 ISLOCA ACCIDENT SEQUENCE DEFINITION AND SELECTION

In the NUREG-1150 terminology, the ISLOCA sequences at Sequoyah are defined by plant damage state (PDS) Group 4 - V. The accident sequences in this PDS group are initiated by failure of the check valves between the reactor coolant system (RCS) and the low pressure injection system (LPIS). The LPIS is then overpressurized and ruptures, resulting in a release path from the RCS to the auxiliary building, bypassing the containment. For Sequoyah, there are two types of V sequences: wet and dry. For the wet V srquences the fire sprays in the auxiliary building scrub the release. The dry V sequences are not scrubbed by the fire sprays.

The starting point of the analysis in this report was the results obtained from analyzing the Sequoyah V sequence, PDSG 4, accident progression event tree (APET). This APET was analyzed as part of the Containment Performance Improvement (CPI) project studying improvements for the ice condenser containment. Therefore, reanalyzing this APET was not necessary as

G - 3

the results were readily available. Using the NUREG-1150 methodology, this APET was evaluated 200 times, once for each latin hypercube sample (LHS). Each evaluation or observation resulted in hundreds of different accident sequences. For each observation, the accident sequences were sorted and grouped into a smaller number of similar as ident progression bins (APBs). Because the APET analyzed the wet and dry V sequences together, each observation contained both scrubbed (wet) and unscrubbed (dry) APBs. However, For this analysis, the consequences of the wet and dry sequences were calculated separately. Therefore, the APET analysis results were manually edited to separate the dry APBs from the wet APBs. The total number of APBs for all observations was 2,642 and 2,941 for the dry and wet V sequences, respectively. The number of unique APBs over all observations was 42 and 46, respectively. Listings of the unique APBs for the dry and wet V sequences are given in Tables G.1 and G.2. For the sake of brevity, listings of the APBs over all 200 observation, are not provided here, but have been stored on magnetic tape and can be made available upon request.

dry	V sequences at	Sequoyah.	 	
ANADDCAADAAAAA				
ANADDCAADAAAAA				
AHAUDUAAUAADDD				
ANADDCAADDAAAA				
ANADDCAADDAAAD				
AUADDCAAUDADDD				
AUADDCAADCAAAAA				
ΔΗΔΟΒΕΔΔΟΕΔΔΔΟ				
AHADRCAADCARRA				
AHADRCAADCARRR				
AHADRCAADDAAAA				
AHADBCAADDAAAB				
AHADBCAADDAAAD				
AHADBCAADDABBA				
AHADBCAADDABBB				
AHADBCABDAAAAA				
AHADBCABDAAAAB				
AHADBCABDAABBB				
AHADBCABDBAAAA				
AHADBCABDBAAAB				
AHADBCABDBABBB				
AHADBCABDCAAAA				
AHADBCABDCAAAB				
AHADBCABDCAAAD				
AHADBCABDCABBA				
AHADBCABDCABBB				
AHADBCABDDAAAA				
AHADBCABDDAAAB				
AHADBCABDDAAAD				
AHAUBCABUUABBA				
AHAUSCABUUABBB				
AHAUUCAAUDDACA				
AUADDCAADDDACD				
AUADDCAADEDALU				
AUADOCAADDDCA				
AUADDCARDBBBCD				
AHADDCARDRRACR				
AHAODCARDRRACD				
AHADDCARDRRPCA				
AHADDCABDRBBCR				

Table G.1 Listing of the unique APBs over all 200 LHS observations for the dry V sequences at Sequoyah.

and the surger with property descent starting and the second starting of the second starting of	And in case of the second s	o of around series	In successful ways of a successful way of a succes	and the second	
BHADBCAADAAAAA					
RHADRCAADAAAAR					
DUADDCAADAADDA					
DHAUDLAADAADBA					
BHADBCAADAABBB					
BHADBCAADBAAAA					
BHADBCAADBAAAB					
RHADRCAADRARRA					
DUADDCAADDADDO					
BHAUBCAADBABBB					
DHADBCAADCAAAA					
BHADBCAADCAAAB					
RHADRCAADCAAAD					
PUADDCAADCADDA					
DUADDCAADCADDA					
BHADBCAADCABBB					
BHADBCAADDAAAA					
BHADBCAADDAAAB					
RHADRCAADDAAAD					
DUADDCAADDADDA					
DNAUDLAAUUABBA					
BHADBCAADDABBB					
BHADBCABDAAAAA					
BHADBCABDAAAAB					
RHADRCHRDAARRA					
DUADDCADDAADDA					
DHAUBLADDAABBB					
BHADBCABDBAAAA					
BHADBCABDBAAAB					
BHADSCARDRARRA					
RUADCAPDRADER					
DUADDCADDCAAAAA					
BHA. BLABULAAAA					
BHADBCABDCAAAB					
BHADBCABDCAAAD					
BHADRCARDCARRA					
RUADECADDCADDD					
DIADDCARUCADDD					
SNAUBLE DUAAAA					
BHADBCABDDAAAB					
BHADBCABDDAAAD					
RHADRCARDDARRA					
RUADDCADDDADDD					
DHADDCADDDADDD					
BHADUCAADBBACA					
BHADDCAADBBACB					
BHADDCAADBBACD					
RHADDCAADRRCA					
DUADOCAADDDDCA					
BHAUUCAADBBBCB					
BHADDCAEDBBACA					
BHADDCABDEBACB					
BHADD ABDE ACD					
RUADDCARD					
DHADDUFBULSSUA					
BHADDCABDBBBCB					

Table G.2 Listing of the unique APBs over all 200 LHS observations for the wet V sequences at Sequoyah.

3.0 SOURCE TERM ANALYDIS

3.1 Source Term Description

The next step in the consequence analysis was to determine the source term release information for each AF8 in each observation. The source term for a given bin contains the release fractions for nine radionuclide classes released during the early and late thases of the accident, and additional information about the release timings, the energy of the releases, and the height of the releases. The nine radionuclide classes are: noble mases, iodine, cesium, tellurium, strontium, ruthenium, lanthanum, cerium, and barium.

The source term analysis was performed with the SEQSOR code. SEQSOR is a parametric computer code used to calculate the source terms for each APB in each observation for Sequoyah. In order to study the effect of varying the V sequence DF, minor modifications were mere to the SEQSOR source code and its input files. These modifications are described below.

3.2 Decontamination Factors

For the dry V sequences, the base case DF for all release classes had a uniform value of 1.0 and the release elevation was 0.0 m. The sensitivity analysis involved calculating the source terms with uniform-valued DFs of 5, 10, 50, and 100 for all release classes except the noble gases, for which the DF remained at the base case value of 1.0. For the first two calculations, with DFs of 5 and 10, the release elevation remained at the base case value of 0.0 m. For the last two calculations, with DFs of 50 and 100, the release elevation was specified as 10.0 m. For the dry V sequences, the DFs are specified within the SEQSOR code. Therefore, minor code modifications were required to model the DFs used in the sensitivity analyses. A sample of these code modifications is listed in Table G.3 for the dry V sequence with a DF of 5.0. The release elevation parameter is also specified within the SEQSOR code. The SEQSOR code modification required to specify different elevations is listed in Table G.4.

statistic rest in such	more another second as a subscription of the second s
Modifi	ication_1:
C DJP C DJP	VDF~1.0 VDF~5.0
Modif	ication 2:
C DJP C DJP	DFE=1.0 DFE=5.0
Modif	ication 3:
C DJP C C C C C C C C C C C C C C C C C C C	FOR V-SEQUENCE WITH WATER: IF(CHH.EQ.'B')THEN DFE=VDF DO 22621 ISP=2,NSP DFL(ISP)=AMAX1(VDF,VPS(ISP)) CONTINUE
22621 C DJP	FOR DRY V-SEQUENCE: IF(CHH.EQ.'A')THEN DFE=VDF DO 22621 ISP=2,NSP UFL(ISP)=:MAX1(VDF,VPS(ISP)) CONTINUE

Table G.3 Listing of SEQSOR code modifications for the dry V sequence with DF = 5.0.

Table G.4	Listing of	SEQSOR	code i	modifications	required	to	specify	a release
	elevation (of 10.0	m.					

Modification 1:

C DJP C ELEV = 0. ELEV = 10.0 C DJP

Table G	.5 Dis as	tribution specified	VDF for ia the	r the ba SEQSOR	se case input fi	analysis le ISTDA	of the L.DAT.	wet V sequences
0%	1%	5%	25%	50%	75%	95%	9.9%	100%
5.1E+03	4.52+03	4.12+03	1.3E+02	6.2E+00	3.0E+00	1.8E+00	1.7E+00	1.6E+00

For the wet V sequences, the base case DF was specified as a distribution to be sampled upon. This distribution is input into SEQSOR as the parameter VDF in the SEQSOR input file, ISTDAT.DAT. The distribution VDF is given in Table G.5 for the uase case analysis of the wet V sequences. As for the dry V sequences, sensitivity analyses were performed with uniform-valued DFs of 5, 10, 50, and 100 for all release classes except the noble gases, for which the DF remained at the base case value of 1.0. For the first two sensitivity calculations, with DFs of 5 and 10, the release elevation remained at the base case value of 0.0 m. The calculations with DFs of 50 and 100 were performed with the release elevation specified both at the base case value of 0.0 m and at 10.0 m. No SEQSOR code modifications were necessary for these wet V sequences. Instead, the distribution of parameter VDF in the SEQSOR input file ISTDAT.DAT was replaced with a uniform-valued DF.

3.3 Partitioning Into Source Term Groups

Because source terms are calculated for each APB for each observation in the sample (a total of 2,642 and 2,941 source terms for the dry and wet V sequences, respectively), the output from the SEQSOR calculations is quite voluminous and will not be reproduced here. However, this output has been stored on magnetic tape and can be made available upon request. Also, because of the very large number of APBs, performing consequence calculations for each APB is not practicable. Instead, APBs with similar source terms were grouped together using the PARTITION computer program. PARTITION takes the source term information output from SEQSOR and estimates the early and chronic health effects of each source term for each APB. Source terms with similar effects are then grouped together into source term groups. PARTITION is an interactive program and the number of source term groups is specified by the user. The number of source term groups can be further reduced by repooling small source term groups. For the analyses in this report, the final number of source term groups for each sensitivity was 2 except for the base case wet V sequence for which 3 source term groups were used. Table G.6 lists the record of the interactive computer session for the dry V sequence with DF = 5.0 sensitivity.

When the partitioning and repooling is completed, PARTITION creates a single frequency-weighted source term for each source term group. This source term information is then output by PARTITION under the file name MACCS.INP.

4.0 CONSEQUENCE ANALYSIS

A PC version of the MELCOR Accident Consequence Code System Version 1.5.11 (MACCS 1.5.11) was utilized for the consequence analysis. The computer code comprises a single FORTRAN77 program consisting of three basic modules, ATMOS, EARLY, and CHRONC, which are exercised in sequence. This program has been developed for the purpose of evaluating the severe accident consequences at commercial LWR power plants. Table G.6 Listing of PARTITION interactive computer record for the dry V sequence with DF = 5.0.

DO YOU WANT SUMMARY CONSEQUENCE RESULTS FOR EACH SAMPLE ELEMENT (NO PARTITIONING PERFORMED)? (Y OR N)

LIST OF FILES READ FROM PARTITION.INP: MASTER BIN LIST FILE =../dry_nbr_mas.kep SOURCE TERM WEIGHT FILE =../seqwgt.inp SOURCE TERM FILE =../dry5/release.out APB COND PROB FILE =../dry_seq.frq PDS FREQ FILE =../temac.isT

SOURCE TERM FILE CONTAINS SOURCE TERMS WITH 9 RELEASE FRACTIONS

	NUMBER OF SOURCE TERMS	PERCENT OF TOTAL WEIGHTED FREQUENCIES
EF>0 AND CF:	>0 2642	100.00
EF=0 AND CF:	>0 0	0.00
EF=0 AND CF:	=0 0	0.00
TOTAL	2642	100.00

<PRESS RETURN TO CONTINUE>

THE GRID SIZE FOR THE DATA IS DETERMINED BY THE RATIO: (MAX VALUE IN A GRID CELL)/(MIN VALUE IN A GRID CELL) GRID SIZES CORRESPONDING TO POSSIBLE RATIOS USED WITH THESE DATA ARE AS FOLLOWS:

R/	ATIC)	GRID SIZE	
1.34	TO	1.37	7 X 12	
1.38	TO	1.40	7 X 11	
1.41	TO	1.41	6 X 11	
1.42	TO	1.47	6 X 10	
1.45	TO	1.50	6 X 9	
1.51	TO	1.54	5 X 9	
1.55	TO	1.64	5 X 8	
1.65	TO	1.66	5 X 7	
1.67	TO	1.78	4 X 7	
1.79	TO	1.96	4 X 6	
1.97	TO	2.00	3 X 6	
2.01	TO	2.38	3 X 5	
2.39	TO	2.75	3 X 4	
2.76	TO	3.18	2 X 4	
3.19	TO	5.67	2 X 3	
5.68	TO	7.60	2 X 2	
7.61	TO	10.00	1 X 2	

Table G.6 Continued

SELECT GRID SIZE BY ENTERING DESIRED RATIO 9.000000 THE SPECIFIED RATIO PRODUCES A GRID SIZE OF 1 X 2 IS THIS GRID SIZE ACCEPTABLE? (Y OR N) Y CELL COUNTS WITHIN THE GRID FOR A TOTAL COUNT OF 2642:

1 2 1 681 1961

<PRESS RETUPN TO CONTINUE>

PERCENTAGE OF WEIGHTED FREQUENCIES CONTAINED IN EACH CELL:

1 2 1 |16.12 |83.88 |

DO YOU WISH TO POOL THE VALUES IN ANY CELLS WITH ADJACENT CELLS? (Y OR N) N

MEAN OF AND EF COORDINATES FOR EACH CELL:

1 2 CF EF CF EF 1 3.39 0.95 4.04 1.35

FOR EF=0 AND CF>0, RANGE OF X=LOG10(CF)=******* TO ******* ENTER THE NUMBER OF GROUPS TO BE USED TO PARTITION BASED ONLY ON CF (0 < NUMBER OF GROUPS < 13)

CELL COUNTS WITHIN THE GRID FOR A TOTAL COUNT OF 0:

PERCENTAGE OF WEIGHTED FREQUENCIES CONTAINED IN EACH CELL:

1 1 |

Table G.6 Continued

DO YOU WISH TO POOL THE VALUES IN ANY CELLS WITH ADJACENT CELLS? (Y OR N) N DO YOU WISH TO REDO POOLING? (Y OR N) N ENTER THE EVACUATION ESCAPE TIME (SEC) 1800.000 ENTER THE EVACUATION DELAY TIME (SEC) 7200.000 DO YOU WISH TO ENTER A SUBGROUP 2 CUTOFF TIME FOLLOWING START OF RELEASE 1? (Y OR N) Y ENTEP CUTOFF TIME (S) 3600.000 DO YOU WISH TO POOL THE SUBGROUP 3 SOURCE TERMS WITH SUBGROUP 4 SOURCE TERMS? (Y OR N) Y ENTER THE CONDITIONAL PROBABILITY BELOW WHICH SUBCELLS WILL BE COMBINED 0.000000E+00 DO YOU WISH TO GENERATE LINE PRINTER PLOTS FOR INDIVIDUAL SOURCE TERM PARAMETERS FOR SOURCE TERM CELLS AND SUBCELLS? (Y OR N) N DO YOU WISH TO GENERATE FILE FOR USE IL CALCULATION OF PARTIAL CORRELATION COEFFICIENTS? (Y OR N) ¥

4%

6

D

60

Table G.7 Listing of the PARTITION output file MACCS.INP containing the source term information of the source term groups for the dry V sequence with DF = 5.0.

SEQSOR SOURCE TERMS FOR MACCS

9	3			
9 1 -4799.998 10250.00 2.1361155E-02 3.5504778E-05 2.2498136E-02	4 2.108 8449.999 21599.99 2.1777147E-02 1.4072780E-04 9.7113749E-04	82160E-08 0.16 1250.000 0.0000000E+00 2.3204859E-03 7.4236863E-04 1.0220926E-02	12279 3649.999 3699999. 6.5133086E-04 169999.9 4.1008261E-03	1800.000 0.9902838 1.5444255E-04 9.7354446E-03 4.6844118E-05
4.1119891E-04 1 0. 2 0. 3 0. 4 2 -4799.998 10250.00 2 1361155E-02	4.2428359E-04 .0000000E+00 0. .0000000E+00 0. .0000000E+00 0. 1082160E-08 1 8449.999 21599.99 2 1777147E-02	3.1941610E-03 .0000000E+00 0 .0000000E+00 0 .0000000E+00 0 1.000000 0 1250.000 0.0000000E+00 2.3204859E-03	.0000000E+00 .0000000E+00 .1612279 .3649.999 .3699999. .6.5133086E-04	1800.000 0.9902838 1.54442555.04
2.13011352-02 3.5504778E-05 2.2498136E-02 4.1119891E-04 2 -4800.000	1.4072780E-04 9.7113749E-04 4.2428359E-04 4 1.096 8449.995	7.4236863E-04 1.0220926E-02 3.1941610E-03 57797E-07 0.838 1250.000	169999.9 4.1008261E-03 87731 3650.001	9.7354446E-03 4.6844118E-05
10250.01 7.8708678E-02 1.9877709E-03 7.0597276E-02 1.0439638E-03 1 0.	1600.01 /.9747535E-02 1.2582530E-02 1.1536309E-02 8.7401050E-04 0000000E+00 0.	0.0000000E+00 1.9105045E-02 1.9511884E-02 3.9632775E-02 1.0390015E-02 0000000E+00 0	3700003. 1.9408649E-02 170000.1 1.2491394E-02	0.9866727 3.5171590E-03 1.3340072E-02 1.3667987E-04
2 0. 3 0. 4 1. -4800.000 10250.01 7.8708678E-02 1.9877709E-03 7.0597276E-02 1.0439638E-03 3	0000000E+00 0. 0000000E+00 0. 0967797E-07 1 8449.995 21600.01 7.9747535E-02 1.2582630E-02 1.1536309E-02 8.7401050E-04 4 0.000	0000000E+00 0. 0000000E+00 0. 1250.000 0.0000000E+00 1.250.000 0.0000000E+00 1.9105045E-02 1.9511884E-02 3.9632775E-02 1.0390015E-02 0000E+00 0.000	.0000000E+00 .0000000E+00 .8387731 .3650.001 .3700003. 1.9408649E-02 .170000.1 1.2491394E-02	1800.001 0.9866727 3.5171590E-03 1.3340072E-02 1.3667987E-04

MACCS requires six input files. The EARLY and CHRONC modules each require their own input files, as well as two auxiliary data files: a site data file and a dose conversion file. The June 1989 draft NUREG-1150 EARLY and CHRONC input files for Surry were obtained from Sandia National Laboratories (SNL). They are not reproduced here but are available upon request. These two input files did not require modification for this analysis.

The dose conversion file used was "MACCS DOSE CONVERSION FILE: MOD SER #32, 6-JUL-89, 15:59:19 SANDIA NATIONAL LABORATORIES, J. JOHNSON." This is the most recent dosimetry file provided by SNL for MACCS 1.5. The site data file used was also from the NUREG-1150 MACCS 1.5.11 Surry model. These files are not reproduced here but are available upon request.

The ATMOS module requires a meteorological file for the site and an ATMOS input deck. The meteorological data file used was the NUREG-1150 MACCS 1.5.11 METSUR data file for the Surry site and was obtained from SNL. The ATMOS input deck contains the source term release information and a number of other input parameters which are site and plant-specific. Because this analysis combines the Sequoyah plant with the Surry site, the ATMOS input deck obtained from SNL had to be modified. Also, the source term information output from PARTITION was reformatted and then appended to the MACCS ATMOS input deck. A detailed listing of the ATMOS input file used for this analysis follows:

ATMOS Input Description

```
Variable Name - FILE25
Purpose - Specify EARLY User Input File.
FILE25='C:\MACCS\ISLOCA\V1 EAR.INP', STATUS='OLD', ACTION='READ'
Variable Name - FILE26
Purpose - Specify CHRONC User Input File.
FILE26='C:\MACCS\ISLOCA\V1 CHR.INP', STATUS='OLD', ACTION='REAU'
Variable Name - FILE27
               - Specify DOSE Factors Input Data File.
Purpose
FILE27='C:\MACCS\ISLOCA\dosdat8', STATUS='OLD', ACTION='READ'
Variable Name - FILE28
Purpose - Specify Meteorological Input Data File.
FILE28='C:\MACCS\ISLOCA\metsur', STATUS='OLD', ACTION='READ'
Variable Name - FILE29
Purpose - Specify Site Input Data File.
FILE29='C:\MACCS\ISLOCA\sursilo', STATUS='OLD', ACTION='READ'
Variable Name - FILEO6
Purpose - Specify MACCS Output File (Case Specific).
CASE 1, STG 1: FILEO6='C:\MACCS\ISLOCA\DRY\V1_MACCS.01', STATUS='UNKNOWN',
                     CARRIAGE CONTROL= 'FORTRAN'
CASE 1, STG 2: FILEO6='C:\MACCS\ISLOCA\DRY\V1 MACCS.02', STATUS='UNKNOWN',
                     CARRIAGE CONTROL='FORTRAN'
```

CASE 2, STG 1: FILEO6='C:\MACCS\ISLOCA\DRY5\V5_MACCS.01', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 2, SIG 2: FILE05='C:\MACCS\ISLOCA\DRY5\V5_MACCS.02', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 3, STG 1: FILEO6='C:\MACCS\ISLOCA\DRY10\V10MACCS.01', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 3, STG 2: FILEO6='C:\MACCS\ISLOCA\DRY10\V10MACCS.0L', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 4, STG 1: FILEO6='C:\MAUCS\ISLOCA\DRY50\V50MACCS.01', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 4, STG 2: FILEO6='C:\MACCS\ STALDRY50\V50MACCS.02', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 5, STG 1: FILEO6='C:\MACCS\ISLOCA\DRY100\V100MACC.01', STATUS' JNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 5, STG 2: FILEO6='C:\MACCS\ISLOCA\DRY100\V100MACC.02', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 6, STG 1: FILEO6='C:\MACCS\ISLOCA\WET\W1_MACCS.01', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 6, STG 2: FILEO6='C:\MACCS\ISL\CA\WET\W1_MACC^.02', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 6, STG 3: FILEO6='C:\MACCS\ISLOCA\WET\W1_MACCS.O3', STATUS='UNKNOWN', CARRIAGE CONTRO.='FORTRAN'

CASE 7, STG 1: FILEO6='C:\MACCS\ISLOCA\WET5\W5_MACCS.01', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 7, STG 2: FILE06='C:\MACCS\ISLOCA\WET5\W5_MACCS.02', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 8, STG 1: FIL 106='C:\MACCS\ISLOCA\WETIO\WIOMACCS.01', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 8, STG 2: FILE06='C:\MACCS\ISLOCA\WETIO\W10MACCS.02', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 3 STG 1: FILEO5='C:\MACCS\ISLOCA\WET50\W50MACCS.01', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 9, STG 2: FILEO6='C:\MACCS\ISLOCA\WET50\W50MACCS.02', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 10, STG 1: FILE06='C:\MACCS\ISIOCA\WET50\W51MACCS.01', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 10, STG 2: FILE06='C:\MACCS\ISLOCA\WET50\W51MACCS.02', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 11, STG 1: FILE06='C:\MACCS\ISLOCA\WET100\W100MACC.01', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 11, STG 2: FILEO6='C:\MACCS\ISLOCA\WET100\W100MACC.02', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 12, STG 1: FILEO6='C:\MACCS\ISLOCA\WET100\W110MACC.01', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

CASE 12, STG 2: FILE06='C:\MACCS\ISLOCA\WET100\W110MACC.02', STATUS='UNKNOWN', CARRIAGE CONTROL='FORTRAN'

Variable Nam	e - ATNA	M1				
Purpose	- Iden	tifier for	specific	ATMOS case		
RIATNAM1001	'SURRY SI	TE, SEQUOY	AH SOURCE	TERMS FOR	ISLOCA'	
Variable Nam	e - ENDA	۲1				
Purpose	- Flag to b	to indica e run.	te that th	is is the	last nrogra	m in the series
Source	- NURE	G-1150 MAC	CS 1.5.11	Surry Mode	1.	
OCENDAT1001	.FALSE. (SET THIS V	ALUE TO .T	RUE. TO SK	IP EARLY AN	D CHRONC)
Variable Nam	e - NUMR	AD				
Purpose	- Numb	er of radi	al spatial	elements	defined in	the model.
Source	- NURE	G-1150 MAC	CS 1.5.11	Surry Mode	1.	
GENUMRAD001	26					
Variable Nam	e - SPAE	ND				
Purpose	- Dist	ance in me	ters to th	e end of t	he spacial	intervals.
Source	- NURE	G-1150 MAC	CS 1.5.11	Surry Mode	1.	
* SURRY						
*						
GESPAEND001	.16	. 52	1.21	1.61	2.13	
GESPAEND002	3.22	4.02	4.83	5.63	8.05	
GESPAEND003	11.27	16.09	20.92	25.75	32.19	
GESPAEND004	40.23	48.28	64.37	80.47	112.65	
GESPAEND005	160.93	241.14	321.87	563,27	804.67	
CESPAENDODA	1600 34					

Variable Name	- NUMISO	
Purpose	Number	of nuclides defined in the model.
Source	- NUREG-	1150 MACCS 1.5.11 Sequoyah Model.
ISN/4150001 6	0	
Variable Name	- MAXGRP	
Purpose	- Number	of nuclide groups defined in the model.
Source	- NUREG-	1150 MACCS 1.5.11 Sequeyah Model.
ISMAXGRP001 9		
Variable Name	- WETDEP	
Purpose	- Logica	1 flag for each of the nuclide groups that indicate
	whethe	r they are subject to wet deposition.
Source	- NUREG-	1150 MACCS 1.5.11 Surry Model.
Variable Name	- DRYDEP	
Purpose	- Logica	1 flag for each of the nuclide groups that indicate
	whethe	r they are subject to dry deposition.
Sr	- NUREG-	1150 MACCS 1.5.11 Surry Model.
*	WETDEP	DRYDEP
*		
ISDEPFLA001	.FALSE.	,FALSE.
ISDEPFLA002	.TRUE.	. TRUE .
ISDEPFLA003	.TRUE.	. TRUE .
ISDEPFLA004	. TRUE .	. TRUE .
ISDEPFLA005	.TRUE.	. TRUE .
ISDEPFLA006	, TRUE .	. TRUE .
ISDEPFLA007	.TRUE.	.TRUE.
ISDEPFLA008	, TRUE .	. TRUE .
ISDEDEL MOOD	TRUE	TRUE.

Variable Name	- NUCNAM				
Purpose	- Identi	fying name a	ssociated	with each on th	e nuclides.
Source	- NUREG-	1150 MACCS 1	.5.11 Sur	ry Model.	
Variable Name	- PARENT				
Purpose	- Name o	f parent nuc	lide if a	ny.	
Source	- NUREG-	1150 MACCS 1	.5.11 Sur	ry Model.	
Variable Name	- IGROUP				
Purpose	- Chemic	a' group to	which nuc	lide is assigned	14. ¹
Source	- NUREG-	1150 MACCS 1	1.5.11 Sur	ry Model.	
Variable Name	- HAFLIF				
Purpose	- Half-1	ife of the i	isotope in	seconds.	
Source	- NU (G-	1150 MACCS 1	1.5.11 Sur	ry Model.	
*	NUCNAM	PARENT	IGROUP	HAFLIF(S)	
*					
ISOTPGN/001	CO-58	NONE	6	6.160E+06	
ISOTPGRP002	CO-60	NONE	6	660E+08	
I SOTPGRP003	KR-85	NONE	1	3.386E+08	
ISOTPGRP004	KR-85M	NONE	1	1.613E+04	
I SOTPGRP005	KR-87	NONE	1	4.560E+03	
ISOTFGRP006	KR-88	NONE	1	1.008E+04	
ISOTPGRP007	RB-86	NONE	3	1.611E+06	
I SOTPGRP008	SR-89	NONE	5	4.493E+06	
ISOTPGRP009	SR-90	NONE	5	8.865E+08	
ISOTPGRP010	SR-91	NONE	5	3.413E+04	
ISOTPGRP011	SR-92	NONE	5	9.756E+03	NEW
ISOTPGRP012	Y-90	SR-90	7	2.307E+05	
ISOTPGRP013	Y-91	SR-91	7	5.080E+06	
ISOTPGRP014	Y-92	SR-92	7	1.274E+04	NEW
ISOTPGRP015	Y-93	NONE	7	3.636E+04	NEW

ISOTPGRP016	ZR-95	NONE	7	5.659E+06	
ISOTPGRP017	ZR-97	NONE	7	6.048E+04	
ISOTPGRP018	NB-95	ZR-95	7	3.033E+06	
ISOTPGRP019	MO-90	NONE	6	2.3772+05	
ISOTPGRP020	TC-99M	MO-99	6	2.167E+04	
1SOTPGRP021	RU-103	NOME	6	3.4218+06	
ISOTPGRP022	RU-105	NONE	6	1.598E+04	
ISI TRP023	RU-106	NONE	6	3.188E+07	
ISGTPGRP024	RH-105	RU-105	6	1.278E+05	
ISOTPGRP025	SB-127	NONE	4	3.283E+05	
ISOTPGRP026	SB-129	NONE	4	1.562E+04	
ISOTPGRP027	TE-127	SB-127	4	3.366E+04	
ISOTPGRP028	TE-127M	NONE	4	9.418E+06	
ISOTPCRP029	TE-129	SB-129	4	4.200E+03	
ISOTPGRP030	TE-129M	NONE	4	2.886E+06	
ISOTPGRP031	TE-131M	NONE	4	1.080E+05	
1SOTPGRP032	TE-132	NONE	4	2.808E+05	
ISOTPGRP033	1-131	TE-131M	2	6.9475+05	
ISOTPGRP034	I-132	TE-132	2	8.226E+03	
ISOTPGRP035	1-133	NONE	2	7.488E+04	
ISOTPGRP336	I-134	NONE	2	3.156E+03	
ISOTPGRP037	1-135	NONE	2	2.371E+04	
ISOTPGRP038	XE-133	I-133	1	4.571E+05	
ISOTPGRP039	XE-135	I-135	1	3.301E+04	
ISOTPGRP040	CS-134	NONE	3	6.501E+07	
ISOTPGRP041	CS-136	NONE	3	1.123E+06	
ISOTPGRP042	CS-137	NONE	3	9.4952+08	
ISOTPGRP043	BA-139	NONE	9	4.986E+03	NEW
ISOTPGRP044	BA-140	NONE	9	1.105E+06	
ISOTPGRP045	LA-140	BA-140	7	1.448E+05	
ISOTPGRP046	LA-141	NONE	7	1.418E+04	NEW
1SOTPGRP047	LA-142	NONE	7	5.724E+03	NEW
ISOTPGRP048	CE-141	LA-141	8	2.811E+06	PARENT ADDED

ISOTPGRP049	CE-143	NONE	8	1.188E+05
ISOTPGRP050	CE-144	NONE	8	2.457E+07
ISOTPGRP051	PR-143	CE-143	7	1.173E+06
ISOTPGRP052	ND-147	NONE	7	9.495E+05
ISOTPGRP053	NP-239	NONE	8	2.030E+05
1SOTPGRP054	PU-238	CM-242	8	2.809E+09
ISOTPGRP055	PU-239	NP-239	8	7.700E+11
ISOTPGRP056	PU-240	CM-244	8	2.133E+11
ISOTPGRP057	PU-241	NONE	8	4.608E+08
ISOTPGRP058	AM-241	PU-241	7	1.366E+10
ISOTPGRP059	CM-242	NONE	7	1,408E+07
ISOTPGRP060	CM-244	NONE	7	5.712E+08

Variable	Name	ж.	CWASH1						
Purpose			Linear	term	of	the	washou	t factor.	
Source		*	NUREG-1	150	MACC	S 1.	5.11 S	urry Model	×

WDCWASH1001 9.5E-5 (JON HELTON AFTER JONES, 1986)

Variable Name	CWASH2	
Purpose	The exponential term for the washout factor	k
Source	NUREG-1150 MACCS 1.5.11 Surry Model.	

WDCWASH2001 0.8 (JON HELTON AFTER JONES, 1986)

Variable Name	- NPSGRP
Purpose	- The number of particle size groups that are used for dry
	deposition.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.

ODNPSGRP001 1

Variable Name - VDEPOS Purpose - The representative dry deposition velocities associated with each of the particle size groups. - NUREG-1150 MACCS 1.5.11 Surry Mode). Source DDPDEPOSOO1 0.01 Variable Name - CYSIGA Purpose - The linear term in the expression for sigma-y for 5 stability classes. Source - NUREG-1150 MACCS 1.5.11 Surry Model. * STABILITY CLASS: A B C D E F DPCYSIGA001 0.3658 0.2751 0.2089 0.1474 0.1046 0.0722 Variable Name - CYSIGB - The exponential term of the expression for sigma-y, 6 Purpose stability classes. Source - NUREG-1150 MACCS 1.5.11 Surry Model. * STABILITY CLASS: A B C D E F DPCYSIGB001 .9031 .9031 .9031 .9031 .9031 .9031 Variable Name - CZSIGA Purpose - The linear term of the expression for sigma-z, 6 stability classes. Source - NUREG-1150 MACCS 1.5.11 Surry Model. * STABILITY CLASS: A B C D - E F DPCZSIGA001 2.5E-4 1.9E-3 .2 .3 . 4 .2

Variable Name	- CZS168
Purpose	- The exponential term of the expression for sigma-z, 6
	stability classes.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
* STABILITY CL	ASS; A U C D E F
*	
DPCZSIGB001	2.125 1.6021 .8543 .6532 .6021 .6020
Variable Name	- YSCALE
Purpose	- The linear scaling factor for the sigma-y function.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
DPYSCALE001	1.
Variable Name	- ZSCALE
Purpose	- The linear scaling factor for the sigma-z function.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
DPZSCALE001	1.27
Variable Name	- TIMBAS
Purpose	- The time base for the expansion factor (seconds).
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
PMTIMBAS001	600. (10 MINUTES)
Variable Name	- BRKPNT
Purpose	- The break point in the formula used for calculating the
	plume meander expansion factor.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
PMBRKPNT001	3600. (1 HOUR)

u-26

Variable Name	- XPFAC1
Purpose	- Exponential expansion factor number 1.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
PMXPFAC1001	0.2
Variable Name	- XPFAC2
Purpose	- Exponential expansion factor number 2.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
PMXPFAC2001	0.25
Variable Name	- SCLCRW
Purpose	 Scaling factor for the critical wind speed for entrainment of a buoyant plume.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
PRSCLCRW001	1.
Variable Name	- SCLADP
Purpose	- Scaling factor for the a-d stability plume rise formula.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
PRSCLADP001	1.
Variable Name	- SCLEFP
Purpose	- Scaling factor for the e-f stability plume rise formula.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
PRSCLEFP001	1.

Variable Name	BOILDW
Purpose	- Width of the reactor building in meters.
Source	- NUREG-1150 MACCS 1.5.11 Sequoyah Model.
WEBUILDWOO1 40	. * SEQUOYAH
Variable Name	- BUILDH
Purpose	- Height of the reactor building in meters.
Source	- NUREG-1150 MACCS 1.5.11 Sequoyah Model.
WEBUILDHOO1 40). * SEQUOYAH
Variable Name	- IDEBUG
Purpose	- Debug output flag (0 - no debug).
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
OCIDEBUG001 0	
Variable Name	- NUCOUT
Purpose	- Specifies which nuclide will appear on the dispersion
	listing if one is produced. The dispersion listing is only
	produced if IDEBUG is greater than zero.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
	이는 그는 것 같은 것은 것 같은 것 같은 것 같은 것 같은 것 같은 것 같은
OCNUCOUTOO1 CS	-137
Variable Name	METCOD
Purposo	Motoovological campling option code
rurpose	- meteororogical sampling option code.
	metcod = 1, user specified day and nour in the year,
	2, weather category bin sampling,
	3, 120 nours of weather specified on the atmos user
	input file,
	4, constant met,

	stratified random samples for each day of the year.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
MIMETCODO01 2	
Variable Name	- ISTRDY
Purpose	- Day in the year on which the weather sequence is to begin (Not used if METCOD is specified as 2.)
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
M3ISTRDY001 1	
Variable Name	- ISTRDY
Purpose	- Hour of day on which the weather sequence is to begin (Not used if METCOD is specified as 2.)
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
M3ISTRHR001 1	
/ariable Name	- LIMSPA
Purpose	- Last Spacial Interval for Measured Weather
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
42LIMSPA001 25	
/ariable Name	BNDMXH
urpose	- Boundary weather mixing layer height.
Source	- NUREG-1150 MACUS 1.5.11 Surry Model.
and an available to a state when the lat	

M2BNDMXH001 1000. (METERS)

Variable Name	- IBDSTB
Purpose	- Boundary weather stability class index.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
M2IBDSTB001 4	(D-STABILITY)
Variable Name	- BNDRAN
Purpose	- Boundary weather rain rate.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
M2BNDRAN001 5.	O (MM/HR)
Variable Name	- BNDWND
Purpose	- Boundary weather wind speed.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model specifies 4.0 M/S.
	However, 5.0 was used in these calculations. Sensitivity
	calculations indicate this difference in wind speed has no
	effect on the consequence results.
M2BNDWND001 5.	0 (M/S)
Variable Name	- NSMPLS
Purpose	- Number of samples per bin.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
M4NSMPLS001 4	(THIS NUMBER SHOULD BE SET TO 4 FOR RISK ASSESSMENT)
Variable Name	- NRNINT
Purpose	- Number of rain distance intervals for binning.
Source	- NUREG-1150 MACCS 1.3.11 Surry Model.
MANENINTOOL 6	

Variable Name	RNDSTS
Purpose	- Endpoints of the rain distance intervals (kilometers).
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
M4RNDSTS001 3	.22 5.63 11.27 20.92 4 80.47
Variable Name	- NRINTN
Purpose	- Number of rain intensity breakpoints.
Source	NUREG-1150 MACCS 1.5.11 Surry Model.
M4NRINTNO01	3
Variable Name	- RNRATE
Purpose	- Rain intensity breakpoints for weather binning
	(millimetors per hour).
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
M4RNRATE001	2. 4. 6.
Variable Name	- IRSEED
Purpose	- Initial seed for random number generator.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
M4IRSEED001	79
Variable Name	- CORINV
Purpose	- Defines the total core inventory for each nuclide, NUCNAM.
Source	- 3412 MWTH PWR core inventory, end-of-cycle, supplied by D.
	E. Bennett, 5/14/36, NUREG-1150 MACCS 1.5.11 Model.
*	NUCNAM CORINV(CI)
*	
RDCORINV001	CO-58 3.223E+16

		ATMOS Input (cont	.)
RDCORINVOOZ	CO-60	2.465E+16	
RDCORINV003	KR-85	2.475E+16	
RDCORINV004	KR-85M	1.159E+18	
RDCORINV005	KR-87	2.118E+18	
RDCORINV006	KR-88	2.864E+18	
RDCORINV007	RB-86	1.888E+15	
RDCORINV008	SR-89	3.590E+18	
RDCURINV009	SR-90	1.938E+17	
RDCORINV010	SR-91	4.616E+18	
RDCOR1NV011	SR-92	4.803E+18	
RDCORINV012	Y-90	2.079E+17	
RDCORINV013	Y-91	4.374E+18	
RDCORINV014	Y-92	4.821E+18	
RDCORINV015	Y-93	5.454E+18	
RDCORINV016	ZR-95	5.526E+18	
RDCORINV017	ZR-97	5.759E+18	
RDCORINV018	NB-95	5.224E+18	
RDCORINV019	MO-99	6.098E+18	
RDCORINV020	TC-99M	5.263E+18	
RDCORINV021	RU-103	4.542E+18	
RDCORINV022	RU-105	2.954E+18	
RDCORINV023	RU-106	1.032E+18	
RDCORINV024	RH-105	2.046E+18	
RDCORINV025 SB-127		2.787E+17	
RDCORINV026 SB-129		9.872E+17	
RDCORINV027	TE-127	2.692E+17	
RDCORINV028	TE-127M	3.564E+16	
RDCORINV029	TE-129	9.267E+17	
RDCORINV030	TE-129M	2.443E+17	
RDCORINV031	TE-131M	4.680E+17	
RDCORINV032	TE-132	4.658E+18	
RDCORINV033	RINV033 I-131 3.206E+18		
RDCORINV034	I-132	I-132 4.725E+18	

		ATMUS Input (cont.)		
RDCORINV035	1-133	6.779E+18		
RDCOR1NV036	1-134	7.440E+18		
RDCORINV037	1-135	6.392E+18		
RDCORINV038	XE-133	6.782E+18		
RDCORINV039	XE-135	1.273E+18		
RDCORINV040	CS-134	4.324E+17		
RDCOR1NV041	CS-136	1.316E+17		
RDCORINV042	CS-137	2.417E+17		
RDCORINV043	BA-139	6.282E+18		
RDCORINV044	BA-140	6.216E+10		
RDCORINV045	4-140	6.352E+18		
RDCORINV046	LA-141	5.826E+18		
RDCORINV047	LA-142	5.616E+18		
RDCOR1NV048	CE-141	5.651E+18		
RDCORINV049	CE-143	5.494E+18		
RDCORINV050	CE-144	3.405E+18		
RDCOR1NV051	PR-143	5.395E+18		
RDCORINV052	ND-147	2.412E+18		
RDCOR1NV053	NP-239	6.464E+19		
RDCORINV054	PU-238	3.664E+15		
RDCORINV055	PU-239	8.263E+14		
RDCORINV056	PU-240	1.042E+15		
RDCORINV057	PU-241	1.755E+17		
RDCORINV058	AM-241	1.159E+14		
RDCORINV059	CH-242	4.436E+16		
RDCORINV060	CM-244	2.596E+15		
Variable Name	- SCLCRW	- SCLCRW		
Purpose	- Scaling factor to adjust the core inventory.			
Source	- NUREG-1150 MACCS 1.5.11 Sequoyah Model.			

RDCORSCA001 1.003

0,
.

.

ø

is

able Name	- PSDIST
Pur, ose	- Particle size distribution for each nuclide group.
Source	- NUREG-1150 MACCS 1.5.11 Surry Model.
	1.0
2	1.0
Sc 561 - 253	1.0
R!w	1.0
¥005	1.0
RD-12 06	1.0
RUPSDIS 107	1.0
DPSDIST008	1.0
RDPS01ST009	1.0
Variable Nam	ATNAM2
Purnase	- Descriptive text identifying the source term. This text
Fullyose	used to identify specific source terms in the output (cas
	the fic.)
Case 1, STG	1;
RDATNAM2001	'SEQUOYAH SOURCE TERM AS OBTAINED FROM'
RDATNAM2002	'SEQSOR, FULL APET DRY BASE CASE, DFV=1.0'
RDATNAM2003	'SOURCE TERM-001, GROUP-003'
Case 1, 516	Z:
RDATMAM2001	SEQUOYAH SOURCE TERM AS OBTAINED FROM
RUATNAM2002	SEQSOR, FULL APET DRY BASE CASE, DFV=1.0
RDATNAM2003	SOURCE TERM-DUZ, GROUP-DU4
Case 2, STG	1:
RDATNAM2001	'SEQUOYAH SOURCE TERM AS OBTAINED FROM'
RDATNAM- 302	'SEQSOR, FULL APET DRY CASE, DFV=5.0'
RDATNAM2000	SOURCE TERM-001, GROUP-001'

P

Ð

Case 2, STG 2: RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET DRY CASE, DFV=5.0' RDATNAM2003 'SOURCE TERM-002, GROUP-002'

Case 3, STG 1: RDATNAM2001 'SEQUOYAH SOURCE TERM AS ODTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET DRY CASE, DFV=10.0' RDATNAM2003 'SOURCE TERM-001, GROUP-001'

Case 3, STG 2:

RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET DRY CASE, DFV=10.0' RDATNAM2003 'SOURCE TERM-002, GROUP-002'

Case 4, STG 1: RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET DRY CASE, DFV=50.0' RDATNAM2003 'RELEASE ELEVATION = 10.0 m.' RDATNAM2004 'SOURCE TERM-001, GROUP-002'

Case 4, STG 2: RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET DRY CASE. DFV=50.0' RDATNAM2003 'RELEASE ELEVATION = 10.0 m.' RDATNAM2004 'SOURCE TERM-002, GROUP-003'

Case 5, STG 1: RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL A; IT DRY CASE, DFV=100.0' RDATNAM2003 'RELEASE ELEVATION = 10.0 m.' RDATNAM2004 'SOURCE TERM-001, GROUP-002'

Case 5, STG 2: RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET DRY CASE, DFV=100.0' RDATNAM2003 'RELEASE ELEVATION = 10.0 m.' RDATNAM2004 'SOURCE TERM-002, GROUP-003'

Case 6, STG 1: RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET WET BASE, DFV SAMPLED' RDATNAM2003 'SOURCE TERM-001, GROUP-006'

Case 6, STG 2: RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM'

RDATNAM2002 'SEQSOR, FULL APET WET BASE, DFV SAMPLED' RDATNAM2003 'SOURCE TERM-002, GROUP-007'

Case 6, STG 3: RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FRCM' RDATNAM2002 'SEQSOR, FULL APET WET BASE, DFV SAMPLED' RDATNAM2003 'SOURCE TERM-003, GROUP-008'

Case 7, STG 1: RDAINAM2001 'SEQUOYAH SOURCE FERM AS OBTAINED FROM' RDAINAM2002 'SEQSOR, FULL APET WET CASE, DFV=5.0' RDAINAM2003 'SOURCE TERM-001, GROUP-001'

Case 7, STG 2: RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RUATNAM2002 'SEQSOR, FULL APET WET CASE, DFV=10.0' RDATNAM2003 'SOURCE TERM-002, GROUP-002'

Case 8, STG 1: RDATNAM2001 'SEQUOVAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET WET CASE, DFV=10.0' RDATNAM2003 'SOURCE TERM-001, GROUP-001'

Case 8, STG 2:

RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET WET CASE, DFV=10.0' RDATNAM2003 'SOURCE TERM-002, GROUP-002'

Case 9, STG 1:

RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET WET CASE, DFV=50.0' RDATNAM2003 'RELEASE ELEVATION IS 0.0 m.' RDATNAM2004 'SOURCE TERM-001, GROUP-002'

Case 9, STG 2:

RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET WET CASE, DFV=50.0' RDATNAM2003 'RELEASE ELEVATION IS 0.0 m.' RDATNAM2004 'SOURCE TERM-002, GROUP-003'

Case 1J, STG 1:

RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET WET CASE, DFV=50.0' RDATNAM2003 'RELEASE ELEVATION IS 10.0 m.' RDATNAM2004 'SOURCE TERM-001, GROUP-002'

Case 10, STG 2:

RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET WET CASE, DFV=50.0' RDATNAM2003 'RELEASE ELEVATION IS 10.0 m.' RDATNAM2004 'SOURCE TERM-002, GROUP-003'

Case 11, STG 1:

RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET WET CASE, DFV=100.0' RDATNAM2003 'RELEASE ELEVATION IS BASE VALUE, 0.0 m.' RDATNAM2004 'SOURCE TERM-001, GROUP-002'

Case 11, STG 2:

RDATNAM2001 'SEQUOYAH SOURCE TERM ...S OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET WET CASE, C '=100.0' RDATNAM2003 'RELEASE ELEVATION IS BASE VALUE, 0.0 m.' RDATNAM2004 'SOURCE TERM-002, GROUP-003'

Case 12, STG 1:

RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' RDATNAM2002 'SEQSOR, FULL APET WET CASE, DFV=100.0' RDATNAM2003 'RELEASE ELEVATION IS 10.0 m.' RDATNAM2004 'COUPCE TERM-001, GROUP-002'

Case 12, STG 2:

RDATNAM2001 'SEQUOYAH SOURCE TERM AS OBTAINED FROM' PDATNAM2002 'SEQSOR, FULL APET WET CASE, DFV=100.0' RDATNAM2003 'RELEASE ELEVATION IS 10.0 m.' PDATNAM2004 'SOURCE TERM-002, GROUP-003'

Variable Name	- CALARM
Purpose	- Time after accident initiation when the accident reaches
	general emergency conditions (as defined in NUREG-0654), or
	when plant personnel can reliably predict that general
	emergency conditions will be attained
Source	- NUREG-1150 SEQSOR source term information for Sequoyah
	Model.

RUUALARMOUI 12	50.0
Variable Name	- MAXRIS
Pumphase	- Selection of risk dominant plume.
Sour e	- NUREG-1150 MACCS 1.5.11 Sequoyah Model (for small release
	fractions, on the order of 1.0E-02 or less, plume selection
	has little effect on consequence results.)
RDMAXRISO01 1	
Variable Name	- REFTIM
Purpose	- Specifies the representative time point of each plume
	segment (0.0 = leading edge, 0.5 = midpoint, 1.0 = trailing
	edge). The radioactive decay, dry deposition, and dispersion
	calculations are all performed as if the entire contents of
	the plume segment are located at this point.
Source	- A value of 0.0 is assumed for the first plume in this
	analysis. This results in a conservative calculation. For
	the second plume, a value of 0.5 is used in accordance with
	the NUREG-1150 MACCS 1.5.11 Surry Model.
RDREFTIMOO1 0.	00 0.50
Variable Name	- NUMRFL
Purnose	- Number of plume segments that are released.
Source	- MUREG-1150 SEOSOR source term information for Seouovah
500100	Model
	nogera
RDNUMREL001 2	
Variable Name	- PLHEAT
Purpose	- Heat content of plume release (w).
Source	- NUREG-1150 SEQSOR source term information for Sequoyah
	Model.

Dry v sequences: OPLHEAT001 3.70E+06 1.70E+05

Wet V sequences: RDPLHEATO01 1.85E+06 1.70E+05

Variable Name - PLHITE
Purpose - Height of plume segments at rele se (m).
Source - NUREG-1150 SEQSOR source term information for Sequovah Model for cases 1, 2, 3, 6, 7, 8, 9, and 11. MELCOR calculations for cases 4, 5, 10, and 12.

Cases 1, 2, 3, 5, 7, 8, 9, and 11: RDPLHITEOC: 0. 0.

Cases 4, 5, 10, and 12: RDPLHITE001 10.0 10.0

Variable Name - PLUDUR Purpose - Duration of plume segments (s). Source - NUREG-1150 SEQSOR source term information for Sequoyah Model.

RDPLUDUR001 1800. 21600.

Variable Name	- PDELAY	
Purpose	- Specifies the start time of each plume segment in seconds	P
	from the time of accident initiation, e.g., reactor scram	١.
Source	- NUREG-1150 SEQSOR source term information for Sequoyah	
	Model.	

RDPDELAY001 3650. 10250.

Variable Name	- RELFAC
Purpose	 Release fractions for isotope groups in release (Case Dependant).
Sources	- Cases 1 and 6 represent the base case analyses for the dry and wet V sequences, respectively, as obtained from the NUREG-1150 SEQSOR source term information for Sequoyal. The remaining cases are obtained by modifying the V sequence DFs of the NUREG-1150 SEQSOR source term code for Sequoyah. For all cases except one, the accident progression bins have been partitioned into two source term groups. The one exception is case 6 for which three source term groups were used.
* £ GROU	IPS: XE, R I CS TE SR RU LA CE BA
Case 1, Source * Dry V sequenc	Term Group 1: e, DF=1.0, Release Elevation=0.0
RDRELFRC001 9.8 7.88E-02 1.22E-	4E-01 4.56E-01 4.63E-01 1.16E-01 1.21E-01 2.19E-02 1.24E-02 01
RDRELFRC002 1.6 3.46E-03 5.52E-	2E-02 3.32E-02 4.31E-03 1.84E-01 6.56E-02 7.73E-04 4.71E-03 02
Case 1, Source	Term Group 2:
* Dry V sequenc	e, DF·1.0, Release Elevation=0.0
RDRELFRC001 9.9 4.72E-04 2.92E-	4E-01 1.29E-01 1.27E-01 1.25E-02 2.48E-03 6.69E-04 1.25E-04 03
RDRELFRC002 5.7 5.12E-03 2.78E-	3E-03 6.17E-02 2.10E-02 8.03E-02 3.56E-02 2.73E-04 4.72E-03 02
Case 2, Source	Term Group 1:
* Dry V sequence	e, DF=5.0, Release Elevation=0.0

RDRELFRC001 9.90E-01 2.14E-02 2.18E-02 2.32E-03 6.51E-04 1.54E-04 3.55E-05 1.41E-04 7.42E-04 RDRELFRC002 9.74E-03 2.25E-02 9.71E-04 1.02E-02 4.1CE-03 4.68E-05 4.11E-04 4.24E-04 3.19E-03

Case 2, Source Term Group 2: * Dry V sequence, DF=5.0, Release Elevation=0.0

RDRELFRC001 9.87E-01 7.87E-02 7.97E-02 1.91E-02 1.94E-02 3.52E-03 1.99E-03 1.26E-02 1.95E-02 RDRELFRC002 1.33E-02 7.06E-02 1.15E-02 3.96E-02 1.25E-02 1.37E-04 1.04E-03 8.74E-04 1.04E-01

Case 3, Source Term Group 1: * Dry V sequence, DF=10.0, Release Elevation=0.0

RDRELFRC001 9.91E-01 1.39E-02 1.43E-02 1.42E-03 2.65E-04 6.55E-05 1.39E-05 5.40E-05 3.11E-04 RDRELFRC002 8.73E-03 2.14E-02 1.14E-03 1.12E-02 6.01E-03 1.23E-04 8.06E-04 7.03E-04 5.21E-03

Case 3, Source Term Group 2: * Dry V sequence, DF=10.0, Release Elevation=C.0

RDRELFRC001 9.86E-01 4.09E-02 4.14E-02 1.02E-02 1.06E-02 1.91E-03 1.08E-03 6.85E-03 1.06E-02 RDRELFRC002 1.40E-02 7.84E-02 1.24E-02 2.33E-02 5.44E-03 4.27E-05 3.71E-04 3.11E-04 4.44E-03

Case 4, Source Term Group 1: * Dry V sequence, DF=50.0, Release Elevation=10.0

RDRELFRC001 0.85E-01 7.75E-03 7.89E-03 1.92E-03 1.97E-03 3.56E-04 2.02E-04 1.28E-03 1.98E-03 RDRELFRC002 1.50E-02 5.72E 02 3.46E-03 4.72E-03 1.14E-03 1.35E-05 8.39E-05 6.40E-05 9.52E-04

Case 4, Source Term Group 2: * Dry V sequence, DF=50.0, Release Elevation=10.0

RDRELFRC001 9.98E-01 3.04E-03 2.96E-03 2.86E-04 3.65E-05 1.28E-05 1.60E-06 5.35E-06 4.62E-05 RDRELFRC002 1.79E-03 1.16E-01 4.06E-02 3.01E-02 1.00E-03 6.15E-06 1.44E-04 1.58E-04 7.82E-04

Case 5, Source Term Group 1: * Dry V sequence, DF=100.0, Release Elevation=10.0

RDRELFRC001 9.85E-01 3 88E-03 3.94E-03 9.61E-04 9.86E-04 1.78E-04 1.01E-04 6.38E-04 9.91E-04 RDRELFRC002 1.50E-02 5.74E-02 3.43E-03 3.14E-03 5.68E-04 6.7*E-06 4.20E-05 3.20E-05 4.76E-04

Case 5, Source Term Group 2: * Dry V sequence, DF=100.0, Release Elevation=10.0

RDRELFRC001 9 ~9E-01 1.52E-03 1.48E-03 1.43E-04 1.82E-05 6.38E-06 7.96E-07 2.66E-06 2 30E-05 RDRELFRC002 1.79E-03 1.16E-01 4.06E-02 2.92E-02 5.02E-04 3.07E-06 7.18E-05 7.87E-05 3.92E-04

Case 6, Source Term Group 1: * Wet V sequence, Base case sampled DF, Release Elevation=0.0

RDRELFRC001 9.86E-01 1.81E-02 1. 4E-02 1.87E-03 2.06E-04 6.24E-05 9.88E-06 3.38E-05 2.73E-04 RDRELFRC002 1.40E-02 4.18E-02 3.26E-03 1.66E-02 9.26E-03 2.09E-04 1.34E-03 1.22E-03 8.13E-03

Case 6, Source Term Group 2: * Wet V sequence, Base case sampled DF, Release Elevation=0.0

RDRELFRC001 9.94E-01 1.82E-01 1.81E-01 5.77E-02 5.17E-02 9.37E-C3 5.32E-03 3.38E-02 5.20E-02 RDRELFRC002 5.91E-03 4.64E-02 1.66E-03 4.12E-02 1.34E-02 7.55E-04 5.70E-04 2.29E-03 1.21E-02

Case 6, Source Term Group 3: * Wet V sequence, Base case sampled DF, Release Elevation=0.0

RDRELFRC001 9.98E-01 5.06E-02 4.98E-02 4.39E-03 4.39E-04 1.65E-04 1.83E-05 5.98E-05 5.66E-04 RDRELFRC002 1.95E-03 1.03E-01 3.80E-02 5.18E-02 1.13E-02 6.89E-05 1.62E-03 1.77E-03 8.87E-03

Case 7, Source Term Group 1: * Wet V sequence, DF=5.0, Release Elevation=0.0

RDRELFRC001 9.90E-01 2.12E-02 2.14E-02 1.87E-03 3.64E-04 9.49E-05 2.09E-05 8.32E-05 4.31E-04 RDRELFRC002 9.56E-03 2.38E-02 1.05E-03 1.13E-02 6.23E-03 1.06E-04 7.64E-04 8.10E-04 5.00E-03

Case 7, Source Term Group 2: * Wet V sequence, DF=5.0, Release Elevation=0.0

RDRELFRC001 9.91E-01 7.19E-02 7.25E-02 1.53E-02 1.17E-02 2.15E-03 1.18E-03 7.44E-03 1.19E-02 RDRELFRC002 8.95E-03 6.48E-02 1.19E-02 3.73E-02 1.24E-02 2.56E-04 1.28E-03 1.42E-03 1.06E-02

Case 8, Source Term Group 1: * Wet V sequence, DF=10.0, Release Elevation=0.0

KDRELFRC001 9.91E-01 1.41E-02 1.43E-02 1.27E 03 1.52E-04 4.12E-05 8.11E-06 3.13E-05 1.88E-04 RDRELFRC002 8.61E-03 2.16E-02 1.08E-03 1.20E-02 7.20E-03 1.61E-04 1.02E-03 9.26E-04 6.27E-03

Case 8, Source Term Group 2: * Wet V sequence, DF=10.0, Release Clevation=0.0

RDRELFRC001 9.912-01 3.81E-02 3.84E-02 8.48E-03 6.71E-03 1.23E-03 6.78E-04 4.27E-03 6.80E-03 RDRELFRC002 9.30E-03 7.48E-02 1.35E-02 2.21E-02 4.79E-03 8.90E-05 3.82E-04 5.17E-04 3.97E-03

Case 9, Source Term Group 1: * W : V sequence, DF=50.0, Release Elevation=0.0

RDRELFRC001 9.89E-01 6.81E-03 6.90E-03 1.46E-03 1.13E-03 2.06E-04 1.14E-04 7.16E-04 1.14E-03 RDRELFRC002 1.07E-02 4.75E-02 2.56E-03 3.81E-03 1.09E-03 2.57E-05 1.04E-04 1.16E-04 9.41E-04

Case S, Source Term Group 2: * Wet V sequence, DF=50.0, Release Elevation=0.0

RDRELFRC001 9.98E-01 2.92E-03 2.85E-03 2.77E-04 4.07E-05 1.28E-05 1.79E-06 6.12E-06 5.08E-05 RDRELFRC002 1.70E-03 1.14E-01 4 09E-02 3.01E-02 1.20E-03 7.39E-06 1.73E-04 1.89E-04 9.37E-04

Case 10, Source Term Group 1: * Wet V sequence, DF=50.0, Release Elevation=10.0

RDRELFRC001 9.89E-01 6.81E-03 6.90E-03 1.46E-03 1.13E-03 2.06E-04 1.14E-04 7.16E-04 1.14E-03 RDREFFRC002 1.07E-02 4.75E-02 2.56E-03 3.81E-03 1.09E-03 2.57E-05 1.04E-04 1.16E-04 9.41E-04

Case 10, Source Term Group 2: * Wet V sequence, DF=50.0, Release Elevation=10.0

RDRELFRC001 9.98E-01 2.92E-02 2.85E-03 2.77F-04 4.07E-05 1.28E-05 1.79E-06 6.12E-06 5.08E-05 RDRELFRC002 1.70E-03 1.14E-01 4.09E-02 3.01E-02 1.20E-03 7.39E-06 1.73E-04 1.89E-04 9.37E-04

Case 11, Source Term Group 1: * Wet V sequence, DF=100.0, Release Elevation=0.0

RDRELFRC001 9.89E-01 3.41E-03 3.45E-03 7.28E-04 5.63E-04 1.03E-04 5.69E-05 3.58E-04 5.71E-04 RDRELFRC002 1.07E-02 4.77E-02 2.54E-03 2.43E-03 5.45E-04 1.29E-05 5.22E-05 5.79E-05 4.70E-04

Case 11, Source Term Group 2: * Wet V sequence, DF=100.0, Release Elevation=C.0

RDRELFRC001 9.98E-01 1.46E-03 1.43E-03 1.38F-04 2.03E-05 6.40E-03 8.93E-07 3.05E-06 2.1.7-05 RDRELFRC002 1. /E-03 1.14E-01 4.08E-02 2.92E-02 6.01E-07 3.69E-06 3.61E-05 9.44E-05 4.69E-04

Case 12, Source Term Group 1: * Wet V sequence, DF=100.0, Release Elevation=10.0

RURELFRC001 9.89E-01 3.41E-03 3.45E-03 7.28E-04 5.63E-04 1.03E-04 5.69E-05 3.58E-04 5.71E-04 RDRELFRC002 1.07E-02 4.77E-02 2.54E-03 2.43E-03 5.45E-04 1.29E 05 5.22E-05 5.79E-05 4.70E-04

Case 12, Source Term Group 2: * Wet V sequence, DF=100.0, Release Elevation=10.0

RDRELFRC001 9.98E-01 1.46E-03 1.43E-03 1.38E-04 2.03E-05 6.40E-06 8.93E-07 3.05E-06 2.54E-05 RDRELFRC002 1.70E-03 1.14E-01 4.08E-02 2.92E-02 6.01E-04 3.69E-06 8.61E-05 9.44E-05 4.69E-04

5.0 RESULTS

The consequence results for the dry and wet V sequences are summarized in this section. The complete output files from the 25 MACCS calculations are stored on magnetic tape and are available upon request. For the dry V sequence, five cases were analyzed, one base case and four sensitivity cases. For each case, the APBs were partitioned into two source term groups. For each source term group, a MACCS calculation was performed - a total of ten for the ory V sequence cases. Table G.8 reports three consequence measures for each source term group for each dry V sequence case. Table G.8 also includes the conditional frequency weighting for each source term group for each case as obtained from P⁺ TITION. The three consequence measures reported here are the mean number of early fatalities and latent cancer fatalities within a 1000-mile radius, and the mean population dose, in person-rem, within a 50mile radius. The uncertainty distributions calculated by MACCS for each measure are not reported here but can be obtained from the stored MACCS output.

Table G.9 reports similar results for the wet V sequence cases. For the wet V sequence, seven cases were analyzed, one base case and six sensitivity cases. Again, the reported consequence measures are mean values.

The mean consequences for each sensitivity case, conditional upon the occurrence of that particular V sequence, are obtained by multiplying the mean consequence measure for each source term group by the conditional probability of the source term group and summing over all source term groups:

$$CONS_{k} = \sum_{i=1}^{n} CSTGP_{i} * CONS_{ik} (FP_{i})$$

where,

CONSk		mean magnitude of consequence measure k
CSTGP,		the conditional probability of source term group i
FP,	72	fission product source term for cource term group i

 $CONS_{ik}$ = mean magnitude of consequence measure k for source term group i, given fission product source term (FP_i)

Table G.8 Conditional probabilities and MACCS consequence results for each source term group for each dry V sequence sensitivity.

<u>Sensi</u> DF	<u>tivity Case</u> Release <u>Elevation</u>	Source Term Group	Conditional <u>Probability</u>	Mean Early <u>Fatalities</u>	Mean Latent <u>Fatalities</u>	Mean 50-Mile Dose (person vem)
1.0	0.0	1	6.69E-01 3.311-01	1.49E+02 8.08E-01	6.79E+03 2.47E+03	8.05E+06 2.22E+06
5.0	0.0	1 2	1.61E-01 8.39E-01	1.33E-01 1.92E+00	6.06E+02 2.05E+03	9.33E+05 2.38E+06
10.0	0.0	1	2.29E-01 7.71E-01	1.11E-01 4.90E-01	5.04E+02 1.37E+03	8.59E+05 1.76E+06
50.0	10.0	1 2	8.29E-01 1.71E-01	7.38E-02 5.86E-02	4.34E+02 9.14E+02	8.20E+05 1.19E+06
100.0	10.0	1 2	8.28E-01 1.72E-01	6.20E-02 5.63E-02	2.99E+02 8.70E+02	6.41E+05 1.12E+06

<u>Sensi</u> DF	tivity Case Release Elcvation	Source Term	Cc.ditional <u>Probability</u>	Mean Early <u>Fatalities</u>	Mean Latent <u>Fatalities</u>	Mean 50-Mile Dose (person-rem)
Base Case ^a	0.0	1 2 3	4.87E-01 3.17E-01 1.96E-01	2.09E-01 1.11E+01 3.94E-01	6.33E+02 3.42E+03 1.6 E+03	1.04E+06 3.88E+06 1.77E+06
5.0	0.0	1 2	2.14E-01 7.86E-01	2.28E-01 1.20E+00	5.94E+02 1.73E+03	9.48E+05 2.02E+06
10.0	0.0	1 2	3.16E-01 6.84E-01	1.96E-01 4.96E-01	4.98E+02 1.21E+03	8.63E+05 1.61E+06
50.0	0.0	1 2	8.17E-01 1.83E-01	1.80E-01 1.59E-01	3.40E+02 9.01E+02	6.90E+05 1.17E+06
50.0	10.0	1	8.17E-01 1.83E-01	1.21E-01 1.06E-01	3.48E+02 9.13E+02	6.98E+05 1.18E+06
100.0	0.0	1	8.17E-01 1.83E-01	1.60E-01 1.54ē-01	2.37E+02 8.60E+02	5.37E+05 1.10E+06
100.0	10.0	1 2	8.17E-01 1.83E-01	1.07E-01 1.03E-01	2.41E+02 8.70E+02	5.42E+05 1.11E+06
-						

Table G.9 Conditional probabilities and MACCS consequence results for each source term group for each wet V sequence sensitivity.

. Wet V sequence base case DF is a sampled distribution as given in Table $\rm G,S$

Tables G.10 and G.11 present the mean consequence results for the dry and wet V sequence sensitivity cases, respectively. These results are also presented graphically on Figures G.1 through G.6. Comparisons of the dry and wet V sequence consequence results are also presented graphically in Figures G.7, G.8, and G.9.

<u>Sensi</u>	tivity Case			Mean 50-Mile
DF	Release	Mean Early	Mean Latent	Dose
	Elevation	<u>Fatalities</u>	<u>Fatalities</u>	(person-rem)
1.0	0.0	9.99E+01	5.36E+03	6.12E+06
5.0	0.0	1.63E+00	1.82E+03	2.15E+06
10.0	0.0	4.03E-01	1.17E+03	1.55E+06
50.0	10.0	7.12E-02	5.16E+02	8.83E+05
100.0	10.0	6.10E-02	3.97E+02	7.24E+05

Table G.10 Mean MACCS consequence results for each dry V sequence sensitivity.

Table G.11 Mean MACCS consequence results for each wet V sequence sensitivity.

<u>Sensit</u> DF	<u>tivity Case</u> Release <u>Elevation</u>	Mean Early <u>F</u> a <u>talities</u>	Mean Latent <u>Fatalities</u>	Mean 50-Mile Dose (person-rem)	
Base ^a 5.0 10.0 50.0 50.0 100.0 100.0	0.0 0.0 0.0 10.0 10.0 10.0	3.69E+00 9.92E-01 4.01E-01 1.76E-01 1.18E-01 1.59E-01 1.06E-01	1.71E+03 1.49E+03 9.85E+02 4.43E+02 4.51E+02 3.51E+02 3.56E+02	2.08E+06 1.79E+06 1.37E+06 7.78E+05 7.86E+05 6.40E+05 6.46E+05	

a. Wet V sequence base case DF is a sampled distribution as given in Table G.5.



Figure G.1 Mean early fatality consequence results as a function of decontamination factor for the dry V sequences.



Figure G.2 Mean latent fatality consequence results as a function of decontamination factor for the dry V sequences.



Figure G.3 Mean 50-mile population dose consequence results as a function of decontamination factor for the dry V sequences.















Figure G.7 Comparison of the dry and wet V sequences mean early fatality conseq ence results as a function of decontamination factor (release elevation = 10.0 m for DFs of 50.0 and 100.0, otherwise the elevation = 0.0 m).

,



Figure G.8 Comparison of the dry and wet V sequences mean latent istality consequence results as a function of decontamination factor (release elevation = 10.0 m for DFs of 50.0 and 100.0, otherwise elevation = 0.0 m).



Figure 6.9 Comparison of the dry and wet V sequences mean 50-mile population dose consequence results as a function of decontamination factor (release elevation = 10.0 m for DFs of 50.0 and 100.0, otherwise elevation = 0.0 m).

Appendix H

Component Failure Analysis

Pressure-Dependent Fregilities for Piping Components -Reference Combustion Engineering Plant

Prepared by D. A. Wesley*, D. K. Nakaki, H. Hadidi-Tamjed*

ADB Impell Corporation 27401 Los Altos, Suite 480 Mission Viejo, CA 92691

Under Subcontract to: Idaho National Engineering Laboratory EG&G Idaho, Inc. Idaho Falls, ID 83415

June, 1991

*EQE Inc. Costa Mesa, CA



27401 Los Altos, Suite 480 Mission Viejo, California 92691 Telephone (714) 582-9178 Kox (714) 582-1083

TABLE OF CONTENTS

Section			Title	Page			
	Title	Page					
	Report Approval Cover Sheet						
	Tabl	e o' 70	ontents	H-4			
	List	of Tabi	es	H-6			
	List	of Figu	ires	H-9			
	Rec	ord of F	Revisions	H-10			
1	INTE	RODUC	TION	H-11			
2	PIP	PIPE		H-15			
	2.1	Media	an Cylinder and Pipe Fallure Pressure Criteria	H-15			
	2.2	Cyline	der and Pipe Fallure Pressure Variability	H-18			
	2.3	Cylind	der and Pipe Failure Stress and Variability	H-19			
	2.4	Pipe (Capacities	H-20			
3	GAS	H-29					
	3.1	Introd	luction	H-29			
	3.2	Variat	bles Affecting Flanged Joint Leakage	H-29			
		3.2.1	Bolt/Stud Preload	H-30			
		3.2.2	Bolt/Stud Temperature	H-30			
		3.2.3	Bolt/Stud Yield Strength	H-31			
		3.2.4	Bolt/Stud Stress-Strain Relationship	H-31			
		3.2.5	Bolt Relaxation	H-32			
		3.2.6	Flange Flexibility	H-32			
		3.2.7	Initial Gasket Stress	H-33			

TABLE OF CONTENTS (Continued)

Section

4

n

2

.

Title

Page

	3.2.8	Gasket Loading Stiffness	H-33
	3.2.9	Gasket Unlo. (1)/Reloading Stiffness	H-34
	3.2.10	Gasket Creep and Relaxation	H-35
	3.2.11	Pipe Bending Moments	H-36
3.3	Flangs	Joint Behavior	H-36
3.4	3.4 Calculation of Leak Rate and Leak Area		H-37
3.5	5 Gasketed-Flange Connection Capacities and Variabilities		H-41
	3.5.1	150lb Flanges	H-41
	3.5.2	300lb Flanges	H-43
	3.5.3	400lb Flanges	H-45
VA	LVES		H-84
RE	FEREN	CES	H-86

1

No No

ä

LIST OF TABLES

٩.

Table	Description	Page
2-1	304 Stainless Steel Mate ul Properties	H-21
2-2	316 Stainless Steel Material Properties	H-21
2-3	Failure Stresses and Variability for 304 SS	H-22
2-4	304 Stainless Steel Pipe Failure Pressures	H-23
2-5	304 Stainless Steel Pipe Failure Pressures	H-25
2-6	304 Stainless Steel Pipe Failure Pressures	H-27
3-1	Stiffnesses for Asbestos-Filled Spiral Wound Gaskets	H-47
3-2	150# ANSI Flange and Gasket Data	H-48
3-3	150# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 20000 psi, JR = 0%)	H-49
3-4	150# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, JR = 0%)	H-50
3-5	150# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 30000 psi, $JR = 0\%$)	H-51
3-6	150# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, JR = 15%)	H-52
3-7	150# Flange Gasket Stress, Gross Leak Pressure, chid Leak Rate (IBS = 25000 psi, JR = 25%)	H-53
3-8	150# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, JR = 33%)	H-54
3-9	150# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, JR = 50%)	H-55
3-10	150# Flange Gasket Stress Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, $JR = 0\%$)	H-56
3-11	300# ANSI Flange and Gasket Data	H.57

LIST OF TABLES (Continued)

Table	Description	Page
3-12	300# Flange Gasket Stress, Gross Leak Prossure, and Leak Rate (IBS = 20000 psi, JR = 0%)	H-58
3-13	300# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, $JR = 0$)	H-59
3-14	300# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 30000 psi, JR = 0%)	H-60
3-15	300# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, JR = 15%)	H-61
3-16	300# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, JR = 25%)	H-62
3-17	300# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, $JR = 33\%$)	H-63
3-18	300# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, JR = 50%)	H-64
3-19	300# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, $JR = 0\%$)	H-65
3-20	300# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 50000 psi, JR = 0%)	H-66
3-21	300# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 60000 psi, JR = 0%)	H-67
3-22	300# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 60000 psi, JR = 25%)	H-68
3-23	300# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 60000 psi, JR = 0%)	H-69
3-24	400# ANSI Flange and Gasket Data	H-70
3-25	400# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 20000 psi, JR = 0%)	H-71
3-26	400# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, JR = 0%)	H-72

Ф

LIST OF TABLES (Continued)

Table	Description	Page
3-27	400# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 30000 psi, JR = 0%)	H-73
3-28	400# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, JR = 15%)	H-74
3-29	400# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, JR = 25%)	H-75
3-30	400# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, JR = 33%)	H-76
3-31	400# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, JR = 50%)	H-77
3-32	400# Flange Gasket Stress, Gross Leak Pressure, and Leak Rate (IBS = 25000 psi, JR = 0%)	H-78
4-1	Valves	H-85

۵

сЭ

-8

ð

LIST OF FIGURES

è

igure	Title	Page
3-1	Incremental Gasket Load to Total Pressure Load Ratio for 150 to 600 Lb. Rated Flanges	H-79
3-2	Typical Load-Deflection Diagram Showing A Standard Test Sequence	H-80
3-3	A Typical Mass-Leak-Rate Vs. Gasket Stress Plot	H-81
3-4	Gasket Stress Vs. Tightness Parameter for Two Spiral- Wound Asbestos Gaskets, Cyclic Tests with Water	H-82
3-5	Leak Rate Versus Pressure for 8"-150# Flange	H-83

.
RECORD OF REVISIONS

Revision No.

Description

0

1 21

Issue of final report

....

1. INTRODUCTION

The work presented in this report is described in Revision 4 of the Task Action Plan for Generic Issue 105, *Interfacing System LOCA in LWRs*, dated February 13, 1990. It is anticipated that the results documented here will facilitate the resolution of this high priority generic issue in a timely manner as well as the effort being undertaken by EPRI/NUMARC.

A probabilistic risk assessment of a reference Combustion Engineering plant is being conducted by EG&G Idaho, Inc., to evaluate the probability of plant damage from Interfacing System Loss of Coolant Accidents (ISLOCA). ABB Impell Corporation is under subcontract to EG&G idaho, Inc., to establish the pressure capacities of several low pressure systems to withstand k sures and temperatures above the design levels. The probability of failure as a function of internal pressure has been developed for the critical modes of failure of two fluid systems subject to potential overpressurization. The two segments identified by EG&G for the reference Combustion Engineering plant are the low pressure safet, injection line (LPSI) on the suction side of the safety injection pump back to the refueling water storage pool, and the LPSI discharge lines to the reactor coolant system cold legs. Included in this evaluation are the pumps, valves, and flanged connections for this system. No vessels or heat exchangers were identified for evaluation. The variability in the probability of failure is included, and the estimated leak rates or leak areas are given for the controlling modes of failure. For this evaluation, all failures are based on quasi-static pressures since the probability of dynamic effects resulting from such causes as water hammer have been initially judged to be negligible for the reference Combustion Engineering plant.

The pressure capacities of the pipes and vessels are evaluated using limit state analyses for the various failure modes considered. The capacities are dependent on several factors, including the material properties, modeling assumptions, and the postulated failure criteria. A major source of uncertainty in the failure criteria is the expected strain resulting in failure. All weids are full penetration and the probability of failure at membrane strains below yield is considered to be quite low. On the other hand, biaxial strains and gage length effects as well as strain concentrations and bending significantly reduces the expected hoop strain at failure when compared to elongation data developed from standard specimen ultimate tests. Since test data from vessel tests are extremely limited, considerable variability is introduced,

not only in the failure criteria but in analytical modeling and other assumptions. In particular the limited data that do exist are related to finite length cylinders with internal pressure loading only, and no test results are available for such effects as thermal or bending strains in pipe, strain concentrations at branch connections, or nozzle loads on tanks. Since many of the base parameters are random and the methods used to evaluate the capacities are subject to some uncertainty, the pressure capacity for any failure mode is also considered to bulk random variable.

A different approach must be used to evaluate the pressure capacities for gasketed flange connections, valves, and pumps. Unlike the failure modes for piping, vessels, and heat exchangers, which lend themselves to evaluation by general structural mechanics techniques, the failure modes for gasketed flange connections, valves, and pumps are very complex and the evaluation must rely primarily on the results from ongoing gasket research test programs and available vendor information and test data.

It is assumed that the pressure capacities have a lognormal distribution. This assumption is made because a lognormal distribution has been shown to be a valid description of the variability in material strengths. In addition, for a random variable that can be expressed as the product and quotient of several random variables, the distribution of the dependent variable tends to be lognormal regardless of the distributions of the independent base variables.

With the pressure capacity assumed to be a lognormal random variable and denoting it as *P*, the probability of failure occurring at a pressure less than or equal to a specific value *p* is expressed as:

 $P_{q} = \Pr(p \leq p) =$

$$\Phi\left[\frac{\ln\left(p/\tilde{P}\right)}{\beta_{c}}\right]$$

(1 - 1)

where:

= probability that failure occurs at a pressure $P \leq p$

random pressure capacity

logarithmic standard deviation of

P = median pressure capacity

In Equation 1-1, the pressure capacity for a given failure mode is probabilistically described by the following expression,

 $P = \hat{P} \cdot M \cdot S \tag{1-2}$

in which β is the median pressure capacity, M is a lognormally distributed random variable having a unit median and a logarithmic standard devia. In β_M representing the uncertainty in modeling, and S is also a lognormally distributed random variable with a unit median value and a logarithmic standard deviation β_S representing the uncertainty in the material properties. The overall uncertainty in the median capacity is obtained by taking the square root of the sum of the squares of β_M and β_S .

The median pressure capacity represents the internal pressure level for which there is a 50% probability of failure (leakage or burst) for a given failure mode. The median values are evaluated from limit state analyses for the different failure modes. The uncertainties, β_{M} and β_{S} , are associated with variability due to a lack of knowledge related to differences between the analytical model and the real structure. Modeling uncertainties are associated with the assumptions used to develop analytical models and their ability to properly represent the failure condition. The strength uncertainties are associated with variabilities related to the material resistance. Examples of the sources of strength uncertainties include: variability in steel yield and ultimate strengths, stress-strain relationships, and the influence of elevated temperatures on material strength.

Uncertainties will exist in the estimated pressure capacities due to differences between the analytical idealization of the structure and the real conditions. There are numerous possible sources of modeling uncertainties. Examples of the sources of modeling uncertainties include: assumptions used to develop the internal force distributions, failure criteria, and the use of empirical formulae. Moreover, since the uncertainties are dependent on the particular

failure mode under consideration, they must be evaluated on a case-by-case basis. However, in many instances, the evaluation of these uncertainties would require very detailed analysis and/or extensive data which may not be available. As a result, it was necessary to use subjective evaluation and engineering ,udgment to estimate these uncertainties.

The evaluation of the median capacities and the associated variabilities for the postulated failure modes for piping is discussed in Section 2 while the evaluation of median capacities and variabilities for flanged connections and valves are discussed in Sections 3 and 4, respectively.

2. PIPE

The piping of pertinence for ISLOCA consideration at the reference Combustion Engineering plant is fabricated from 304 Stainless Steel (i.e., SA 312, TP 304). The material properties for 304 Stainless Steel from Reference 1 were used for the pipe material. Ultimate strength values at room temperature up to 800°F (References 2 and 3) available in the literature were used to establish expected median strengths for the 304 SS. Table 2-1 shows the expected median ultimate strengths for room temperature, 400°F, 600°F, and 800°F, as well as the corresponding values for yield strength and elongation. Also shown for comparison are the corresponding ASME Code values. Table 2-2 shows similar expected median and code values for 316 SS (SA 312, TP 316, and TP 316 H for pipe).

As is apparent from these tables, a significant margin of safety often exists when the median material properties are compared with the corresponding code (lower bound) values. In a few instances, however, the median values available in the literature are close to, or even slightly lower than, the Code value; particularly for the 800°F temperature range. In all cases, however, since the systems and components being evaluated here were designed, for the most part, for relatively low pressures and temperature, ultimate pressure capacities for these components, governed by stresses near median ultimate values rather than the code design stress intensity, S_m, or allowable stress, S, values, are expected to be well in excess of the system design pressures.

2.1 Median Cylinder and Pipe Failure Pressure Criteria

Design stresses in piping system pressure vessels include provision for stresses resulting from deadweight, thermal expansion, nozzle loads, earthquake and other loads as well as internal pressure. Stresses from other than internal pressure may constitute a major or even the controlling portion of the design allowable stress. At overpressure conditions, however, the percentage of available strength required to resist the nonpressure loads may be expected to decrease (i.e., the deadweight stress in a piping system does not increase with an increase in pressure, and while thermal stresses may increase above the design case, they are not expected to be the controlling load). Thus, the failure criteria developed for pipe and pressure vessel burst is concentrated on the internal pressure effects, as reflected in the

(2-1)

hoop stress in a cylinder, while still retaining some consideration for other loads such as bending or branch connections in pipe, or nozzle loads in tanks. The goal was to develop criteria which could reasonably include these additional effects without requiring a detailed evaluation in order to obtain the actual magnitudes of the bending stresses, etc., at every location and temperature.

Failure due to hoop stress in a basically unflawed cylinder can be expected when the failure strain is reached. Several factors influence the failure strain when compared to the elongation predicted from a simple uniaxial 2-inch gage specimen, however. First, an unrestrained cylinder is in a state of biaxiai stress and it is known that failure strains are reduced for multiaxial stress states. Manjoine (Reference 4) suggests a triaxial reduction factor of the form:

$$T \cdot F \cdot = \frac{\sqrt{2}(\sigma_1 + \sigma_2 + \sigma_3)}{[(\sigma_1 - \sigma_2)^2 + (\sigma_1 - \sigma_3)^2 + (\sigma_2 - \sigma_3)^2]^{1/2}}$$

where σ_i (i = 1,2,C) are the principal stresses.

For an unrestrained cylinder with no bending subjected to internal pressure, the hoop stress is twice the axial stress and the maximum radial stress equal to the internal pressure is small, compared to the hoop and axial stresses. For this condition,

and T. F. = 1.73

Including provision for bending such that

 $\sigma_1 = \sigma_2$ and $\sigma_3 = 0$

results in T.F. ~2.

(2-2)

Thus, failure strain in a cylinder fabricated from a material with uniaxial elongation, ϵ , due to hoop stress could be expected in the range of:

$$\epsilon_{\tau,\tau} = \frac{\epsilon}{1.73}$$
 to $\frac{\epsilon}{2}$

Another factor which can influence the failure strain in a pipe or vessel is the effective gage length. Part of the elongation reported for a 2-inch uniaxial test specimen results from necking. For longer specimens, the necking purtion of the elongation remains essentially constant so that the total elongation for longer length specimen is less (i.e., the elongation for a uniaxial 8-inch test specimen is less than for a 2-inch specimen). For a segment of pipe, the effective gage length may be expected to be of the order of the pipe circumterence, and a further reduction on failure strain of the order of 1.5 to 2 is estimated. For pressure vessels with strain concentrations due to under- or over-reinforced nozzles and other discontinuities, failure is expected at somewhat lower average strains. Based on failure strain values which have been reduced to account for the triaxial and gage length effects and based on the typical stress-strain relationships for strain hardening carbon and austenitic steels, failure stresses can therefore be expected at about 0.9 and 0.85 of the ultimate uniaxial stress, respectively. Based on the limited available data, these ratios are assumed independent of temperature. Therefore, including the increased radius at failure can be found, from the simple relationship:

$$p_{i} = \frac{\hat{\sigma}_{i} t}{r(1 + \epsilon_{i})}$$

p.

ő,

t

10

where:

is the median failure stress

is the median failure pressure

- is the wall thickness (nominal)
- is the initial inside radius

and ϵ , is the median hoop strain at failure

This approach defines the failure pressure in terms of hoop stress which is an easy parameter to quantify but includes some provision for the biaxial stress-state and strain concentration effects.

2.2 Cylindor and Pipe Failure Pressure Variability

The variability associated with the calculated cylinder failure pressure results from a number of sources including material strength, wall thickness, variability in the stress-strain relationship, uncertainty related to the biaxial stress condition, necking and strain concentration effects on the failure strain and bending or discontinuity stresses. Most important, however, may be the existence, size, and orientation of partial through-wall flaws in the cylinder wall. To conduct a rigorous, probabilistic fracture mechanics evaluation of all the piping and vessels important for ISLOCA requires a knowledge of the crack depth, crack length, crack orientation, and fracture toughness including biaxial load and corrosion effects, both now and at end of plant life. In addition to a mean or median value of the above parameters, the statistical distribution and coefficient of variation would be required. Current evaluations are being conducted to accumulate and evaluate this type of data, but a rigorous evaluation based on this type of data requires consideration of the stress-state at every pipe weld and component over the entire range of temperatures which is beyond the scope of the current program.

Therefore, for the current investigation, the variability was developed for a lognormal distribution by assuming a probability of failure of 0.001 corresponding to yield in the cylinder. This approach, in essence, assumes the possibility that a large flaw may exist which would be required to fail the cylinder at yield but eliminates the need to determine how the flaw was developed. Since the controlling failure condition is based on hoop stress, the above assumption of failure at yield implies the controlling flaw is an axial crack although a larger circumferential crack could also lead to the same failure at yield. Whether the crack was present at fabrication or whether it was initiated and grew due to causes such as thermai fatigue, water hammer, or corrosion, is unknown and is not required with this assumption.

In essence then, the assumption of 0.001 probability of failure at yield implies a 0.001 probability of the existence of a very large flaw or combination of a large flaw with high thermal stress, etc. This is believed to be a very conservative (probably overconservative) assumption. If it is found that a significant contribution to the lotal risk occurs due to pipe or

vessel burst at pressures well below the median burst pressures, then this assumption should be reevaluated. This would require that burst, as opposed to flange leakage, is determined to be the dominant failure mode but could easily be evaluated by sensitivity studies where the variability is based on probability of failure at yield is assumed to be 10⁻⁴ or 10⁻⁵, for instance, or by truncating the tails of the distribution in the range of .02 to .04. If the overall risk is found to be sensitive to these assumptions, further research into the probability of expected flaw size may be necessary.

2.3 Cylinder and Pipe Failure Stress and Variability

Using the methods described in Sections 2.1 and 2.2 above, median hoop failure stresses and variabilities were developed for the materials and temperature range for both pipes and vessels. Due to the possibility of strain concentrations due to nozzles and other discontinuity stresses in tanks, somewhat lower median failure stresses, along with decreased variabilities, were used for tanks and vesses compared to pipe. Table 2-3 shows the failure stresses and corresponding lognormal standard deviations for the material used in this investigation.

In general, the failure stresses decrease with temperature, but due primarily to change in the ratio of yield-to-ultimate with temperature, the lognormal standard deviation tends to increase with temperature. Note, however, the tendency of the failure stress for the low carbon steels to increase in the 400° to 600°F range. This characteristic results from the corresponding increase in ultimate strength in same temperature range, although the yield strength of the same material shows essentially a monotonic decrease with temperature (c.f., Reference 5). The lognormal standard deviations, shown in Table 2-3, are considered to be representative (although probably conservative) for cylinders with median pressure determined essentially for unflawed vessels but admitting the possibility of the presence of very large flaws. As such, a lognormal distribution is considered reasonable for failure pressure of the order of plus one standard deviation and below. However, at failure pressures in the high end of the distribution, the use of the lognormal distribution is inappropriate and some upper bound cutoff is required. This occurs physically since the presence of flaws can significantly reduce the failure pressure below the median unflawed cylinder, but the absence of flaws cannot further strengthen the cylinder above the assumed median unflawed cylinder capacity. This upper bound cutoff is controlled essentially by the ultimate strength of the material without including the biaxial load and strain concentration reduction effects. In essence, the upper bound cutoff is expected to correspond more closely to the failure of a cylinder with no bending, nozzles, branch connections, or flaws, and is representative of the results obtained from finite length unflawed cylinder test results. The cutoff is also not a discrete value but has a median with associated uncertainty governed by the material strength properties.

2.4 Pipe Capacities

All pipe of concern for the reference Combustion Engineering plant ISLOCA investigation is fabricated from TP 304 stainless steel. Median burst pressures for the temperature range of interest are presented for various assumed levels of corrosion in Tables 2-4 through 2-6. The associated variabilities at these temperatures are given in Table 2-3. Note that the variabilities given in Table 2-3 are for failure stress but can also be used for failure pressure since the contribution of other elements to the total variability is expected to be negligible.

Temperature (°F)	Ultimate Strength (ksi)		Yield Strength (ksi)		Elongation (%) (2-inch gage)	
	Median	Code*	Median	Code	Median	Code
R.T.	86	75	37	30	60	35 (long.) 25 (trans.)
400	74	64.4	23	20.7	53	
600	70	63.5	19.5	18.2	49	
800	65.5	62.7	16.5	16.8	46	

Table 2-1. 304 Stainless Steel Material Properties

* SA 312, TP 304, and TP 304 H

Table 2-2. 316 Stainless Steel Material Properties

Temperature (°F)	Ultimate Strength (ksi)		Yield Strength (ksi)		Elongation (%) (2-inch gage)	
	Median	Code*	Median	Code	Median	Code
R.T.	86.5	75	42	30	46	35 (long.) 25 (trans.)
400	81.5	71.8	35	21.4	43	
600	77.5	71.8	31	18.8	41.5	
800	73	70.9	.28	17.6	40	

* SA 312, TP 316, and TP 316 H

	P	ipe	Vessel		
Tempi rature (F)	σ, (ksi)	β	o, (ksi)	β	
R.T.	74	0.22	67	0.19	
400	62.9	0.33	57.9	0.30	
600	59.5	0.36	54.6	0.33	
800	55.7	0.39	51.2	0.37	

Table 2-3. Failure Stresses and Variability for 304 SS

where \hat{o} , denotes the median hoop failure stress

Table 2-4. 304 Stainless Steel Pipe Failure Pressures

CORROSION ALLOWANCE = 0.000

Pipe			a sea office of the second second second	Construction of the other states of the other	And the party of the loss of the loss of the		
Size (in)	Schedule	OD (in)	ID (in)	70°F	400*F	600°F	800°F
1-1/2	405 80 160	1.900 1.900 1.900	1.610 1.500 1.337	9968 14757 23303	8556 12666 20001	7963 11788 18615	7525 11140 17591
2	405 80 160	2.375 2.375 2.375	2.067 1.939 1.889	8246 12443 22476	7078 10680 19292	6587 9940 17955	6225 9393 16967
3	105 405 60 160	3.500 3.500 3.500 3.500	3.260 3.068 2.900 2.624	4074 7792 11449 18474	3497 6688 9827 15857	3254 6225 9148 14758	3075 5882 8643 13946
4	105 405 80 160	4.500 4.500 4.500 4.500	4.260 4.026 3.826 3.438	3118 6515 9749 17094	2676 5592 8367 14672	2490 5205 7787 13655	2353 4918 7359 12904
6	105 405 80 120 160	6 625 6 525 6 625 6 625 6 625 6 625	6 357 6 065 5.761 5.501 5.189	2333 5110 8299 11307 15314	2002 4386 7123 9705 13145	1864 4082 6630 9032 12234	1761 3857 6265 8536 11561
8	105 20 405 80 120 140 160	8.625 8.725 8.625 8.625 8.625 8.625 8.625 8.625	8.329 8.125 7.981 7.625 7.189 7.001 6.813	1967 3405 4465 7258 11054 12837 14718	1688 2923 3833 6229 9488 11018 12633	1571 2720 3567 5798 8830 10254 11757	1485 2571 3371 5479 8344 9590 11110
10	105 20 405 80 120 140	10.750 10.750 10.750 10.750 10.750 10.750 10.750	10.420 10.250 9.564 9.064 8.750 8.500	1753 2699 4032 6862 10294 12649 14649	1504 2317 3460 5890 8835 10857 12573	1400 2156 3221 5482 8223 10104 11702	1323 2038 3043 5180 7770 9548 11058
12	10\$ 20 51d 40 80 120 140 160	12 750 12 750 12 750 12 750 12 750 12 750 12 750 12 750 12 750	12.390 12.250 11.938 11.376 10.750 10.500 10.125	1608 2259 3459 3764 6884 10296 11858 14340	1380 1939 2969 3221 5737 8837 10178 12308	1284 1804 2763 3007 5339 8224 9473 11455	1214 1705 2611 2841 5046 7772 8952 10825
14	105 20 510 40 120 140	14.000 14.000 14.000 14.000 14.000 14.000 14.000	13.624 13.375 13.250 13.125 12.500 11.814 11.500 11.188	1527 2556 3132 3689 6641 10240 1201 1201 13909	1311 2220 2689 3167 5700 8785 10326 11938	1220 2066 2502 2947 5305 8180 9610	1153 1952 2365 2785 5013 7730 9081 10500

Pipe				MEDIAN FAILURE PRESSURES				
Size (in)	Schedule	OD (in)	ID (in)	70°F	400*F	600°F	800*F	
16	105 20 51d 40 80 120 140 160	16.000 16.000 16.000 16.000 16.000 16.000 16.000 16.000	15.624 15.375 15.250 16.000 14.314 13.564 13.124 12.814	1332 2250 2722 3689 6518 9939 12127 13759	1143 1931 2336 3167 5595 8530 10409 11810	1064 1797 2174 2947 5207 7939 9687 10991	1005 1698 2054 2785 4920 7502 9154 10387	
18	105 20 51d 40 80 120 140 160	18.000 18.000 18.000 18.000 18.000 18.000 18.000 18.000	17.622 17.375 17.250 16.876 16.126 15.250 14.876 14.433	1181 1991 2406 3686 6431 9979 11621 13677	1013 1709 2065 3164 5520 8565 9975 11739	943 1590 1922 2944 5137 7972 9283 10925	891 1503 1816 2782 4853 7533 8773 1032	
50	105 205 40 120 140 160	20.000 20.000 20.000 20.000 20.000 20.000 20.000 20.000	19.564 19.250 18.814 17.938 17.000 16.500 16.5054	1233 2156 3488 6361 9786 11739 13559	1059 1851 2994 5480 8382 10075 11838	985 1722 2787 5082 7801 9337 10831	931 1628 2633 4800 7377 898 1023	
24	10S 20S 40 80 120 140	24.000 24.000 24.000 24.000 24.000 24.000	23.500 23.250 22.626 21.564 21.564 25.376 19.376	1177 1785 3361 6251 9842 11482	1011 1532 2884 5366 8448 9855	941 1426 2684 4994 7862 9172	889 134 253 471 743 866	

Table 2-4. 304 Stainless Steel Pipe Failure Pressures (Continued)

Calculation 304P000

i

C

Table 2-5. 304 Stainless Steel Pipe Failure Pressures

Participant and a second	Dies.				MEDIAN FAILLIRE PRESSLIRES				
Size (in)	Schedule	00 (in)	10 (in)	70°F	400°F	600°F	BOO*F		
Province of some set of the law and the set of the set	No. of Concession, Name of Street, or other		and set office of the set of the set of						
1.10	4/39	1.900	1.610	8593	7378	6864	6487		
	80	1,900	1.500	13281	11399	10609	10025		
	180	1,900	1.337	21647	18580	17292	16341		
Second second	100								
5	405	2.375	2.067	7175	6158	5732	5418		
1 C	80	2 375	1.930	11302	9700	9028	8532		
	160	2 375	1,689	21166	18167	16908	15978		
	1000								
3	105	3.500	3.260	3395	2914	2712	2563		
	405	3.500	3.068	7071	6069	5648	5338		
	80	3.500	2.900	10686	9172	85.36	8067		
	160	3.500	2.624	17631	15133	14084	13309		
4	105	4.500	4.260	2598	2230	2075	1961		
	40S	4.500	4.026	5966	5120	4765	4503		
	80	4.500	3.826	9170	7871	7325	6322		
	160	4.500	3.438	16450	14119	13141	12418		
						1000			
6	105	6.625	6.357	1985	1704	1585	1.45946		
	405	6.625	6.065	4745	4072	3790	3582		
	80	6.6	5.761	7915	6794	6323	5975		
	120	6,	5.501	10905	9360	8711	8232		
	160	6.1	5.189	14888	12778	11893	11239		
8	105	8.625	8.329	1701	1460	1359	1284		
	20	8.625	8.125	3133	2688	2503	2365		
	405	8.625	7.981	4188	3595	3345	3161		
	8C	8.625	7.525	6967	5980	5560	57.238		
	120	8.625	7.189	10746	9223	8584	8112		
	140	5.625	7.001	12521	10747	10002	9452		
	160	8.625	6.613	14383	123094	11490	10865		
	1		10.000	10.40	11000	1000	1163		
10	105	10,750	10.420	1540	1366	16.00	1075		
	20	10.750	10.250	Susa	2136	1204	10/2		
	405	10.750	10.020	3811	0671	3044	60//		
	80	10.750	9.564	0631	2091	0201	70.00		
	120	10.750	9.064	10049	6620	Dicield	1100		
	140	10.750	8.750	12396	10040	Statute	3 JU (
	160	10,750	8.500	14388	12349	11494	1 CIEND 1		
					1000	11.475	1070		
12	105	12.750	12,390	1429	16.00	1142	1075		
	20	12.750	12.250	2178	1784	1000	1,0639		
	Std	12.750	12.000	3274	2810	2010	6978		
	40	12.750	11,938	3578	3072	2000	4000		
	80	12.750	11,376	5489	5570	0.1.84	4080		
	120	12.750	10.750	10090	8660	8060	200		
	140	12.750	10,500	11648	9997	9304	0.0000		
	160	12,750	10.126	14122	12121	11281	TUDERU		
		And Marca	10.000	1200	1171	1000	1030		
14	1015	34.000	13,024	1,36,5	11/1	10000	1807		
	20	14.000	13.375	2420	CUT I	1000	2238		
	210	14.000	13.250	2360	2040	2003	2000		
	40	14.000	13.125	3521	JUKE	2012	4070		
	80	14.000	12.500	0464	5548	0163	70/00		
	120	14.000	11.814	10052	8525	8030	/ 000		
	140	14.000	11.500	11838	10160	9430	53.30 + 5555		
		1.0.002	1.1.2.848	19711	1.1.7888				

CORROSION ALLOWANCE = 0.020

Pipe				MEDIAN FAILURE PRESSURES			
Size (in)	Schedule	OD (in)	ID (in)	70*F	400°F	600°F	BOO"F
16	105 20 51d 40 80 120 140	16.000 16.000 16.000 16.000 16.000 16.000 16.000 16.000	15.824 15.375 15.250 14.314 13.564 13.124 12.814	1190 2106 2576 3542 6364 9775 11958 13586	1021 1807 2211 3040 5462 8390 10264 11661	951 1682 2058 2829 5083 7809 9553 10853	896 1589 1945 2674 4504 7379 9027 10256
18	105	18.200	17.624	1055	906	843	796
	20	18.000	17.375	1863	1599	1488	1407
	51d	18.000	17.250	2278	1955	1819	1719
	40	18.000	16.876	3555	3051	2839	2683
	80	18.000	16.126	6294	5402	5028	4751
	120	18.000	15.250	9834	8441	7856	7424
	140	18.000	14.876	11473	9647	9165	8860
	160	18.000	14.433	13523	11607	10803	10208
20	105	20.000	19.56 /	1120	961	895	846
	205	20.000	19.250	2041	1752	1630	1541
	40	20.000	18.814	3371	2893	2693	2545
	80	20.000	17.938	6238	5354	4983	4709
	120	20.000	17.000	9636	8270	7697	7274
	140	20.000	16.500	11604	9960	9270	8760
	50	20.000	16.064	13421	11520	10721	1013
24	105	24.000	23.500	1083	930	865	618
	205	24.000	23.250	1690	1450	1350	1275
	40	24.000	22.626	3263	2800	2606	2463
	80	24.000	21.564	6149	5278	4912	4642
	120	24.000	20.376	9734	8355	7776	7348
	140	24.000	19.876	11371	9760	9083	8584
	160	24.000	19.876	13312	11426	10634	1004

Table 2-5. 304 Stainless Steel Pipe Failure Pressures (Continued)

CORROSION ALLOWANCE = 0.020

Table 2-6. 304 Stainless Steel Pipe Failure Pressures

Pipe				MEDIAN FAILURE PRESSURES			
Size (in)	Schedule	OC (in)	iD (in)	70*F	400%F	BOC"F	800°F
1-1/2	405 80 160	1.900 1.900 1.900	1.610 1.500 1.337	7218 11506 19992	6195 10133 17159	5766 9431 15970	5449 8912 15091
2	40S 80 160	2 375 2 375 2 375 2 375	2.067 1.939 1.689	6104 10160 10855	5239 6721 17042	4876 8116 15861	4608 7670 14988
3	105 405 80 160	3.500 3.500 3.500 3.500	3,260 3,066 2,900 2,624	2716 6349 9923 16787	2331 5450 8517 14409	2170 5072 7927 13410	2050 4793 7491 ***672
4	105 405 80 160	4.500 4.500 4.500 4.500	4.260 4.026 3.826 3.438	2078 5416 8592 15807	1784 4648 7374 13567	1660 4326 6863 12627	1569 4088 6486 11932
6	105 405 80 120 160	6.625 6.625 6.625 6.625 6.625	6.357 6.065 5.7.61 5.301 5.189	1637 4380 7531 10502 14461	1405 3759 6464 (014 12412	1307 3499 6016 8390 11552	1235 3306 5685 7928 10917
8	105 20 405 80 120 140 160	8.625 8.625 8.625 8.625 8.625 8.625 8.625 8.625	8.329 6.125 7.981 7.625 7.189 7.001 6.813	1435 2961 3911 6677 10438 12204 14068	1232 2455 3357 5731 8959 10475 12075	1146 2285 3124 5334 8338 9749 11238	1083 2159 2952 5040 7880 9213 10520
10	105 20 405 80 120 140 160	10.750 10.750 10.750 10.750 10.750 10.750 10.750	10.420 10.250 9.564 9.064 8.750 8.500	1328 2268 3590 6400 5805 12143 14125	1140 1946 3081 5493 8416 10422 12126	1061 1811 2858 5112 7833 9700 11266	1002 1712 2710 4831 7402 9166 10665
12	105 20 51d 40 80 120 140 160	12,750 12,750 12,750 12,750 12,750 12,750 12,750 12,750	12.390 12.250 11.000 11.938 11.376 10.755 10.500 10.126	1251 1897 3090 3393 6295 9884 11437 13903	1073 1628 2652 2912 5403 8483 9816 11933	999 1516 2468 2711 5028 7895 9136 11106	944 1432 2332 2561 4752 7461 9633 10495
14	105 20 51d 40 120 140 160	14.000 14.000 14.000 14.000 14.000 14.000 14.000 14.000	13.624 13.375 13.250 13.125 12.500 11.814 11.500 11.188	1202 2255 2798 3352 6287 9865 11645 13513	1032 1935 2402 2877 5396 8467 8995 11599	960 1801 2235 2678 5022 7880 9302 10795	908 1702 2530 4746 7447 8791 10201

CORROSION ALLOWANCE = 0.040

Table 2-6. 304 Stainless Steel Pipe Failure Pressures (Continued)

Pipe Size (in)	Schedule	OD (in)	ID (in)	70"F	MEDIAN FA	TE PRESSURES	800"F
16	105 20 Std 40 80 120 140	16.000 16.000 16.000 16.000 16.000 16.000 16.000	15.624 15.375 15.250 15.000 14.314 13.564 13.124 12.814	1048 1962 2431 3394 6209 9612 11790 13414	900 1684 2087 2913 5329 8250 10119 11513	837 1567 1942 2711 4960 7678 9416 10715	791 1481 835 2562 4687 7256 8900 10126
18	105 20 516 40 50 120 140 160	18.000 18.000 18.000 18.000 18.000 18.000 18.000 18.000 18.000	17.624 17.375 17.250 16.876 16.126 15.250 14.876 14.433	929 1736 2149 3423 6156 9689 11324 13370	798 1490 1845 2938 5284 3316 9719 11475	742 1387 1717 2735 4918 7740 9046 10680	702 1310 1623 2584 4647 7314 8548 10093
20	105 205 40 80 120 140 150	20.000 20.000 20.000 20.000 20.000 20.000 20.000	19.564 19.250 18.814 17.938 17.000 16.500 16.500	1007 15/26 3253 6115 9505 11470 13284	864 1853 2792 5248 8158 9845 11401	804 1539 2599 4884 7593 9163 10611	760 1454 2456 4616 7175 8659 10026
24	105 205 40 80 120 140 160	24.000 24.5 24.50 24.000 24.000 24.000 24.000 24.000	23.500 23.250 22.626 21.564 20.376 19.876 19.314	989 1595 3165 6046 9625 11259 13197	849 1369 2716 5189 8261 9664 11327	790 1274 2528 4830 7689 8994 10542	747 1204 2389 4564 266 8499 9962

CORROSION ALLOWANCE = 0.040

3. GASKETED-FLANGE CONNECTIONS

3.5 Introduction

Although most of the piping joints in the safety injection system are full penetration butt welds, a number of gasketed-flange connections are required for the installation and maintenance of flow-restricting orifices, flow elements, and major equipment components. The elements of the flanged joints include standard ANSI B16.5 flanges with asbestos-filled, spiral-wound gaskets.

The lines designer for 150, 300, and 400lb rated service employ raised-face flanges fabricated from Type 304 stainless steel and are secured by means of SA-193 B8 bolts or studs and SA-564 Grade 630-HT 1100 studs. The minimum room temperature yield and ultimate strengths for the SA-193 B8 material are 30,000 and 75,000 psi, respectively. Similarly, for the SA-564 Grade 630 material, the minimum room temperature yield and ultimate strengths are 115,000 and 140,000 psi, respectively.

3.2 Variables Affecting Flanged Joint Leakage

The behavior of gasketed-flanges under pressure and temperature conditions is quite complex. The propensity for leakage, under a given pressure loading, is as much or more dependent upon the previous history of the joint than it is on its state at the time the pressure is applied. As a result, numerous variables are introduced. These include:

- Bolt/Stud Preload
- Bolt/Stud Temperature
- · Bolt/Stud Yield Strength
- Bolt/Stud Stress-Strain Relationship
- Bolt Relaxation
- Flange Flexibility
- Initial Gasket Stress
- Gasket Loading Stiffness

- Gasket Unloading/Reloading Sülfness
- Gasket Creep and Relaxation
- Pipe Bending Moments

3.2.1 Bolt/Stud Preload

Reference 6 provides bolt torquing procedures for flanged connections. The procedure indicates that the bolt torque values are to be obtained from vendor information or other approved site documents. However, for cases when the torque value cannot be obtained from vendor information or approved site documents, Reference 6 does provide recommended torque values for various bolt sizes and bolt materials. Using these data and noting that there is variability in the expected bolt torque values since vendors typically provide a recommended range of bolt torques, the SA-193 B8 bolts were considered to have a median initial prestress of 25,000 psi. In the parameter studies, a range of bolt preload stresses from 20,000 to 30,000 psi was considered and the range from 25,000 to 30,000 psi was taken to represent a 2.33ß variation. For the SA-564 Grade 630 bolts, the median bolt preload stress was taken as 60,000 psi and the range from 50,000 to 60,000 was taken as a 2.33ß variation. On the pipelines and components of interest, SA-193 B8 bolts are used with the 150lb flanges, both SA-193 B8 and SA-564 Grade 630 bolts are used with 300lb flanges, while the 400lb flanges are installed with the SA-193 B8 bolts.

It should be remembared that the uncertainty variability is related to the parameter of interest such as leak rate or leak area. Thus, the variability is calculated from the variation in say, leak rate resulting from the variations in initial bolt stress noted above.

3.2.2 Bolt/Stud Temperature

The maximum operating temperatures for the low pressure portions of the safety systems under consideration range from 120 to 350°F. As a result, the bolt temperature is likely to be less than 200°F during normal operation. However, during the ISLOCA event, the reactor coolant system pressure and temperature conditions of 2250 psi and 650°F can propagate back through the initially cold, non-operating systems. Based on preliminary analyses of the system, it is our understanding that the pressure propagates much more rapidly than the fluid temperature. Specifically, a relatively large leak at the flange connection or

MV-47.2.1 2011-FL98.4 Pastyl ()

downstream (reverse flow condition) of the flange connection must occur in order for the flange or flange bolting temperatures to rise substantially. In addition, for higher temperature conditions in the pipe, flow must continue for a relatively long period of time before the bolting temperature will rise substantially. Thus, it was judged that potential flange leakage will most likely occur under high-pressure, low-temperature conditions and that leak rates and leak areas will increase somewhat as the flange and bolt temperatures increase. Based on these considerations, a bolt temperature of approximately 140°F was taken to represent the median case and the effect of higher bolt temperatures was not considered. Evaluation of higher bolt temperatures, should they occur, can be modeled employing lower elastic modulus, yield strength, and ultimate strength values.

3.2.3 Bolt/Stud Yield Strength

The SA-193 Grade B8 bolt material is a Type 304 stainless steel. The material properties of Type 304 stainless steel are given in Table 2-1 of this report. Consistent with the selected 140°F median value for the fiange bolt/stud temperature, a value of 33,000 psi was taken as the median bolt yield strength. The ASME code minimum value of 27,500 psi was taken to represent a -2.338 variation.

The SA-564 Grade 630 bolt/stud material is a precipitation-hardened, high alloy steel. At the 140°F median temperature, the median bolt/stud yield strength was taken to be 122,000 psi. The ASME code minimum yield strength of 111,000 psi was taken as a -2.33ß variation.

3.2.4 Bolt/Stud Stress-Strain Relationship

References 2 and 3 provide limited data on the stress-strain curve to failure for Type 304 stainless steel at room and elevated temperatures. These data were interpolated to estimate the strain at failure for a bolt temperature of 140°F and were scaled to median material properties. The resulting curve was approximated in a piecewise linear fashion. A similar approach was used for the SA-564 Grade 630bolt/stud material to arrive at a piecewise linear stress-strain relation.

3.2.5 Bolt Relaxation

For stainless steel bolts which are initially torqued to prestress levels exceeding the material yield strength, some relaxation will occur. This relaxation requires substantial time, particularly at the relatively low bolt temperatures experienced during normal plant operation. Relaxation during the course of the ISLOCA event is judged to be negligible. Since the bolts used at the reference Combustion Engineering plant are not initially torqued to prestress ievels exceeding the material yield strength, bolt relaxation was not included in this study.

3.2.6 Flange Flexibility

The flanged joint consists of the flange, the flange bolting, and the gasket. To study specific variables most test programs isolate one or more of the joint elements, such as the flexibility of the flange. However, to properly characterize the overall joint behavior, all three elements must function as part of an integral unit. In order to evaluate the joint behavior, three axisymmetric elastic finite element joint models (Reference 8) were developed. The models were for a 4"-300lb flange, a 12"-300lb flange, and a 4"-150lb flange and included the flange and pipe structure, bolt stiffness, and gasket unloading stiffness. Material properties were taken at 200°F. As expected, it was found that fiance flexibility affects the way that the pressure load is carried by the gasket and bolting. As the relative flange stiffness increases, a greater portion of the pressure load goes to loading the bolting and a lesser portion goes to unloading the gasket. Thus, for larger flanges or more flexible flanges (lower pressure rating), the portion of the pressure load going to the unloading of the gasket increases. This can be seen in Figure 3-1 which plots the results of the three analyses. In this figure, the ratio of the load removed from the gasket to the total pressure load is plotted vorsus nominal flange size. For the elastic case of the 4"-300lb flange, about 43% of the total pressure load goes to unloading the gasket while the remaining 57% is carried by increased bolt load. The ratio for the larger 12"-300lb flange is approximately 86% and the ratio for the lighter weight 4"-150lb flange is about 59%. It is likely that the variation of the ratio with flange size is more complex. than the linear variation shown, but project time constraints limited the scope of this supporting study. Due to the lack of specific data, the maximum value of the incremental gasket load to total pressure load ratio was set at 1.0. However, for flanges which are flexible relative to the gasket, experience has shown that the ratio can be greater than 1.0 such that an increase in precsure results in a decrease in flange bolt tension.

Both SA-193 B8 and SA-564 Grade 630 flange bolts are used for the gasketedflange connections at the reference Combustion Engineering plant. Since the SA-193 B8 material is a relatively low strength Type 304 stainless steel, leak areas at higher pressure are primarily governed by the inelastic deformation of the bolts with the flange remaining essentially elastic. On the other hand, the SA-564 Grade 630 material has a high yield strength and the flanges are stressed to higher levels. At high pressures, some inelastic behavior in the flange could be expected, in which case me inelastic deformation of the flange at the gasket could be a significant contributor to the leak areas. However, further detailed studies of flange flexibility with high strength bolts was not pursued because the flanges joined by the high strength bolts were found to have high gross leak pressure capacities.

3.2.7 Initial Gasket Stress

Over the past several years, the Pressure Vessel Research Committee (PVRC) of the Welding Research Council has sponsored a major ongoing gasket test program as part of its Long Range Flanged Joint Improvement Program. Most of the tests have been conducted with nitrogen or helium as the test fluid; however, a limited data set exists for tests using water. Some of the results of the test program are reported in References 9 through 12. These results clearly indicate that the leak resistance of a gasketed-flange joint is a function of the initial level to which the gasket is stressed during the preloading of the flange bolting. The higher the initial gasket stress, the greater the leak resistance. The gasket stress versus deflection curve (Figure 3-2) and the mass leak rate versus gasket stress curve (Figure 3-3) for a typical spiral-wound gasket are both characterized by the presence of a "knee" at a gasket stress of approximately 5,000 psi. Above 5,000 psi, the leak rate drops more rapidly with increasing stress indicating improved sealing performance. Thus, although the controlled variable in assembling a flanged connection is bolt preload, it is the resulting gasket stress which determines the leak resistance of the joint for the pressure loading.

3.2.8 Gasket Loading Stiffness

The loading stiffness parameter is of importance since it determined whether or not the flange raised face bottoms out on the 0.125" thick compression gauge ring due to the bolt preload. Spiral-wound gaskets used for virtually all applications are fabricated to the requirements of Military Specification MIL-G-21032E, including Amendment 2 (Reference 13).

This standard specifies the test load and corresponding deflection for each gasket size and service rating. Caskets with an initial nominal thickness of 0.175" are to be compressed to a thickness of 0.130 ± 0.005" under the specified test load while gaskets with an initial nominal thickness of 0.125" are to be compressed to a thickness of 0.100 ± 0.005". Figure 3-2 shows the gasket stress versus deflection for the loading sequence of a typical spiral-wound gasket and Table 3-1 presents the range of gasket stiffness (expressed as gasket stress per inch of deflection) which meet the specification. The Reference 13 specified test load is shown together with the resulting bolt stress (based on bolt stress area). Test loads greater than 557000 lbs are not specified since this load is judged to be the practical limit of testing facilities. However, since the specified test load corresponds to the load resulting from prestressing the flange bolts to approximately 30,000 psi (based on bolt thread root area), the stiffness for larger size gaskets can also be computed. 300 and 600lb rated gaskets are interchangeable for sizes 3" and smaller and therefore, the stiffnesses are identical. The stiffnesses associated with the nominal compressed thickness (Tg=0.130) were taken to be median centered and the range from a compressed thickness of 0.125 to 0.135" was considered to represent a ±2,33B variation. In this study, for the sake of economy in the computing effort, the gasket stiffnesses for 300% rated flanges were conservatively used for the 400lb rated gaskets, as well.

Based upon the definition of the gasket test load noted above, prestressing the flange bolts to less than 30,000 psi will not result in a lock-up between the flange and the compression gauge ring. However, prestressing the bolts to 30,000 psi or greater may or may not result in lock-up depending on the gasket loading stiffness.

3.2.9 Gasket Unloading/Reloading Stiffness

The unloading stiffness characterizes the recovery of the gasket as the gasket stress is reduced due to increase in pressure, bolt relaxation, or other means. Figure 3-2 depicts the loading and unloading/reloading behavior of a typical spiral-wound gasket. Review of the available test data indicates that the recovery behavior for all spiral wound gaskets is quite similar and can reasonably be expressed as an unloading/reloading stiffness of about 1,000,000 psi/inch. This appears to be a reasonable value for new gaskets such as those used in the test program. However, hardening due to gasket aging may substantially increase

the gasket unloading/reloading stiffness, resulting in a decrease of the Gross Leak Pressure and an increase in the joint leak rate. It is important to note that reloading of the gasket due to removal of the pressure load, for example, follows the unloading stiffness curve.

3.2.10 Gasket Creep and Relaxation

It is understood from experience that gaskets behave nonlinearly and that they creep, even at room temperature. Until recently, information available on the creep and relaxation of commonly-used fabricated gaskets was scarce. Such information is vital for the proper understanding of the behavior of bolled flange joints. The previously mentioned PVRC-sponsored test program provided a vehicle for gathering creep and relaxation data for spiral-wound gaskets which is reported in Reference 12. The results indicate that maximum creep occurs at the lower stress levels and is particularly extensive at about 5,000 psi which coincides with the yield plateau in the stress-deflection diagram. For constant stress, most of the creep occurs in the first 10 to 15 minutes, while for cyclic stress, 20 to 25 stress cycles are required. Cyclic creep exhibits nearly the same overall behavior as constant stress level but is reasonably constant in terms of percent relaxation. Most of the gasket relaxation also occurs in the first 10 to 15 minutes after initial prestressing of the bolts. It is also of interest to note that lock-up of the flange and compression gauge ring significantly limits the gasket relaxation.

In a real life gasketed-bolted-flange connection, the gasket is subjected to neither pure creep or pure relaxation, even under steady-state operating conditions, since gasket creep causes the bolt load and deflection to change when there is no lock-up between the flange and the compression gauge ring. It is expected that, in many cases, the flange bolting was initially tightened and then retightened some minutes or hours later or possibly during or after the preservice hyractic pressure test. Retightening eliminates much of the effect of initial short term creep and relaxation. Thus to account for relaxation and cyclic creep for those cases where lock-up between the flange and compression gauge ring does not occur (150lb flanges with low initial bolt stress and 300 and 400lb flanges with low bolt stress and high gasket loading stiffness), a joint relaxation of 25% was taken to be median-centered while the range from 0 to 25% was taken to represent a + 3.0ß variation.

3.2.11 Pipe Bending Moments

Bending moments in the piping at the flange connection due to deadweight or thermal loads are carried by tension in the bolting. Based on normal practice, the piping supports are placed such that deadweight pipe stresses are relatively low and thus, the additional bolt stress is small. If the flange bolting is elastic, it was felt that the reduced gasket stress on the one side is balanced by an increased gasket stress on the other such that the joint mass leak rate will be about the same whether or not the bending moment is considered. On the other hand, if the flange bolting is inelastic, the bending moment could result in some increase in the calculated leak area but the increase would be limited by redistribution of the bending moment. Thus, pipe bending stress was not considered specifically in the evaluation of the flange joint. It is judged that the variabilities from other sources and conservatisms introduced into the approach cover the potential effect of pipe bending moments.

3.3 Flange Joint Behavior

The behavior of gasketed-flange connections, due to increasing pressure of the ISLOCA event, is characterized in the manner described in the following steps.

- The flange bolts are torqued to prestress levels satisfying the requirements of Reference 6 resulting in an initial gasket stress.
- 2 Over the course of normal operation, gaskets sustain cyclic creep and relaxation. If Step 1 produced lock-up between the flange and compression gauge ring, the relaxation reduces the gasket stress with a corresponding increase in the lock-up stress and negligible change in the bolt stress. If Step 1 did not produce lock-up, the creep and relaxation reduces the gasket stress with a corresponding reduction in the bolt stress.
- At the initiation of the ISLOCA event, the increasing pressure must first overcome the lock-up load, if any, with no reduction of the gasket stress or increase in bolt stress.

(3-1)

- 4 Further increase in pressure to the Gross Leak Pressure is shared by the gasket and bolts in accordance with Figure 3-1 resulting in a decrease in the gasket stress and an increase in the bolt stress. Gross Leak Pressure is defined as the point at which the gasket stress and the pressure are equal. However, for cases in which joint lock-up does not occur, Gross Leak Pressure was defined as either the point at which the gasket stress and the pressure are equal. are equal or the point at which the flange bolts yield, whichever occurs first.
- Further increase in pressure above the Gross Leak Pressure results in a corresponding increase in the bolt stress accompanied by increases in bolt length up to the bolt failure strain in accordance with the bolt stress-strain diagram.

These steps were used consistently in the evaluation of the myriad of cases covering the sizes, pressure ratings, and ranges of the variables affecting leakage.

3.4 Calculation of Leak Rate and Leak Area

 $T_{k} = \frac{p}{p^{2}} \left[\frac{L_{kM}}{L_{kM}} \right]^{\alpha}$

The definition of the onset of gross leakage, or the Gross Leak Pressure, as the point at which the gasket stress is equal to the pressure being retained, is used quite generally throughout the gasket industry. This definition has come about, it appears, from gasket tests where some "O"-ring and flat face gaskets have suffered blowout. Although it is doubtful that spiral-wound gaskets are on the verge of catastrophic failure when the gasket stress is reduced to the point that it equals the pressure, the potential certainly exists. For pressures less than the Gross Leak Pressure, the mass leak rate is calculated from the results of the gasket leakage test with water reported in Reference 11. Leakage of this form is related to the pressure of seams and crevasses in the flange/seal joint rather than any apparent leak area. In this test, 4"-600lb rated gaskets were subjected to both standard and cyclic load pressure sequences. The results are presented in Figure 3-4 which is a plot of Gasket Stress versus the Tightness Parameter (T_{e}). T_{e} is defined as:

where p = Internal Fluid Gauge Pressure (psig)

p* = Reference Atmospheric Pressure (14.7 ps:a)

 $L_{RM}^* = \text{Reference Mass Leak Rate (1 mg/sec)}$

L_{RM} = Total Mass Leak Rate through the Gasket (mg/sec)

and a = Tightness Parameter Exponent (1.0 for water)

Thus, the total mass leak rate for the water case is computed as:

$$L_{RM} = \frac{P}{(14.7 \cdot T_{P})}$$

Since the leak rate data correspond to the total mass leakage from the 4"-600lb rated gasket and not, for example, the leakage per unit mean circumference, a correction must be made to the calculated mass leak rate to account for the various gasket sizes. Since the probability of leakage increases with gasket perimeter, it is reasonable to assume that leakage through a larger diameter gasket will increase in proportion to the gasket diameter. In addition, a correction factor must be introduced to account for variations in the gasket width. It should be noted that the calculation of gasket width and gasket area should not include the outer 1/8" which is ineffective in the sealing process. The leak rate at the Gross Leak Pressure is then determined as:

$$LR_{GLP} = \frac{(D_o + D_1)(p)}{348.6W_o} \left[\frac{1}{e^{(6.1872 - 6.2361(9.2103 - \ln 50_o) - 7913(1(\ln 50_o - \ln 50))}} \right]$$
(3-3)

where $D_{s} =$ Gasket Outside Diameter (in)

 $D_1 =$ Gasket Inside Diameter (in)

(3-2)

W c = Gasket Width (in)

SG. = Initial Gasket Stress (psi) = Actual Gasket Stress/(1 - J R/100)

SG = Current Gasket Stress (psi)

and JR = Joint Relaxation expressed in percent of SG.

Note that in Equation 3-3 the quantity, 348.6, is the product of 14.7 from Equation 3-2 and 23.714 which is the value of $(D_e + D_i)/W_e$ for a 4"-600lb rated gasket. The quantity in the denominator of the term within the brackets in Equation 3-3 represents the Tightness Parameter (T_D) which is obtained by curve-fitting using the curves shown in Figure 3-4.

It is difficult to determine the significance of a leak rate in milligrams per second. However, if a drop of water is idealized as a 1/8 inch diameter solid sphere, a leak rate of 1 mg/sec would correspond to 3.5 drops per minute or about one drop every 17 seconds. By the time the leak rate increased to 17 mg/sec, the joint leakage would be about 1 drop per second which could be of concern for nuclear operation depending on the location of nearby electrical equipment. Leak rates of 200 to 500 mg/sec would constitute a spray of water which could possibly inhibit some operator actions in the vicinity.

For pressures above the Gross Leak Pressure, it was judged that the leakage is no longer due to seams and crevasses in the flange/seal joint but due to actual separation of the flange and gasket. Thus, a leak area is calculated which is intended to be in addition to the leak rate calculated at Gross Leak Pressure. The leak area is calculated as the mean gasket perimeter times the separation distance at the gasket. The separation distance is affected by bolt extension, gasket recovery, and flange flexibility. Of these, the contribution of bolt extension is by far the most dominant one. Therefore, the separation distance calculated in this study includes the effect of bolt extension only. Note that excluding the effect of gasket recovery from the leak area calculation is conservative and leads to slightly higher leak area values. The leak area at pressures above the Gross Leak Pressure is equated as shown in Equations 3-4 and 3-5, respectively, for the r ase where the bolt stress is less than or equal to the bolt material yield stress and for the cast, where the bolt stress exceeds the material yield. The term, $f(p_{oL}, JR, SG_{o}, SG_{poL}, K_{o})$, represents the remaining recoverable gasket deflection beyond GLP. It should be noted that some gasket recovery occurs prior to GLP. Due to the difficulties involved in arriving at a reasonably accurate estimate of the term, $f(p_{oL}, JR, SG_{o}, SG_{poL}, K_{o})$, and recognizing the fact that the effect of bolt extension on the leak area far exceeds that of the gasket recovery, this term was conservatively neglected.

For bolt stress < yield

$$A_{i} = \frac{\pi(D_{s} + D_{i})}{2} \left[L_{s} \left\{ \frac{(p - p_{ci})(A_{s})}{N_{s}(A_{s})(E_{s})} \right\} - f(p_{ci}, JR, SG_{s}, SG_{pci}, K_{c}) \right]$$
(3-4)

For bolt stress > yield

$$s_{1} = \frac{\pi(D_{x} + D_{y})}{2} \left[L_{x} \left\{ \frac{(S_{xy} + S_{xy})}{E_{x}} + \left[S_{xy} + \frac{(p + p_{yy})(A_{y})}{N_{y}(A_{y})} + S_{xy} \right] / E_{xy} \right\} + f(p_{yy}, JK, SC_{y}, SC_{yyy}, K_{y}) \right]$$
(3-5)

where $L_{s} = \text{Bolt/Stud Length}(in)$

per = Gross Leak Pressure (psi)

 A_p = Pressure Area (in²) - based on gasket inside diameter

 $N_B =$ Number of Fiange Bolts

 $A_s = Bolt Tensile Stress Area (in²) - per bolt$

E = Bolt Material Elastic Modulus (psi)

SG PEI = Gasket Stress at Gross Leak Pressure (psi)

I (poi, JR. SG , SG , ci, Kc) = Recoverable Gasket Deflection (in)

K = Gasket Unloading/Reloading Stiffness (psi/in)

S ... = Bolt Material Yield Stress (psi)

 S_{ss} = Actual Bolt Stress (psi) = (1 - JR/100) S_{ss} for no lockup case = S_{ss} for lockup case

S ... = Initial Bolt Stress (psi)

and E', = Bolt Modulus for appropriate inelastic portion of the stress-strain diagram (psi)

3.5 Gasketed-Flange Connection Capacities and Variabilities

Based on the gasket loading stiffness of Table 3-1 (for median case, Tg=0.130) and gasket load to total load ratios of Figure 3-1, the gasketed- flange connection capacities and variabilities of 150, 300, and 400lb rated flanges were evaluated and are discussed in the following sections. As mentioned in section 3.2.8, for the sake of savings in the computation effort, the gasket stiffnesses for the 300lb flanges were conservatively used for the 400lb rated flanges as well.

3.5.1 150lb Flanges

Tables 3-2 through 3-10 present the results of the analyses for the 150lb rated flanges. Table 3-2 shows the dimensional data for the gaskets and flange bolting which are standard for 150lb ANSI flanges. The numerical results for the Gross Leak Pressures, leak rates (pressure \leq Gross Leak Pressure), and leak areas (pressure \geq Gross Leak Pressure) are shown in Tables 3-3 through 3-10 for the various cases studied. For the particular components of interest with 150lb rated flanges, the flanges are joined by SA-193 Grade B8 bolts. In the tables of results, the parameter variation cases studied include variation in the initial bolt stress (Tables 3-3 through 3-5), variation in the amount of joint relaxation (Tables 3-6 through 3-9), and variation in the bolt yield strength (Table 3-10). In reviewing these tat \leq s, recall that the median case corresponds to an initial bolt stress of 25,000 psi and joint relaxation of 25%. with a median bolt yield strength of 33,000 psi.

In each table, the effective gasket stress, the actual gasket stress, and the gasket deflection are shown. The effective gasket stress is defined here as the gasket stress due to the bolt prestress as unaffected by the gasket compression gauge ring, while the actual gasket stress represents the stress limited by the compression gauge ring. That is, in cases where the bolt preload is great enough to cause the flange to bear upon the compression gauge ring (i.e., the condition of gasket bottom-out or joint lock-up), the gasket stress is limited by the maximum deflection of the gasket. The gasket deflections are computed using the gasket stiffnesses shown in Table 3-1.

In Tables 3-3 through 3-10, the effective and actual gasket stresses are equal for most of the flanges, which indicates that the bolt initial stress is not large enough to bottom-out the flange on the compression gauge ring and the resulting gasket deflection is less than the maximum deflection of the nominal 0.175" gasket. Only for the cases with an initial bolt stress of 30,000 psi (Table 3-5) do some of the flanges bottom-out on the gasket. As previously described, the Gross Leak Pressure was calculated as the point at which the gasket stress equals the internal pressure. The exception is for cases in which the bolt preload is not sufficient to bottom-out the gasket, where Gross Leak Pressure was taken as the k- ser between the pressure required to yield the flange bolts and the previously mentioned definition of Gross Leak Pressure. This criterion was based on the evaluation methods developed for Reference 1. In general, the calculated Gross Leak Pressure is greater for the smaller flanges and trends lower as the flange size increases. The mass leak rates and leak areas are calculated as described in Section 3.4.

From Tables 3-3 through 3-9, the Gross Leak Pressures for the smaller flanges in which joint lock-up does not occur (initial bolt stress equal to 20,000 and 25,000 psi) is governed by the yielding of the flange bolts. However, for the larger flanges, the bolt yield stress occurs after the gasket stress has decreased to match the retained pressure. As a result, by decreasing the initial bolt stress from 25,000 to 20,000 psi (Tables 3-3 and 3-4), the Gross Leak Pressures for the smaller flanges show an increase since a greater internal pressure is required to reach the bolt yield stress, while the Gross Leak Pressures for the larger flanges show a decrease since the gasket seating stress is smaller. From Table 3-5, increasing the bolt preload to 30,000 psi increases the Gross Leak Pressure for all flange sizes, since it produces joint lock-up for some flanges while also increasing the gasket seating stress. In Tables 3-6 through 3-9, increases in the amount of joint relaxation shows an increase in the Gross Leak Pressure for

the smaller flanges, but a decrease in the Gross Leak Pressure for the larger flanges. The mass leak rates decrease with increasing initial bolt stress and, at a given pressure, increasing the joint relaxation tends to increase the leak rate. The leak areas increase with increasing initial bolt stress since the increased bolt stress approaches the bolt yield stress. However, the leak areas reported in Tables 3-3 through 3-10 are given in terms of multiples of the Gross Leak Pressure and not at constant pressures. In Table 3-10, it can be seen that the leak areas are substantially increased with a reduced bolt yield stress, since the leak areas increase rapidly with increasing pressure once the flange bolts have yielded.

The estimation of the leak rate variabilities were based on the results of the 8"-150# flange cases. It was judged that the uncertainties would not vary substantially with flange size and, as a result, a constant uncertainty was used for all flanges. In Figure 3-5, the mass leak rates are plotted as a function of pressure for the varial 38"-150# flange cases. As shown in Figure 3-5, the variations in the leak rate due to the initial bolt stress are greater than that due to the variations in the amount of joint relaxation. The uncertainty in the leak rate due to the joint relaxation was estimated based on the judgment that the range from 25% to 0% joint relaxation reasonably represented a -33 variation. An uncertainty variability of $\beta_{JE} = 0.20$ was selected to represent the variability over the entire pressure range. In a similar manner, the uncertainty variability due to the initial bolt preload was evaluated based on the judgment that the range from 25000 psi to 30000 psi reasonably represented a 2.33 β variation. An uncertainty variability of $\beta_{JE} = 0.59$ was selected to represent the variability for the leak rate of the 150# rated was taken as:

 $B_{10} = (0.20^2 + 0.59^2)^{1/2} = 0.62$

3.5.2 300lb Flanges

The results of the analyses for the 300lb rated flanges are presented in Tables 3-11 through 3-23. Table 3-11 shows the dimensional data for the gaskets and flange of iting which are standard for 300lb ANSI flanges while Tables 3-12 through 3-23 present the calculated Gross Leak Pressures, leak rates (pressure < Gross Leak Pressure), and leak areas (pressure < Gross Leak Pressure), and leak areas (pressure < Gross Leak Pressure) for the various cases studied. For the 300lb flanges, both SA-193 B8 and SA-564 Grade 630 bolts/stude are used on the pipelines of interest. Tables 3-12 through

3-19 provide results for the flanges joined by the SA-193 B8 bolts while Tables 3-20 through 3-23 provide results for the flanges joined by the SA-564 Grade 630 bolts. Among the tables, the cases studied include variation in the initial bolt/stud stress (Tables 3-12 through 3-14 and 3-20), variation in joint relaxition (Tables 3-15 through 3-18 and 3-21 through 3-22), and variation in bolt yield strength (Tables 3-19 and 3-23). Recall that for the 300lb flanges with the SA-193 B8 bolts, the median case corresponds to an initial bolt stress of 25,000 psi and a joint relaxation of 25%. For the flanges with the SA-564 Grade 630 bolts, the median case corresponds to an initial bolt stress of 25%.

Each table includes the effective gasket stress, defined here as the gasket stress due to bolt preload unaffected by the preserve of the compression gauge ring, the actual gasket stress, limited by the presence of the compression gauge ring, and the resulting gasket loading deflection. For the flanges with the SA-193 B8 bolts (Tables 3-12 through 3-19), the effective and actual gasket stresses for most of the flanges are equal indicating that the bolt preload is insufficient to bottom-out the flange on the compression gauge ring. Only for the cases with an initial bolt stress of 30,000 psi (Table 3-14) do some of the flanges bottom-out on the compression gauge ring.

Reviewing Tables 3-12 through 3-18, the results for the 300lb flanges are similar to those for the 150lb flanges. The Gross Leak Pressures for the smaller flanges in which joint lock-up does not occur (initial bolt stress equal to 20,000 and 25,000 psi) are governed by yield of the fiange bolts. However, for the larger fianges, bolt yield occurs after the gasket stress has decreased to match the pressure retained. As a result, by decreasing the initial bolt stress from 25,000 to 20,000 psi (Tables 3-12 and 3-13), the Gross Leak Pressures for the smaller flanges show an increase since a greater internal pressure is required to reach the bolt yield stress while the Gross Leak Pressures for the larger flanges show a decrease since the gasket seating stress is smaller. Increasing the joint relaxation shows a similar trend. The Gross Leak Pressure for the smaller fianges increases while it decreases for the larger flanges with increasing joint relaxation. In the cases for the SA-193 B8 bolts, the computed leak rates are quite low. The leak areas are dependent on the pressure required to develop yield in the flange bolts. Once yielding occurs in the flange bolts, the leak area increases substantially. The influence of reducing the bolt yield stress can be seen by comparing Tables 3-13 and 3-19. Decreasing the bolt yield stress resulted in lower Gross Leak Pressures for the smaller flances. With the decreased bolt yield stress, the leak areas are increased.

Tables 3-20 through 3-23 show the results for the 300lb flanges with the SA-564 Grade 630 bolts. For these cases, the initial bolt stress is much higher than for the SA-193 B8 bolts. As a result, the prestress is sufficiently high to properly seat the gasket and bottom-out the flange on the gasket compression gauge ring. As expected, the Gross Leak Pressures are much higher than for the cases with the softer bolts. The computed leak rates are very low. In addition, the leak areas beyond Gross Leak Pressure are low since bolt yielding does not occur until very high pressures.

The mass leak rates are typically low for pressures less than Gross Leak Pressure and the Gross Leak Pressures are relatively high for the median case shown in Table 3-16. Therefore, the specific value of the uncertainty variability for the leak rate is relatively unimportant. Based on results from Reference 1 for 300lb rated flanges, a combined uncertainty variability of 0.51 was used to represent the variability due to the initial bolt stress and the bolt relaxation.

3.5.3 400lb Flanges

The results of the analyses for the 400lb rated flanges are presented in Tables 3-24 through 3-32. Table 3-24 shows the dimensional data for the gaskets and flange bolting which are standard for 400lb ANSI flanges while Tables 3-25 through 3-32 present the calculated leak rates (pressure < Gross Leak Pressure) and leak areas (pressure > Gross Leak Pressure) for the various cases studied. The cases include variation in initial bolt/stud stress (Tables 3-25 through 3-27), variation in joint relaxation (Tables 3-26 and 3-28 through 3-31), and variation in bolt yield strength (Tables 3-26 and 3-32). For the 400lb rated flanges on the lines of interest, only the SA-193 B8 bolts are used. The median case corresponds to an initial bolt stress of 25,000 psi and a joint relaxation of 25%.

For most of the cases in Tables 3-25 through 3-31, the effective gasket stress exceeds the actual gasket stress indicating that the bolt preload is sufficient to bottom-out the fiange on the compression gauge ring. For those cases, the gasket deflection is equal to the maximum deflection of the nominal 0.175" gasket. The cases for which joint lock-up does not occur correspond to the smallest flanges with the median initial bolt stress and all flanges with the lower bound initial bolt stress. For these cases, similar trends as for the 150lb and 300lb flanges with respect to the initial bolt stress and joint relaxation can be seen.
For the flanges with joint lock-up, it can be seen that the Gross Leak Pressure increases with increasing initial bolt stress. Joint relaxation has little influence on the flange leak resistance. This is due to the lock-up between the flange and the compression gauge ring. The effect of variation in the bolt material yield stress can be seen in Tables 3-26 and 3-32. For the cases with joint lock-up, decreasing the bolt yield stress has no effect on the Gross Leak Pressure or the mass leak rate, but does increase the leak area.

The computed leak rates for the 400lb flanges are very low and the Gross Leak Pressures are high for the median case (Table 3-29). As noted for the 300lb flanges, the specific value of the uncertainty variability for the leak rate is of little importance. Therefore, the uncertainty variability of 0.51 used for the 300lb flanges was also used for the 400lb flanges.

STIFFNESSES FOR ASBESTOS-FILLED SPIRAL WOUND GASKETS

	-	Winnty	Acces	Marrier	Dimmin		1 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	Casses!	Bolt Stress		WELS LEFT	······
	(au)	(iui)	(up bs)	twomper	(in)	(sq in)	(in)	Fest Load (Ibs)	et GRL frach	Tg=0.125 footfinb	Tg=0.136	Tg=0.135 Includ
									1	for any of	fram Marrielle	haberth
	2 125	0.2500	3 865	4	1/2	0.1416	1 625	15100	26680	151902	1798427	2023 '8
	2 125	0.2500	1 865		3/4	0.3340	1 875	36200	27096	3661337	431263	485171
	1010	00520	1 865		3/4	0 3340	2375	36200	27096	388137	431263	485171
	2 260	00020	1000	ę (200	0.2256	1.750	24200	26817	205416	228240	256770
	2 250	0 254 KD	2 366	c q	0.0	0,0000	CAN'S I	48400	26817	410832	456480	513540
	3250	0.2500	2749		a all	00000	0000	INUHSIAN C	11007	410932	456480	513540
	3 250	0.2500	2749	- 00	3/4	0 2340	1756 6	TOSON	27112	COTABIC	190904	22/XX189
	3 2 30	0.2500	2749	-	3/4	0.3340	2875	12564	27430	507400	ACCOUNT A	000000
	4 000	0.3125	4 234	4	5/8	0 2256	2 525	24200	266.07	114318	FCOLCE.	0000000
	4,000	0.3125	4 234	8	Rus.	0.3340	2500	127500	271.03	342483	380537	428104
	4.000	0.3125	4 234	00	3/14	0.3340	3.125	72500	27133	342483	380537	426104
	5 000	0.3750	6 332	80	5/8	0.2256	2.125	48500	2687.)	153184	170204	191479
	5 000	0.3750	6,332	10	3/4	0 3340	2.750	72500	27133	228985	254479	286232
	052.5	0 5000	8.247	80 G	1/18	0.4612	3 625	101000	27374	244947	272163	306184
	1000	0.4088	11.25	117	3/4	0 3340	2,256	72500	27133	128606	142695	180757
	1 188	0.4688	5/211	23	ar in	03340	3 125	100/00	25125	178623	221961	223286
	0101	0.5200	14.7.25	2		0.6051	4 375	190000	26165	258043	286715	322554
	1001	0 5313	15.220	10	19/10	0 3340	2 500	72500	27133	89394	1/2066	211743
	9 188	0.5313	16 220		1/8	0 4512	2 B00	150800	27248	- 185340	206600	22425
	8 875	0.6975	20.654	12	1-1/8	0.7627	5 000	262000	28626	253710	261500	たわれたいの
0.0	61313	11650	19.767	12	2/18	0.4512	2 625	151000	27294	152781	163756	19,376
2 5	01010	01000	101 101	0.1		1004.0	4 00.00	(Presch)	2/310	267518	297242	000000
2.0	202011	0,6250	27 489	2 2	778	C1 26 C1	0.440	0.000000	26/192	5134/4	400,040	200000
0	13 375	0.6250	27 489	19	1.12	0 7637	1250	14040M	286.25	CELEC E	100000	101000
0	12 875	0.8750	37 797	20	5-3/4	0.9684	5875	557000	28759	294730	307479	TURNET .
2	14 625	0.6250	29 943	12		0.6051	3 000	198000	27268	132250	146944	165312
8	14 625	0.6250	29.943	62	1.1/8	0.7627	4 500	436000	28583	291217	323574	364021
8	14 250	0.8125	38 448	8	1-3/8	1.1538	6 125		27408	328398	365553	411247
8	16 625	0.7500	40.333	16	÷	0.6051	3.425	264000	27258	128973	143303	161276
3	16.525	0.7500	40.939	20	2-2/4	0 9684	# 750	557000	28759	272113	302348	340741
2	16 250	0 9375	50.621	8	1.1/2	1.4041	6.6.25	l	27643	306701	340778	383376
	18 688	0 9688	59 822	9	1-1/8	0.7627	5257	ł	27262	111226	123564	139032
	18 688	0.9688	59 822	58	8.174	0.9684	2 000	£	27565	214187	237985	261733
9	18 500	1 06:25	662 539	59	1-5/8	1 7723	7 125		28439	#52(約0円)	343060	0000000
8	20 688	0.9688	628 303	29	1-1/8	0.7627	3 625		27262	126192	140213	157740
8	20.688	0 9638	62 303	24	2-2/4	0.9684	5.250		27565	194406	2160	243008
	20.500	1 0625	71 974	24	1-5/8	1.7723	7.625	1	26439	336140	373488	420175
8	24.750	1 0625	86 161	50	1-1/4	0 9684	4.000		27565	(23327	137636	1549.08
2	24 750	1.0625	86.161	54	1-1/2	1 4041	5,750		27643	216232	240258	270290
8	24 750	1 0625	86 161	5.4	1-7/8	2.4107	929 8	j.	29467	385000	427778	481230
ý	45.625	0 9375	BC: LEF	OF-	000	1.1538	10.215	1	男のひとで	109200	NOT OF CONCEL	451185

MV-4237-001-R004 Rev. 0

r.g. - Compressed Gasket Thickness

Flange	¥	GAS	KET		Pressure	*	-BOLT	S	*
Diameter (in)	OD (in)	ID (in)	Width (in)	Area (sq in)	Area (sq in)	Number	Diameter (in)	Area (sq in)	Length (in)
Flanges									
1-1/2 2-1/2 3 4 6 8 10 12 14 16 18 20 24	2.750 3.375 3.875 4.750 5.875 8.250 10.375 12.500 14.750 16.000 18.250 20.750 22.750 27.000	2.125 2.750 3.250 4.000 5.000 7.188 9.188 11.313 13.375 14.625 18.688 20.688 24.75	0.2500 0.2500 0.3125 0.3750 0.4688 0.5313 0.5313 0.5313 0.6250 0.6250 0.9688 0.9688 1.0625	1.865 2.356 2.749 4.234 6.332 11.275 16.220 19.767 27.489 29.943 40.939 59.822 65.909 86.161	3.547 5.940 8.296 12.566 19.635 40.574 66.298 100.509 140.500 167.989 217.077 274.279 336.129 431.105	4444888822226600	1/2 5/8 5/8 5/8 3/4 7/8 7/8 1 1-1/8 1-1/8 1-1/4	0.1416 0.2256 0.2256 0.2256 0.3340 0.3340 0.4612 0.4612 0.6051 0.6051 0.7627 0.7627 0.9684	1.625 1.750 2.000 2.125 2.125 2.250 2.500 2.625 2.750 3.000 3.125 3.375 3.625 4.000

TABLE 3-2 150# ANSI FLANGE AND GAGKET DATA

Bolt areas correspond to tensile stress area

H-48

TABLE 3-3

150# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

.....

INITIAL BOLT STRESS = 20000 psi

JOINT RELAXATION = 0%

Boll Stress at 2 001.P (ps)	CATCA	CAURD -	Same	53913	50904	46588	42167	40061	36727	36974	36827	36413	36721	36962	
Leek Area at 2.0GLP (sq in)		100	2 11	192	3.18	363	346	3 42	227	287	3.28	3 53	4.61	6.44	
Leak Aree at 1.75GLP (sq ln)		0.03	1.41	0.46	2 2 1	227	102 5	124	0.06	10.0	0.08	0.09	0.11	0.15	and the second second
Leck Area et 1.5GLP (sq in)		0.52	0.80	1.12	80.1	10.01	0.03	200	0.04	0.04	0.05	0.06	0.07	0.10	and the second se
Leak Area at 125GLP (sq in)		0.21	0.34	0000	X	0.01	100	000	000	0.00	0.03	200	0.01	0.05	
Leak Rate et 75GLP (mg/sec)		55	0)	1	20	0.00	00	240	0.0	2 66	100	s X	1.1	48	
Leak Rete et SOGLP (mg/sec)		40	ev .	*	31	n ş	R	14.5	-	20	0.5		00	20	
Loak Rate et 25GLP (mg/sec)		¢i		CV.	16	es 1	80 :	ŋ :		0.5	<i>.</i> ,		÷.	. 0	
Leak Rate et GLP {mg/sec}		14	2	13	118	8	74	535	38	213	38	Ba	183	201	121
Gross Leaf Pres.ure (psi)		3273	3427	2617	1857	2015	1347	757	1004	659	462	150	131	159	600
Gasket Dellect (in)		0.034	0.034	0.033	0.034	0 034	0.033	0.033	0.033	0 033	0.033	0 033	0.033	0-033	0.033
Act Gasket Stresn (psi)		6073	7660	6566	4263	5700	4740	3295	5600	4027	4850	4730	4080	4629	4496
Eff Gasket Stress (psi)		1000	7680	6566	4263	5700	4740	3295	5600	4027	4850	4730	4080	4629	4495
Flange Diameter (m)	Flanges			0.10		12	d	8	10	12	1.4	16	18	202	24

- H-49

Bult Vield Stress - 33,000 psi (SA193 Bil)

DGmax = 0.050 in

Calculation 1502000

JOINT RELAXATION = 0%

150# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE TABLE 3-4

INITIAL BOLT STRESS = 25000 psi

	L.			And a start	I a de Data	1 auto Rata	t ank fiate	Leak Rate	Look Area	Leek Area	Leek Area	Leak Area	Bolt Stress
Fiange Diameter	Eff Gasket Stress	Act Caskel Stress	Deflect	Pressure fresh	et GLP	et 25GUP (mg/sec)	et 50GLP (mg/sec)	et .75GLP (mg/sec)	at 1.25GLP (sq.fn)	et 1.5GLP (sq in)	et 1750LP (sq in)	at 2.06LP (sq in)	at 2.004.P
(14)	(bad)	6sc/h	land	Lund									
Flanows													
									0.00	0.44	0.66	0.88	47545
C1110	1631	1652	0.042	2323	1	0			10.40	0.64	180	1.29	49696
	9670	9575	0.042	2383		0			N. B.B.	0.07	1.21	1 75	49000
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	2002	8207	0.042	1740	3	0			10.04	10.0	1 82	243	50021
- 4 ME	1000	5329	0.042	1222	33	67	6		10.00	1.72	250	361	52512
	2426	7125	0.042	1794	8		CN S	55	1 000	12.2	203	8.87	58235
- 12	5925	5325	1100	1684	23	CH .	0	1.6	10.0	3 00	5.20	7.82	52708
	4118	4118	0.043	348	166	15	8, 4	11	CU U	274	5.47	\$ 20	50076
- 10	7000	7000	0.041	1255	12	- 1	N	5.2	000	1 43	4.59	1.69	45909
	5033	5033	0.041	824	88		N a	10.7	0.02	1 44	561	9.35	46218
	1 6062	6062	0.041	917	27	N	0 W		0.03	2.15	654	10.93	46033
. 10	5912	5912	0.041	938	2	2	n :	10	0.04	28.6	7.60	12.83	45524
10	5100	5100	0.041	616	57			100	2000	3.63	9.33	15.64	45902
20	5786	5786	0.041	949	32	N	01	14	2010	422	12.63	21.04	45203
24	5620	5620	0.041	854	04	2							

Bolt Yield Stress = 33 (000 psi (SA 193 88)

Calculation 1502500 DGmax = 0.050 in

- H-50

.

150# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 30000 psi

JOINT RELAXATION = 0%

Flange Diameter (in)	Eff Gasket Stress (psi)	Act Gasket Stress (psi)	Gasket Deflect (in)	Gross Leak Fressure (psi)	Leak Bate et GLP (mg/sec)	Leek Rete et .25GLP (mg/sec)	Leak Rate et .50GLP .(mg/sec)	Leak Rate al .75GLP (mg/sec)	Leak Area at 1.25GLP (sq in)	Leak Area et 1.5GLP (sq In)	Leak Area et 1.75GLP (sq in)	Leak Area at 2 OGLP (sq In)	Bolt Stress st 2 0GLP (pai)
Flanges													
1-1/2	9109	9006	0.050	4877	2	6	1.1		0.65	1.77	3.89	6.85	27133
2	11490	11400	0.050	5120	1	0	0	0	1.07	2.94	6.69	11.06	80766
2.1/2	9848	9800	0.050	3916	2	0	0	1.1	1.69	4.58	10.58		83922
3 3	6394	6350	0.050	2477	15	2	5	8	1.76	4.95	11.42	19.17	80607
4	8550	8500	0.050	3015	3	0		2	1.45	4.68	10.20	10.36	1 76171
6	7110	7110	0.050	2020	9		2	5	2.94	5.00	10.05	19.65	69882
8	4942	4942	0.050	1136	64	6	15	30.	2.64	5.28	10.00	17.05	63250
10	8400	8400	0.049	1506	5	0	1 T T	2	3.15	6.43	10.58	17 98	60091
12	6040	6040	0.050	988	25	2	5	10	1.95	5.67	3 39	15.6	55091
14	7275	7275	0.049	1101	10	1.1.1	2	1.1.1	2.38	6.67	11.35	19	55462
16	7095	7095	0.050	1126	51	1	2	4	2.78	8.05	13.32	22.21	55240
18	6120	6120	0.049	1096	22	2	4	9	3.24	9.52	15.80	25.71	54628
20	6943	6943	0.050	1138	12	1	2	5	3.97	11.54	19.10	31.65	55082
24	6744	6744	0 049	1024	15	1.1	3	6	5.37	15.46	25.54	42.91	55443

* Bolt Ultimate Strength Exceeded

Bolt Ultimate Strength = 82,500 psi (SA193-88)

Bolt Yield Stress = 33,000 psi (SA193-88)

DGmax = 0.050 in

Calculation 1500000

150# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 25000 psi JOINT RELAXATION = 15%

-

H-52

Flange Diameter	Ett Gasket Stress (psil	Act Gasket Stress (csi)	Gasket Deflect (m)	Gross Leak Pressure (psi)	Leak Rate at GLP (mg/sec)	Leak Rate et 25GLP (mg/sec)	Leak Rate et .50GLP (mg/sec)	Leak Rate at .75GLP (mg/sec)	Leak Aree at 1.25GLP (sq.in)	Leek Area et 1.5GLP (sq in)	Leak Area el 1.75GLP (sq in)	Leak Area ef 2:0GLP 11-1 in)	Bolt Stress at 2.0GLP (psi)
Flangos 1-1/2 2-1/2 3 4 6 8 10 12 14 16 18 20 24	6453 8139 6976 4529 6057 5036 3501 5950 4278 5153 5025 4335 4918 4777	6453 6139 6976 4529 6057 5036 3501 5950 4278 5153 5025 4335 4918 4777	0 041 0 041 0 041 0 041 0 041 0 041 0 041 0 040 0 040 0 040 0 040 0 040 0 040 0 040	3412 3500 2556 1760 2141 1431 804 1067 700 780 797 776 806 725	4 1 35 8 22 161 12 64 28 55 31 38	10051251322425	201112632212451167	3 1 2 2 4 11 7 4 5 5 5 10 11 2 3 12 12 14	0.32 0.47 0.64 0.64 0.06 0.01 0.02 0.02 0.02 0.02 0.03 0.03 0.04 0.05	0 65 0 95 1 28 1 71 1 57 1 51 0 71 0 03 0 04 0 05 0 06 0 07 0 08 0 11	0.97 1.42 2.62 3.09 0.36 2.58 2.30 0.99 1.31 1.46 1.45 2.01 2.94	1.49 2.33 3.21 4.44 4.25 4.40 4.45 4.62 3.62 4.49 5.19 5.19 5.90 7.36 10.09	54364 56039 56500 57283 54086 49500 44802 42565 39023 39285 39128 38695 39016 39272

· .*

Bolt rield Stress - 33,000 par (SA193-E9)

DGmax = 0.050 in

Calculation 1502515

150# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 25000 psi JOINT RELAXATION = 25%

Bolt Stress at 2 00LP (psi)		486.70	CONCRED 1	KOCAN	SOSA	A7723	A'3676	206.24	376.67	24432	24665	345.76	24142	3640K	34652
Leek Area et 2.0GLP (sq in)		0.04	1.45	200	2 50	265	2 840	2.45	223	160	1.2%	1.36	121	1 85	2.79
Leak Area at 1.75GLP (sq in)		0.65	1 00	152	E2 1	121	1 5.0	0.83	0.18	0.05	0.06	0.07	0.09	0.11	0.14
Leek Area at 15GLP (sq in)		0.36	0.59	160	0.96	0.80	0.31	0.02	0.03	0.03	0.04	0.05	0.05	0.07	0.05
Leek Area at 1.25GLP (sq in)		0.07	0.15	0.29	0 19	0.01	0.01	0.01	0.02	0.02	0.00	6.22	0.03	0.04	0.05
Leek Pate at 75GLP (mg/sec)		-	-	cv	20	**	11	13	10	24	ø	11	22	EV.	14
Leck Rate et 50GLP (mg/sec)		N	0		11	¢4	9	38	-	11	*	ŝ	10	10	¢
Leek Rate at 25GLP [mg/sec]			0		5	*	2	14		*	EV.	CN	*	0	e
Leak Rate at GLP (m.)/sec)		4		4	34	8	22	157	11	55	25	58	54	30	37
Gross Leak Pressure (psi)		3068	3213	2453	1553	1869	1263	710	941	618	688	704	685	727	540
Gaoket Deflect (in)		0.040	0 040	0.040	140 0	0.040	0 0 00	0.040	0.039	0040	0.040	0.040	0.040	0.040	0.039
Act Gasket Stress [psi]		56.33	7181	6155	3996	5344	4444	6906	5250	3775	4547	\$434	33255	4340	4215
Eff Gasket Stress (psi)		5633	7181	6155	98560	5344	4444	3089	5250	3775	#542 ·	44.34	3825	4340	4215
Flange Diameter (in)	Flanges	11/2	2	2.122	-			8	- 30	12	14	16	18	00	24

8-11 Yorld Shess - 33,000 psi (CA193-88)

DGmise = 0.050 in Calculation 1502525

MV-4237-001-R004 Rev. 0

150# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 25000 psi

JOINT HELAXATION = 33%

Budt Stress et 2 0/GLF (psi)		43352	45274	21684	#5152	42632	39017	35314	33551	30759	30860	30842	30501	30754	30356	
L wak Area et 2 0GLP (sq in)		0.63	101	153	1.74	172	1.62	0.00	0.32	0.06	200	0.09	0.10	0 13	0.17	
Leak Arse at 1.75GLP (sq in)		0.37	0.62	0.96	105	0.30	0.48	50.0	0.04	0.05	0.06	100	0.08	0.00	0.13	
Lesk Area at 1.5GLP (sq in)		0.11	80.04	2.43	0.36	100	000	000	0.03	0.03	0.04	0.04	0.05	0.06	0.08	
Leak Area at 1259LP [sq in]		0.20	0.00	0.00	0.01	100	0.01	0.01	201	2010	000	000	0.04	0.03	and a	1000
Lesk Rate et 75GLP (mg/sec)		~		. 0	in the second					24	0	01	10			
Lesk Rate et SOGLP (mg/sec)				5.*		t e	0.44	o X	30	14			100	2.4	0.4	0
Leak Rate et 25GLP (mg/sec)				5 6	22 M	n v		• 1	÷.•				4	* 3		N
Leak Rate at GLP (mg/sec)					a	5	100	1.4	1015	1 A A	10	11	12	XI	62	R
Gross Leak Pressure (psi)			1617	58/97	2612	1387	1991	SZLL.	63%	941	700	514	679	6.2	636	572
Gasket Defiect. (in)			060.0	600	0 033	0.040	0 0 40	0.039	0.040	0.039	0.040	0.039	0 033	0.039	0.039	0.039
Act Gasket Stress (pel)		-	20296	6415	5439	3570	4224	3970	2759	4690	3372	4062	3961	といせの	3877	3765
L.1 Gasket Stress (psi)			5086	6415	5493	3570	4774	0/60	2759	4630	33722	4062	3961	3417	3877	3765
Flange Diameter (in)	Flatines		318.4	2	2.1/2	10	4	9	10	10	12	14	16	18	20	24

- H-54

Bolt Yield Stress = 33,000 psi (SA193 BB)

DGmax = 0.050 in

Calculation 1502533

5

MV-4237-001-R004 Rev. 0

150# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE TABLE 3-9

INITIAL BOLT STRESS = 25000 psi

0,

JOINT RELAXATION = 50%

Joht Stress et 2.0GLP (psi)	32352	33787	35053	33696	31815	23118	26354	25038	22955	23109	23017	22762	22951	23101	
Leak Area at 2:0GLP (sd in)	 0.01	0.07	0.23	0.11	0 02	0.03	0 03	0.04	0.05	90.0	0.01	0.08	0.09	0.12	
Leak Aren et 1.75GLF (sq in)	000	0.01	0.01	0.01	0.01	0.02	30.0	0.03	0.03	0.04	0.05	0.06	20.07	60.0	
Leek Area at 1.5GLP (sq in)	0.00	0 00	n 01	0.31	0.01	0.01	000	000	0.02	0.03	0.03	0.04	0.06	90.0	
a (25GLP (sq in)	0.00	000	0.00	000	000	0.01	10.0	100	100	0.01	000	COLO	0000	0.03	And and the second
Leak Rate at 75GLP (mg/sec)	0			10	0.4			20	-	3 .	ur ur	101		13	A
Leri, Rate et 50GLP (mg/sec)		- 0			2 4	N U	0	33	N	Dr.	e i	n	10	n w	
Leak Rate at 25GLP (mg/sec)							N	13		4	0	N	*	~ ~	4
Leak Rate et GLP (m3/sec)		4		9	32	1	20	344	10	57	23	25	64	28	54
Gross Leak Pressure (psi)		2045	2142	1636	1035	1259	842	E14	528	412	459	469	457	474	176
Gasket Deflect. {in}		0.038	0.037	0 038	0.039	0.039	0 038	0 039	0.038	0 039	0.038	0.038	0.039	0.038	0.038
Act Gaskel Strøss (psi)		3796	4787	4103	2664	3563	0960	2054	3500	2517	3031	2056	DEGO	2893	281J
Eff Gasket Stress (psi)		3796	4787	A103	2664	7552	0000	00200	16000	2647	2024	2000	- NEED	2893	2810
Flange Diameter (in)	Hanges	CIT I	0	0 110	201.2			p e	p 5	2 9	4 7	- 4	2 4	50	24

Bolt Yield Stress = 33,000 psi (SA193-B8)

DGmax = 0.050 in Calculation 1502550

5

18. 18

MV-4237-001-R004 Rev. 0 1

11

.

150# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRFSS = 25000 psi

JOINT RELAXATION = 0%

Flange Diamersr (in)	Eft Gasket Stress (psi)	Act Gasket Stress (psi)	Gasket Deflect (in)	Gross Lesk Preceure (psi)	Lesk Rate at GLP (mg/sec)	Leak Refs st 25GLP (mg/seu)	Leak Rete et 50GLP (mg/sec)	Leak Rate et 75GLP (mg/sec)	Leak Area at 1.25GLP (sq In)	Lesk Area at 1.5GLP (sq In)	Leek Area et 1.75GLP (sq in)	Leak Area at 2.0GLP (sq in)	Bolt Stress et 2.0GLP (psi)
Flanges													
6/5-1	7591	7591	0.042	726	0	0	0	0	0.07	0.14	0.21	0.28	32045
	06.76	95.75	0.042	745		0	0	0	0.10	0.20	0:30	0.40	32402
0110	1000	TUCA	0.042	Sad		0	0	0	0.14	0.27	0.41	0.55	32500
311.2	0023	5100	0.042	387	8		- 04	3	0.19	0.38	0.57	0.76	32619
2.8	7495	3632	0000	660	9	0	0	*	027	054	0.81	1.08	32598
r ()	6036	Raph	0.041	549	18		0	2	0.55	1.11	1.66	2.22	35833
	4110	8448	0.045	560	143	80	19	32	130	2.60	3.90	5.20	41389
0 5	2000	2000	6.041	12555	12	•	2	5	2.62	5.35	8.79	14.36	50076
	ECO1	ED13	0.041	824	88	5	12	26	1.63	4.73	7.82	12.94	45309
14	cana 1	COOD	0.041	917	27	2	S	10	1.99	5.72	9.45	15.87	46213
	5013	5412	0.041	938	50	~	5	11	2.32	8.71	11.10	18.47	45033
a	1002	2100	0.041	619	57	4	11	23	270	7.94	13.17	21.38	45524
000	6786	S.7AR	0.041	949	32	2	9	12	331	9.62	15.92	26.33	45902
100	00100	6600	0.041	BC.A	- WU		1	15	4.47	12,88	2129	35.59	46203

Boft Yield Stress = 27,500 psi (SA193-B8)

DGmax = 0.05i; in

Calculation 1502500Y

MV-4237-001-R004 Rev. 0

- H-56

TABLE 3-11

300# ANSI FLANGE AND GASKET DATA

Flance	*	GAS	SKET-	*	Pressure	*	BOL	TS	*
Diameter (in)	OD (in)	ID (in)	Width (in)	Area (sq in)	Area (sq in)	Number	Diameter (in)	Area (sq in)	Length (in)
Flanges									
1-1/2	2.750	2.125	0.2500	1.865	3.547	4	3/4	0.3340	1,875
2	3.375	2.750	0.2500	2.356	5.940	8	5/8	0.2256	2.00
2-1/2	4 750	4 000	0.2500	4 234	12,566	8	3/4	0.3340	2.500
4	5,875	5.000	0.3750	6.332	19.635	8	3/4	0.3340	2.750
6	8.250	7.188	0.4688	11.275	40.574	12	3/4	0.3340	3.125
8	10.375	9.188	0.5313	16.220	66.296	12	7/8	0.4612	3.500
10	12.500	11.313	0.5313	19.767	100.509	16	1	0.6051	4.000
12	14.750	13.375	0.6250	27.489	140.500	16	1-1/8	0.7896	4.250
14	16,000	14.625	0.6250	29.943	167.989	20	1-1/8	0.7896	4.500
16	18.250	16.625	0.7500	40.939	217.077	20	1-1/4	0.9985	4.750
18	20.750	18.688	0.9688	59.822	274.279	24	1-1/4	0.9985	5.000
20	22.750	20.688	0.9688	65.909	336.129	24	1-1/4	0.9985	5.250
24	27.000	24.750	1.0625	86.161	431.105	24	1-1/2	1.4899	5.750

Bolt areas correspond to tensile stress area

300# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE **TABLE 3-12**

INITIAL BOLT STRESS = 20000 psi JOINT RELAXATION = 0%

	and the second second second second		_		_	_											
the second s	Bolt Stress at 2 0GLP {psl}		51310	52118	52697	53635	55807	55029	49082	45923	41208	36974	36827	36419	20204	12100	296(5)
And and a second	Leek Aree at 2.0GLP (sq in)		3.59	1.83	2.56	3.85	6.38	9.65	8.46	9.48	7.59	4.30	4.98	5.31	6.0.0	20.01	8.25
	Lesk Arec et 1.75GLP (sq In)		0.96	1.35	1.81	2 59	3.92	5.52	5 59	5.66	3 29	0.10	0.12	0.14	1000	0.10	0.22
	Leak Aree at 1.5GLP (sq in)		0.64	0.90	121	1.73	2.61	3.68	272	1.85	0.05	0.67	0.06	0.09	20.00	0.11	0.14
and the second second	Leek Ares at 1 25GLP (sq ir:)		0.32	0.45	0.60	0.86	131	1.45	0.02	0.03	0.03	0.03	0.04	0.04		0.00	0.01
	Lesk Rate at 75GLP (mg/sec)		0	0	0	0	2	4	40	1				0	4.	4	2
	Leak Rate et .50GLP (mg/sec)		0	0	0	0		6	•	0		0	0			2	-
	Leak Rate at 25GLP (mg/sec)		0	0	0	0	0			0	0	0	0		2		0
	Leak Rata et GLP (mg/sec)		0	0	0	0	0	0	10	0			- 0		n	10	5
	Gross Leak Pressure (psi)		6897	6,809	6344	A 294	3104	2386	IROS	ALIK	1698	15.05	ar an	2000	1430	1192	1261
	Gasket Deflect. (in)		0.033	0.03d	0.033	0.033	1 DOT	0.036	0.033	0.033	0.033	2000	0.000	2000	0.034	0.034	0 035
	Aut Gasket Stress (psi)		14325	15320	10441	00802	8430	0.14	PCB3	0702	0100	2010	10766	Della	8012	7272	8300
	Eff Gasket Stress (psl)		14325	16320	10441	10000	BA'30	1110	10.04	0000	0100	or ac	10040	0012	8012	7272	8300
	Flange Diarmeter (in)	 ca6upiu	4.2.13		1111	×11.2	5.4	r 4	2 0	0.5	2.64	¥ 3		0	82	20	24

Bolt Yield Stress = 33,000 psi (SA193-89) DGmax = 0.050 in 1

Calculation 3002000

MV-4237-001-R004 Rev. 0

H-58

300# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 25000 psi JOINT RELAXATION = 0%

Bolt Stress at 2:00LP (psi)		AADER	44765	45121	AGRAR	ATOTA	50778	67242	57404	51510	46219	46033	45524	45902	46203
Lesk Area at 2.0GLP (sq in)		0.79	1.16	5.49	2 13	322	8.57	16.31	22.87	16.39	14.02	16.62	19.01	22.65	30.24
Leek Area at 175GLP (sq In)		0.59	0.83	21.1	1 60	180	4 93	954	13.05	11.61	8.42	9.94	11.26	13 52	18.15
Leek Area at 1.5GLP (sq in)		0.39	0.55	0.74	1.06	161	3.29	6.36	8.28	626	284	3.27	3.51	4.39	6.07
Leak Area at 1.25GLP (sq in)		0.20	0.28	0.37	0.53	0.80	1.64	3.18	351	0.85	0.04	0.05	0.06	0.07	0.09
Leak Rate at .75GLP (mg/sec)		0	0	0	0	0		¢v	¢	0	0	0		-	-
Leak Rate et 50GLP (mg/sec)		0	0	0	0	0	0		0	0	0	0	0		0
Leek Rate et 25GLP (mg/sec)		0	0	0	0	0	0	0	0	0	c	0	0	0	0
Leak Rate at GLP (mg/sec)		0	0	0	0		6	4		-	0	-	2	3	2
Gross Leak Pressure (psi)		4245	3575	3904	2700	1910	1756	2024	2517	2103	1995	1935	1793	1490	1576
Gasket Deflect. (in)		0.042	0.042	0.041	0.041	0.042	0.045	0.041	0.041	0.041	0.041	0.040	0.042	0.042	0.043
Act Gasket Stress (psi)		17906	19150	24301	15778	10549	9887	8530	12245	11490	13185	12195	10015	9060	10375
Eff Gasket Stress (psi)		17906	19150	24301	15778	10549	8887	8530	12245	11490	13185	12195	10015	90806	10375
Flange Diameter (in)	Flanges	1-1/2	CV	2-112	en,	4	9	80	10	12	14	96	18	20	24

Bolt Vield Stress = 33,000 pst (SA193 B8)

DGmax = 0.050 in Calculation 3002500

MV-4237-001-R004 Rev. 0

300# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 30000 psi

JOINT RELAXATION = 0%

Flange	Eff Gasket	Act Gasket	Gasket	Gross Leak	Leak Rate	Leak Rate	Leak Rate	Leak Rate	Leak Area	Leak Area	Leak Area	Leak Area	Bolt Stress
Dlamater	Stres	Stress	Deflect.	Fressure	at GLP	et .25GLP	at .50GLP	et .75GLP	at 1.25GLP	at 1.SGLP	at 1.75GLP	at 2.0GLP	et 2 0GLP
(in)	(psi)	(psi)	(in)	(psi)	(mg/sec)	(mg/sec)	(mg/sec)	(mg/sec)	(sq in)	(sq In)	(sq in)	(sq in)	(psi)
Flanges 1-1/2 2-1/2 3 4 6 8 10 12 14 16 19 20 24	21487 22979 29161 18933 12659 10665 10236 14694 13788 15822 14634 12018 10908 12450	21487 22800 25161 18933 12659 9900 10236 14694 13788 15822 14634 11900 10800 12000	0 050 0 050 0 050 0 050 0 050 0 050 0 049 0 049 0 049 0 049 0 049 0 049 0 048 0 040 0 048 0 050 0 050 0 050	13850 12652 14393 9024 5425 3464 2738 3021 2531 2394 2394 2394 2322 2152 1788 1891	0 0 0 0 2 1 0 0 0 0 1 1 1			0 0 0 0 1 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1 83 3 62 5 99 5 71 5 67 4 07 4 30 5 72 6 45 3 58 4 23 4 80 5 75 7 71	5.64 9.78 14.65 16.93 17.56 12.06 12.19 12.74 12.89 10.30 12.24 14.11 16.71 22.22	* * 28 09 24 80 25 83 22 77 17 02 20 25 23 41 27 67 36 72	* 48.53 50.26 50.51 38.39 28.61 33.76 38.09 45.84 61.68	92871 99790 104178 99174 92591 79881 73622 68895 61812 55462 55462 55462 55240 54828 55082 55443

* Bolt Littimate Strength Exceeded

Bolt Ultimate Strength = 82,500 pci (SA193-88) Bolt Yield Stress = 33,000 pci (SA193-88)

DGmax = 0.050 in

Calculation 3003000

300# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 25000 psi JOINT RELAXATION = 15%

Bolt Stress at 2.0GLP 'pol}		49549	50279	50803	51651	53614	58468	52149	48794	43783	39285	39128	38695	33016	39272
Leak Area st 2.0GLP (sq in)		1.16	1.62	2 19	3.13	5.25	12.52	10.29	11.56	9.94	6.73	7.83	874	10.67	14.50
Lesk Area et 1.75GLP (sq in)		0.87	122	1.64	2.34	354	7.16	7.02	7.51	5.37	1.97	222	2.15	2.90	4.23
Leek Area et 1.5GLP (sq in)		0.58	0.81	1.09	1.56	2.36	4 67	397	3.46	0.81	0.07	0.08	0.10	0.12	0.15
Leak Area at 1.25GLP (sq in)		0.29	0.41	0.55	0.78	1.18	2.30	0.92	0.03	0.03	0.04	0.04	0.05	90.0	0.08
Leak Rate et 75GLP (mg/sec)		0	0	0	0	0	CV.	67	0	C	0	0			-
Leak Rate et SOGLP (mg/sec)		0	0	0	0	G	-	-	0	0	0	0	0		0
Lesk Rate et 25GLP (mg/sec)		0	0	0	0	0	0	0	0	0	0	0	0	0	0
Leak Rate at GLP (mg/sec)		0	0	0	0	1	0	4		*	0	-	2	5	2
Gross Leak Pressura (psi)		6234	5231	5734	3966	2805	2536	1933	2140	1793	1695	1645	1524	1267	1339
Gasket Defect. (In)		0.042	0.042	0.041	0.041	0.042	0.045	0.041	0.041	0.041	0.041	0.040	0.042	0.042	0.043
Act Gasket Stress (psl)		15220	16277	20656	13411	1968	7554	7251	10408	9166	11267	10366	8512	7726	8519
Eff Gasket Stress (psi)		15220	16277	20656	13411	1968	7554	7251	10408	9766	11207	10366	8512	7726	6819
Flange Diameter (in)	Flanges	1-1/2	ev.	2-112	m	1	9	8	10	12	14	16	18	20	24

Bolt Yield Stress = 33,000 psi (SA 193-89)

DGmax = 0.050 in Calculation 3002515

MV-4237-001-R004 Rev. 0

300# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE **TABLE 3-16**

INITIAL BOLT STRESS = 25000 psi JOINT RELAXATION = 25%

P at 50GLP at 75GLF c) (mg/sec) (mg/sec)	at 25Gl (mg/sec	et GLP (mg/sec)	Pressure (ps)	Gasket Dellect. (in)	Act Gasket Stress (psi)	Eff Gasket Stress (psi)
---	--------------------	--------------------	------------------	----------------------------	-------------------------------	-------------------------------

provident statements				uni-s			-	-	-	-	-	-	-	-	-
Bolt Stress at 2 0GLP (psl)		53070	53956	54591	55619	57870	51590	46014	43053	38632	34663	34525	34143	24426	34652
Leak Ares et 2.0GLP (sq in)		1.52	222	3.08	4.60	7,45	6.88	6.86	7.39	5.24	187	2.07	1.89	2 68	4.01
Lesk Area at 1.75GLP (sq in)		1.05	1.48	1.99	2.84	4.27	4.79	4.17	3.81	1.21	0.09	0.11	0.13	0.15	0.20
Leek Area et 1.5GLP (sq in)		0.70	0.98	1.33	1.90	2.84	2.70	1.48	0.24	0.06	0.05	0.07	0.09	0.10	0.13
Lesk Area at 1.25GLP (sq in)		0.35	0.49	0.66	0.95	1.42	0.60	0.02	0.03	0.03	0.03	0.04	0.04	0.05	0.07
Lesk Rate at 75GLF (mg/sec)		0	0	0	0		-	~	0	0	0	0	-		
Leek Rate at .50GLP (m3/sac)		0	0	0	0	0			0	0	0	0	0		0
Leek Rate at 25GLP (mg/sec)		0	0	0	0	0	0	0	0	0	c	0	c		0
Leak Rate et GLP (mg/sec)		0	0	0	0		67	4			0		c	40	0.04
Gross Leak Pressure (psi)		7561	6368	6954	4810	3391	1800	1711	1688	1583	ADE	1921	3464	0.000	1182
Gasket Deflect. (in)		0.642	0.042	0.041	0.041	CFUU	0.045	0.041	0.043	0.041	0.041	0.040	0000	2000	0.043
Act Gasket Stress (psi)		62463	CSEVE	18226	11833	CIEL	RAAK	E age	0104	B617	0000	01440	2140	1101	7781
Eff Gasket Stress (psi)		PCAF1	14262	18226	11033	7410	SCO.	6308	0100	0017	1100	0000	0510	1101	7701
Flange Dizmeter (in)	langes	0111	10	0.110	2010	5 3	7 9	0 0	0		4.4	7 0	01	0	02

Bolt Yield Stress = 33,000 psi (SA193-88) DGmax = 0.050 in

Calculation 3002525

MV-4237-001-R004 Rev. 0

- H-62

300# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 25000 psi JOINT RELAXATION = 33%

Bolt Stress at 2.0GLP (psi)		541 27	56014	57621	65372	51697	46087	31106	38461	34511	30966	30842	39501	30754	30956
Leak Aran at 2.0GLP (sq in)		5.33	2.66	3 80	453	4.29	4 86	4 29	\$ 05	1.48	0.11	0 13	0.15	0.18	0.24
Leak Area at 1.75GL ^o (sq in)		96.0	1.61	202	276	3 02	2 89	1 89	0.85	0.06	0.08	0.10	0.12	0.14	0.16
Lesk Area at 1.5GLP (sq In)		0.80	1.06	151	177	1.74	112	0.04	0.05	0.05	50.0	10.0	0.08	0.09	0.12
Leak Area at 1.25GLP (sq. ¹ -1)		0.24	0.52	0.75	0.77	0.46	100	0.02	0.02	0.03	0.03	0.03	0.04	0.05	0.06
Leak Rate at .75GLP (n:g/sec)		0	0	0	e			CV	0	0	0	0		*	
I sak Rate at 50GLP (mg/sec)		0	0	¢.	c	0			0	2	0	0	0	-	0
Leak Rate at 25GLP (mg/sec)		0	0	0	c	0	0	0	0	0	0	0	0	0	0
Leek Pate at GLP (mg/sec)		0	0	a	0	4	0	0			0		04	0	*
Gross Leek Pressura (psi)		7733	1102	19597	5038	3029	1999	1529	1687	1413	1336	1296	1201	966	1056
Gasket Deflect. (in)		0.042	0.042	0.041	0.041	0.042	0.045	0.041	0.041	0.041	0.041	0.040	0.042	0.042	0.043
Act Gaske! Stress (psl)		1997	12830	15281	10521	1000	5954	5715	8204	7698	8834	8171	6710	0603	6951
Eff Gasket Stress (psi)		11997	12830	16281	10571	7068	5954	5715	8204	7698	8834	8171	6710	6090	6951
Flange Diameter (in)	langes	1-1/2	N	2-1/2	5	4	9	80	10	12	14	16	18	- 20	24

Bolt Yield Stress = 33,000 psi (\$A193-86)

DGmax = 0.050 in Calculation 3002533

MV-4237-001-R004 Rev. 0

300# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 25000 psi JOINT RELAXATION = 50%

	(B a)		to the character	****		-	encies.				-	-		-	(Arrest)	
	Bolt Stres at 2.0GUF (psi)		38696	41801	43407	41322	38580	24393	30676	28702	25755	23109	23017	22762	22951	23101
and the second second	Leek Area at 2.0GLP (sq in)		0.40	0.83	1.29	1.41	130	0.55	0.05	0.07	0.08	0.08	0.10	0.12	0.14	0.18
and the second se	Leak Area et 1.75GLP (sq in)		0.14	0.42	0.71	0.67	0.35	0.03	0.04	0.05	30.08	0.06	0.07	0.03	0.10	0.13
	Leek Area et 1.5GLP (aq in)		0.00	10.0	0.14	0.01	0.01	0.02	0.03	0.04	0.04	D.04	0.05	0.06	200	0.09
	Leek Area at 1.25GLP (sq in)		0.00	0.00	0000	0.01	0.01	1010	0.01	20.02	0.02	0.02	0.05	0.03	0.03	0.04
	Leak Rate at 75GLP (mg/sec)		0	0	0	0		1 5	cv	0	0	0	0	1	1	
	Leak Rete at 50GLP (ms/sec)		0	0	a	0	0			0	0	0	0	0	0	0
	Leak i-ate et 25GLP (mg/sec)		0	0	0	0	0	0	0	0	0	0	0	0	0	0
	Leak Rate at GLP (mg/sec)		0	0	c	0		2	0	-		0	1		5	
	Gross Leek Pressure (psi)		5771	5300	2665	3760	2264	1492	1141	1259	1055	997	67	897	745	788
	Gasket Deflect. (in)		0.042	0.042	0.341	0.041	0.042	0.045	0.041	0.041	0.041	0.041	0.040	0.042	0.042	0.043
	Act Gasket Stress (psi)		8953	9575	12150	7889	5275	4444	4265	6122	STAS	6592	6038	5007	4545	5186
	Eff Gasket Stress (psi)		8953	9575	12150	7889	5275	4444	4265	6122	5745	65,32	RUGR	5007	45.45	5188
	Flange Diameter (in)	Flanges	1.1/2	0	2.112			- 42	. 60	10	10	1.4	16	69	00	24

Brit Yield Stress = 33,000 psi (SA 193-89) DGmax = 0.050 in

Calculation 3002550

MV-4237-001-R004 Rev. 0

H-64

300# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 25000 psi

JOINT RELAXATION = 0%

Flange Diameter	Eff Gasket Stress (osi)	Act Gasket Stress (psi)	Gasket Deflect (in)	Gross Laak Pressure (psi)	Leak Rate at GLP (mg/sec)	Leak Rate at .25GLP (mg/sec)	Leak Rate at .50GLP (mg/sec)	Leak Rate at .75GLP {mg/sec}	Leak Arna at 1.25GLP (sq in)	Leak Area at 1.5GLi (sq In)	Leak Area at 1.75GLP (sq in)	Leak Area et 2.0GLP (sq in)	Bolt Stress at 2 0GLP (psi)
Flanges 1-1/2 2 2-1/2 3 4 6 8 10 12 14 16 18 20 24	17906 19150 24301 15778 10549 8887 8530 12245 11490 13185 12195 10015 9090 10375	17906 19150 24301 15778 10549 8887 8530 12245 11490 13185 12195 10015 9090 10375	0.042 0.042 0.041 0.041 0.045 0.041 0.041 0.041 0.041 0.041 0.042 0.042 0.042 0.042	1326 1117 1220 844 597 549 632 1003 1729 1995 1935 1793 1490 1576	0 0 0 1 2 3 1 1 0 1 2 3 2		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.06 0.09 0.12 0.17 0.25 0.51 0.99 1.90 4.40 2.98 3.52 4.00 4.79 6.43	0.12 0.17 0.23 0.33 0.50 1.03 1.99 3.80 8.81 8.58 10.20 11.76 13.93 18.51	0.18 0.26 0.35 0.50 0.75 1.54 2.98 5.70 13.21 14.18 16.87 19.5 ¹ 23.06 30.60	0.25 0.35 0.47 0.67 1.01 2.05 3.97 7.60 21.74 23.86 28.16 31.78 38.24 51.45	31176 31288 31468 31886 33056 35076 37917 46731 46218 46033 46524 45524 45524 45524

Bott Yield Stress = 27,500 psi (SA193-B8)

DGmax = 0.050 in

Calculation 3002500Y

MV-4237-001-R004 Rev. 0

H-65

ŧ.

300# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

0

a

INITIAL BOLT STRESS = 50000 psi JOINT RELAXATION = 0%

Bolt Stress at 2.0GLP (psi)		111020	121192	128119	124022	118305	110147	106778	103963	97844	92436	92067	010.67	00000	2 FOND	82400
Leak Area at 20GLP (sq in)		0.02	0.03	0.78	0.38	0.08	0.13	0.18	024	0.29	0.32	0.38	0.44		20.0	0.68
Leak Area et 1.75GLP (sq in)		0.02	0.03	0.04	0.05	90.08	60.0	0.13	0.18	0.22	0.24	0.28	0.99	0000	0.39	0.51
Lesk Arsa et 1.5GLP (aq in)		0.01	0.02	0.02	0.03	0.04	0.06	0.09	0.12	0.14	0.16	0.10	00.00	0.64	0.26	0.34
Lesk Aree at 1.25GLP (sq in)		0.01	100	0.01	0.02	0.02	0.03	0.04	0.06	0.07	D DA	000	***	0.11	0.13	0.17
Leek Rate et .75GLP (mg/sec)		C	0	0	00	0			0					0	0	0
Lesk Rate at .50GLP (mg/sec)		0						5 10			. c		0	0	0	0
Leak Rate at .25GLP (mg/sec)		0									5 0		n	0	0	0
Leak Rate at GLP (mg/sec)							20	ų •	- 0		0.0	0	0	-		-
Gross Leak Pr	NORMAL (*** -	40003	10001	00001	NU111	11/04	1020	4111	1997	ECC+	4001	3989	3870	3586	2980	3152
Gaskel Deflect. (in)		o new	0.000	0000	0.000	0 000	0 000	0000	0 0 2 0 2 0	0000	0.000	0 0 0 0 0	0.050	0.050	0.050	0.050
Act Gasket Stress (psl)		and the second	21550	228(30)	29300	19050	127030	(3)(3)	10350	14850	141(0)	16200	15100	11900	1 DRADO	12000
Eff Gasket Stress (psi)			35812	38299	48601	31556	21098	17774	17060	24483	22979	26370	24390	80000	10100	20751
Flange Diameter (in)	Fianges		1-1/2	C4	2-1/2	0	Ŧ	9	cn)	10	12	14	16	18	00	24
THE OWNER AND ADDRESS OF	THE OWNER WHEN THE PARTY NAMES	and strength of the local division of the lo	1000	1000	-	and the second	and stores	and states	and in case	and in case	1000	and statements			-	

- H-66

194

Bolt Yield Stress = 122,000 psi (SA564.630)

DGmax = 0.050 in

Calculation 3005000 WR1

MV-4237-001-R004 Rev. 0

- 1 - 1

300# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 30000 psi JOINT RELAXATION = 7%

Bolt Stress at 2.0GLP (pel)		110008	PERILLI I	130026	136363	120677	196980	123220	121414	*15770	110923	110480	109257	110164	110887
Leak Area at 2.0GLP (sq in)		000	080	214	5 33	02.0	1 29	0.81	0.29	0.34	0.38	0.45	0.53	0.62	0.82
Leek Area at 1.75GLP fsq in)		0.00	0.03	0.38	0.05	0.07	0.11	0.15	0.21	0.26	0.29	0.34	0.39	0.46	0.62
Leek Arec et 1.5GLP (sq in)		0.01	0.02	0.03	0.03	0.05	0.07	0.10	0.14	0.17	0.19	0.23	0.26	0.31	0.41
Leek Area at 1.25GLP (sq in)		0.01	0.01	0.01	0.02	0.02	0.04	0.05	0.07	0.09	0.10	0.11	0.13	0.15	0.21
Leak Rate et .75GLP (mg/sec)		0	0	0	0	0			0	0	0	0	0	0	0
Leek Rste et SOGLP (mg/sec)		0	0	0	0	0		0	0	0	0	0	0	0	0
Leak Rate at 25GLP (mg/sec)		0	0	0	0	0	0	0	0	0	0	0	0	0	0
Leak Rate at GLP (mg/sec)		0	0	0	0	0	0	54	0	0	0	0	1		
Gross Leak Pressure (psi)		17896	16722	19332	12397	7768	5433	4583	5324	4741	4787	4644	4304	3576	3782
Gaskel Deflect (in)		0.050	0.050	0 0 0 0 0	0.050	0.050	0.050	0 050	0.050	0.050	0 0 2 0	0 050	0.050	0.050	0.050
Act Gasket Stress (psi)		21550	22800	29300	19050	12700	9900	10350	14850	14100	16200	15100	11900	10800	12000
Eff Gaskei Stress (psi)		42974	45959	58322	37867	25318	21329	20472	29387	27575	316.4	29268	24035	21816	24901
Flange Diamater (in)	Flanges	1-1/2	çu	2.12	(1)	17	9	60	10	12	14	16	18	20	24

Bolt Yield Stress = 122,000 psi (SA564.630)

DGmax = 0.050 in

Calculation 3006000 WR1

MV-4237-001-R004 Rev. 0

300# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 60000 psi

JOINT RELAXATION = 25%

Bolt Stress at 2.0GLP {psl}		103465	114972	122655	120537	119340	116659	117078	117236	113715	110923	110480	109257	440404	101012	196011
Leek Area et 2.0GLP (sq.in)		0.02	0.03	0.12	0.06	0.08	0.13	0.19	0.28	0.34	0.38	0.45	0.53	1000	20.0	0.82
Leak Area at 1.75GLP (sq in)		0.02	0.02	0.03	0.05	0.06	0.10	0.15	0.21	0.25	6.29	0.34	0.70	0.40	0.40	0.62
Leak Area et 1.5GLP (sq in)		0.01	0.02	0.02	0 03	0.04	0.07	0.10	0.14	0.17	0.19	0.23	0.96	1000	0.31	0.41
Leak Area at 1.25GLP (sq In)		0.01	0.01	0.01	0.02	0.02	0.03	0.05	0.07	0.08	010	0.11	0.13	21.0	0.15	0.21
Leak Rate at 75GLP (mg/sec)		0	0	0	0	0		-	0	0	0	0	0	2 1	0	0
Leak Rate at 50GLP (mg/sec)		0	0	0	0	0		0	0	0	c				0	0
Lesk Rate et 25GLP (mg/sec)	•	0	0	0	0	0	0	0	0	0	0				0	0
Leak Nate et GLP (mg/sec)		o	G	0	0	0	0	10	4 C							*
Gross Leak Pressure (psi)		15,430	14576	ROAR	19967	6993	RNGG	2352	54A1	4657	1004	40.44		4304	3578	3782
Gasket Deflect. (in)		0.050	n nen	0.050	0.050	0.050	0.050	0000	0.060	D DED	0.000	North N	0 000	0.050	0.050	0.050
Act Gaskei Stress (psi)		16163	17100	91075	DUCEN.	06.36	7495	1955	0110	11130	C /COL	10121	11360	6258	8100	0006
Eff Gasket Stress {psi}		ATOCA	ACORD	60000	100260	00000	20010	C 13C3	20416	10000	C1010	50000	59767	240355	21816	24901
Flange Diameter (in)	Flanges	4 4 10	2014	N and	211.2	2 *	7 4	0 0	n 5	01	2	4	16	18	UC	24

Bolt Yield Stress = 122,000 psi (SA564-630)

DGmax = 0.050 in

Calculation 3006025.WR1

MV-4237-001-R004 Rev. 0

H-68

300# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE **TABLE 3-23**

INITIAL BOLT STRESS = 60000 psi JOINT RELAXATION = 0%

Bolt Stress at 2.0GLP (psi)		119998	131894	139926	136253	132577	125280	123220	121414	115770	110923	110480	109257	110164	110887
Leak Area at 200LP (sq in)		0.62	1.89	3.41	4.08	1.77	5.14	6.27	7.48	4.48	0.38	0.45	0.53	0.62	0.82
Leak Area ** 1.75GLP (sq In)		0.02	0.66	1.66	1.75	1.65	0.29	0.15	0.21	0.26	0.29	0.34	0.39	0.45	0.62
Leek Area et 1.50LP (sq in)		0.01	0.02	0.03	0.03	0.05	0.07	0.13	0.14	0.17	0.19	0.23	0.26	0.31	0.41
Leak Area at 1.25GLP (sq in)		0.01	0.01	0.01	0.02	0.02	0.04	0.05	10.07	60.0	0.10	0.11	0.13	0.15	0.21
Leak Rate et .75GLP (mg/sec)		0	0	0	0	0		1	0	0	0	0	0	0	0
Leek Rate et .50GLP (mg/sec)		0	0	0	0	0		0	0	0	0	0	0	0	0
Leak Rate et 25GLP (mg/sec)		0	0	0	0	0	0	0	0	0	0	0	0	0	0
Leak Bate at GLP (mg/soc)		0	0	0	0	0	CV	2	0	0	0	0	*		ł
Gross Leak Pressure (psi)		17896	15722	19332	12397	7768	5433	4583	5324	4741	4787	4644	4304	3576	3782
Gasket Deflect. (in)		0.050	0 050	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0.050
Act Gasket Stress (ps!)		21550	22800	29300	13050	12700	0066	10350	14850	14100	16200	15100	11900	10800	12000
Eff Gasket Stress (psi)		42974	45355	58322	37867	25318	21329	20472	29387	27575	31644	29268	24035	21816	24901
Flange Diameter (in)	Flanges	1-1/2	¢4	2-1/2	0	7	9	80	10	12.	14	16	18	20	24

Bolt Vield Stress = 111,000 psi (SA564 630)

DGmax = 0.050 in

Calculation 30060009,WR1

MV-4237-001-R004 Rev. 0

H-69

TABLE 3-24

400# ANSI FLANGE AND GASKET DATA

Flance	*	*GASKET			Pressure	*	*		
Diameter (in)	OD (in)	ID (in)	Width (in)	Area (sq in)	Area (sq in)	Number	Diameter (in)	Area (sq in)	Length (in)
Flanges									
1-1/2 2-1/2 3 4 6 8 10 12 14 16 18 20	2.750 3.375 3.875 4.750 5.875 8.250 10.375 12.500 14.750 16.000 18.250 20.750 22.750	2.125 2.750 3.250 4.000 5.000 7.188 9.188 11.313 13.375 14.625 16.625 18.688 20.688 20.688	0.2500 0.2500 0.3125 0.3750 0.4688 0.5313 0.5313 0.6250 0.6250 0.7500 0.9688 1.0625	1.865 2.356 2.749 4.234 6.332 11.275 16.220 19.767 27.489 29.943 40.939 59.822 65.909 86.161	3.547 5.940 8.296 12.566 19.635 40.574 66.296 100.509 140.500 167.989 217.077 274.279 336.129 481.105	4 8 8 8 8 2 2 2 6 6 0 0 4 4 4	3/4 5/8 3/4 7/8 7/8 1 1-1/8 1-1/4 1-1/4 1-1/4 1-3/8 1-3/8 1-1/2 1-3/4	0.3340 0.2256 0.3340 0.3340 0.4612 0.4612 0.6051 0.7896 0.9985 0.9985 1.2319 1.2319 1.2319 1.4899 1.8983	2.125 2.375 2.625 2.875 3.125 3.625 4.125 4.625 4.625 5.125 5.375 5.625 5.875 6.375

Bolt areas correspond to tensile stress area

400# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE TABLE 3-25

INITIAL BOLT STRESS = 20000 psi JOINT RELAXATION = 0%

Bolt Stress at 2.0GLP {psi}		51310	52118	52637	53635	55807	55029	49082	45923	41208	36974	36827	36419	36721	36962
Leek Ares et 2.0GLP (sq.ic)		4.07	2.18	2.98	4.42	7.25	11.20	16.8	10.96	8.71	4.90	5.64	5.98	7.47	10.26
Lesk Area at 1.75GLP (sq in)		1.09	1.60	2.12	2.98	4.45	6.86	6.59	6.55	3.76	0.11	0.13	0.16	0.18	0.24
Leak Aree et 1.5GLP (sq in)		0.73	1.07	1.41	1.99	2.97	4.27	321	2.13	200	0.08	60.0	0.10	0.12	0.18
Leak Area at 1.25GLP (sq in)		0.36	0.53	11/0	0.39	1.48	1.68	0.03	0.03	0.04	0.04	0.04	0.05	30.6	0.08
Leak Hate at 750° P (mg/sec)		0	0	0	0	0	8 41		0	0	0	0		0	-
Leek Rate at 50GLP (mg/sec)		0	0	0	0	0	0		0	0	0	0	0	0	¢
Leek Rate at 25GLP (mg/sec)		0	0	0	0	0	0	Û.	0	0	þ	0	c	0	0
Leak Rate at GLP (mg/sec)		0	0	0	c	**	C4	9	-		0	**	2	**	*
Gross Leak Pressure (ps.)		2683	6023	6344	4388	4286	3295	2395	2628	2134	2018	1910	1770	1779	1806
Gasket Deflect. (in)		0.033	0 634	0 033	0.033	0.046	0.050	0.043	0.043	0.041	0 C41	0.040	0 042	0.050	0.044
Act Gasket Stress (psi)		14325	15320	19441	12622	11653	5817	8963	12783	11624	13339	12036	9884	10800	10575
Eff Gasket Strass (psi)		14325	15320	19441	12622	11653	9817	8953	12783	11624	13339	12036	9884	10851	10575
Tlange Diameter (in)	Flanges	1-1/2	CN.	2-1/2	e')	4	φ	80	10	12	14	16	18	20	24

Bolt Yield Stress = 33,000 psi (SA193-88)

DGmax = 0.050 in Calcutation 4002000

H-71

MV-4237-001-P:004 Rev. 0

400# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 25000 psi

	х.	
	ς.	
200	2	
6.2	2.	
-		
14.4		
11		
23		
-	۰	
£.,	2	
pre-	ε.	
	J.	
	Ε.	
	2	
-	е.	
	<i>#</i> -	
62	2	
24	E.	
	2.	
62		
	٦.	
-	ż.	
1.8	Υ.	
LL.	2	
~		
1.		
	۰.	
	۳.	
-	μ.	
ź,	1	
	2	
per	8	
	3	
1994	с.	

Bolt Stress at 2.0GLP (psi)		44268	44765	45121	45698	71524	62804	59830	56431	51312	46218	46033	ACROA	1000	422605	46203
Leak Area at 2.0GLP (sq in)		0.69	131	174	2.45	21.52	18.78	22.73	24.71	19.26	15.87	18.81	06.10	100°1.2	25.34	33.53
Leek Area at 1.75GLP (sq in)		0.67	0.98	130	1.84	10.89	11.09	13.42	14.37	13.14	959	11.25	1967	10.21	15.13	20.13
Leak Area at 1.5GLP (sq in)		0.45	0.66	0.87	1.22	5.43	5.91	324	8 95	7.00	321	3 70	2000	3,83	4.91	6.73
Leek Area at 1.25GLP (sq in)		0.22	0.33	0.43	0.61	2 00	2.96	4 12	25E	0.83	0.05	0.06	0000	0.00	0.08	0.10
Leek Rate at .75GLP (mg/sec)		0	0	0	0	0				0	c		2 4	0	0	0
Leak Rate at 50GLP (mg/sec)		0	0	0	0		0	0		0	0			0	0	0
Leek Rate at 25GLP (mg/sec)		0	C	0		0	0			, c				0	0	0
Leak Rate at GLP (mg/sec)		0	G				20					2 0	0		-	
Groes Leak Prossure (psi)		4246	3676	3004	0000	E707	2721	10100	2107	9000	1002	2000	Z367	2212	- FCCC	2008
Gasket Definct. (in)		0.042	CPU O	1000	1000	0.000	0000	0.000	0000	0000	0000	nco.n	0,050	0.050	0.070	0.050
Act Gasket Stress (psi)		A TON'SC	*GTEU	10 HOO	24201	07/01	1C/UN	NULLAN N	10320	14000	14100	16200	15046	11900	+COOM	12000
Eft Gasket Stress {ps}		17005	10000	00101	24301	13/161	1905:	21221	11192	123/10	140.23	166/3	15046	12356	60200	13010
Flange Diamater (in)	Flanges		211-1		2-1/2		4	20 1	8	01	12	4	16	ц,		

Bolt Yield Stress = 33,000 psi (SA193-88)

DGmax = 0.050 in

Calculation 4002500

MV-4237-001-R004

Rev. 0

400# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE TABLE 3-27

INITIAL BOLT STRESS = 30000 psi JOINT RELAXATION = 6%

Bolt Stress et 2 0GLP (psi)		92871	06166	104178	99174	78159	70371	68051	65157	80274	55462	55240	54628	55082	55.243
LCSK Area et 20GLP (sq in)			*	*	*	31.44	32.86	39.41	43.61	58 BE	32.58	38.20	42.85	6130	0-239
Leak Area at 1.75GLP (sq In)		•	•			17.62	+8.72	20.81	25.09	23.58	19.38	22.91	26.34	30.96	4C.7 +
Leek Arra et 1.5GLP (sq in)		6.39	11.62	17.09	19.47	7.66	831	10.22	12.52	14.24	EZ.11	13.85	15.87	18.70	24.63
Leak Area at 1250LP (sq in)		2.07	430	6.99	6.57	5.54	19.91	4.69	6.26	7.03	4.07	4.79	5.40	6.44	8.55
Lesk Rate et .75GUP (mg/sec)		0	0	0	0	0	+	-	0	0	0	0	0	2.	ę
Leek Rate et .50GLP (mg/sec)		0	0	0	0	0	0	0	0	0	0	0	0	0	0
Leak Rete et 25GLP (mg/sec)		0	0	0	0	0	0	0	0	0	0	0	0	0	0
Leak Rele et GLP (mg/sec)		0	0	0	0	0	2		0	0	0	0		**	5
Gross Leak Pressure (psi)		1,3850	12652	\$4393	9024	6324	4214	3321	3728	3121	3027	2865	2655	2668	2409
Gasket Deflect (in)		0:020	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0 050	0.050	0.050	0.050	0.050
Act Gasket Stress (psi)		21487	22800	29161	18933	12700	9900	10350	14850	14100	16200	15100	11900	10800	12000
Eft Gasket Stress (psi)		21487	22979	29161	18933	17480	14726	13430	19174	17435	20008	18055	14827	16278	15863
Flange Diameter (in)	Flanges	1-1/2	2	2.1/2	0	-	9	8	10	12	14	10	18	20	24

* Bolt Uttimate Strength Exceeded

Bolt Ultimate Strength = 82,500 psi (SA193-B8)

Bolt Yield Stress = 33,000 psi (SA193-88)

DGmax = 0.050 in

Calculation 4003000

MV-4237-001-R004

Rev. 0

400# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 25000 psi JOINT RELAXATION = 15%

Flange Diameter (in)	Eff Gasket Stress (psi)	Act Gasket Stress (psi)	Gasket Deflect (in)	v oss Leak essure (psi)	Leak Rate at G ^{i p} (mg/sec)	Lesk Rate et .25GLP (mg/sec)	Leek Rate et .50GLP (mg/sec)	Leak Rate at .75GLP (mg/sec)	Leak Area at 1.25GLP (sq in)	Leak Arze et 1.5GLP (sq in)	Lesk Area at 1.75GLP (sq in)	Leak Area at 2.0GLP (sq in)	Bolt Stress at 2.0G' (psi)
Flanges													
1.1/2	15200	15220	0.042	6234	0	0	0	0	0.33	0.66	0.99	1.31	64580
1-1/6	16277	16077	0.042	5251	0	e	0	0	0.48	0.96	1 44	1.93	50279
2.1/2	20656	20656	0.041	5734	õ	0	õ	0	0.64	1.28	1.91	2.55	50803
3	13411	13411	0.041	3966	0	0	0	Ö	0.90	1.80	2 70	3 60	51651
4	14567	10795	0.050	5321	0	0	0	0	1.84	3 69	7.79	13.50	65772
6	12272	8415	0.050	3537	2	9	0	1	2.78	5.58	2.76	15.04	590.8
8	1 11192	8798	0.050	2782	1	0	0	1	3.08	7.01	10.93	18.92	57021
10	15978	12623	0.050	3119	0	0	C	0	2.47	7.70	12.34	21.05	54511
12	14529	11985	0.050	2607	0	0	0	0	0.15	6.21	12.24	19.26	50337
14	16673	13770	0.050	2522	0	0	0	0	0.05	5.21	9.59	15.97	46218
16	15046	12789	0.050	2387	0	0	0	0	0.06	3 70	11.25	18.81	46033
18	12356	10115	0.050	2212	1	0	0	0	0.06	3.99	12.67	21.39	45524
20	13563	9180	0.050	2224	1	0	0	0	0.08	4.91	15.13	25.34	45902
24	13219	10200	0.050	2008	1	0	0	0	0.10	6.73	20.13	33.53	46203

Bolt Yield Stress = 33.000 psi (SA193-B8)

DGmax = 0.050 in

H-74

Calculation 4002515

400# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 25000 psi JOINT RELAXATION = 25%

Rch S ess at 2.0GLP (psi)		53070	53956	54591	55619	61937	56561	55149	53230	49687	46218	46033	45524	45902	46203
Leek Area et 2.0GLP (sq in)		1.72	2.64	3.60	5.29	10.61	12.68	16.31	18.62	17.53	15.97	18.81	21.39	25.34	33 53
Leok Area at 1.75GLP (sq in)		1.19	1.75	2.32	327	6.35	7.45	16.6	11.98	11.63	9.59	11.25	12.67	15.13	20.13
Lesk Area at 1.5GLP (sq in)		0.60	1.17	1.55	2 18	3.47	4.78	6.11	6.87	5.69	321	370	3.95	4.91	6.73
Leek Area at 1.25GLP (sq in)		0.40	0.58	0.77	1.09	1.74	2.12	2.31	1.76	0.04	0.05	0.06	0.06	0.08	0.10
Leak Rate at 75GLP (mg/sec)		0	0	0	0	0	-		0	0	0	0	0	0	0
Leek Rate at 50GLP (mg/sec)		0	0	0	0	0	0	0	0	0	0	0	0	0	0
Lenk Rete at 25GLP (mg/sec)		0	0	0	0	0	0	0	0	0	0	0	0	0	0
Leah Rate at GLP (mg/sec)		0	0	0	0	0	2		0	0	0	0			-
Gross Leak Pressure (psi)		7561	6368	6354	4810	5011	3387	2691	3046	2573	2522	2387	2212	2224	2008
Gasket Defect. (in)		0.042	0.042	0.041	0.041	0.050	0.050	0 0 2 0	0.050	0.050	0.050	0.050	0.050	0.050	0.050
Act Gasket Stress (psi)		13429	14362	18226	11833	9525	7425	7763	11138	10575	12150	11284	8925	8100	9006
Eff Gasket Stress (psi)		13429	14362	18226	11833	14567	12272	11192	15978	14529	16673	15046	12356	13563	13219
Fiange Diameter (in)	Planges	1-1/2	2	2.1/2	(*)	4	9	00	10	12	14	16	18	20	24

Bolt Yield Stress = 33,000 psi (SA 193-88)

DGmax = 0.050 in

Calculation 4002525

MV-4237-001-R004 Rev. A

400% FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE **TABLE 3-30**

INITIAL BOLT STRESS = 25000 psi

JOINT RELAXATION = 33%

Bolt Stress at 2 0GLP {psi}		51853	55014	57821	55372	58870	54564	53351	52206	49:67	46218	46033	45524	45902	46203
Leak Area at 2.0GLP (sq in)		1.50	3.16	4.58	5.18	8.90	10.	14.22	16.67	17.03	16.31	18.61	21.39	25.34	33-53
Leak Area et 1750LP (sq in)		1.09	1.91	2.65	3.17	5.19	6.69	60.6	11.22	11.15	9.59	11.25	12.67	15 13	20.13
Leek Area at 1.5GLP (sq in)		0.58	1.26	1.76	2.03	3.30	\$ 12	5.39	6.21	5.27	3.21	3.70	3.95	491	6.73
Leek Area at 1 25GLP (sq in)		0.28	0.61	0.58	0.83	1.65	1.55	1.70	1.19	0.04	0.05	0.06	0.06	0.08	0.10
Leek Rate et 75GLP (mg/sec)		0	0	0	0	0		-	0	0	0	0	0	0	0
Leak Rate at 50GLP (mg/sec)		0	0	0	0	0	0	0	0	0	0	0	0	0	0
Leak Rate at 25GLP (mg/sec)		0	0	0	0	0	0	0	0	0	0	0	0	0	0
Leek Rate et GLP (mg/sec)		0	0	0	0	0	54	-	0	0	0	0			
Gruss Leak Pressure (psi)		1733	7102	7930	5038	4763	3267	2618	2987	2546	2522	2387	2212	2224	2008
Gasket Deftact. (m)		0.042	0.042	0.041	0.041	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0.050
Act Gasket Stress (psi)		11997	12830	16291	10571	8509	6633	6935	9950	9447	1005.4	100341	7973	7236	8040
Eff Gesket Stress (psi)		1997	12830	16281	10571	14567	CLCC1	11192	15978	14529	16673	150AG	12265	126.613	13219
Flange Litameter (in)	Flanges	1.1/2	0	2112	9	4	12	đ	U1	-	14	41	1	00	24

Bolt Yield Stress = 33,000 psi (SA193-88) DGmax = 0.050 in

Calculation 4002533

MV-4237-001-R004 Rev. 0

H-76

400# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE TABLE 3-31

INITIAL BC: TSTRESS = 25000 psi

JOINT RELAXATION = 50%

Bolt Stress at 2.0GLP {p.1}		-	35550	41801	LOWER	\$1322	52351	50318	50468	50029	48063	46218	46033	45524	45902	46203	
Leak Area at 2.0GLP (sq in)			0.46	0.99	150	1.62	5.23	7 44	10.82	74.43	15.88	15.97	18.81	21.39	25.34	33.53	and the second second
Leak Aree #1.75GLP (sq In)			0.15	0.50	0.83	0.77	3.58	5.07	7.35	9.60	10.13	9.59	11.25	12.67	15.13	20.13	Second second second
Leak Aree et 1.5GLP (sr] in)			0:00	0.02	0.17	0.01	241	271	3.87	4.80	4 36	321	370	3.95	4.91	673	
Leek Area at 1.25GLP (sq in)			000	0.00	000	0.01	0.64	0.34	0.60	0.04	004	0.05	0.06	0.06	D DB	110	and a second sec
Leak Rate et .75GLP (mg/sec)			0	0	0					- 0			3 0		2 4	5 6	~
Leak Rate et SOGLP (mg/sec)			0						5 6			5 4	5 ¢		2 0	o c	2
Leak Rate at 23(3LP (mg/sec)			c			5 0	0.0			0 4	0		0	0	0.0	0 0	a
Leak Rate et GLP (mg/sec)					2 0		0	0	2	-	5	0	0	0	-		-
Gross Leek Pressure (psi)				2110	2006	2665	3760	4236	3013	2463	2863	2483	2552	2387	2212	2224	2008
Gasket Deflect. (in)				0.042	0.042	0.041	0.041	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0.050
Act Gasket Stress (csi)				6363	8575	12150	7889	6350	4950	5175	7425	7050	8100	7523	5950	5400	6000
Eft Gasket Stress	1000			8953	9575	12150	7889	14567	12272	11192	15978	14529	16673	15046	12356	13563	13219
Flarige Dismeter	feast	Flanges		2/1-1	2	2.1/2	0	4	9	st.	10	12	1.4	16	191	00	24

H-77

Bolt Yield Stress = 33,000 psi (SA193-88)

1....

DGm.ex = 0.050 in

Calculation 4002550

MV-4237-001-R004 Rev. 0 519 M

.

0

S

400# FLANGE GASKET STRESS, GROSS LEAK PRESSURE, AND LEAK RATE

INITIAL BOLT STRESS = 25000 psi JOINT RELAXATION = 0%

Bolt Stress at 2.0GLP (psl)		31021	31176	31268	31468	71524	62804	59830	56431	51312	46218	46033	45524	45902	ACTURE	40,603
Leek Area at 2 OULP (sq in)		0.28	0.41	0.54	0.76	35.76	37.66	43.98	43.99	36.12	27.18	31.86	35.75	42.79	67.04	51.00
Leak Area et 1.75GLP (sq in)		0.21	0.31	0.41	0.57	23.12	19.02	22.00	23.55	21.48	16.15	19.09	21.85	25.80	200.000	03.30
Leak Area et 1.5GLP (sq in)		0.54	0.20	0.27	0.38	10.47	906	10.85	52.20	祝い	9.77	11.54	13.23	15.58	in the second	502
Leak Area at 1 25GLP (sq in)		0.07	01.0	0.34	0.13	3.73	2.96	4.12	5.42	6.14	3.40	3.99	4 50	5.37	1111	6.13
Leak Rate at 75GLP (mg/sec)		0	0	0	0	0	-	1	0	0	0	0	0	0		0
Lesk Rate st 50GLP (mg/sec)		0	0	0	0	0	0	0	0	0	0	0	0	0	2.1	0
Leek Rate et 25GLP (mg/sec)		0	0	0	0	0	0	0	0	0	0	0	0		2.1	0
Leek Rate at GLP (mg/sec)		0	6	0	0	0	cy		0	0	3	0				
Gross Leak Pressura (psi)		1326	1117	1220	R44	5787	3761	2010	8000	2657	0620	2387	CICC	0004	6264	2008
Gasket Deflect. (in)		6.042	0.042	0.041	0.041	0.050	0.050	0.050	0.050	0.050	0.050	0.050	0.060	noon n	nonn	0.050
Act Gasket Stress (psi)		17906	19150	24301	15274	12700	0006	10350	14950	14100	UNC31	15046	140000	100011	Incini	12000
Eff Gasket Strass (psi)		17906	19150	102.96	15,773	14567	01001	00111	15,078	145.00	10001	10010	00000	00021	13063	13219
Flange Diamster (in)	Flanges	1.10		2.110		2 10	- 12	o a	101			10.	0.	2	R.	24

Bolt Yield Stress = 27,500 psi (SA193-88)

DGmax = 0.050 in Calculation 4002500Y

MV-4237-001-R004 Rev 0



Flange Size (in)

FIGURE 3-1

INCREMENTAL GASKET LOAD TO TOTAL PRESSURE LOAD RATIO

(KSI)



GASKET DEFLECTION, DG (Am)

FIGURE 3-2

TYPICAL LOAD-DEFLECTION DIAGRAM SHOWING A STANDARD TEST SEQUENCE



GASKET STRESS, SG (MPo)

FIGURE 3-3

A TYPICAL MASS-LEAK-RATE VERSUS GASKET STRESS PLOT


FIGURE 3-4

GASKET STRESS VERSUS TIGHTNESS PARAMETER FOR TWO SPIRAL-WOUND ASBESTOS GASKETS, CYCLIC TESTS WITH WATER MV-4237-001-R004 Rev. 0

MV-4237-001-R004 Rev. 0



.

- 25000, 25% - 25000, 33% - 25000, 50%

FIGURE 3-5

LEAK RATE VERSUS PRESSURE FOR 8*-150# FLANGE

MV-4237-001-R004 Rev. 0

4. VALVES

Three failure modes are postulated for the various valves present in the reference Combustion Engineering plant safety system under consideration. These include failure of the stem packing, failure of the bolted bonnet seal, and failure of the valve body. The types of valve stem packing currently used in most nuclear plants tend to compress under high pressure conditions providing greater leak resistance. Although it is possible that the stem packing for some valves could deteriorate in response to service conditions, it was judged that any resulting leak rate or leak area would be quite smail and have a negligible effect on both valve and system operation. Failure of the bolked bonnet seal is a credible failure mode. However, information on the valve bolted bonnets was not available and, therefore, estimates of pressure capacities for this failure mode could not be made. Since the valve body thickness is typically greater than that for the adjacent piping, it was judged that failure of the adjacent piping will occur prior to the failure of the valve body. Table 4-1 lists the valves in the lines under consideration identified by EG&G.

A particular valve of interest is SI-107A(B), a valve that checks revulse flow from the suction side of the Low Pressure Safety Injection pump back into the Refueling Water Storage Pool. For this valve, the failure mode of interest is an internal failure allowing reverse flow into the Refueling Water Storage Pool. Adequate information on the dimensions of the internal hinged plates was not available. As a result, an evaluation of the pressure capacity for internal failure could not be made. However, based on the limited available information, it was judged that the pressure capacity for this failure mode would be high.

Table 4-1

Valves

SI-107A(B)	SI-208A
SI-201B	SI-216
SI-604B	SI-219A
SI-602B	SI-135A
SI-201A	SI-407A
SI-1071B	SI-405A
SI-108B	SI-133A
SI-109B	SI-138A
SI-410B	SI-228B
SI-122B	SI-244
SI-124B	SI-332B
SI-125B	SI-330B
SI-203A	SI-336B
SI-207A	

REFERENCES

- Wesley, D. A., T. R. Kipp, D. K. Nakaki, and H. Hadidi-Tamjed, Pressure Dependent Fragilities for Piping Components - Pilot Study on Davis-Besse Nuclear Power Station, prepared for EG&G Idaho, Inc. by ABB Impell Corporation, September 1990.
- Harvey, P. D. Ed., Engineering Properties of Steel, American Society for Metals, Metals Park, Ohio, 1985.
- Weiss, V. and J. G. Sessler, Eds., Aerospace Structural Metals Handbook, Vol. I: Ferrous Alloys, ASD-TDR-67-741, Air Force Materials Laboratory, Wright-Patterson Air Force Base, Ohio, March 1963.
- Manjoine, M. H., Ductility Indices at Elevated Temperature, J. Engineering Materials and Technology, Trans. ASME, Vol. 97, Series H, No. 2, 1975 pp. 156-161.
- Metals Handbook, Vol. 1, Properties and Selection: Iron and Steel, 8th Ed., American Society for Metals, Metals Park, Ohio, 1961.
- General Torquing and Detensioning, Waterford 3 SES, Plant Operating Manual, Procedure No. MM-06-011, Rev. 4.
- Response to INEL/USNRC Information Request, Bolt Type to Line Number from RWSP to LPSI Pump Suction to the MOV Isolation Valve, Transmittal from Entergy Operations, Inc., to Dana Kelly, INEL, Fabruary 13, 1991.
- 8. Algor Finite Element System, Algor Interactive Systems, Inc., Pittsburgh, PA, 1989.
- Bazergui, A. and L. Marchand, PVRC Milestone Gasket Tests First Results, Bulletin 292, Welding Research Council, New York, NY, February 1984.
- Bazergui, A., L. Marchand, and H. Raut, *Development of A Production Test Procedure for Gaskets Bulletin* 309, Welding Research Council, New York, NY, November 1985.
- 11. Bazergui, A., L. Marchand, and H. Raut, *Further Gasket Leakage Behavior Trends,* Bulletin 325, Welding Research Council, New York, NY, July 1987.
- Bazergui, A., Short Term Creep and Relaxation Behavior of Gaskets, Bulletin 294, Welding Research Council, New York, NY, May 1984.
- MIL-G-21032E, Military Specification for Metallic-Asbestos, Spiral Wound Gaskets, through Amendment 2, 18 May 1979.

BIBLIOGRAPHIC DATA SHEET	NUREG/CR-5745
TITLE AND SUBTITLE	DOU- SOON
Assessment of ISLOCA Risk-Methodology and Application to a Combustion Engineering Plant	April 1992
	A FIN OR GRANT NUMBER B5099
D. L. Kelly J. L. Auflick L. N. Haney	Technical
	7. PERIOD COVERED Interview Dates
PERFORMING ORGANIZATION - NAME AND ADDRESS (1) NRC provide Division, Differ of Region, U.S. Nuclear Regulatory C name and mailing address)	ommigsion, and mailing address in contractor provi
EGAG Idaho. Inc.	
Idaho Falls, ID 85415	
SPONSORING ORGANIZATION - NAME AND ADDRESS (If NRC Type State as obout "If contractor, provide NRC Durison, D and mailing address.) Division of Safety Issue Resolution Office of Nuclear Regulatory Research U.S. Nuclear Regulatory Commission Washington, D.C. 20555	rius ur Region, U.S. Nucleer Regulators Commun
D. SUPPLEMENTARY NOTES	and a second second . The data of the local second s
Inter-system loss-of-coolant accidents (ISLOCAs) have been identified as important	contributors to offsite risk for
some nuclear power plants. A methodology has been developed for identifying and e hardware designs, human factors issues, and accident consequence factors relevant to core damage frequency and risk. This report presents a detailed description of the app methodology to a Combustion Engicieering plant.	valuating plant-specific the estimation of ISLOCA dication of this analysis
some nuclear power plants. A methodology has been developed for identifying and echardware designs, human factors issues, and accident consequence factors relevant to core damage frequency and risk. This report presents a detailed description of the apprenthodology to a Combustion Engicteering plant.	Valuating plant-specific the estimation of ISLOCA dication of this analysis
 some nuclear power plants. A methodology has been developed for identifying and e hardware designs, human factors issues, and accident consequence factors relevant to core damage frequency and risk. This report presents a detailed description of the apprenthodology to a Combustion Engicteering plant. 12 KEY WORDS/DESCRIPTORS /Lat words or persons that will auto remember 1: Meaning the report.) 13 KEY WORDS/DESCRIPTORS /Lat words or persons that will auto remember 1: Meaning the report.) 14 NET-system loss-of-coolant accident V-sequence probabilistic risk assessment. 	Valuating plant-specific the estimation of ISLOCA dication of this analysis Tr. AVALABILITY STATEM Unlimited TA SECURITY CLAINTER Thur Page: Unclassified
some nuclear power plants. A methodology has been developed for identifying and echardware designs, human factors issues, and accident consequence factors relevant to core damage frequency and risk. This report presents a detailed description of the apprenthodology to a Combustion Engicteering plant. 12 KEY WORDS/DESCRIPTORS /Lat words or present that will auto researches in detailing the report.) 14 KEY WORDS/DESCRIPTORS /Lat words or present that will auto researches in detailing the report.) 15 KEY WORDS/DESCRIPTORS /Lat words or present that will auto researches in detailing the report.)	Valuating plant-specific the estimation of ISLOCA olication of this analysis Transformed Instruction Statem Unlimited Instruction Transform Unclassified Transform Unclassified
some nuclear power plants. A methodology has been developed for identifying and echardware designs, human factors issues, and accident consequence factors relevant to core damage frequency and risk. This report presents a detailed description of the apprenthodology to a Combustion Engicieering plant. 12. KEY WORDS/DESCRIPTORS /List words or phrases that will assue researchers in locating the report. Inter-system loss-of-coolant accident V-sequence probabilistic risk assessment core damage frequency	valuating plant-specific the estimation of ISLOCA dication of this analysis TY. AVALABILITY STATEM Unlimited The Report Unclassified The Report Unclassified The Report Unclassified

THIS DOCUMENT WAS PRINTED USING RECYCLED PAPER

